

### PROJECT MANUAL FOR:

### G-8 ALLIED HEALTH LAB REMODEL

TAFT, CALIFORNIA

06/04/25

TAFT COLLEGE WEST KERN COMMUNITY COLLEGE DISTRICT PROJECT NO: 430-0083

| SET NO: |  |
|---------|--|
|---------|--|

### DIVISION 00 PROCUREMENT AND CONTRACTING REQUIREMENTS

| 000101      | Title  |  |
|-------------|--|--|
| 001113      | Advertisement Bid                                    |  |
| 002113      | Instructions To Bidders                              |  |
| 004113      | Bid Proposal   |  |
| 004313      | Bonds And Certificates Required                      |  |
| 004318      | Prevailing Wage Rages & Apprenticeship Requirements  |  |
| 006001      | General Conditions                                   |  |
| 006002      | Supplementary Conditions                             |  |
| 006003      | Contract Agreement                                   |  |
|             |  |  |
| DIVISION 01 | GENERAL REQUIREMENTS                                 |  |
| 011000      | Summary Of Work                                      |  |
| 012100      | Allowances   |  |
| 012600      | Modification Procedures                              |  |
| 013100      | Project Meetings                                     |  |
| 013300      | Submittals   |  |
| 015050      | Temporary Facilities (Short Form)                    |  |
| 017750      | Project Closeout (Short Form)                        |  |
| 017800      | Warranties And Bonds                                 |  |
| 018000      | Construction Waste Reduction, Disposal and Recycling |  |
| DIVISION 02 | EXISTING CONDITIONS                                  |  |
| 024119      | Selective Demolition                                 |  |
| DIVISION 06 | WOOD, PLASTICS AND COMPOSITES                        |  |
| 062000      | Finish Carpentry                                     |  |
| DIVISION 07 | THERMAL AND MOISTURE PROTECTION                      |  |
| 072100      | Building Insulation                                  |  |
| 079200      | Joint Sealants                                       |  |
| DIVISION 08 | OPENINGS   |  |
| 081113      | Standard Steel Doors And Frames                      |  |
| 081416      | Flush Wood Doors                                     |  |
| 087100      | Door Hardware  |  |
| 088000      | Glass And Glazing                                    |  |
| DIVISION 09 | FINISHES   |  |
| 092900      | Gypsum Board Assemblies                              |  |
| 095113      | Acoustical Panel Ceilings                            |  |
| 096513      | Resilient Wall Base And Accessories                  |  |
| 096519      | Resilient Tile Flooring                              |  |
| 099000      | Painting Sherwin Williams                            |  |
|             | · · · · · · · · · · · · · · · · · · ·                |  |

Attachments:

End Of Table Of Contents

TABLE OF CONTENTS 1

# G-8 ALLIED HEALTH LAB REMODEL TAFT COLLEGE

# WEST KERN COMMUNITY COLLEGE DISTRICT TAFT, CALIFORNIA KERN COUNTY

APPROVED: WEST KERN COMMUNITY COLLEGE DISTRICT

BY: BOARD RESOLUTION

J. PATRICK FOGARTY, AIA AP Architects 3434 Truxtun Ave., Suite 240 Bakersfield, CA 93301 C – 19670

CHEN Engineers, Inc Richard Chen 729 W. Enterprise Ave. Clovis, CA 93619 E - 17213

TITLE 000101 – 1

04/06/17

## NOTICE TO CONTRACTORS (ADVERTISEMENT FOR BIDS)

Notice is hereby given that sealed bids will be received by the WEST KERN COMMUNITY COLLEGE DISTRICT for the furnishing of all labor, material, equipment, transportation and service for **G-8 ALLIED HEALTH LAB REMODEL- TAFT COLLEGE**. Project budget is \$\$300,000.00. Scope of work includes, but not limited to, interior remodel of existing modular G8 for nursing use.

Pre-Bid Non-Mandatory Job Walk/Conference is scheduled on Tuesday 6/17/25 at 10:00 a.m. in front of Taft College Gym, 29 Cougar Court, Taft, CA 93268.

Construction time - 90 calendar days.

Bid information and documents can be viewed and downloaded for free at the Taft College Vendor Portal https://vendors.planetbids.com/portal/66036/bo/bo-search

WEST KERN COMMUNITY COLLEGE DISTRICT will receive sealed bids for the award of the contract for the identified project up to, but not later than the specified deadline. Bids received by the deadline shall be opened and publicly released on the Taft College Vendor Portal <a href="https://vendors.planetbids.com/portal/66036/bo/bo-search">https://vendors.planetbids.com/portal/66036/bo/bo-search</a>

Bids will be received on **Tuesday**, **6/24/25 up until 3:00 p.m.**, via the Taft College Vendor Portal bid posting located on <a href="https://vendors.planetbids.com/portal/66036/bo/bo-search">https://vendors.planetbids.com/portal/66036/bo/bo-search</a>

Each bid must conform and be responsive to the Contract Documents.

Each bid must be accompanied by the security referred to in the Contract Documents and a list of proposed subcontractors.

The DISTRICT reserves the right to reject any or all bids and/or waive any irregularities or informalities in any bids or in the bidding process.

The DISTRICT has determined the general prevailing rate of per diem wages in the locality in which this work is to be performed for each craft or type of worker needed to execute this contract. These rates are available on the Internet at www.dir.ca.gov/DLSR/statistics\_research.html. Hardcopies may be downloaded by Contractor. A copy of these rates shall be posted at the job site. District shall make special determinations when requested, in writing, minimum of seven (7) days before time of bid opening.

The schedule of per diem wages is based upon a working day of eight hours. The rate for holiday and overtime work shall be at least time and one half.

It shall be mandatory upon the contractor to whom the contract is awarded (CONTRACTOR), and upon any subcontractor under them, to pay not less than the specified rates to all workers employed by them in the execution of the contract. It is CONTRACTOR's responsibility to determine any rate change, which may have or will occur during the intervening period between each issuance of written rates by the Director of Industrial Relations.

This project is subject to a Labor Compliance Program approved by the California Department of Industrial Relations and is on file at the district office for review. A limited exemption from prevailing wage does not apply. The Labor Compliance Program will be enforced as required by state law and regulations and the Director of the Department of Industrial Relations. The District will be monitoring prevailing wages on this project. Under the Labor Compliance Program, DISTRICT is required to monitor, audit and investigate the practices of CONTRACTOR, and any subcontractor on the project, concerning payment of prevailing wage, maintenance of proper payroll records, and observation of apprenticeship requirements under the Labor Code, as further described in the Contract Documents. The CONTRACTOR and subcontractors will be required to submit documents on-line and hard copies to confirm compliance for this requirement in the

format and system procedures as selected by the DISTRICT.

This project is not subject to a Labor Compliance Program. The CONTRACTOR and subcontractors will be required to submit documents to confirm compliance with prevailing wage requirements as prescribed in the project manual.

No CONTRACTOR or subcontractor may be listed on a bid proposal for a public works project (submitted on or after March 1, 2015) unless registered with the Department of Industrial Relations pursuant to Labor Code section 1725.5 [with limited exceptions from this requirement for bid purposes only under Labor Code section 1771.1(a)]. No CONTRACTOR or subcontractor may be awarded a contract for public work on a public works project (awarded on or after April 1, 2015) unless registered with the Department of Industrial Relations pursuant to Labor Code section 1725.5. This project is subject to compliance monitoring and enforcement by the Department of Industrial Relations.

No bidder may withdraw their bid for a period of 120 days after the date set for the opening of bids.

Each bid must be accompanied by a cashier's check, certified check or a bid bond issued by Surety Company for at least ten percent (10%) of the amount bid, payable to said Owner. The said check or bid bond shall be retained by the District as liquidated damages if the bidder whose bid is accepted fails or neglects to sign the Agreement and file the required bonds and insurance within ten (10) days from and after the Notice of the Award.

A payment bond and a performance bond will be required prior to execution of the contract. The payment bond shall be in the form called for in the Contract Documents.

Pursuant to the provisions of Public Contract Code Section 22300, CONTRACTOR may substitute certain securities for any funds withheld by DISTRICT to ensure their performance under the contract. At the request and expense of CONTRACTOR, securities equivalent to any amount withheld shall be deposited, at the discretion of DISTRICT, with either DISTRICT or a state or federally chartered bank, as the escrow agent, who shall then pay any funds otherwise subject to retention to CONTRACTOR. Upon satisfactory completion of the contract, the securities shall be returned to CONTRACTOR.

Securities eligible for investment shall include those listed in Government Code Section 61430, bank and savings and loan certificates of deposit, interest bearing demand deposit accounts, standby letters of credit, or any other security mutually agreed to by CONTRACTOR and DISTRICT. CONTRACTOR shall be the beneficial owner of any securities substituted for funds withheld and shall receive any interest on them. The escrow agreement and requirements for implementation shall be as indicated in the Project Manual and as approved by the DISTRICT.

The lowest bid shall be the lowest total of the bid prices on the base contract and all alternates as being used for the purpose of determining the lowest bid price.

To perform the work required by this Notice, CONTRACTOR must possess one of the following type of contractor's license: B.

By Order of the Board of Trustees of the West Kern Community College District

Leslie Minor, Ph.D.,
Interim Superintendent/President

Publish Dates: 06/09/25

06/16/25

END OF SECTION 001113

### PART 1 - GENERAL

### A. SECURING DOCUMENTS:

1. Bid information and documents can be viewed and downloaded for free at the Taft College Vendor Portal <a href="https://vendors.planetbids.com/portal/66036/bo/bo-search">https://vendors.planetbids.com/portal/66036/bo/bo-search</a>

### B. EXAMINATION OF PLANS, SPECIFICATIONS AND SPECIAL PROVISIONS:

- Each bidder shall examine the bidding documents carefully and not later than three days prior to date for receipt for bids, shall make written request to the Architect for interpretation and/or correction of any ambiguity, inconsistency or error therein which he may discover. Any interpretation and/or correction will be issued as an Addendum by the Architect. Only a written interpretation and/or correction by Addenda shall be binding. No bidder shall rely upon any interpretation and/or correction given by any other method.
- 2. By submitting a bid, the bidder implies that he has thoroughly investigated and is satisfied as to the character quality and quantities of work to be performed and materials to be furnished, and as to all the stipulations and requirements of the Contract and construction documents.
  - a. Submission of a bid without requesting a clarification shall be incontrovertible evidence that the bidder has determined that the plans, specifications and documents are sufficient for bidding and completing the work, that the bidder is capable of reading, following and completing the work in accordance with the plans, specifications and drawings, and that the plans, specifications and drawings fall within an acceptable standard for these items and that bidder agrees that the project can and will be completed according to the owner's time lines and according to the allocated days of construction indicated herein.
- 3. The Bidder shall diligently investigate existing conditions to ascertain work required and include all necessary cutting and patching and refinishing in his bid. He shall provide for and exercise every precaution to protect the existing facilities against dust, dirt, water, trash, interruption of personnel activities, interruption of other construction on site, etc., due to operations under this Contract.
  - a. The existing plans of the campus are on file at the Architect's office. All contractors shall review plans prior to bidding during Architect's normal working hours. Copies for Contractor's use may be obtained at printing plus shipping/handling cost, (paid by contractor). Not reviewing plans will not be a reason for claim for additional cost and/or additional time. It shall be the responsibility of the bidders to confirm existing conditions above grade visible and review existing plans for existing conditions not visible from surface, prior to bidding, to confirm impact on scope of work to complete project.
- 4. Neither the Owner or the Architect will be responsible for any omissions, errors, etc., which may result from the Contractor's procurement of incomplete documents. It shall be the Contractor's responsibility to review and ascertain all of the required work, materials, etc., to be provided by him in performing all work as required and/or called for by the Contract Documents.
- 5. Contractor shall review Section 011000 Summary of Work, and provide manpower, resources, etc., as required to complete each phase of project on or before the date required for project completion. Contractor shall allow in Proposal weekend workers, shifts of workers and additional productivity not limited to workers, materials, temporary facilities and equipment as required to meet project schedule duration with limited access times as indicated herein.

- 6. Demolition of existing items are shown diagrammatically on construction documents. Contractor shall review existing plans to confirm configuration and scope to facilitate project.
- 7. The Contract Documents show and describe the existing conditions as they are believed to have been used in the design of the work and are only provided as information for the bidder. The Owner and Architect are not making any warranties regarding this information. The Owner and Architect shall not be liable for any loss sustained by the successful bidder resulting from any variance between the conditions and design data given in the Contract Documents and the actual conditions revealed during the bidder's pre-bid examination or during the progress of the work. Bidder agrees that the submission of a bid shall be incontrovertible evidence that the bidder has complied with and agrees to further comply with all the requirements of this section.

### 8. Allowances:

- a. Selected materials, services and equipment, and in some cases, their installation are shown and specified in the Contract Documents by allowances herein. Allowances have been established in lieu of additional requirements and to defer selection of actual materials, miscellaneous additional work scope and equipment to a later date when additional information is available for evaluation.
- b. Special allowances have been established for unforeseen conditions, latent conditions and related items to be authorized by the architect for use.
- c. The contractor shall include in his base bid all overhead, profit, supervision, bonds, insurance and all other indirect costs for allowance items. No costs for overhead, profit, supervision, bonds, insurance and all other indirect costs will be added to lump sum and miscellaneous allowance as it is used by the Owner and directed by the Architect.
  - In the event the allowance is required in an Alternate, the contractor shall include in his alternate bid all overhead, profit, supervision, bonds, insurance and all other indirect costs for allowance items specific to that alternate.
- d. Where use of the allowances is for additional scope where the basis of payment is Time and Material, the overhead, profit, supervision, insurance and related indirect costs shall be included in the contractor's base bid.
  - 1) In the event the allowance is in and Alternate, the same costs shall be included in the costs of the alternate as bid to the Owner.

### C. INTERPRETATION OF DOCUMENTS:

Should a bidder find discrepancies in, and/or omissions from the drawings and specifications, and/or should he be in doubt as to their meaning, he shall at once notify the Architect and should it be found necessary, a written addendum or clarification will be sent to all bidders. The Architect will not be responsible for oral instructions.

- Questions during bidding shall be submitted in writing to Architect's office. Use Pre-Bid Request For Information Form at the end of this section. Fax copies will be accepted, but Architect is not responsible for incomplete or missed transmissions. Upon receipt of Pre-Bid Request For Information, Architect shall contact sender with a response.
- 2. No questions will be answered (3) working days prior to bid opening.
- 3. E-mail questions will not be accepted.

### D. BID PROPOSALS:

Bids to receive consideration shall be made in accordance with the following instructions:

1. Bids shall be made upon the bid forms (provided by the Architect in the Project Manual – (acceptable to make copies), properly executed and with all items filled out; numbers shall be stated both in writing and in figures. In the event of inconsistency between words and

- figures in the bid, figures shall control words. The complete bid form shall be without alterations to content, form and project scope; and the signatures of all persons signing shall be in longhand and original wet signature.
- 2. Alternate proposals will not be considered unless called for. No oral, telegraphic or telephonic proposals or modifications will be considered.
- 3. Before submitting proposals for this work, including RFP's, each bidder will be held to have examined the project premises and satisfied himself as to the existing conditions under which he will be obliged to operate, and that no changes will be made subsequently in this connection or in behalf of the Contractor for any error or negligence on his part, and he shall include in the bid a sum to cover the cost of all items included in the contract and/or subsequent RFP's. No additional cost will be considered for price increases of any materials, labors, methods and/or procedures. The Contractor shall make allowances for any and all price changes occurring during this project from bidding through final completion and project acceptance by Owner.
  - a. Each bidder shall make such surveys and investigations, including investigation of subsurface or latent physical conditions at the site or where work is to be performed, as it may deem necessary for performance of the work at the price being bid. Each bidder shall determine the character, quality, and quantities of the work to be performed and the materials and equipment to be provided, and shall correlate its observations, investigations, and determinations with all requirements of the project.
  - b. The Contract Documents show and describe the existing conditions as they are believed to have been used in the design of the work and are only provided as information for the bidder. The Owner is not making any warranties regarding this information. The Owner shall not be liable for any loss sustained by the successful bidder resulting from any variance between the conditions and design data given in the Contract Documents and the actual conditions revealed during the bidder's prebid examination or during the progress of the work. Bidder agrees that the submission of a bid shall be incontrovertible evidence that the bidder has complied with and agrees to further comply with all the requirements of this section.
- 4. Bids will be delivered to the Owner at locations and time noted on "Bid Proposal" on or before the day and hour set for the opening of bids. Bid forms (provided by the Architect), shall be enclosed in a envelope, and bear the title of work and the name of the bidder. It is the sole responsibility of the bidder to see that his bid is received in proper time. Any bid received after the schedule closing time for receipt of bids will be returned to the bidder unopened, in accordance with Government Code Section 53068.
- 5. Each bidder shall include with Proposal a Non-Collusion Affidavit, Prevailing Wage Compliance Certificate, required bond documents and Certification of Workmen's Compensation to be executed by bidder and submitted with bid. Form is included in "Bid Proposal" section. Form shall be fully executed and included with Bid Proposal.
  - a. Anti-Discrimination: The successful bidder will be required to sign a contract, which contains the following provisions prohibiting discrimination:
    - 1) "Anti-Discrimination in Employment: In connection with the performance of work under this Contract, the Contractor agrees (as prescribed in Chapter 6 of Division 3 of Title II of the Government Code of the State of California, commencing at Section 12900 and by Labor Code Section 1735) not to discriminate against any employee or applicant for employment because of race, religious creed, color, national origin, ancestry, physical disability, mental disability, medical condition, marital status, gender, or age. The aforesaid provisions shall include, but are limited to, the following: hiring, upgrading, demotion or transfer, recruitment or recruitment advertising, layoff or termination, rates of pay or other forms compensation, and selection for training, including apprenticeship. The Contractor agrees to post hereafter in conspicuous places, available for employees and applicants for employment,

notices setting forth the provisions of this non-discrimination clause. The Contractor further agrees to insert the foregoing provisions in all subcontracts entered into hereunder, except subcontracts for standard commercial supplies of raw materials."

- 6. Each bidder shall submit a list of the proposed subcontractors on the project as required by the Subletting and Subcontracting Fair Practices Act (Public Contract Code Section 4100 and following sections) on the form furnished with the Contract Documents.
  - a. If alternate bids are called for and the bidder intends to use different or additional subcontractors, a separate list of subcontractors must be submitted for each such alternate bid. The Owner may request that bidder submit information to assess the responsibility of the bidder's proposed subcontractors. The apparent low bidder shall, within 24 hours of the bid opening, provide a complete listing of all subcontractors, including full name, address, telephone numbers and contractor's license number and type.

### 7. General Information:

- a. The District reserves the right to reject any or all bids and/or waive any irregularities or informalities in any bids and/or in the bidding process.
- b. The District has determined the general prevailing rate of per diem wages in the locality in which this work is to be performed for each craft or type of worker needed to execute this Contract. These rates are on file on the Internet at the following address: <a href="www.dir.ca.gov/DLSR/statistics research.html">www.dir.ca.gov/DLSR/statistics research.html</a>. Copies may be downloaded by the Contractor. A copy of these rates shall be posted at the job site by the Contractor. Refer to section Prevailing Wage Rates & Apprenticeship Requirements for additional requirements. District will be monitoring prevailing wages on this project.
- c. The schedule of per diem wages is based upon a working day of eight hours. The rate for holiday and overtime work shall be at least time and one half.
- d. It shall be mandatory upon the Contractor to whom the Contract is awarded (Contractor), and upon any subcontractor under him, to pay not less than the specified rates to all workers employed by them in the execution of the Contract. It is Contractor's responsibility to determine any rate change which may have or will occur during the intervening period between each issuance of written rates by the Director of Industrial Relations.
- e. A Payment Bond and a Performance Bond will be required prior to execution of the Contract refer to Section "Bonds and Certificates Required" for additional requirements. The executed Bonds shall be on the forms herein Project Manual.

### E. WITHDRAWAL OF BID PROPOSAL:

Any bid may be withdrawn, either personally, by written request, or by telegraphic or facsimile request confirmed in the manner specified herein for bid modifications, at any time prior to the scheduled closing time for receipt of bids. In accordance with this paragraph, the bid security shall be returned for bids withdrawn prior to the scheduled closing time for receipt of bids. No bidder may withdraw any bid for a period indicated in the project manual after the award of the contract.

 A bidder's unawarded alternative bids remain open for a period of 120 calendar days after award of contract or acceptance of completed project, whichever come first, as irrevocable offers to enter into either change orders or separate contracts for the stated price adjustment.

### F. CORPORATION AS BIDDER:

In case a bid is submitted by a corporation, it shall be signed in the name of such corporation by a dully authorized officer or agent thereof.

### G. SALES TAXES:

Sales taxes and any or all taxes and any other City, County, State, or Federal, except property taxes shall be included in the bid. All bids shall include all license fees, permit fees, and other fees to complete this project. See herein for permits, inspections, and assessments required for this project.

### H. ADHERENCE:

No bid will be considered that does not strictly adhere to all requirements of these instructions to Bidders. The District reserves the right to waive any irregularities or informalities in any bids or in the bidding process.

### COMPETENCY OF BIDDERS

In selecting the lowest responsive and responsible bidder, consideration will be given not only to the financial standing but also to the general competency of the bidder for performance of the work. By submitting a bid, each bidder agrees that in determining the successful bidder and its eligibility for the award, the Owner may consider the bidder's experience, facilities, conduct, and performance under other contracts, financial condition, reputation in the industry, and other factors relating to or which could affect the bidder's performance of the project.

### J. BIDS TO BE ACCEPTED BY OWNER:

The lowest bid shall be the lowest total of the bid prices on the base contract and all alternates.

- 1. After the lowest bidder is determined, the District will select which alternates to award based on which are most advantageous to the District.
- 2. A bidder's un-awarded alternative bids remain open for a period of up to 120 calendar days or project duration whichever is less, after date of award of contract as irrevocable offers to enter into either change orders or separate contracts for the stated price adjustment.

### K. OPENING OF BIDS:

Bids will be opened and publicly read aloud at the time and place set in the Notice to Contractors. The Owner may require such bidder or representative to affirmatively state that he knows the contents of his own bid and that he then believes that such bid is complete and correct in all particulars.

At the time set for the opening of bids, the sealed bids will be opened and publicly read aloud at the place indicated in the Notice to Contractors Calling for Bids. However, if this project calls for prequalification of bidders pursuant to Public Contract Code Section 20111.5, only those sealed bids received from bidders who have been prequalified when required herein, for at least one day prior to bid opening shall be opened and publicly read aloud.

### L. AWARD OR REJECTION OF BIDS:

The contract shall be awarded to the responsible bidder complying with these instructions. The Owner reserves the right to reject any and all bids and to waive any informality or irregularity in any bids or bidding process. The award, if made, will be made within one hundred twenty (120) calendar days after the opening of the bids.

- 1. If two identical low bids are received from responsive and responsible bidders, the Owner will determine which bid will be accepted pursuant to Public Contract Code Section 20117.
- 2. If made by the Owner, award of the contract will be by action of the governing board or other governing body to the lowest responsive and responsible bidder. In the event an award of the contract is made to a bidder and that bidder fails or refuses to execute the Agreement and provide the required documents within the time required, the Owner may award the

contract to the next lowest responsive and responsible bidder or release all bidders. An election by the Owner to reject all bids does not release the bid security of any bidder who has previously been awarded the contract and failed or refused to execute the Agreement and provide the required documents.

### M. EXAMINATION OF SITE:

The Bidder shall carefully examine the site of the contemplated work prior to submitting a bid proposal and shall have satisfied himself as to the existing conditions and the conditions under which he will be obligated to operate, and/or that will in any manner affect the work under the contract. No allowance will be made subsequently in this connection for items that could be reasonably be inferred to be required to complete project scope from a careful examination of site of the contemplated work as well as plans of existing facilities. Refer to item (B) above for additional requirements.

### N. ADDENDA AND BULLETINS:

Any addenda or bulletin items issued during the time of bidding shall be an integral part of the Contract Documents used by the Bidder for the preparation of his bid, all items of addenda and/or bulletins shall be included in the Bid Proposal and shall be made part of the Contract. Delivery of any Addenda or Bulletin in person to the individual or to a member of the firm or to an officer of the corporation for whom it is intended, delivery by mail to the last known business address of the Contractor, or fax transmittal with telephone confirmation of complete receipt will be considered to be proper service of said documents.

### O. PERMITS, INSPECTIONS SPECIAL REPORTS AND ASSESSMENTS, ETC.:

The Contractor, under the Agreement to be executed for this Contract, shall obtain and pay for all permits required for the complete execution of the work and for all inspections that are required by Federal, State or local (City or County) laws, special District requirements and/or ordinances. The Contractor shall be responsible for determining the necessity for said Permits and Inspections.

### 1. <u>List of Permits required</u>:

a. None.

### 2. List of Fees/Assessments:

a. None.

### P. FORMAL PROTEST OF BID PROPOSAL:

Any bidder having submitted a bid on the project or third party may file a protest against the proposed contract award or challenging the validity of other bids. The protest must meet all of the following requirements:

- 1. The protest shall be submitted in writing and shall contain all the materials required by these provisions; one that does not contain all the required material shall not be recognized.
- 2. The protest shall be received by the Owner no later than close of business on the second business day after bid opening; protest(s) received after that time shall not be recognized.
- 3. Each protest shall contain the following:
  - a. Identification by name, address, and telephone number of the protesting person(s), company and/or organization and identification of the project to which the protest pertains.
  - b. The protest shall set forth in detail all grounds for the protest, including without limitation all facts, identification by name of any other bids or bidders involved in the

protest, all supporting documentation, together with any legal authorities and/or argument in support of the grounds for the protest. Any matters not set forth in the written protest shall be deemed waived. All factual contentions must be supported by competent, admissible, and credible evidence.

- 4. Any protest not conforming to the requirements of this section shall be rejected as invalid.
- 5. Where a protest is filed in conformity with this section, the Owner's staff or such individual(s) as may be designated by the Owner shall review and evaluate the basis of the protest and provide a written decision to the protesting bidder. The written decision shall either concur with or deny the protest.
- 6. Submission of a written protest to and receipt of a written decision from the Owner staff shall be considered an administrative remedy, and failure to follow this procedure shall be a bar to any legal action.
- 7. The written decision by the Owner's staff may be appealed to the Owner. The appeal must be filed with the Owner's governing board within two business days of the protesting bidder's receipt of the written decision of the Owner's staff.
- 8. The appeal must clearly state the reasons and basis for appealing the decision of the Owner's staff, making specific reference to any portions of the material submitted with the protest required.
- 9. A hearing on the appeal shall be held before the Owner's governing board within 45 days or less of receipt of the appeal.
- 10. The Owner's governing board or other governing body will make a decision within seven days following the hearing. The decision of the Owner's governing board or other governing body is not subject to arbitration, mediation, reconsideration, or further appeal.
- 11. Submission of an appeal to and receipt of a decision from the Owner's governing board or other governing body shall be considered an administrative remedy, and failure to follow this procedure shall be a bar to any legal action.

### Q. PROCEDURE FOR PROTESTING BEING DEEMED A NON-RESPONSIVE BIDDER.

Any bidder or prospective bidder deemed non-responsible after having submitted a bid may file an appeal of the action to the Owner's governing board or other governing body. The protest must meet all of the following requirements:

- 1. The appeal shall be submitted in writing, and shall contain all the materials required by these provisions; one that does not contain all the required material shall not be recognized.
- 2. The appeal must be received by the Owner's governing board or other governing body within two business days of the action by Owner giving rise to the protest; one received after that time shall not be recognized.
- 3. A hearing on the appeal shall be held before the Owner's governing board prior to the award of contract.
- 4. The decision of the Owner's governing board or other governing body is not subject to arbitration, mediation, reconsideration, or further appeal.
- 5. Submission of a protest to and receipt of a decision from the Owner's governing board or other governing body shall be considered an administrative remedy, and failure to follow this procedure shall be a bar to any legal action.
- R. PRE-AWARD MEETINGS: The Architect may conduct pre-award meetings to discuss the project with various bidding Contractors, listed Subcontractors, Surety and Owner. Meetings will be held in the Architect's office.
- S. SCHEDULE OF VALUE: The Prime Contractor shall provide a draft detailed cost breakdown two (2) calendar days after time of bid opening. Failure to provide this complete information shall be grounds of bidder being declared non-responsive.

Breakdown cost by at least each Project Manual section and as indicated herein (refer to 006002-9.2.1.1 for minimum list of items).

### T. STARTING WORK:

g.

 Pre-Award Documents: All documents shall be submitted and approved prior to starting work. Failure to provide complete information prior days indicated shall be considered nonresponsive and bid bond will be forfeited, and Contractor shall be declared in default. (All days below are calendar days).

| a. | Performance bond and data -  | 10 days |
|----|------------------------------|---------|
| b. | Payment bond and data -      | 10 days |
| C. | All insurance certificates - | 10 days |

d. Schedule of Values final - 2 days prior to starting work

e. Proof of Contractor's/Subcontractor's license - 10 days

f. DVBE, MBE and WBE listing of Contractors/

Subcontractors by dollar volume for

this project - 10 days
Drug-free workplace certification 10 days

- 2. Notice of Award: If awarded, the Owner will prepare a Notice of Award. The Contractor shall provide all the above documents no later than 10 calendar days from the Notice of Award preparation date.
- Contract: If awarded and Contractor provides the required documents as prescribed herein
  and documents are accepted and approved by the Owner, the Owner will prepare a contract
  for execution (copy enclosed herein). The Contractor shall execute and return the contract
  on or before 7 calendar days of receipt of contract.
- 4. Notice to Proceed: The Owner shall prepare and deliver to the Contractor a Notice to Proceed. The Contractor shall commence the work within fifteen (15) calendar days after the Notice to Proceed document date, or by special arrangements with Architect.
  - Subsequent to receipt of Notice to Proceed, no work may start in site until receipt of SJVAPCD approval of Construction Notification Form, Fingerprint Clearances, Dust Control Plan (if required), permits, Hazardous Material Notifications and Storm Water Prevention Plan Approval.
  - b. Contractor shall use best endeavors to submit items to required agencies.

### PART 2 - PERFORMANCE OF WORK UNDER CONTRACT

### A. SUPERVISION:

- 1. The General Contractor and all subcontractors engaged by general contractor will be required to designate one responsible on-site person with authority to receive directions and issue instructions for the orderly prosecution of the work.
  - a. The General Contractor shall designate a project superintendent, who shall not carry tools, unless approved by Architect, prior to project starting.

### B. LIQUIDATED DAMAGES:

Should the contractor fail to complete this contract within the time limits fixed for such completion, or within the time limits as may be extended as provided elsewhere in this Project Manual, damages will be sustained by the Owner. It is hereby understood and agreed that it is and will be impracticable or extremely difficult to determine the actual amount of damages which the Owner will sustain in the event of any by reason of such delays, and it is therefore agreed that the Contractor will pay to the Owner the sum stipulated below for each and every delay beyond the time limit specified or as may be extended, as and for liquidation damages; and in case the Contractor fails to make such payment, the Owner may deduct the amount thereof from any money due or that may become due the Contractor under the contact. Refer to Division Section "Contract Agreement" for amount per

calendar day.

### C. QUALIFICATIONS OF BIDDERS AND PROOF OF LICENSE:

Plans, Project Manual and Bid Proposal forms will only be issued to those Contractors who have the following qualifications:

- 1. Valid State Contractors license of the type required to perform the work of this Contract, and
- 2. Written proof of ability to provide the required Bonds and Insurance Certificate.
- A generally recognized record for satisfactory execution of Contracts of similar sizes and character.
- 4. Contractor's License: Prior to award of this contract for work to be performed by the Contractor, as defined by Section 7026 of the Business and Professions Code, the Contractor shall prove he or she is licensed in a classification appropriate to the work to be undertaken. Contractor shall present his or here pocket license or certificate of licensure and provide a signed statement which swears, under penalty of perjury, that the pocket license or certificate of licensure presented is his or hers, is current and valid, and is in a classifications appropriate to the work to be undertaken.
  - Within 10 days of award of contract, Contractor shall provide same information for all subcontractors who will be working on or at this project. First Pay Request will not be approved until this is completed and documented accepted by the Architect.
- 5. "Pursuant to the requirements of SB 1362 and California Labor Code section 3099.2, all employees performing electrical work for a subcontractor holding a C-10 license must be certified. If employees working on project are found to be not certified, they shall be immediately removed. Failure to provide proof of this documentation on all employees will be considered a violation and subject the subcontractor to corrective action up to and including being removed from the project."

### D. BUILDING CODE REQUIREMENTS:

- 1. All work performed under this Contract shall conform with the applicable portions and editions of the following codes:
  - a. California Code of Regulations--CCR, T24 Parts 1-9.
  - b. Public Health Code of the California State Department of Public Health and Local Health Department.
  - c. California Occupational Safety and Health Act (CAL/OSHA).
  - d. Rules and regulations of the State and Local Fire Marshals.
  - e. Safety Orders of the Industrial Accident Commission, State of California.
  - f. California Electric Code.
  - g. California Plumbing Code.
  - h. All laws governing the employment of labor, posting of minimum wage rates, and accident prevention.
  - i. Refer to Section 014300 Governing Agency for additional items.
  - j. American Disability Act, Federal law.
- 2. Requirements of enforcing authorities may supersede requirements of the above laws and regulations, and nothing in the Contract Documents shall be construed to permit work not conforming to applicable codes.
- 3. All of the above laws and regulations, are as much a part of this contract as if they were incorporated in their entirety herein.

### E. BID, PERFORMANCE AND PAYMENT BONDS:

General Contractor shall take out and maintain Bid, Performance and Payment bonds as indicated in Section 004313 "Bonds and Certificates Required."

### F. WORKER'S COMPENSATION INSURANCE:

The Contractor shall take out and maintain during the life of this Contract, Worker's Compensation Insurance for all of his employees employed on the project and he shall require all subcontractors similarly to provide Worker's Compensation Insurance. Refer to Section 004313 "Bonds and Certificates Required".

### G. LIABILITY INSURANCE:

The Contractor shall take out and maintain at all times during the performance of the work under this Contract, through companies and Agencies approved by the Owner, and containing provisions satisfactory to the Owner, the following insurance (refer to Section 004313 "Bonds and Certificates Required"):

- 1. Public Liability Insurance.
- 2. Automobile Liability Insurance.
- 3. Contractual Liability Insurance.

### H. "ALL RISK" INSURANCE:

The Contractor shall purchase and maintain "All Risk" property insurance upon the entire Work at the site. Refer to Section 004313 "Bonds and Certificates Required" for type required by this project.

### I. CERTIFICATES OF INSURANCE:

Certificates of Insurance acceptable to the Owner shall be filed with the Owner prior to commencement of the Work. These Certificates shall contain a provision that coverages afforded under the policies will not be canceled until at least thirty (30) days' prior written notice has been given to the Owner. Refer to Section 004313 "Bonds and Certificates Required".

## J. ASSIGNMENT OF ANTITRUST AND UNFAIR BUSINESS PRACTICE CLAIMS IN PUBLIC WORKS CONTRACTS:

In accordance with Section 7103.5(b) of the Public Contract Code, the Contractor and subcontractor shall conform to the following requirements. In entering into public works contract or subcontract to supply goods, services, or materials pursuant to a public works contract, the Contractor or subcontractor offers and agrees to assign to the awarding body all rights, title and interest in and to all causes of action it may have under Section 4 of the Clayton Act (15 U.S.C. Sec. 15) or under the Cartwright Act (Chapter 2(commencing with Section 16700) of Part 2 of Division 7 of the Business and Professions Code), arising from purchases of goods, services, or materials pursuant to the public works contract or the subcontract. This assignment shall be made and become effective at the time the awarding body tenders final payment to the Contractor, without further acknowledgement by the parties.

## K. ASSIGNMENT OF ANTITRUST AND UNFAIR BUSINESS PRACTICE CLAIMS IN PUBLIC PURCHASING CONTRACTS:

In accordance with Section 4552 of the Government Code, the bidder shall conform to the following requirements. In submitting a bid to a public purchasing body, the bidder offers and agrees that if the bid is accepted, it will assign to the purchasing body all rights, title, and interest in and to all causes of action it may have under Section 4 of the Clayton Act (15 U.S.C. Sec. 15) or under the Cartwright Act (Chapter 2(commencing with Section 16700) of Part 2 of Division 7 of the Business and Professions Code), arising from purchases of goods, services, or materials by the bidder for sale to the purchasing body pursuant to the bid. Such assignment shall be made and become effective at the time the purchasing body tenders final payment to the bidder.

### L. FINGERPRINTING REQUIREMENTS:

 Contractors and their employees shall comply with the Michelle Montoya School Safety Act, Section 45125.1 of the Education Code. Refer to Section 004317 – Fingerprinting Requirements.

### M. DVME, MBE, WBE REPORTING REQUIRMENTS:

The California Community Colleges requires each District to annually report Disabled Veteran Business Enterprises, Minority Business Enterprises and Women Business Enterprises that receive construction contracts either as a prime contractor, sub contractor, sub-sub contractor or material provider.. Each successful bidder shall provide a listing of DVBE, MBE and WBE contracts that are included in bid as part of Pre-Award Documents. List Name of Enterprise, type of contract, contact person, dollar amount and percentage of total contract to be awarded. (California Community Colleges have no specific goal or requirement of dollar amount for any specific project.)

### M. DISABLED VETERANS BUSINESS ENTERPRISE CONTRACTING GOALS

To the extent required by law, the Contractor shall meet requirements of law relating to the participation of Disabled Veterans Business Enterprises Contracting Goals, pursuant to AB 1633, and shall comply with Public Contract Code Section 10115 et seq., Education Code Section 17076.11 and all applicable regulations. Contractor further agrees that, when required, Contractor will endure compliance by all subcontractors and will complete all forms required by the Owner, the Office of Public School Construction, the State Allocation Board or other agencies exercising jurisdiction over the project. The Owner will reject all bids not complying with the provisions. All Contractors shall complete and submit with their bids, the Declaration of Good Faith Efforts to use Disabled Veteran Business Enterprises (DVBE)" form as included in bid form.

### N. IMMIGRATION REFORM AND CONTROL ACT OF 1986:

The bidder hereby certifies that it is, and at all times during the performance of work hereunder shall be, in full compliance with the provisions of the federal Immigration Reform and Control Act of 1986 ("IRCA") in the hiring of its employees, and the bidder shall indemnify, hold harmless, and defend the Owner against and all actions, proceedings, penalties or claims arising out of the bidder's failure to comply strictly with the IRCA.

### O. DRUG FREE WORKPLACE CERTIFICATION:

Pursuant to Government Code Section 8350 and following sections, the successful bidder will be required to execute and return to Owner the Drug-Free Workplace Certificate contained in the Contract Documents with the executed Construction Agreement. The bidder will be required to take positive measures outlined in the certificate to ensure the presence of a drug-free workplace. Failure to abide with the conditions set forth in the Drug-Free Workplace Act could result in penalties, including termination of the Construction Agreement or suspension of payment under the Construction Agreement.

### P. ASBESTOS-CONTAINING PRODUCTS:

- 1. Contractor agrees that asbestos-containing products or materials will not be used or substituted in performing work under the Agreement.
- 2. At the completion of work under this Agreement, Contractor will certify in writing to the Owner that, to the best of Contractor's knowledge, no asbestos-containing products or materials were used or substituted in performing work under the Agreement.

### Q. PCB-CONTAINING PRODUCTS & LEAD PLUMBING ITEMS:

1. Contractor agrees that lead plumbing domestic water items, other lead containing products, asbestos, PCB-containing products or materials will not be used or substituted in performing

work under the Agreement.

- 2. At the completion of work under this Agreement, Contractor will certify in writing to the Owner that, to the best of Contractor's knowledge, no lead plumbing domestic water items, no other lead containing products, asbestos/PCB-containing products or materials were used or substituted in performing work under the Agreement.
- R. PRELIMINARY NOTICE, STOP NOTICE, STOP NOTICE RELEASES AND ALL OTHER LEGAL NOTIFICATIONS:

Preliminary Notices and all other legal notifications must be served with the Contractor, Owner and Architect:

Owner: West Kern Community College District Office

29 Cougar Court Taft, CA 93268

Attn.: Leslie Minor, Ph.D. Phone: (661) 763-7700

Architect: AP Architects

3434 Truxtun Avenue - Suite 240

Bakersfield, CA 93301

Attn.: Jose Vargas – Project Manager

Phone: (661) 327-1690

Contractor: - verify -

END OF SECTION 002113

## SAN JOAQUIN VALLEY AIR POLLUTION CONTROL DISTRICT CONSTRUCTION NOTIFICATION FORM

|   |                 | Project Informati     | on                                     |                                       |
|---|-----------------|-----------------------|--|---------------------------------------|
| Project Name: G   | G-8 ALLIED HEAL | ΓΗ LAB REMODEL        |  |                                       |
| Project Address:  |                 |                       |  |                                       |
| Major X-Streets:  |                 |                       |  |                                       |
| City:   | County:         |                       | Today's Date:                          |                                       |
| Expected Construction   |                 |                       | Total project site area:               |                                       |
| •   | End Date:       | <br>Total             |  | Acres                                 |
| The project is:   | Residential [   | Non-residential (comm | nercial, industrial, institutional, pu | ublic, etc.)                          |
| Construction activities on any site that will include ten acres or more of disturbed surface area for residential developments, five acres or more for non-residential developments, or involve moving, depositing or relocating more than 2,500 cubic yards per day of bulk materials on at least three days are required to submit a Dust Control Plan to the District prior to the start of any construction activity. The Construction Notification Form may not be used to circumvent any Dust Control Plan requirement. |                 |                       |  |                                       |
|   |                 | Contacts              |  |                                       |
| Property Owner:   |                 |                       |  |                                       |
| Address:  |                 |                       |  |                                       |
| City/State/Zip:   |                 |                       |  |                                       |
| Phone:  |                 | Fax:                  | Cell:                                  | <del></del>                           |
| Developer:  |                 |                       |  |                                       |
| Address:  |                 |                       |  |                                       |
| City/State/Zip:   |                 |                       |  |                                       |
| Contact Person:   |                 |                       |  |                                       |
| Phone:  |                 | Fax:                  | Cell:                                  |                                       |
| General Contractor:   |                 |                       |  |                                       |
| Address:  |                 |                       |  | <del></del>                           |
| City/State/Zip:   |                 |                       |  | <del></del>                           |
| Contact Person:   |                 |                       |  | · · · · · · · · · · · · · · · · · · · |
| Phone:  |                 | Fax:                  | Cell:                                  |                                       |
|   |                 |                       |  |                                       |
| Property Owner:   |                 |                       |  |                                       |
| Address:  |                 |                       |  | · · · · · · · · · · · · · · · · · · · |
| City/State/Zip:   |                 |                       |  | <del></del>                           |
| Contact Person:   |                 |                       |  | · · · · · · · · · · · · · · · · · · · |
| Phone:  |                 | Fax:                  | Cell:                                  |                                       |
| For District Use:   |                 |                       |  |                                       |
|   |                 |                       |  |                                       |
| Notification received or  | ):<br>          | No                    | tification accepted by:                |                                       |



## PRE-BID REQUEST FOR INFORMATION

| Send to:      | AP Architects                  | Sent by:               |                 |
|---------------|--------------------------------|------------------------|-----------------|
|               | 3434 Truxtun Avenue, Suite 240 | Address:               |                 |
|               | Bakersfield, CA 93301          |                        |                 |
| Attention:    | Jose Vargas                    |                        |                 |
| Project:      | G-8 ALLIED HEALTH LAB          | Contact Name:          |                 |
|               | REMODEL                        | Contact #:             |                 |
|               |                                |                        |                 |
| Project No:   | 430-0083                       | Fax #:                 |                 |
| DSA Appl #:   |                                | Email:                 |                 |
| DSA File #:   |                                |                        |                 |
|               |                                |                        |                 |
| Subject For ( | Clarification:                 | Project Manual Section |                 |
|               |                                | Drawing Sheet          | #:              |
| Description   |                                |                        |                 |
| Description:  |                                |                        |                 |
|               |                                |                        |                 |
|               |                                |                        |                 |
|               |                                |                        |                 |
|               |                                |                        |                 |
| -             |                                |                        |                 |
|               |                                |                        |                 |
|               |                                |                        |                 |
|               |                                |                        |                 |
|               |                                |                        |                 |
|               |                                |                        |                 |
| ARCHITE       | CT'S RESPONSE:                 | Included in Addendum # |                 |
|               |                                | PB RF                  | l #:            |
|               |                                |                        | ·               |
|               |                                |                        |                 |
|               |                                |                        |                 |
|               |                                |                        |                 |
|               |                                |                        |                 |
|               |                                |                        |                 |
|               |                                |                        |                 |
|               |                                |                        |                 |
|               |                                |                        |                 |
|               |                                | Attachmer              | nts:            |
|               |                                | Da                     | ate:            |
|               |                                |                        | Bv <sup>.</sup> |

via

tha

received

Ride

\A/ill

hΦ

### **BID PROPOSAL**

\/endor

Portal

hid

College

Taft

02/29/16

located

noeting

| Submit  |  | Submitted by:   |
|---|--|---|
|   | of Trustees  |   |
| West Kern Community College District          | Name of Firm   |   |
|   | , California   |   |
| Board   | Members:   |   |
| Having  | ງ carefully examined the Advertisement fo  | or Bids, Instructions to Bidders, General Conditions to the   |
| Contra ALLIE the co                           | ct, Supplementary Conditions, Special Co D HEALTH LAB REMODEL – Taft Collegenditions affecting the work, including Accepted proposes to furnish all material and ance with said documents for the sum of:  | or Bids, instructions to Bidders, General Conditions to the inditions, Specifications and Drawings entitled - <b>TC – G-8</b> ge - located in Taft, California, as well as the premises and iddendum (a) No (s), inclusive, the displayed labor called for by all documents for the "entire work", in |
| Contra ALLIE the co Unders accord             | ct, Supplementary Conditions, Special Co D HEALTH LAB REMODEL – Taft Collegenditions affecting the work, including Accepted proposes to furnish all material and ance with said documents for the sum of:  | nditions, Specifications and Drawings entitled - <b>TC – G-8</b> ge - located in Taft, California, as well as the premises and ddendum (a) No (s), inclusive, the   |
| Contra<br>ALLIE<br>the co<br>Unders<br>accord | ct, Supplementary Conditions, Special Co D HEALTH LAB REMODEL – Taft Collegenditions affecting the work, including Accepted proposes to furnish all material and ance with said documents for the sum of:  | nditions, Specifications and Drawings entitled - <b>TC – G-8</b> ge - located in Taft, California, as well as the premises and ddendum (a) No (s), inclusive, the   |
| Contra ALLIE the co Unders accord             | ct, Supplementary Conditions, Special Co D HEALTH LAB REMODEL – Taft College and the Mark, including Acts and proposes to furnish all material and ance with said documents for the sum of:  BID  Base Bid amount [written in words]  \$10,000.00 Owner's allowance (to be use | ed at Owner's discretion)  inditions, Specifications and Drawings entitled - TC - G-8  ge - located in Taft, California, as well as the premises and dendum (a) No (s), inclusive, the dilabor called for by all documents for the "entire work", in  |

The undersigned understands that all documents required prior to starting work shall be provided no later than 10 days from the Notice of Award preparation date.

The undersigned understands that the time required to complete the work is the essence of the Contract and agrees to commence the work within fifteen (15) calendar days of the Notice to Proceed date, unless noted otherwise. The undersigned further agrees that this bid may not be withdrawn for a period of one hundred twenty (120) days after the date set for the opening thereof unless otherwise required by law.

The undersigned agrees, if awarded the Contract, to complete it within  $\underline{90}$  calendar days plus any extensions of time, as provided for in the General Conditions of the Contract; failing to complete the work within the above stipulated time, he agrees to be bound by the conditions as set forth in the Supplementary Conditions, Article 17, and Instructions to Bidders - Section 002113.

The undersigned has checked carefully all the above figures and understands that the Owner will not be responsible for any errors or omissions on the part of the undersigned in making up this bid.

The undersigned confirms there are no clerical errors in preparation of this bid proposal.

The undersigned hereby certifies that this bid is genuine and not sham or collusive or made in the interest or in behalf of any person not herein named, and that the undersigned has not directly or indirectly induced or solicited any other bidder to refrain form bidding, and that the undersigned has not in any manner sought by

collusion to secure for himself any advantage over any other bidder.

Enclosed find ( ) Bid Bond ( ) Certified Check ( ) Cashier's Check for 10% of the amount bid.

### SUBCONTRACTOR LIST:

Pursuant to the Provisions of the Public Contracts Code Sections 4100 to 4107 inclusive, every bidder shall set forth the name and location of the place of business of each subcontractor who will perform work or labor in or about the construction of the work or improvement in an amount in excess of one-half of one percent (1/2 of 1%) of the Bidder's total bid. If a Contractor is not listed and the work is more than one-half of one percent (1/2 of 1%) of the Bidder's total bid, he agrees to perform that portion himself. The following is the list of subcontractors:

| PORTION<br>OF WORK | SUBCONTRACTOR | <u>DIR</u><br>REGISTRATION<br><u>NUMBER</u> | LOCATION<br>OF<br>BUSINESS |
|--------------------|---------------|---|----------------------------|
|                    |               |   |                            |
|                    |               |   |                            |
|                    |               |   |                            |
|                    |               |   |                            |
|                    |               |   |                            |
|                    |               |   |                            |
|                    |               |   |                            |
|                    |               |   |                            |
|                    |               |   |                            |
|                    |               |   |                            |
|                    |               |   |                            |
|                    |               |   |                            |
|                    |               |   |                            |
|                    |               |   |                            |
|                    |               |   |                            |
|                    |               |   |                            |
|                    |               |   |                            |
|                    |               |   |                            |
|                    |               |   |                            |
|                    |               |   |                            |
|                    |               |   |                            |
|                    |               |   |                            |
|                    |               |   |                            |

|  |                              | _        |
|--|------------------------------|----------|
|  |                              |          |
|  |                              |          |
|  |                              |          |
| I declare, under penalty of perjury, that information and correct and that this declaration was executed |                              |          |
| and correct and that this declaration was executed , California.   | (date)                       | (city)   |
| (county)   |                              |          |
|  |                              |          |
| Respectfully submitted,  | Corporate Se                 | al       |
|  | If Applicable                | aı       |
| Name of Firm   |                              |          |
| Individual, Partnership, Corp.**   | Phone()                      |          |
| · ·  | Fax()                        |          |
| Ву   |                              |          |
| Address  | License Type & Number/Ex     | кр. date |
|  | -                            |          |
|  | DIR Registration # & Exp. of | date     |

### No bid is valid unless signed by the person making the bid.

TC – G-8 ALLIED HEALTH LAB REMODEL

<sup>\*\*</sup> State whether your firm is a corporation, a co-partnership, private individual, or individuals, doing business under a firm name. If the bidder is a partnership, the bid should be signed with the partnership name and by one of the authorized partners. If the bidder is a corporation, it should be signed by a person authorized to execute bids on behalf of the corporation.

Dated:

### **BID BOND** Be advised that we, \_\_\_\_\_ as Principal ("Principal") and \_\_\_\_\_ a corporation duly licensed to transact business under the laws of the State of California as Surety ("Surety") are firmly bound to \_\_\_\_\_\_ as Obligee ("Obligee") in the sum \_\_\_\_\_ for the payment of which the Principal and the Surety bind ourselves, our heirs, executors, administrators, successors, and assigns, jointly and severally, by this Bond. The Principal has submitted a bid for \_\_\_\_\_\_. The condition of our obligation is this: if the Principal is awarded the contract upon its proposal, and shall, within the required number of days after the notice of award, execute a contract with the Obligee in accordance with the contract documents, submit the required payment and performance bonds, and provide all other required documents, then this obligation shall be null and void; but in the event that the Principal fails and/or refuses to execute and deliver those documents, this bond will be charged with the costs of the damages experienced by the Obligee as a result of that refusal, including but not limited to, publication costs, the difference in money between the amount of the bid of the Principal and the amount for which the Obligee may legally contract with another party to perform the work if the amount is in excess of the former; building lease or rental costs, transportation costs, professional service costs, and additional salary costs that result from the delay due to the Principal's default on the awarded contract. In no event, however, shall the Surety's liability exceed the penal sum indicated above. The Surety, for value received, stipulates and agrees that its obligations and its bond shall not be impaired or affected by an extension of the time within which the Obligee may accept such bid; and Surety waives notice of any time extension. Dated: **PRINCIPAL**

BID PROPOSAL 004113 - 4

SURETY

### NON-COLLUSION AFFIDAVIT

### TC - G-8 ALLIED HEALTH LAB REMODEL

To: WEST KERN COMMUNITY COLLEGE DISTRICT 29 Cougar Court Taft, CA 93268 State of California County of , being duly sworn, deposes and says: That he or she is the \_\_\_\_\_ (position) of (name of bidder), the party making the bid; that the bid is not made in the interest of, or on behalf of any undisclosed person, partnership, company, association, organization or corporation; that the bid is genuine and not collusive or sham; that the bidder has not directly or indirectly induced or solicited any other bidder to put in a false or sham bid, and has not directly or indirectly colluded, conspired, connived, or agreed with any bidder or anyone else to put in a sham bid, or that anyone shall refrain from bidding; that the bidder has not in any manner, directly or indirectly, sought by agreement, communication, or conference with anyone to fix the bid price of the bidder or any other bidder, or to fix any overhead, profit, or cost element of the bid price, or of that of any other bidder, or to secure any advantage against the public body awarding the contract of anyone interested in the proposed contract; that all statements contained in the bid are true; and, further, that the bidder has not, directly or indirectly, submitted his or her bid price or any price breakdown, or their contents, or divulged relative information or data, or paid, and will not pay, any fee to any corporation, partnership, company, association, organization, bid depository, or to any member or agent thereof to effectuate a collusive or sham bid. (Firm Name) (Printed Name - Authorized Agent) (Signature - Authorized Agent) Subscribed and sworn to before me on \_\_\_\_\_\_, 20 \_\_\_\_\_.

NOTARY SEAL

BID PROPOSAL 004113 - 5

**Notary Public** 

## PUBLIC WORKS CONTRACTOR REGISTRATION LAW (SB-854) AND PREVAILING WAGE COMPLIANCE CERTIFICATION:

### TC - G-8 ALLIED HEALTH LAB REMODEL

To: WEST KERN COMMUNITY COLLEGE DISTRICT 29 Cougar Court Taft, CA 93268

In submitting this proposal, I hereby certify that I will conform to the State of California Public Works Contract Requirements regarding wages; benefits; on-site audits with 48-hour notice; payroll records; registration with the DIR per SB-854 guidelines and, apprentice and trainee employment requirements. I will submit CPR's online electronically as prescribed by the Labor Commissioner.

| <br>Contractor                 | (type or print |
|--------------------------------|----------------|
|                                |                |
| <br><br>Contractor's signature |                |
|                                |                |
| <br><br>Dated                  |                |

## CONTRACTOR'S CERTIFICATE REGARDING WORKER'S COMPENSATION

### TC - G-8 ALLIED HEALTH LAB REMODEL

To: WEST KERN COMMUNITY COLLEGE DISTRICT 29 Cougar Court Taft, CA 93268

### Labor Code Section 3700 provides:

"Every employer except the State and all political subdivision or institutions thereof, shall secure the payment of compensation in one or more of the following ways:

- "(a) By being insured against liability to pay compensation in one or more insurers duly authorized to write compensation insurance in this State.
- "(b) By securing from the Director of Industrial Relations a certificate of consent to self-insure, which may be given upon furnishing proof satisfactory to the Director of Industrial Relations of ability to self-insure and to pay any compensation that may become due to his employees."

I am aware of the provisions of Section 3700 of the Labor Code which require every employer to be insured against liability for Workers Compensation or to undertake self-insurance in accordance with the provisions of that Code, and I will comply with those provisions before commencing the performance of the work of this contract.

| Dated:   | CONTRACTOR |
|--|------------|
|  | By:        |
|  | Title:     |
| (In accordance with Article 5 [commencing at Section 186 this certificate must be signed and filed with the award contract.) | •          |
| END OF SECTION 004113  |            |

### PART 1 - GENERAL

### 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and other Division 0 & 1 Specification Sections, apply to this Section.

### 1.2 SUMMARY

- A. This Section includes, but is not limited to, the following:
  - 1. Performance Bond.
  - 2. Payment Bond.
  - 3. Bid Bond.
  - 4. Worker's Compensation Insurance.
  - 5. Comprehensive General Liability Insurance.
  - 6. All-Risk Builder's Risk Insurance.
  - 7. Remodel/Repair Installation Floater Insurance.
  - 8. Automobile Insurance.
  - 9. Evidence of Insurance.
  - 10. Additional information as required to proof bonds/insurance meets requirements.
  - 11. Performance Bond and Payment Bond forms.
  - 12. Drug free workplace certification.
  - 13. Preliminary Notices.
- B. See Division 0 Section "Performance Bond" and Section "Payment Bond" for bonds and insurance requirements that related to this section and related and complimentary requirements for the project.
- C. See Division 0 Section "Drugfree Workplace Certification" for additional requirements that related to this section and related and complimentary requirements for the project

### 1.3 PRESUMPTION OF QUALIFICATIONS

A. All surety companies with a minimum rating of "A-VIII," as rated by the current edition of Best's Key Rating Guide, published by A.M. Best Company, Oldwick, New Jersey 08858 and qualified to do business in California shall be presumed to be satisfactory to the DISTRICT for the issuance of insurance and bonds. In the alternative, any surety company who satisfies the requirements set forth in California Code of Civil Procedure Section 995.660 shall be accepted and approved for the issuance of bonds.

### 1.4 SUBMITTALS

- A. General: Submit the following in accordance with conditions of the Contract, Division 0, and Division 1 Specification Sections.
- B. Provide the following additional items with Performance and Payment Bonds:
  - 1. Proof of Surety to do business in California as a California-admitted Surety for Bonds and Insurance.

- 2. Telephone number, with area code, to contact Surety direct and name of person to contact
- 3. Surety/Insurance meets requirements of California Code of Civil Procedures Section 995.660.

### C. Evidence of Insurance.

- 1. Before the work is started, the Contractor shall forward to the Architect three (3) copies of a Certificates of Insurance and all the Contractual Liability coverage called for in the Contract Documents is in force, and specifically covers this particular Contract with the Owner, including the hold harmless requirements. In addition, the Certificates shall contain the following:
  - a. "No cancellation of this policy or endorsement of same shall be effective until; until the thirtieth (30th) day following the receipt of notice of such cancellation of the policy or endorsements by the Owner."
  - b. Provide additional insured on (Form B) as issued by Insurance Services Office, Inc. List names of those required here in this Project Manual.
  - c. Certificates of Insurance shall contain transcripts from the policies authenticated by the proper office of the Insurer, evidencing in particular those insured, the extent of the insurance, the location of and the operations to which the insurance applies, the expiration date and the thirty (30) day NOTICE OF CANCELLATION CLAUSE.
- 2. The Contractor must certify to the Owner that he has obtained similar certificates or memorandum evidence of insurance from each of his Subcontractors before their work commences. Each subcontractor must be covered by insurance of the same character and in the same amounts as the Contractor. All policies shall name architect and consulting engineers as additional insured.
- D. Provide the following items with evidence of insurance for all insurance policies for this project:
  - 1. Proof insurance Underwriter is qualified to do business in California.
  - 2. The insurance required must be written by a Best's Key Rating Guide "A-VIII" or better rated carrier admitted to write insurance in the state where the work is located at the time the policy is issued or satisfies requirements set forth in California Code of Civil Procedures Section 995.660.
- E. Acceptance of the Certificates of Insurance shall not relieve or decrease the liability of the Contractor, subcontractors and/or others doing work or providing materials and service for this project.

### 1.5 INDEMNIFICATION

A. CONTRACTOR shall defend, indemnify and hold harmless DISTRICT, Architect, Inspector, the State of California and their officers, employees, agents and independent contractors from all liabilities, claims, actions, liens, judgments, demands, damages, losses, costs or expenses of any kind arising from death, personal injury, property damage or other cause based or asserted upon any act, omission, or breach connected with or arising from the progress of Work or performance of service under this Agreement or the Contract Documents. As part of this indemnity, CONTRACTOR shall protect and defend, at its own expense, DISTRICT, Architect, Inspector, the State of California and their officers, employees, agents and independent contractors from any legal action including attorneys' fees or other proceeding based upon such act, omission, or breach.

Furthermore, CONTRACTOR agrees to and does hereby defend, indemnify and hold

harmless DISTRICT, Architect, Inspector, the State of California and their officers, employees, agents and independent contractors from every claim or demand made, and every liability, loss, damage, expense or attorneys' fees of any nature whatsoever, which may be incurred by reason of:

- Liability for (1) death or bodily injury to persons; (2) damage or injury to, loss (including theft), or loss of use of, any property; (3) any failure or alleged failure to comply with any provision of law or the Contract Documents; or (4) any other loss, damage or expense, sustained by any person, firm or corporation or in connection with the Work called for in this Agreement or the Contract Documents, except for liability resulting from the sole or active negligence, or the willful misconduct of the DISTRICT.
- 2. Any bodily injury to or death of persons or damage to property caused by any act, omission or breach of CONTRACTOR or any person, firm or corporation employed by CONTRACTOR, either directly or by independent contract, including all damages or injury to, loss (including theft), or loss of use of, any property, sustained by any person, firm or corporation, including the DISTRICT, arising out of or in any way connected with Work covered by this Agreement or the Contract Documents, whether said injury or damage occurs either on or off DISTRICT property, but not for any loss, injury, death or damages caused by the sole or active negligence or willful misconduct of the DISTRICT.
- 3. Any dispute between CONTRACTOR and CONTRACTOR's subcontractors/ supplies/sureties, including, but not limited to, any failure or alleged failure of the Contractor (or any person hired or employed directly or indirectly by the Contractor) to pay any Subcontractor or Material man of any tier or any other person employed in connection with the Work and/or filing of any stop notice or mechanic's lien claims.

CONTRACTOR, at its own expense, cost, and risk, shall defend any and all claims, actions, suits, or other proceedings that may be brought or instituted against the DISTRICT, its officers, agents, architects or employees, on account of or founded upon any cause, damage, or injury identified herein and shall pay or satisfy any judgment that may be rendered against the DISTRICT, its officers, agents or employees in any action, suit or other proceedings as a result thereof.

- B. In any and all claims against the Owner or the Architect or Architect's Consultants, or any of their employees by any employee of the Contractor, any Subcontractor, anyone directly or indirectly employed by any of them or anyone for whose acts any of them may be liable, the indemnification obligation under this Paragraph shall not be limited in any way by any limitation on the amount or type of damages, compensation or benefits payable by or for the Contractor or by any Subcontractor under workers compensation acts, disability benefit acts or other employee benefit acts.
- C. The obligations of the Contractor under this Paragraph shall not extend to the liability of the Architect, the Architect's consultants or any of their agents or employees arising out of 1) The preparation or approval of maps, drawings, opinions, reports, surveys, change orders, designs or specifications, or, 2) the giving of or the failure to give directions or instructions by the Architect, the Architect's consultants or any of their agents or employees, provided such giving or failure to give is the primary cause of the injury or damage.
- D. Additional Insured: The Contractor to name the following as additional insured on the Contractor's policy or policies of comprehensive general liability insurance:

Self-Insured Schools of California

- West Kern Community College District
   Board of Trustees-officers-officials
- 2. Taft College employees, consultants and volunteers
- 3. AP Architects and Architect's consultants, its agents and employees

Such insurance shall include products and completed operations and contractual liability coverages, shall be primary and non-contributing with any insurance maintained by Architect, Architect's consultants, or its agents and employees, and shall provide that the Architect be given thirty days, unqualified written notice prior to any cancellation thereof.

### 1.6 WAIVER OF SUBROGATION

Contractor hereby agrees to waive subrogation which any insurer or contractor may acquire from contractor by virtue of the payment of any loss. Contractor agrees to obtain any endorsement that may be necessary to affect this waiver of subrogation. The Workers' Compensation policy shall be endorsed with a waiver of subrogation in favor of the district for all work performed by the contractor, its employees, agents and subcontractors.

### 1.7 GENERAL

In the absence of contrary written instructions from the Owner, the Contractor at the Contractor's expense, shall obtain and maintain insurance at all times during the prosecution of the Contract, in companies and through agencies approved by the Owner, and with limits not less than those stated hereinafter.

### PART 2

### 2.1 PERFORMANCE BOND/PAYMENT BOND

A. Coincident with the signing of the agreement by the Contractor, he shall sign with his sureties and deliver with said Agreement to the Owner for approval, a bond in the standard form with good and sufficient sureties, in an amount of not less than One Hundred Percent (100%) of the Contract Price, conditioned for the completed and Faithful Performance of the entire contract under and in accordance with each and every one of the conditions and specifications and a bond in the amount of One Hundred Percent (100%) of the Contract Price covering Labor and Material. The Contractor's Bonds shall include reimbursement to the Owner for whatever additional Architect's fees may be incurred by reason of the delinquency or insolvency of the Contractor.

Performance Bond shall insure the Owner of full and prompt performance of Contract during construction and for one (1) year after recording Notice of Completion.

- 1. Surety shall be California admitted.
- Surety shall provide proof of ability to bond amount equal to or exceeding project cost.
- 3. Refer to Presumption of Qualification for additional requirements.

### 2.2 BID BOND

A. Each bid shall be accompanied by a certified or cashier's check payable to the OWNER, or a satisfactory bid bond in favor of company as surety, in an amount not less than ten (10%) percent of the maximum amount of the bid. The check or bid bond shall be given as a guarantee that the Bidder shall execute the Contract, if it is awarded to him, in conformity with the Contract Documents and shall provide the surety bond or bonds as specified within ten (10) calendar days after notification of the award of the Contract to the Bidder. The security shall be forfeited to the OWNER should the Bidder to whom the Contract is awarded fail to execute the Agreement and provide the bonds within ten (10) calendar days of award.

Copies or faxes of bid bond signatures will not be accepted. Bid bond shall be executed by contractor principal and surety, both with wet signatures and seal. The attorney-in-fact who executes the Bid Bond on behalf of the surety shall affix to the Bond a certified and current copy of his power of attorney. The Surety Company issuing the Bid Bond shall be California admitted in the State of California and U. S. Treasury listed.

### 2.3 WORKER'S COMPENSATION INSURANCE

- A. Contractor shall provide, during the term of this Contract, Worker's Compensation Insurance for all of his employees engaged in Work under this Contract, on or at the site of the project, and in case any of his work is sublet, Contractor shall require the Subcontractor to provide Worker's Compensation Insurance for all of his employees. Any class of employee or employees not covered by a Subcontractor's insurance shall be covered by the Contractor's insurance. In case any class of employees engaged in work under this Contract, on or at the site of the project, is not protected under the Worker's Compensation laws, Contractor shall provide or cause a Subcontractor to provide, adequate insurance coverage for the protection of those employees not otherwise protected. Contractor shall file, with the Owner, certificates of insurance.
  - 1. The Worker's Compensation Insurance shall be written by a company California admitted in the State of California, and shall be written for not less than the following, as established by the Owner, or greater if required by law.
  - 2. Provide employer's liability endorsements:
    - a. State workers' compensation statutory benefits policy limits of not less than \$1.000.000.00.
    - b. Employer's Liability policy limits of not less than \$1,000,000.00.
      - Per accident for bodily injury or disease. If the contractor maintains higher limits that the minimums shown herein, then, District shall be entitled to coverage for the higher limits maintained by the contractors.

### 2.4 COMPREHENSIVE GENERAL LIABILITY INSURANCE

- A. <u>Commercial General Liability Insurance</u> in Contractor's name, with personal injury limits indicated herein for combined Single Limit per occurrence coverage and annual aggregate. The policy is to be on a Comprehensive General Liability form and must include Contractual Liability endorsed to specifically cover an Indemnity Agreement contained in the Contract. The Comprehensive General Liability coverage may be provided on an "occurrence" form or a "claims made" basis. If the coverage is on a "claims made" basis, the policy shall provide for a non-cancelable 5 year extended reporting period.
  - 1. The Contractor shall carry such public liability and property damage insurance that will protect the Contractor, Owner, Architect and Engineers from claims for damages for bodily injury, including accidental death, as well as for claims for property damages, which may arise from operations under the contract whether such operations be by the Contractor or by any Subcontractor or anyone directly or indirectly employed by either party. The limits of coverage shall be as stated herein.
  - 2. In the event that any suits, actions, or claims are brought against the Owner, Architect, and/or Architect's Consultants, money equal to the "claim amount may be withheld from payments due the Contractor under and by virtue of this contract as may be considered necessary by the Owner for such purpose. Money due the Contractor will not be withheld when the Contractor produces satisfactory evidence that adequate public liability and property damage insurance has been obtained.
  - 3. The policy shall include coverage for the following:

- a. Premises operations
- b. Contractual liability
- c. Products
- d. Completed operations
- e. Broad form PD and including X, C and U coverage
- f. Personal injury
- g. Owners, contractors protective

### 2.5 CONSTRUCTION ALL-RISK BUILDER'S RISK INSURANCE

- A. The Contractor shall purchase and maintain property insurance upon the entire Work at the site to the full insurable value thereof. This insurance shall include the interests of the Owner, the Architect and his Consultants, Contractor, Subcontractors and Subsubcontractors in the Work, and shall be a standard insurance Services Office all physical loss coverage, including fire and extended coverage, vandalism and malicious mischief coverage for an amount equal to One Hundred Percent (100%) of the completed value of entire project. If not covered under the "builder's risk" insurance or otherwise provided for in the Contract Documents, the Contractor shall effect and maintain similar property insurance on portions of the Work stored off the site or in transit when such portions of the Work are to be included in an Application for Payment. The Contractor shall rebuild, repair, restore, and make good all injuries or damages to any portion of the work occasioned by and of the above causes before final acceptance and shall bear the expense thereof. Policy shall have limits equal to 90% of the insurable value thereof, including items of labor and materials connected therewith whether in or adjacent to the structure insured, materials in place or to be used as part of the permanent construction, including surplus materials, shanties, protective fences, bridges, or temporary structures, miscellaneous materials and supplies incident to the work, and such scaffolding, staging, towers, forms, and the equipment as are not owned or rented by the Contractor, the costs or which are included in the cost of the work. Such insurance shall be maintained for the life of the contract.
  - 1. If the Owner finds it necessary to occupy or use a portion or portions of the Work prior to Notice of Completion thereof, such occupancy or use shall not commence prior to a time mutually agreed to by the Owner and Contractor and to which the insurance company or companies providing the property insurance have consented by endorsement to the policy or policies. This insurance shall not be cancelled or lapsed on account of such partial occupancy or use. Consent of the Contractor and of the insurance company or companies to such occupancy or use by the Owner shall not be unreasonably withheld.
  - 2. The policy value for remodel and similar projects shall be the construction contract amount.

### 2.5 REMODEL/REPAIR INSTALLATION FLOATER INSURANCE

A. The Contractor shall purchase and maintain property insurance upon the entire Work at the site to the full insurable value thereof of work for this Contract. This insurance shall include the interests of the Owner, the Architect and his Consultants, Contractor, Subcontractors and Subsubcontractors in the Work, and shall be a standard insurance Services Office all physical loss coverage, including fire and extended coverage, vandalism and malicious mischief coverage for an amount equal to One Hundred Percent (100%) of the completed value of entire project. If not covered under the installation floater insurance or otherwise provided for in the Contract Documents, the Contractor shall effect and maintain similar property insurance on portions of the Work stored off the site or in transit when such portions of the Work are to be included in an Application for Payment. The Contractor shall rebuild, repair, restore, and make good all injuries or damages to any portion of the work occasioned by and of the above causes before final acceptance and shall bear the expense thereof.

1. If the Owner finds it necessary to occupy or use a portion or portions of the Work prior to Notice of Completion thereof, such occupancy or use shall not commence prior to a time mutually agreed to by the Owner and Contractor and to which the insurance company or companies providing the property insurance have consented by endorsement to the policy or policies. This insurance shall not be cancelled or lapsed on account of such partial occupancy or use. Consent of the Contractor and of the insurance company or companies to such occupancy or use by the Owner shall not be unreasonably withheld.

### 2.6 AUTOMOBILE LIABILITY INSURANCE

- A. Automobile Liability Insurance with an Employer's Non-Ownership Liability Endorsement in the Contractor's name. Limits of liability shall not be less than amount indicated herein for Combined Single Limit per occurrence. Provide CSL, BI and PD coverage for owned, non-owned and hired autos.
  - 1. Provide owned, non-owned and hired automobile insurance endorsement.

### 2.7 DRUG FREE WORKPLACE CERTIFICATION

- A. Pursuant to Government Code Section 8350 and following sections, the successful bidder will be required to execute and return to Owner the Drug-Free Workplace Certificate contained in the Contract Documents with the executed Construction Agreement. The bidder will be required to take positive measures outlined in the certificate to ensure the presence of a drug-free workplace. Failure to abide with the conditions set forth in the Drug-Free Workplace Act could result in penalties, including termination of the Construction Agreement or suspension of payment under the Construction Agreement.
  - 1. Submit certification with required documents as prescribed herein on form provided herein.

### PART 3 - EXECUTION

### 3.1 GENERAL

A. The Contractor shall not commence work under this Contract until he has obtained and paid for all insurance required herein and such insurance has been approved by the Owner, nor shall the Contractor allow any Subcontractor to commence work on his Subcontract until such insurance required of the Subcontractor has been so obtained and accepted.

### 3.2 PRELIMINARY NOTICES

- A. Subcontractor and Materialmen with contractual relationship with the contractor:
  - 1. If a subcontractor, material men and lower tier subcontractor files his 20-day preliminary notice and is not paid after 10 days of supplying labor and materials, the subcontractor can file a stop notice with the public entity to withhold the disputed amount from the general contractor. The public entity will only release the withheld amount to the general contractor once a stop notice release is received from the

claimant attesting that payment for service has been rendered.

- a. A prime contractor or subcontractor shall pay to any subcontractor, not later than (7) seven days after receipt of each progress payment, unless otherwise agreed to in writing, the respective amounts allowed the contractor on account of the work performed by the subcontractors, to the extent of each subcontractor's interest therein. In the event that there is a good faith dispute over all or any portion of the amount due on a progress payment from the prime contractor or subcontractor to a subcontractor, the prime contractor or subcontractor may withhold no more than 150 percent of the disputed amount or as prescribed by California Law.
- 2. However, if a subcontractor does not file a 20-day preliminary notice, provides labor and services, and is not paid, the subcontractor is ineligible to file a stop notice with the public entity. The subcontractor must wait until the project is completed or until a NOC is filed to submit a claim to the surety for payment of services. This means, that even if a project is not completed until a year after an unpaid subcontractor provides labor or materials, but the subcontractor did not file a 20-day preliminary notice, that subcontractor will have to wait a year before receiving payment through a surety. In this case, the subcontractor can make a claim within 15 days after recordation of a NOC, or if no NOC has been recorded, up to 75 days after completion of the work of improvement.
- B. Subcontractor and Materialmen with no direct contractual relationship with contractor:
  - 1. If the preliminary notice was required to be given by a person who has no direct contractual relationship with the contractor, and who has not given notice as provided by California Law, that person may enforce a claim by giving written notice to the surety and the bond principal, as provided by California Law, within 15 days after recordation of a notice of completion. If no notice of completion has been recorded, the time for giving written notice to the surety and the bond principal is extended to 75 days after completion of the work of improvement

### C. Notices:

All notices herein provided to be given or which may be given by either party to the other shall be deemed to have been fully given when made in writing and deposited with the United States Postal Service, Registered or Certified, and postage prepaid and addressed as prescribed herein the Project Manual. Nothing herein contained shall preclude the giving of any such notice by personal service.

### **BONDS AND CERTIFICATES SCHEDULE**

A. Bonds: <u>Amount</u>

1. Performance Bond/Payment Bond

100% Contract 10% Contract

Bid Bond

### B. Insurance:

- 1. Worker's Compensation Insurance per State of California policy limits of not less than \$1,000,000.00.
  - a. Employer's Liability Endorsement \$1,000,000 min.
- 2. Comprehensive General Liability:
  - a. Combined single limits for bodily injury and property damage:

\$5,000,000 - Each Occurrence \$10,000,000 - Annual Aggregate

\$1,000,000 - Each Occurrence \$2,000,000 - Annual Aggregate

- b. Personal Injury, with Employment Exclusion deleted.
- c. Include coverage of the following:
  - 1) Premises operations
  - 2) Contractual liability
  - 3) Products
  - 4) Completed operations
  - 5) Broad form PD and including X, C and U coverage
  - 6) Personal injury
  - 7) Owners, contractors protective
- 3. All-Risk Builder's Risk Insurance: Shall be written for the full amount of the Contract amount.
- 3. Installation/Remodel Floater: Shall be written for the full amount of Contract.
- 4. Comprehensive Automobile Liability:
  - a. Combined single limits for bodily injury and property damage: \$1,000,000 Each Occurrence
  - b. Hired Automobile Liability Endorsement.
- C. Bond Forms:
  - 1. Performance Bond (following page).
  - 2. Payment Bond (next following page).
- D. Drug Free Workplace Certification: Submit with executed agreement and related documents.

### **PERFORMANCE BOND**

| BE ADVISED THAT:   |   |  |
|--|---|--|
| The  | of  | County, California ("District") has  |
| awarded to   | as Pı   | rincipal ("Principal"), the contract for the   |
| work described as follows:   |   |  |
| Project Name: Architect's Project No.: District Name:  |   |  |
| The Principal is required to furnish a Bo<br>performance;  |   | n with the contract guaranteeing faithful  |
| We the undersigned Principal and   |   |  |
| as Surety are held and firmly bound to the District i  | in the sum of   |  |
| to be paid t   | to the District; for                                  | which payment we bind ourselves, our   |
| heirs, executors and administrators, successors or   | assigns, jointly ar                                   | nd severally, firmly by this Bond.   |
| The condition of this Bond is this: the successors, or assigns, well and truly perform the cany alterations made as provided in it, at the time harmless the District, its officers, and agents, as strouble and void; otherwise it shall remain in full force a | covenants, condite and in the manitipulated in the co | ions, and agreements in the contract and<br>ner specified, and indemnifies and holds |
| The Surety, for value received, stipulated a addition to the terms of the contract or the work to its obligation on this Bond, and it waives notice of a terms of the contract, the work, or the specifications  | be performed or<br>any change, exter                  | the specifications shall in any way affect   |
| In the event suit is brought upon this bond pay all costs incurred by the District in such suit, Court.  |   |  |
| AS WITNESS, we have affixed our signatu  | res and seals this                                    | s, day of, 20  |
| (Principal Seal)   | PRINCIP   | AL   |
|  | Ву:   |  |
|  | Title:  |  |
| (Surety Seal)  | SURETY<br>By:   |  |
|  | Title:  |  |
|  | <del></del>   |  |
|  |   | Name, Address & Telephone No.<br>of California Agent of Surety                       |

#### TC – G-8 ALLIED HEALTH LAB REMODEL

DE 451/10E5 TILAT

this Bond.

#### **PAYMENT BOND**

| BE ADVISED I                                 | HAT:                           |                  |                                  |            |
|--|--------------------------------|------------------|----------------------------------|------------|
| The  |                                | of               | County, California ("Distr       | rict") has |
| awarded to                                   |                                | as Prin          | cipal ("Principal"), the contrac | t for the  |
| work described as follow                     | /s:                            |                  |                                  |            |
|  | Architect's Project No.:       |                  |                                  |            |
| The Principal is<br>furnish a bond in connec |                                |                  | n 3247), of the California Civil | Code to    |
| Therefore, we,                               | the Principal and              |                  |                                  |            |
| as Surety, are held and                      | firmly bound unto the District | in the penal sum | of:                              |            |
|  |                                |                  | dollars (\$                      | ),         |
| lawful money of the Un                       | ited States of America for the | e payment of whi | ch sum well and truly to be m    | nade, we   |
| bind ourselves, our heir                     | s, executors, administrators,  | successors and   | assigns, jointly and severally,  | firmly by  |
|  |                                |                  |                                  |            |

The condition of this obligation is such that if the Principal, his or its subcontractors, heirs, executors, administrators, successors, or assigns, shall fail to pay any of the persons named in Section 3181 of the California Civil Code, or amounts due under the Unemployment Insurance Code with respect to work or labor performed under the contract, or for any amounts required to be deducted, withheld, and paid over to the Franchise Tax Board from the wages of employees of the contractor and his subcontractors pursuant to Section 18806 of the California Revenue and Taxation Code, with respect to their work and labor the Surety or Sureties will pay for them, in an amount not exceeding the sum specified above, and also, in case suit is brought upon this Bond, all litigation expenses incurred by the District, including reasonable attorney's fees, architect's fees, court costs, expert witness fees, and investigation expenses.

This Bond shall inure to the benefit of any of the persons named in Section 3181 of the California Civil Code, so as to give a right of action to such persons or their assigns in any suit brought upon this Bond.

It is further stipulated and agreed that the Surety on this Bond shall not be exonerated or released from the obligation of this Bond by any change, extension of time for performance, addition, alteration or modification in, to, or of any contract, plans, specifications, or agreement pertaining or relating to any scheme or work of improvement described above or pertaining or relating to the furnishing of labor materials, or equipment for it, nor by any change or modification of any terms of payment or extension of the time for any payment pertaining or relating to any scheme or work of improvement described above, nor by any conditions precedent or subsequent in the bond attempting to limit the right of recovery of claimants otherwise entitled to recover under the contract or agreement or under the Bond, nor by any fraud practiced by any person other than the claimant seeking to recover on the Bond and that this Bond be construed most strongly against the Surety and in favor of all persons for whose benefit it is given, and under no circumstances shall Surety be released from liability to those for whose benefit the Bond has been given, by reason of any breach of contract between the Owner or District and original contractor or on the part of any Obligee named in such Bond, but the sole conditions of recovery shall be that claimant is a person described in Section 3181 of the California Civil Code, and has not been paid the full amount of his claim and that Surety waives notice of any change, extension of time, addition, alteration, or modification mentioned above.

Any claim under this Bond may be addressed to:

| TC – G-8 ALLIED HEALTH LAB REMODEL  | _ (Name, Address and Telephone<br>_ Number of Surety) |
|-------------------------------------|---|
|                                     | Poprocontativo)                                       |
| AS WITNESS, we have affixed our, 20 | signatures and seals this day o                       |
| (Principal Seal)                    | PRINCIPAL By:   |
| (Surety Seal)                       | SURETY By:  |

| TC – G   | G-8 ALLIED HEALTH LAB REMODEL  DRUG-FREE WORKPLACE CERTIFICATION  |
|--|---|
| PROJI  | ECT:  |
| DISTR  | UCT:  |
| following required serviced specifications to suspect to suspect for the suspe | Orug-Free Workplace Certification is required pursuant to Government Code Section 8350 and any sections, and the Drug-Free Workplace Act of 1990. The Drug-Free Workplace Act of 1990 as that every person or organization awarded a contract for the procurement of any property of the section any state agency must certify that it will provide a drug-free workplace by doing certain ed acts. In addition, the Act provides that each contract awarded by a state agency may be subject on the contract and the contractor may be subject to debarment turns contracting, if the state agency determines that Specified acts have occurred. |
|  | ant to Government Code Section 8355, every person or organization awarded a contract from a gency shall certify that it will provide a drug-free workplace by doing all of the following:   |
| <ol> <li>2.</li> </ol>   | Publishing a statement notifying employees that the unlawful manufacture, distribution dispensation, possession, or use of a controlled substance is prohibited in the person's organization's workplace, and specifying actions which will be taken against employees for violations of the prohibition;  Establishing a drug-free awareness program to inform employees about all of the following:   |
|  | <ul> <li>A. The dangers of drug abuse in the workplace;</li> <li>B. The person's or organization's policy of maintaining a drug-free workplace;</li> <li>C. The availability of drug counseling, rehabilitation, and employee assistance programs;</li> <li>D. The penalties that may be imposed upon employees for drug abuse violations;</li> </ul>   |
| 3.   | Requiring that each employee engaged in the performance of work on the Project be given a copy of the statement required by subdivision (a), and that as a condition of employment on the Contract the employee agrees to abide by the terms of the statement.  |
| above<br>substa<br>employ  | undersigned, agree to fulfill the terms and requirements of Government Code Section 8355 listed and will publish a statement notifying employees concerning (a) the prohibition of controlled inces at the workplace, (b) establishing a drug-free awareness program, and (c) requiring that each yee engaged in the performance of the Contract be given a copy of the statement required by a 8355(a) and requiring that the employee agree to abide by the terms of that statement   |

I also understand that if the Owner determines that I have either (a) made a false certification or (b) violated this certification by failing to carry out the requirements of Section 8355, the contract awarded is subject to suspension of payments, termination, or both. I further understand that should I violate the terms of the Drug-Free Workplace Act of 1990, I may be subject to debarment in accordance with the requirements of Section 8350 and following sections.

I acknowledge that I am aware of the provisions of Government Code Section 8350 and following sections, and hereby certify that I will adhere to the requirements of the Drug-Free Workplace Act of 1990.

| Legal Name of Contractor | _     |      |
|--------------------------|-------|------|
| Signature                | Title | Date |
| Print Name               |       |      |
| END OF SECTION 004313    |       |      |

# TC – G-8 ALLIED HEALTH LAB REMODEL

# NOTICE OF AWARD AND REQUEST FOR SUBMITTAL OF BONDS AND INSURANCE

This notice shall confirm your notification of award. The items below are required to be submitted prior to Notice to Proceed being issued. Per Contract Documents, you have 10 days from the Notice of Award date to provide the following:

| Project No: District: District: Date Due: Date Due: Date Due: Date Due: Date Due: Date Due: Date Received - Initial:   |        | Project:   | Notice of Award   | Date: _    |         |   |
|--|--------|--|-------------------|------------|---------|---|
| Contractor:    Date Received - Initial:   Note: Contractor to submit all items in REQ column per Division 0 Section 04313.   | Proj   | ect No:  | Initial Request   | Date:      |         |   |
| Required Documents:  Required Documents:  Required Documents:  1 Performance Bond for 100% of the contract amount 2 Payment Bond for 100% of the contract amount 3 Bond Company State of California Certificate of Authority 4 Bond document US Treasury listing proof 5 Bond document Power of Attorney and Proof to sign documents 6 Bond document proof of ability to bond within capacity (memo) 7 Bond proof/ valid & issued from underwriter (add to memo #6) 8 Bond Best Report 9 Workers Compensation & Employers Liable Endorsement 10 Comprehensive General Liability Insurance - \$ 10,000,000 aggregate  |        | District:  | Date              | Due:       |         |   |
| Required Documents:    Date   REQ   NA   By   Others   | Con    | tractor:   | Date Received - l | nitial:    |         |   |
| Required Documents:    Performance Bond for 100% of the contract amount   Performance Bond Company State of California Certificate of Authority   Performance Bond Company State of California Certificate of Authority   Performance Bond Company Bond Company Bond Proof (Palid & Issued From Underwriter (add to memo #8)   Performance Bond Proof (Valid & Issued from Underwriter (add to memo #8)   Performance Bond Proof (Valid & Issued From Underwriter (add to memo #8)   Performance Bond Proof (Valid & Issued From Underwriter (add to memo #8)   Performance Bond Proof (Valid & Issued From Underwriter (add to memo #8)   Performance Bond Proof (Valid & Issued From Underwriter (add to memo #8)   Performance Bond Proof (Valid & Issued From Underwriter (add to memo #8)   Performance Bond Proof (Valid & Issued From Underwriter (add to memo #8)   Performance Bond Proof (Valid & Issued From Underwriter (add to memo #8)   Performance Bond Proof (Valid & Issued From Underwriter (add to memo #8)   Performance Bond Proof (Valid & Issued From Underwriter (add to memo #8)   Performance Bond Proof (Valid & Issued From Underwriter (add to memo #8)   Performance Bond Proof (Valid & Issued From Underwriter (add to memo #8)   Performance Bond Bond Bond Bond Bond Bond Bond Bond |        |  |                   | r Division | 0 Secti |   |
| Performance Bond for 100% of the contract amount Payment Bond for 100% of the contract amount Bond Company State of California Certificate of Authority Bond document US Treasury listing proof Bond document Power of Attorney and Proof to sign documents Bond document proof of ability to bond within capacity (memo) Bond proof/ valid & issued from underwriter (add to memo #6) Bond Best Report Workers Compensation & Employers Liable Endorsement Comprehensive General Liability Insurance - \$ 10,000,000 aggregate or (pick one – can't be both) Comprehensive General Liability Insurance - \$ 2,000,000 aggregate If Additional insured per item 1.5.D with binders All Risk Builders Insurance, Construction Cost Amount If Memo from Contractor that all subs have same type of insurance  Heroof of License (copy - front and back of pocket license)  DIR Registration verified MBE/ WBE voluntary compliance information and percentages or (pick one – can't be both) Comprehensive automobile liability - \$ 1,000,000 Executed contracts (3 copies) original signature by Contractor  Comprehensive automobile liability - \$ 1,000,000 Executed contracts (3 copies) original signature by Contractor  Other items and notes for complete package a. Drug Free Workplace Certification b. Fingerprinting Clearance from DOJ cc: Contractor/ Owner/ File  For official use only: Bonding Company:  For official use only:   |        |  |                   | REQ        | NA      | - |
| 2 Payment Bond for 100% of the contract amount 3 Bond Company State of California Certificate of Authority 4 Bond document US Treasury listing proof 5 Bond document Power of Attorney and Proof to sign documents 6 Bond document proof of ability to bond within capacity (memo) 7 Bond proof/ valid & issued from underwriter (add to memo #6) 8 Bond Best Report 9 Workers Compensation & Employers Liable Endorsement 10 Comprehensive General Liability Insurance - \$ 10,000,000 aggregate or (pick one – can't be both) Comprehensive General Liability Insurance - \$ 2,000,000 aggregate 11 Additional insured per item 1.5.D with binders All Risk Builders Insurance, Construction Cost Amount or (pick one – can't be both) 12 Installation/ Remodel floater insurance - Construction Cost Amount 13 Memo from Contractor that all subs have same type of insurance 14 Proof of License (copy - front and back of pocket license) 15 DIR Registration verified MBE/ WBE voluntary compliance information and percentages or (pick one – can't be both) 16 DVBE Certification (for K-12) 17 Comprehensive automobile liability - \$ 1,000,000 18 Executed contracts (3 copies) original signature by Contractor 19 W-9 Request for Taxpayers ID Number and Certification 20 Other items and notes for complete package a. Drug Free Workplace Certification b. Fingerprinting Clearance from DOJ cc: Contractor/ Owner/ File   | Requir |  | Completed         |            | _       |   |
| 3 Bond Company State of California Certificate of Authority 4 Bond document US Treasury listing proof 5 Bond document Power of Attorney and Proof to sign documents 6 Bond document proof of ability to bond within capacity (memo) 7 Bond proof/ valid & issued from underwriter (add to memo #6) 8 Bond Best Report 9 Workers Compensation & Employers Liable Endorsement 10 Comprehensive General Liability Insurance - \$ 10,000,000 aggregate or (pick one - can't be both) Comprehensive General Liability Insurance - \$ 2,000,000 aggregate 11 Additional insured per item 1.5.D with binders All Risk Builders Insurance, Construction Cost Amount or (pick one - can't be both) 12 Installation/ Remodel floater insurance - Construction Cost Amount 13 Memo from Contractor that all subs have same type of insurance 14 Proof of License (copy - front and back of pocket license) 15 DIR Registration verified MBE/ WBE voluntary compliance information and percentages or (pick one - can't be both) 16 DVBE Certification (for K-12) 17 Comprehensive automobile liability - \$ 1,000,000 18 Executed contracts (3 copies) original signature by Contractor 19 W-9 Request for Taxpayers ID Number and Certification 20 Other items and notes for complete package a. Drug Free Workplace Certification b. Fingerprinting Clearance from DOJ cc: Contractor/ Owner/ File  For official use only: Bonding Company:   |        |  |                   | _          |         |   |
| 4 Bond document US Treasury listing proof 5 Bond document Power of Attorney and Proof to sign documents 6 Bond document proof of ability to bond within capacity (memo) 7 Bond proof/ valid & issued from underwriter (add to memo #6) 8 Bond Best Report 9 Workers Compensation & Employers Liable Endorsement 10 Comprehensive General Liability Insurance - \$ 10,000,000 aggregate or (pick one – can't be both) Comprehensive General Liability Insurance - \$ 2,000,000 aggregate 11 Additional insured per item 1.5.D with binders All Risk Builders Insurance, Construction Cost Amount or (pick one – can't be both) 12 Installation/ Remodel floater insurance - Construction Cost Amount 13 Memo from Contractor that all subs have same type of insurance 14 Proof of License (copy - front and back of pocket license) 15 DIR Registration verified MBE/ WBE voluntary compliance information and percentages or (pick one – can't be both) 16 DVBE Certification (for K-12) 17 Comprehensive automobile liability - \$ 1,000,000 18 Executed contracts (3 copies) original signature by Contractor 19 W-9 Request for Taxpayers ID Number and Certification 0 Uther items and notes for complete package a. Drug Free Workplace Certification b. Fingerprinting Clearance from DOJ cc: Contractor/ Owner/ File  For official use only: Bonding Company:  | 2      | •  |                   | _          |         |   |
| 5 Bond document Power of Attorney and Proof to sign documents 6 Bond document proof of ability to bond within capacity (memo) 7 Bond proof/ valid & issued from underwriter (add to memo #6) 8 Bond Best Report 9 Workers Compensation & Employers Liable Endorsement 10 Comprehensive General Liability Insurance - \$ 10,000,000 aggregate or (pick one - can't be both) Comprehensive General Liability Insurance - \$ 2,000,000 aggregate 11 Additional insurance per item 1.5.D with binders All Risk Builders Insurance, Construction Cost Amount or (pick one - can't be both) 12 Installation/ Remodel floater insurance - Construction Cost Amount 13 Memo from Contractor that all subs have same type of insurance 14 Proof of License (copy - front and back of pocket license) 15 DIR Registration verified MBE/ WBE voluntary compliance information and percentages or (pick one - can't be both) 16 DVBE Certification (for K-12) 17 Comprehensive automobile liability - \$ 1,000,000 18 Executed contracts (3 copies) original signature by Contractor 19 W-9 Request for Taxpayers ID Number and Certification 20 Other items and notes for complete package a. Drug Free Workplace Certification b. Fingerprinting Clearance from DOJ cc: Contractor/ Owner/ File  For official use only: Bonding Company:   | 3      | Bond Company State of California Certificate of Authority      |                   |            |         |   |
| 8 Bond document proof of ability to bond within capacity (memo) 7 Bond proof/ valid & issued from underwriter (add to memo #6) 8 Bond Best Report 9 Workers Compensation & Employers Liable Endorsement 10 Comprehensive General Liability Insurance - \$ 10,000,000 aggregate or (pick one - can't be both) Comprehensive General Liability Insurance - \$ 2,000,000 aggregate  11 Additional insured per item 1.5.D with binders All Risk Builders Insurance, Construction Cost Amount or (pick one - can't be both) 12 Installation/ Remodel floater insurance - Construction Cost Amount 13 Memo from Contractor that all subs have same type of insurance 14 Proof of License (copy - front and back of pocket license) 15 DIR Registration verified MBE/ WBE voluntary compliance information and percentages or (pick one - can't be both) 16 DVBE Certification (for K-12) 17 Comprehensive automobile liability - \$ 1,000,000 18 Executed contracts (3 copies) original signature by Contractor 19 W-9 Request for Taxpayers ID Number and Certification 20 Other items and notes for complete package a. Drug Free Workplace Certification b. Fingerprinting Clearance from DOJ cc: Contractor/ Owner/ File  For official use only: Bonding Company:  | 4      | Bond document US Treasury listing proof                        |                   | _          |         |   |
| 7 Bond proof/ valid & issued from underwriter (add to memo #6) 8 Bond Best Report 9 Workers Compensation & Employers Liable Endorsement 10 Comprehensive General Liability Insurance - \$ 10,000,000 aggregate or (pick one - can't be both) Comprehensive General Liability Insurance - \$ 2,000,000 aggregate 11 Additional insured per item 1.5.D with binders All Risk Builders Insurance, Construction Cost Amount or (pick one - can't be both) 12 Installation/ Remodel floater insurance - Construction Cost Amount 13 Memo from Contractor that all subs have same type of insurance 14 Proof of License (copy - front and back of pocket license) 15 DIR Registration verified MBE/ WBE voluntary compliance information and percentages or (pick one - can't be both) 16 DVBE Certification (for K-12) 17 Comprehensive automobile liability - \$ 1,000,000 18 Executed contracts (3 copies) original signature by Contractor 19 W-9 Request for Taxpayers ID Number and Certification 20 Other items and notes for complete package a. Drug Free Workplace Certification b. Fingerprinting Clearance from DOJ cc: Contractor/ Owner/ File  For official use only: Bonding Company:   | 5      | Bond document Power of Attorney and Proof to sign documents    |                   | _          |         |   |
| 8 Bond Best Report 9 Workers Compensation & Employers Liable Endorsement 10 Comprehensive General Liability Insurance - \$ 10,000,000 aggregate 0r (pick one - can't be both) Comprehensive General Liability Insurance - \$ 2,000,000 aggregate  11 Additional insured per item 1.5.D with binders All Risk Builders Insurance, Construction Cost Amount 0r (pick one - can't be both) 12 Installation/ Remodel floater insurance - Construction Cost Amount 13 Memo from Contractor that all subs have same type of insurance 14 Proof of License (copy - front and back of pocket license) 15 DIR Registration verified MBE/ WBE voluntary compliance information and percentages or (pick one - can't be both) 16 DVBE Certification (for K-12) 17 Comprehensive automobile liability - \$ 1,000,000 18 Executed contracts (3 copies) original signature by Contractor 19 W-9 Request for Taxpayers ID Number and Certification 20 Other items and notes for complete package a. Drug Free Workplace Certification b. Fingerprinting Clearance from DOJ cc: Contractor/ Owner/ File  For official use only: Bonding Company:   | 6      | Bond document proof of ability to bond within capacity (memo)  |                   | ~          |         |   |
| 9 Workers Compensation & Employers Liable Endorsement 10 Comprehensive General Liability Insurance - \$ 10,000,000 aggregate or (pick one - can't be both) Comprehensive General Liability Insurance - \$ 2,000,000 aggregate  11 Additional insured per item 1.5.D with binders All Risk Builders Insurance, Construction Cost Amount or (pick one - can't be both)  12 Installation/ Remodel floater insurance - Construction Cost Amount 13 Memo from Contractor that all subs have same type of insurance 14 Proof of License (copy - front and back of pocket license) 15 DIR Registration verified MBE/ WBE voluntary compliance information and percentages or (pick one - can't be both) 16 DVBE Certification (for K-12) 17 Comprehensive automobile liability - \$ 1,000,000  18 Executed contracts (3 copies) original signature by Contractor 19 W-9 Request for Taxpayers ID Number and Certification 20 Other items and notes for complete package a. Drug Free Workplace Certification b. Fingerprinting Clearance from DOJ cc: Contractor/ Owner/ File  For official use only: Bonding Company:  | 7      | Bond proof/ valid & issued from underwriter (add to memo #6)   |                   | ~          |         |   |
| 10 Comprehensive General Liability Insurance - \$ 10,000,000 aggregate or (pick one - can't be both) Comprehensive General Liability Insurance - \$ 2,000,000 aggregate  11 Additional insured per item 1.5.D with binders All Risk Builders Insurance, Construction Cost Amount or (pick one - can't be both)  12 Installation/ Remodel floater insurance - Construction Cost Amount 13 Memo from Contractor that all subs have same type of insurance 14 Proof of License (copy - front and back of pocket license) 15 DIR Registration verified MBE/ WBE voluntary compliance information and percentages or (pick one - can't be both) 16 DVBE Certification (for K-12) 17 Comprehensive automobile liability - \$ 1,000,000  18 Executed contracts (3 copies) original signature by Contractor 19 W-9 Request for Taxpayers ID Number and Certification 20 Other items and notes for complete package a. Drug Free Workplace Certification b. Fingerprinting Clearance from DOJ cc: Contractor/ Owner/ File  For official use only:  Bonding Company:   | 8      | Bond Best Report   |                   |            |         | ~ |
| aggregate or (pick one – can't be both) Comprehensive General Liability Insurance - \$ 2,000,000 aggregate  11 Additional insured per item 1.5.D with binders All Risk Builders Insurance, Construction Cost Amount or (pick one – can't be both)  12 Installation/ Remodel floater insurance - Construction Cost Amount 13 Memo from Contractor that all subs have same type of insurance 14 Proof of License (copy - front and back of pocket license) 15 DIR Registration verified MBE/ WBE voluntary compliance information and percentages or (pick one – can't be both)  16 DVBE Certification (for K-12) 17 Comprehensive automobile liability - \$ 1,000,000  18 Executed contracts (3 copies) original signature by Contractor 19 W-9 Request for Taxpayers ID Number and Certification 20 Other items and notes for complete package a. Drug Free Workplace Certification b. Fingerprinting Clearance from DOJ cc: Contractor/ Owner/ File  For official use only:  Bonding Company:   | 9      | Workers Compensation & Employers Liable Endorsement            |                   | ~          |         |   |
| or (pick one – can't be both) Comprehensive General Liability Insurance - \$ 2,000,000 aggregate  11 Additional insured per item 1.5.D with binders All Risk Builders Insurance, Construction Cost Amount or (pick one – can't be both)  12 Installation/ Remodel floater insurance - Construction Cost Amount 13 Memo from Contractor that all subs have same type of insurance 14 Proof of License (copy - front and back of pocket license) 15 DIR Registration verified MBE/ WBE voluntary compliance information and percentages or (pick one – can't be both)  16 DVBE Certification (for K-12) 17 Comprehensive automobile liability - \$ 1,000,000  18 Executed contracts (3 copies) original signature by Contractor 19 W-9 Request for Taxpayers ID Number and Certification 20 Other items and notes for complete package a. Drug Free Workplace Certification b. Fingerprinting Clearance from DOJ  cc: Contractor/ Owner/ File  For official use only:  Bonding Company:  | 10     | ·  |                   | -          |         |   |
| Comprehensive General Liability Insurance - \$ 2,000,000 aggregate  11 Additional insured per item 1.5.D with binders All Risk Builders Insurance, Construction Cost Amount or (pick one – can't be both)  12 Installation/ Remodel floater insurance - Construction Cost Amount 13 Memo from Contractor that all subs have same type of insurance 14 Proof of License (copy - front and back of pocket license) 15 DIR Registration verified MBE/ WBE voluntary compliance information and percentages or (pick one – can't be both) 16 DVBE Certification (for K-12) 17 Comprehensive automobile liability - \$ 1,000,000  18 Executed contracts (3 copies) original signature by Contractor 19 W-9 Request for Taxpayers ID Number and Certification 20 Other items and notes for complete package a. Drug Free Workplace Certification b. Fingerprinting Clearance from DOJ cc: Contractor/ Owner/ File  For official use only:  Bonding Company:  |        |  |                   | ~          |         |   |
| Additional insured per item 1.5.D with binders All Risk Builders Insurance, Construction Cost Amount or (pick one – can't be both)  12 Installation/ Remodel floater insurance - Construction Cost Amount 13 Memo from Contractor that all subs have same type of insurance 14 Proof of License (copy - front and back of pocket license) 15 DIR Registration verified MBE/ WBE voluntary compliance information and percentages or (pick one – can't be both) 16 DVBE Certification (for K-12) 17 Comprehensive automobile liability - \$ 1,000,000  18 Executed contracts (3 copies) original signature by Contractor 19 W-9 Request for Taxpayers ID Number and Certification 20 Other items and notes for complete package a. Drug Free Workplace Certification b. Fingerprinting Clearance from DOJ cc: Contractor/ Owner/ File  For official use only:  Bonding Company:   |        |  |                   |            |         |   |
| All Risk Builders Insurance, Construction Cost Amount or (pick one – can't be both)  12 Installation/ Remodel floater insurance - Construction Cost Amount 13 Memo from Contractor that all subs have same type of insurance 14 Proof of License (copy - front and back of pocket license) 15 DIR Registration verified MBE/ WBE voluntary compliance information and percentages or (pick one – can't be both) 16 DVBE Certification (for K-12) 17 Comprehensive automobile liability - \$ 1,000,000  18 Executed contracts (3 copies) original signature by Contractor 19 W-9 Request for Taxpayers ID Number and Certification 20 Other items and notes for complete package a. Drug Free Workplace Certification b. Fingerprinting Clearance from DOJ cc: Contractor/ Owner/ File  For official use only:  Bonding Company:  | 11     |  |                   | <u>~</u>   |         |   |
| or (pick one – can't be both)  12 Installation/ Remodel floater insurance - Construction Cost Amount  13 Memo from Contractor that all subs have same type of insurance  14 Proof of License (copy - front and back of pocket license)  15 DIR Registration verified  MBE/ WBE voluntary compliance information and percentages or  (pick one – can't be both)  16 DVBE Certification (for K-12)  17 Comprehensive automobile liability - \$ 1,000,000  18 Executed contracts (3 copies) original signature by Contractor  19 W-9 Request for Taxpayers ID Number and Certification  20 Other items and notes for complete package  a. Drug Free Workplace Certification  b. Fingerprinting Clearance from DOJ  cc: Contractor/ Owner/ File  For official use only:  Bonding Company:  | • • •  |  |                   | -          |         |   |
| 13 Memo from Contractor that all subs have same type of insurance  14 Proof of License (copy - front and back of pocket license)  15 DIR Registration verified  MBE/ WBE voluntary compliance information and percentages or  (pick one - can't be both)  16 DVBE Certification (for K-12)  17 Comprehensive automobile liability - \$ 1,000,000  18 Executed contracts (3 copies) original signature by Contractor  19 W-9 Request for Taxpayers ID Number and Certification  20 Other items and notes for complete package  a. Drug Free Workplace Certification  b. Fingerprinting Clearance from DOJ  cc: Contractor/ Owner/ File  For official use only:  Bonding Company:  |        | or (pick one – can't be both)                                  |                   | ~          |         |   |
| 14 Proof of License (copy - front and back of pocket license)  15 DIR Registration verified  |        |  |                   |            | _       | _ |
| DIR Registration verified  MBE/ WBE voluntary compliance information and percentages or  (pick one – can't be both)  16 DVBE Certification (for K-12)  17 Comprehensive automobile liability - \$ 1,000,000  18 Executed contracts (3 copies) original signature by Contractor  19 W-9 Request for Taxpayers ID Number and Certification  20 Other items and notes for complete package  a. Drug Free Workplace Certification  b. Fingerprinting Clearance from DOJ  cc: Contractor/ Owner/ File  For official use only:  Bonding Company:   | 13     | Memo from Contractor that all subs have same type of insurance |                   | _          |         |   |
| MBE/ WBE voluntary compliance information and percentages or  (pick one – can't be both)  16 DVBE Certification (for K-12)  17 Comprehensive automobile liability - \$ 1,000,000  18 Executed contracts (3 copies) original signature by Contractor  19 W-9 Request for Taxpayers ID Number and Certification  20 Other items and notes for complete package  a. Drug Free Workplace Certification  b. Fingerprinting Clearance from DOJ  cc: Contractor/ Owner/ File  For official use only:  Bonding Company:  | 14     | Proof of License (copy - front and back of pocket license)     |                   |            |         |   |
| (pick one – can't be both)  16 DVBE Certification (for K-12)  17 Comprehensive automobile liability - \$ 1,000,000  18 Executed contracts (3 copies) original signature by Contractor  19 W-9 Request for Taxpayers ID Number and Certification  20 Other items and notes for complete package  a. Drug Free Workplace Certification  b. Fingerprinting Clearance from DOJ  cc: Contractor/ Owner/ File  For official use only:  Bonding Company:  | 15     |  |                   |            |         | ~ |
| 16 DVBE Certification (for K-12)  17 Comprehensive automobile liability - \$ 1,000,000  18 Executed contracts (3 copies) original signature by Contractor  19 W-9 Request for Taxpayers ID Number and Certification  20 Other items and notes for complete package  a. Drug Free Workplace Certification  b. Fingerprinting Clearance from DOJ  cc: Contractor/ Owner/ File  For official use only:  Bonding Company:  |        |  |                   | □          |         |   |
| 17 Comprehensive automobile liability - \$ 1,000,000  18 Executed contracts (3 copies) original signature by Contractor  19 W-9 Request for Taxpayers ID Number and Certification  20 Other items and notes for complete package  a. Drug Free Workplace Certification  b. Fingerprinting Clearance from DOJ  cc: Contractor/ Owner/ File  For official use only:  Bonding Company:  | 16     |  |                   | ı.         |         |   |
| 18 Executed contracts (3 copies) original signature by Contractor  19 W-9 Request for Taxpayers ID Number and Certification  20 Other items and notes for complete package a. Drug Free Workplace Certification b. Fingerprinting Clearance from DOJ  cc: Contractor/ Owner/ File  For official use only:  Bonding Company: .  |        |  |                   | <b>✓</b>   |         |   |
| 19 W-9 Request for Taxpayers ID Number and Certification 20 Other items and notes for complete package a. Drug Free Workplace Certification b. Fingerprinting Clearance from DOJ cc: Contractor/ Owner/ File  For official use only:  Bonding Company:   |        | •  |                   | ~          |         |   |
| 20 Other items and notes for complete package a. Drug Free Workplace Certification b. Fingerprinting Clearance from DOJ cc: Contractor/ Owner/ File  For official use only:  Bonding Company: .  |        | ,                        |                   | <u>~</u>   |         |   |
| a. Drug Free Workplace Certification b. Fingerprinting Clearance from DOJ cc: Contractor/ Owner/ File  For official use only:  Bonding Company:  |        |  |                   | <b>~</b>   |         |   |
| b. Fingerprinting Clearance from DOJ cc: Contractor/ Owner/ File  For official use only:  Bonding Company:   | 20     |  |                   | -          |         |   |
| For official use only:  Bonding Company:   |        | b. Fingerprinting Clearance from DOJ                           |                   |            |         |   |
| Bonding Company:   | cc: Cc | ontractor/ Owner/ File   |                   |            |         |   |
| Bonding Company:   |        | For official use only:   |                   |            |         |   |
| AM Best Number:  | E      | Bonding Company: .   |                   |            |         |   |
|  |        | AM Best Number:  |                   |            |         |   |
| Location:  |        |  |                   |            |         |   |
| Documents from: .  |        |  |                   |            |         |   |
| AM Best rating:  |        | AM Best rating: .  |                   |            |         |   |
| US Treasury Listed: .  | U      | S Treasury Listed: .   |                   |            |         |   |

#### PART 1 - GENERAL

### 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and other Division 0 & 1 Specification Sections, apply to this Section.

# 1.2 WAGES RATES AND PAYROLL RECORDS

- A. Pursuant to the provisions of Article 2 (commencing at Section 1770), Chapter 1, Part 7, Division 2, of the California Labor Code, OWNER has ascertained the general prevailing rate of per diem wages for each craft, classification, or type of worker needed to execute the work of the Project in the locality in which this public work is to be performed. The general prevailing rates of per diem wages are available at OWNER's office. CONTRACTOR is responsible to pay those rates determined to be applicable by the Director of Industrial Relations and OWNER shall not be responsible for any damages arising from the error.
- B. When permitted by law, holiday and overtime work shall be paid at a rate of at least one and one-half times the specified rate of per diem wages, unless otherwise specified.
- C. CONTRACTOR shall pay and shall cause to be paid to each worker engaged in work on the Project not less than the general prevailing rate of per diem wages, regardless of any contractual relationship which may exist between CONTRACTOR or any Subcontractor and such workers.
- D. Pursuant to Labor Code Section 1775, CONTRACTOR shall forfeit and OWNER shall withhold from payments to CONTRACTOR not more than \$200 for each calendar day any worker is paid less than the established prevailing wage rates for the work or craft in which the worker is employed by CONTRACTOR on the Project. The difference between the established prevailing wage rates and the amount paid to each worker for each whole or partial calendar day for which each worker was paid less than the established prevailing wage rates shall be paid to each worker by CONTRACTOR.
- E. Any worker employed to perform work on the Project which is not covered by any classification available in OWNER's office, shall be paid not less than the minimum rate of wages specified for the classification which most nearly corresponds with work to be performed by him, and that minimum wage rate shall be retroactive to the time of initial employment of the person in the classification.
- F. Pursuant to Labor Code Section 1773.1, per diem wages are deemed to include employer payments for health and welfare, pension, vacation, travel, subsistence, apprenticeship, and similar purposes.
  - a. Jobsite interviews may be conducted periodically by an independent enforcement agency throughout the duration of the project. The Contractor shall allow the enforcement agency access to the project and access to workers during working hours to confirm prevailing wage rates and apprenticeship requirements are complied with.
  - b. Request for exemption from prevailing wage requirements shall be made (3) three business days prior to bidding in time for the Architect to issue an addendum to communicate information to bidders. Request for exemption will not be accepted after above mentioned date.

- G. At appropriate conspicuous points on the site of the Project, CONTRACTOR shall post a schedule showing all determined minimum wage rates and all authorized deductions, if any, from unpaid wages actually earned.
- H. CONTRACTOR shall submit a breakdown of all labor costs for this Project by trade. This breakdown shall be for all labor that CONTRACTOR or any subcontractor supplies to the Project. This information shall be provided to OWNER before the first payment request after the Notice to Proceed has been issued. Failure to provide the labor cost breakdown will result in delay in processing the payment request until the complete cost breakdown is provided by CONTRACTOR and received and approved by OWNER. No other labor expenses will be considered unless approved in writing by OWNER.
- I. Pursuant to the provisions of Labor Code Section 1776, CONTRACTOR shall keep and shall cause each Subcontractor performing any portion of the work on the Project to keep an accurate payroll record, showing the name, address, social security number, work classification, straight time, and overtime hours worked each day and week, and the actual per diem wages paid to each journeyman, apprentice, worker, or other employee employed by CONTRACTOR in connection with the Work. Each payroll record shall contain or be verified by a written declaration that it is made under penalty of perjury, stating that (1) the information contained in the payroll record is true and correct, and (2) the employer has complied with the requirements of Sections 1771, 1811, and 1815 for any work performed by the employer's employees on the Project.
- J. The payroll records required under this article shall be certified. All contractors and subcontractors must furnish electronic certified payroll records (eCPR) directly to the Labor Commissioner using the Department of Industrial Relations (DIR) electronic certified payroll reporting system. All payroll records shall be available for inspection at all reasonable hours at CONTRACTOR's principal office on the following basis:
  - A certified copy of an employee's payroll record shall be made available for inspection or furnished to the employee or his or her authorized representative on request;
  - A certified copy of all required payroll records shall be made available for inspection or furnished upon request to a representative of OWNER, the Division of Labor Standards Enforcement, and/or the Division of Apprenticeship Standards of the Department of Industrial Relations;
  - c. A certified copy of all payroll records required under this article shall be made available for inspection or copies made upon request by the public; provided, however, that a request by the public shall be made through either OWNER, the Division of Apprenticeship Standards, or the Division of Labor Standards Enforcement. If the requested payroll records have not been provided pursuant to Paragraph above, prior to being provided the records, the requesting party shall reimburse the costs of preparation by CONTRACTOR, Subcontractors, and the entity through which the request was made. The public shall not be given access to the records at CONTRACTOR's principal office.
  - d. All contractors and subcontractors must furnish electronic certified payroll records (eCPR) directly to the Labor Commissioner using the Department of Industrial Relations (DIR) electronic certified payroll reporting system..

| payment   | request:   |                   |         |        |          |           |             |           |
|-----------|------------|-------------------|---------|--------|----------|-----------|-------------|-----------|
| İ,        |            |                   |         |        |          | (         | printed nar | me), the  |
| undersign | ned, am    | the               |         |        |          | _(positio | n in busine | ess) with |
| the autho | rity to ac | t for and on beha | If of _ |        |          |           |             | (name c   |
| business  | and/or     | CONTRACTOR),      | and     | verify | under    | penalty   | of perjury  | that the  |
| records   | or         | copies            | subi    | mitted | aı       | nd        | consisting  | C         |
|           |            |                   |         | (de    | scriptio | n, numb   | er of pages | are the   |

The form of certification shall be as follows and included with each and every

e.

#### TC – G-8 ALLIED HEALTH LAB REMODEL

|    | originals or true, full, and correct copies of the originals which depict the payro record(s) of the actual disbursements by way of cash, check, or whatever form to the individual or individuals named.  |
|----|--|
|    | Dated:Signature:   |
| K. | CONTRACTOR shall file a certified copy of the required payroll records with the entity requesting the records within 10 days after receipt of a written request. In the CONTRACTOR shall forfeit \$100 for each calendar day, or portion of each calendar day, for each worker until strict compliance is effectuated. Upon request by the Division of Apprenticeship Standards or the Division of Labor Standards Enforcement, these penalties shall be withheld from progress payments then due. |
| L. | Payroll records made available for inspection as copies and furnished upon request to the public by OWNER, the Division of Apprenticeship Standards, or the Division of Labo Standards Enforcement shall be marked or obliterated to prevent disclosure of an individual's name, address, and social security number. The name and address of CONTRACTOR shall not be marked or obliterated.   |
| M. | CONTRACTOR shall inform OWNER of the location of the payroll records, including the street address, city, and county, and within five working days shall provide a written notice of a change of location and address.   |
| N. | It shall be CONTRACTOR's responsibility to ensure compliance with the provisions of this article and the provisions of Labor Code Section 1776.  |
| O. | In addition to the above items, the Contractor shall submit a certificate, that the Contractor has in their possession certified weekly payroll records with each pay request for all workers on-site. Contractor shall maintain copies of contractor, subcontractor and subsubcontractor's certified payroll records on file for seven years after date of Notice of Completion.  |
| P. | Contractor shall certify with each and every pay application that current prevailing wage payroll documents are on file at their office.   |
|    | a. The form of certification to accompany each pay request application shall be as :   |
|    | I,   |
|    | StreetCityState  |

# 1.3 APPRENTICES

- A. CONTRACTOR acknowledges and agrees that the Contract Documents are governed by the provisions of Labor Code Section 1777.5 where applicable. It shall be CONTRACTOR's responsibility to ensure compliance with this article and with Labor Code Section 1777.5 for all apprenticing occupations.
- B. Apprentices of any crafts or trades may be employed, and when required by Labor Code Section 1777.5, shall be employed provided they are properly registered in full compliance

with the provisions of the Labor Code.

- C. Every apprentice shall be paid the prevailing rate of per diem wages for apprentices in the trade to which the apprentice is registered, and shall be employed only at the work of the craft or trade to which the apprentice is registered.
- D. Only apprentices as defined in Labor Code Section 3077 who are in training under apprenticeship standards that have been approved by the Chief of the Division of Apprenticeship Standards, and who are parties to written apprentice agreements under Chapter 4 (commencing at Section 3070), Division 3, of the Labor Code, are eligible to be employed on public works. The employment and training of each apprentice shall be in accordance with either (1) the apprenticeship standards and apprentice agreements under which the apprentice is in training, or (2) the rules and regulations of the California Apprenticeship Council.
- E. Pursuant to Labor Code Section 1777.5, CONTRACTOR and any subcontractors employing workers in any apprenticeship craft or trade performing any work under the Contract Documents shall employ apprentices in at least the ratio set forth in Labor Code Section 1777.5, and may apply to any apprenticeship program in the craft or trade that can provide apprentices to the project site for a certificate approving CONTRACTOR or Subcontractor under the applicable apprenticeship standards for the employment and training of apprentices in the area of industry affected.
- F. Prior to commencing work on the Project, CONTRACTOR shall submit contract award information to an applicable apprenticeship program that can supply apprentices to the project site. The information submitted shall include an estimate of journeyman hours to be performed on the Project, the number of apprentices proposed to be employed, and the approximate dates the apprentices would be employed. A copy of this information shall also be submitted to OWNER if requested. Within 60 days after concluding work on the Project, CONTRACTOR and all Subcontractors shall submit a verified statement of the journeyman and apprentice hours performed on the Project to the awarding body, if requested, and to the apprenticeship program. This information shall be public.
- G. If in performing any of the Work, CONTRACTOR employs journeymen or apprentices in any apprenticeable craft or trade, CONTRACTOR shall contribute to the California Apprenticeship Council the same amount that the Director of Industrial Relations determines is the prevailing amount of apprenticeship training contributions in the area of the Project, subject to any credits permitted by law.
- H. If CONTRACTOR or any Subcontractor is determined by the Chief of the Division of Apprenticeship Standards to have knowingly violated Labor Code Section 1777.5, it shall:
  - a. Forfeit as a civil penalty an amount not exceeding \$100 (\$300 for knowing subsequent violations) for each full calendar day of noncompliance. Notwithstanding Labor Code Section 1727, upon receipt of a determination that a civil penalty has been imposed by the Chief of the Division of Apprenticeship Standards, OWNER shall withhold the amount of the civil penalty from contract progress payments then due or to become due.
  - b. In lieu of the monetary penalty, for a first-time violation and with the concurrence of a specified apprenticeship program, the Chief of the Division of Apprenticeship Standards may order CONTRACTOR or any Subcontractor to provide apprentice employment equivalent to the work hours that would have been provided for apprentices during the period of noncompliance.
  - c. In the event CONTRACTOR or any Subcontractor is determined by the Chief of the Division of Apprenticeship Standards to have knowingly committed a serious violation of any provision of Section 1777.5, the Chief of the Division of

#### TC – G-8 ALLIED HEALTH LAB REMODEL

Apprenticeship Standards may also deny CONTRACTOR or any Subcontractor, and their responsible officers, the right to bid on or be awarded or perform work as a subcontractor on any public works contract for a period of up to one year for the first violation and up to three years for a subsequent violation.

CONTRACTOR or any Subcontractor (or responsible officer) shall have the right to obtain a review of the determination imposing a debarment or civil penalty as provided by law.

- I. CONTRACTOR and all Subcontractors shall comply with Labor Code Section 1777.6, which forbids certain discriminatory practices in the employment of apprentices.
- J. CONTRACTOR shall become fully acquainted with the law regarding apprentices prior to commencement of the work. Special attention is directed to Labor Code Sections 1777.5, 1777.6, and 1777.7, and Title 8, California Code of Regulations, Section 200 et seq. Questions may be directed to the State Division of Apprenticeship Standards, 455 Golden Gate Avenue, San Francisco, California.

#### 1.4 HOURS OF WORK

- A. CONTRACTOR shall furnish, and shall require all Subcontractors to furnish, sufficient forces to ensure the Work is prosecuted in accordance with the detailed project schedule without payment of overtime wage rates whenever possible.
- B. As provided in Article 3 (commencing at Section 1810), Chapter 1, Part 7, Division 2 of the Labor Code, eight hours of labor shall constitute a legal day of work. The time of service of any worker employed at any time by CONTRACTOR, or by any subcontractor, upon the Work or upon any part of the work contemplated by the Contract Documents is limited and restricted to eight hours per day and 40 hours during any one week. Upon completion of all hours worked in excess of eight hours per day, work shall be permitted upon this Project at not less than one and one-half times the basic rate of pay.
- C. CONTRACTOR shall keep, and shall cause all subcontractors to keep, an accurate record showing the name and actual hours worked each calendar day and each calendar week by each worker employed in connection with the Work or any part of the Work contemplated by the Contract Documents. The record shall be kept open at all reasonable hours to the inspection of OWNER and to the Division of Labor Standards Enforcement, Department of Industrial Relations.
- D. Saturdays, Sundays, holidays (including all OWNER designated holidays), and any day with work hours before 7:30 a.m. and/or after 4 p.m. shall be considered overtime for OWNER's representatives, consultants, and inspectors, and shall be compensated as such by CONTRACTOR per OWNER's submitted invoice. Such cost shall be billed to CONTRACTOR and deducted from subsequent progress payments or the final payment.
- E. As a penalty, CONTRACTOR shall pay \$100 to the Awarding Body for each worker employed by CONTRACTOR or by any subcontractor in the performance of the Contract Documents for each calendar day during which the worker is required or permitted to work more than eight hours in any calendar day and 40 hours in any one calendar week in violation of the provisions of Article 3 (commencing at Section 1810), Chapter 1, Part 7, Division 2 of the Labor Code.
- F. Any work performed before or after regular working hours or on Saturdays, Sundays, or holidays (including all OWNER designated holidays) shall be performed without additional expense to OWNER. Should inspection or testing services be necessary on a Saturday, Sunday, or holiday (including all OWNER designated holidays), CONTRACTOR shall pay all additional expenses incurred. Such cost shall be billed to CONTRACTOR and deducted from the next payment.

- G. CONTRACTOR shall anticipate work that would occur outside the normal work hours of 7:30 a.m. to 4 p.m. Such activities would include but are not limited to early morning concrete pours (because of hot weather), early or late material deliveries, required off-site inspections, or any other activity that would require the Project Inspector or OWNER personnel to work longer than an eight-hour day.
- H. The Project Inspector cannot be asked to leave the Project after eight hours of work so CONTRACTOR would not have to pay overtime. If the extended work day is a result of CONTRACTOR'S work, the Project Inspector will perform its DSA assigned work as necessary to assure the Project is kept on schedule and CONTRACTOR is responsible to pay all costs associated with fulfilling these DSA assignments, including the Project Inspector's overtime. These costs shall be billed to CONTRACTOR and deducted from subsequent progress payments or the final payment.

#### 1.5 PUBLIC WORKS CONTRACTOR REGISTRATION PROGRAM

- A. All contractors and subcontractors who bid or work on a public works project must register and pay an annual fee to DIR.
  - 1. Contractors will apply and pay the fee online and must meet minimum qualifications to be registered as eligible to bid and work on public works projects:
    - a. Must have workers' compensation coverage for any employees and only use subcontractors who are registered public works contractors.
    - b. Must have Contractors State License Board license if applicable to trade.
    - c. Must have no delinquent unpaid wage or penalty assessments owed to any employee or enforcement agency.
    - d. Must not be under federal or state debarment.
    - e. Must not be in prior violation of this registration requirement once it becomes effective. However, for the first violation in a 12 month period, a contractor may still qualify for registration by paying an additional penalty.
  - 2. The registration fee is not related to any project. It is more like a license that enables the registrant to bid on and perform public works.
  - 3. DIR will post a list of registered contractors and subcontractors on its website so that awarding bodies and contractors will be able to comply with requirements to only use registered contractors and subcontractors.
- B. No contractor or subcontractor may be listed on a bid proposal for a public works project (submitted on or after March 1, 2015) unless registered with the Department of Industrial Relations pursuant to Labor Code section 1725.5 [with limited exceptions from this requirement for bid purposes only under Labor Code section 1771.1(a)].
- C. No contractor or subcontractor may be awarded a contract for public work on a public works project (awarded on or after April 1, 2015) unless registered with the Department of Industrial Relations pursuant to Labor Code section 1725.5.
- D. This project is subject to compliance monitoring and enforcement by the Department of Industrial Relations.

PART 2 - PRODUCTS (Not Applicable)

# TC – G-8 ALLIED HEALTH LAB REMODEL

# PART 3 – EXECUTION

# 3.1 FURNISHING OF ELECTRONIC CERTIFIED PAYROLL RECORDS TO LABOR COMMISSIONER

- A. All contractors and subcontractors must furnish electronic certified payroll records (eCPR) directly to the Labor Commissioner using the Department of Industrial Relations (DIR) electronic certified payroll reporting system.
  - 1. CPRs will be furnished online by Contractors.

END OF SECTION 004318



AIA Document A201

# General Conditions of the Contract for Construction

THIS DOCUMENT HAS IMPORTANT LEGAL CONSEQUENCES; CONSULTATION WITH AN ATTORNEY IS ENCOURAGED WITH RESPECT TO ITS MODIFICATION

# 1987 EDITION TABLE OF ARTICLES

| 1. GENERAL PROVISIONS                               | 8. TIME                                       |
|---|---|
| 2. OWNER  | 9. PAYMENTS AND COMPLETION                    |
| 3. CONTRACTOR                                       | 10. PROTECTION OF PERSONS AND PROPERTY        |
| 4. ADMINISTRATION OF THE CONTRAC                    | T 11. INSURANCE AND BONDS                     |
| 5. SUBCONTRACTORS                                   | 12. UNCOVERING AND CORRECTION OF WORK         |
| 6. CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS | 13. MISCELLANEOUS PROVISIONS                  |
| 7. CHANGES IN THE WORK                              | 14. TERMINATION OR SUSPENSION OF THE CONTRACT |

This document has been approved and endorsed by the Associated General Contractors of America.

Copyright 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1967, 1970, 1976, © 1987 by The American Institute of Architects, 1735 New York Avenue, N.W., Washington, D.C., 20006. Reproduction of the material herein or substantial quotation of its provisions without written permission of the AIA violates the copyright laws of the United States and will be subject to legal prosecutions.

# INDEX

| Acceptance of Nonconforming Work 9.6.6, 9.9.3, 12.3                      | Building Permit  |
|--|--|
| Acceptance of Work 9.6.6, 9.8.2, 9.9.3, 9.10.1, 9.10.3                   | Capitalization   |
| Access to Work   | Carrifficate of Substantial Country  |
| Accident Prevention  | Certificate of Substantial Completion 9.8.2  |
| Acts and Omissions 3.2.1, 3.2.2, 3.3.2, 3.12.8, 3.18, 4.2.3, 4.3.2,      | Certificates for Payment 4.2.5, 4.2.9, 9.3.3, <b>9.4</b> , 9.5, 9.6.1,   |
| 4.3.9, 8.3.1, 10.1.4, 10.2.5, 13.4.2, 13.7, 14.1                         | 9.6.6, 9.7.1, 9.8.3, 9.10.1, 9.10.3, 13.7, 14.1.1.3, 14.2.4  |
| 4.5.9, 0.5.1, 10.1.4, 10.2.5, 15.4.2, 15./, 14.1                         | Certificates of Inspection, Testing or Approval 3.12.11, 13.5.4  |
| Addenda  | Ceruncales of Insurance . 02 2 0 10 2 11 12  |
| Additional Cost, Claims for 4.3.6, 4.3.7, 4.3.9, 6.1.1, 10.3             | Change Orders 1.1.1, 2.4.1, 3.8.2.4, 3.11, 4.2.8, 4.3.3, 5.2.3   |
| Additional Inspections and Testing 4.2.6, 9.8.2, 12.2.1, 13.5            | 7.1, 7.2, 7.3.2, 8.3.1, 9.3.1.1, 9.10.3, 11.3.1.2,   |
| Additional Time, Claims for 4.3.6, 4.3.8, 4.3.9, 8.3.2                   | 11 2 4 11 20 12 12   |
| ADMINISTRATION OF THE CONTRACT 3.3.3.4.9.4.9.5                           | Change Orders, Definition of   |
| Advertisement or Invitation to Bid 1.1.1                                 | Changes 7.1  |
| Aesthetic Effect   | CHANGES IN THE WORK 3.11, 4.2.8, 7, 8.3.1, 9.3.1.1, 10.1.3   |
| Allowances 3.8   | Claim Definition of  |
| All-risk Insurance   | Claim, Definition of   |
| Applications for Payment 4.2.5, 7.3.7, 9.2, 9.3, 9.4, 9.5.1, 9.6.3,      | Claims and Disputes  |
| 0.0.2 0.10.1 0.10.3 0.10.4 21.1.2 14.7.4                                 | 9.3.1.2, 9.3.3, 9.10.4, 10.1.4   |
| 9.8.3, 9.10.1, 9.10.3, 9.10.4, 11.1.3, 14.2.4                            | Ciaims and Timely Assertion of Claims  |
| Approvals 2.4, 3.3.3, 3.5, 3.10.2, 3.12.4 through 3.12.8, 3.18.3,        | Claims for Additional Cost 436 437 430 611 102   |
| 4.2.7, 9.3.2, 11.3.1.4, 13.4.2, 13.5                                     | Claims for Additional Time   |
| Arbitration 4.1.4, 4.3.2, 4.3.4, 4.4.4, 4.5,                             | Claims for Concealed or Unknown Conditions 4.3.6   |
| 8.3.1. 10.1.2. 11.3.9. 11.3.10   | Claims for Damages 3.18, 4.3.9, 6.1.1, 6.2.5, 8.3.2, 9.5.1.2, 10.1.4   |
| Architect 4.1  | Claims Subject to Arbitration  |
| Architect, Definition of   | Cleaning Up  |
| Architect, Extent of Authority 2.4, 3.12.6, 4.2, 4.3.2, 4.3.6,           | Commonoment of Professor Landson B. 15, 6.3  |
| 4.4, 5.2, 6.3, 7.1.2, 7.2.1, 7.3.6, 7.4, 9.2, 9.3.1,                     | Commencement of Statutory Limitation Period 13.7   |
| 9.4, 9.5, 9.6.3, 9.8.2, 9.8.3, 9.10.1, 9.10.3, 12.1, 12.2.1,             | Commencement of the Work, Conditions Relating to 2.1.2,  |
| 7.4, 7.2, 7.0.3, 7.0.2, 7.0.3, 7.10.1, 7.10.3, 12.1, 12.2.1,             | 2.2.1, 3.2.1, 3.2.2, 3.7.1, 3.10.1, 3.12.6, 4.3.7, 5.2.1,  |
| 13.5.1, 13.5.2, 14.2.2, 14.2.4   | 0.2.2. 8.1.2. 8.2.2. 9.2. 11.1.3. 11.3.6. 11.4.1   |
| Architect, Limitations of Authority and Responsibility . 3.3.3, 3.12.8,  | Commencement of the Work Definition of   |
| 3.12.11, 4.1.2, 4.2.1, 4.2.2, 4.2.3, 4.2.6, 4.2.7, 4.2.10, 4.2.12,       | Communications Facilitating Contract   |
| 4.2.13, 4.3.2, 5.2.1, 7.4, 9.4.2, 9.6.4, 9.6.6                           | Administration   |
| Architect's Additional Services and Expenses 2.4, 9.8.2,                 | Completion, Conditions Relating to 3.11, 3.15, 4.2.2, 4.2.9,   |
| 11.3.1.1, 12.2.1, 12.2.4, 13.5.2, 13.5.3, 14.2.4                         | 4.3.2, 9.4.2, 9.8, 9.9.1, 9.10, 11.2.5, 12.2.2, 12.2.1   |
| Architect's Administration of the Contract 4.2, 4.3.6,                   | COMPLETION, PAYMENTS AND 9   |
| 4.3.7, 4.4, 9.4, 9.5   | Completion, Substantial 4.2.9, 4.3.5.2, 8.1.1, 8.1.3, 8.2.3,   |
| Architect's Approvals 2.4, 3.5.1, 3.10.2, 3.12.6, 3.12.8, 3.18.3, 4.2.7  | 9.2.9, 4.5.5.2, 8.1.1, 8.1.3, 8.2.3,   |
| Architect's Authority to Reject Work 3.5.1, 4.2.6, 12.1.2, 12.2.1        | 9.8, 9.9.1, 12.2.2, 13.7   |
| Architect's Copyright 1.3  | Compliance with Laws 1.3, 3.6, 3.7, 3.13, 4.1.1, 10.2.2, 11.1,   |
| Architect's Decisions 4.2.6, 4.2.7, 4.2.11, 4.2.12, 4.2.13,              | 11.3, 13.1, 13.5.1, 13.5.2, 13.6, 14.1.1, 14.2.1.3   |
| 437 436 441 444 45 63 726 720 012 021                                    | Concealed or Unknown Conditions 4.3.6  |
| 4.3.2, 4.3.6, 4.4.1, 4.4.4, 4.5, 6.3, 7.3.6, 7.3.8, 8.1.3, 8.3.1,        | Conditions of the Contract   |
| 9.2, 9.4, 9.5.1, 9.8.2, 9.9.1, 10.1.2, 13.5.2, 14.2.2, 14.2.4            | Consent, Written   |
| Architect's Inspections 4.2.2, 4.2.9, 4.3.6, 9.4.2, 9.8.2,               | 4.3.4, 4.5.5, 9.3.2, 9.8.2, 9.9.1, 9.10.2, 9.10.3, 10.1.2, 10.1.3,   |
| 9.9.2, 9.10.1, 13.5  | . 1131 11314 11311 123 1243  |
| Architect's Instructions 4.2.6, 4.2.7, 4.2.8, 4.3.7, 7.4.1, 12.1, 13.5.2 | CONSTRUCTION BY OWNER OR BY SEPARATE   |
| Architect's Interpretations 4.2.11, 4.2.12, 4.3.7                        | CONTRACTORS  |
| Architect's On-Site Observations 4.2.2, 4.2.5, 4.3.6, 9.4.2,             | Construction Change Directive, Definition of 7.3.1   |
| 9.5.1, 9.10.1, 13.5  | Construction Change Directives 1.1.1, 4.2.8, 7.1, <b>7.3</b> , 9.3.1.1   |
| Architect's Project Representative                                       | Construction Schedules, Contractor's   |
| Architect's Relationship with Contractor 1.1.2, 3.2.1, 3.2.2,            | Contingent Assignment of Subcontracts 5.4  |
| 3.3.3, 3.5.1, 3.7.3, 3.11, 3.12.8, 3.12.11, 3.16, 3.18, 4.2.3, 4.2.4,    | Continuing Contract Devices  |
| 4.2.6, 4.2.12, 5.2, 6.2.2, 7.3.4, 9.8.2, 11.3.7, 12.1, 13.5              | Continuing Contract Performance 4.3.4  |
| Architect's Relationship with Subcontractors 1,1.2, 4.2.3, 4.2.4,        | Contract, Definition of  |
|  | CONTRACT, TERMINATION OR   |
| 4.2.6, 9.6.3, 9.6.4, 11.3.7  | SUSPENSION OF THE 4.3.7, 5.4.1.1, 14   |
| Architect's Representations 9.4.2, 9.5.1, 9.10.1                         | Contract Administration  |
| Architect's Site Visits 4.2.2, 4.2.5, 4.2.9, 4.3.6, 9.4.2, 9.5.1,        | Contract Award and Execution, Conditions Relating to 3.7.1,  |
| 9.8.2, 9.9.2, 9.10.1, 13.5   | 3 10 5 2 9 2 11 1 3 11 2 6 11 4 1  |
| Asbestos 10.1  | Contract Documents, The  |
| Attorneys' Fees  | Contract Documents, Copies Furnished and Use of 1.3, 2.2.5, 5.3  |
| Award of Separate Contracts 6.1.1  | Contract Documents, Definition of  |
| Award of Subcontracts and Other Contracts for                            | Contract Documents, Definition of  |
| Portions of the Work   | Contract Performance During Arbitration 4.3.4, 4.5.3   |
| Basic Definitions  | Contract Sum   |
| Bidding Requirements   | 0.1.3. 7.2. 7.3. <b>9.1.</b> 9.7. 11.3.1. 12.2.4. 12.3. 14.2.4   |
| Poller and Machinery Incomes   | Contract Sum. Definition of  |
| Boiler and Machinery Insurance   | Contract Time 4.3.6, 4.3.8, 4.4.4, 7.2.1.3, 7.3,   |
| Bonds, Lien  | 921 921 97 1711  |
| Bonds, Performance and Payment 7.3.6.4, 9.10.3, 11.3.9, 11.4             | Contract Time, Definition of   |
|  | a not to the state of the state |

| CONTRACTOR 3  | Emergencies   |
|---|---|
| Contractor, Definition of 3.1, 6.1.2                                    | Employees, Contractor's 3.3.2, 3.4.2, 3.8.1, 3.9, 3.18.1  |
| Contractor's Bid  | 3.18.2, 4.2.3, 4.2.6, 8.1.2, 10.2, 10.3, 11.1.1, 14.2.1.1   |
| Contractor's Construction Schedules                                     | Equipment, Labor, Materials and 1.1.3, 1.1.6, 3.4, 3.5.1.   |
| Contractor's Employees 3.3.2, 3.4.2, 3.8.1, 3.9, 3.18, 4.2.3,           | 3.8.2, 3.12.3, 3.12.7, 3.12.11, 3.13, 3.15.1, 4.2.7   |
| 4.2.6, 8.1.2, 10.2, 10.3, 11.1.1, 14.2.1.1                              | 6.2.1, 7.3.6, 9.3.2, 9.3.3, 11.3, 12.2.4, 14  |
| Contractor's Liability Insurance  | Execution and Progress of the Work 1.1.3, 1.2.3, 3.2, 3.4.1.  |
| Contractor's Relationship with Separate Contractors                     | 3.5.1, 4.2.2, 4.2.3, 4.3.4, 4.3.8, 6.2.2, 7.1.3,  |
| and Owner's Forces 2.2.6, 3.12.5, 3.14.2, 4.2.4, 6, 12.2.5              | 7.3.9, 8.2, 8.3, 9.5, 9.9.1, 10.2, 14.2, 14.3   |
| Contractor's Relationship with Subcontractors 1.2.4, 3.3.2,             | Execution, Correlation and Intent of the  |
| 3.18.1, 3.18.2, 5.2, 5.3, 5.4, 9.6.2, 11.3.7, 11.3.8, 14.2.1.2          | Contract Documents  |
| Contractor's Relationship with the Architect 1.1.2, 3.2.1, 3.2.2,       | Extensions of Time  |
| 3.3.3, 3.5.1, 3.7.3, 3.11, 3.12.8 3.16, 3.18, 4.2.3, 4.2.4, 4.2.6,      | Failure of Payment by Contractor 9.5.1.3, 14.2.1.2  |
| 4.2.12, 5.2, 6.2.2, 7.3.4, 9.8.2, 11.3.7, 12.1, 13.5                    | Failure of Payment by Owner   |
| ontractor's Representations 1.2.2, 3.5.1, 3.12.7, 6.2.2, 8.2.1, 9.3.3   | Faulty Work (See Defective or Nonconforming Work)   |
| ontractor's Responsibility for Those                                    | Final Completion and Final Payment 4.2.1, 4.2.9, 4.3.2,   |
| Performing the Work 3.3.2, 3.18, 4.2.3, 10                              | 4.3.5, <b>9.10,</b> 11.1.2, 11.1.3, 11.3.5, 12.3.1, 13.7  |
| ontractor's Review of Contract Documents 1.2.2, 3.2, 3.7.3              | Financial Arrangements, Owner's   |
| ontractor's Right to Stop the Work 9.7                                  | Fire and Extended Coverage Insurance  |
| ontractor's Right to Terminate the Contract                             |   |
| ontractor's Submittals 3.10, 3.11, 3.12, 4.2.7, 5.2.1, 5.2.3,           | GENERAL PROVISIONS 1  |
| 7.3.6, 9.2, 9.3.1, 9.8.2, 9.9.1, 9.10.2,                                | Governing Law   |
| 9.10.3, 10.1.2, 11.4.2, 11.4.3  | Guarantees (See Warranty and Warranties)  |
| ontractor's Superintendent  | Hazardous Materials   |
| ontractor's Supervision and Construction Procedures 1.2.4,              | Identification of Contract Documents  |
| 3.3, 3.4, 4.2.3, 8.2.2, 8.2.3, 10                                       | Identification of Subcontractors and Suppliers 5.2.1  |
| ontractual Liability Insurance  | Indemnification 3.17, 3.18, 9.10.2, 10.1.4, 11.3.1.2, 11.3.7  |
| oordination and Correlation 1.2.2, 1.2.4, 3.3.1,                        | Information and Services Required of the Owner 2.1.2.2.2.   |
| 3.10, 3.12.7, 6.1.3, 6.2.1  | 4.3.4, 6.1.3, 6.1.4, 6.2.6, 9.3.2, 9.6.1, 9.6.4, 9.8.3, 9.9.2,  |
| opies Furnished of Drawings and Specifications 1.3, 2.2.5, 3.11         | 9.10.3, 10.14, 11.2, 11.3, 13.5, 1.13.5, 2  |
| orrection of Work 2.3, 2.4, 4.2.1, 9.8.2,                               | Injury or Damage to Person or Property 439  |
| 001 1717 177 12712  | Inspections   |
| ost, Definition of  | 4.2.6, 4.2.9, 4.3.6, 9.4.2, 9.8.2, 9.9.2, 9.10.1, 13.5  |
| osts 2.4, 3.2.1, 3.7.4, 3.8.2, 3.15.2, 4.3.6, 4.3.7, 4.3.8.1, 5.2.3,    | Instructions to Bidders   |
| 6.1.1, 6.2.3, 6.3, 7.3.3.3, 7.3.6, 7.3.7, 9.7, 9.8.2, 9.10.2, 11.3.1.2, | Instructions to the Contractor 3.8.1.4.2.8.5.2.1.7.12.1.13.5.2  |
| 11.3.1.3, 11.3.4, 11.3.9, 12.1, 12.2.1, 12.2.4, 12.2.5, 13.5, 14        | Insurance 4.3.9.6.1.1, 7.3.6.4, 9.3.2, 9.8.2, 9.9.1, 9.10.2, 11   |
| tting and Patching  | Insurance, Boiler and Machinery 11 3 2  |
| mage to Construction of Owner or Separate Contractors 3.14.2,           | Insurance, Contractor's Liability 11 1  |
| 6.2.4, 9.5.1.5, 10.2.1.2, 10.2.5, 10.3, 11.1, 11.3, 12.2.5              | Insurance, Effective Date of 822 11 12  |
| mage to the Work 3.14.2, 9.9.1, 10.2.1.2, 10.2.5, 10.3, 11.3            | Insurance, Loss of Use  |
| mages, Claims for 3.18, 4.3.9, 6.1.1, 6.2.5, 8.3.2, 9.5.1.2, 10.1.4     | Insurance, Owner's Liability  |
| mages for Delay 6.1.1, 8.3.3, 9.5.1.6, 9.7                              | Insurance, Property   |
| te of Commencement of the Work, Definition of 8.1.2                     | Insurance, Stored Materials 9.3.2, 11.3.1.4   |
| te of Substantial Completion, Definition of 8.1.3                       |   |
| y, Definition of  | INSURANCE AND BONDS   |
| cisions of the Architect 4.2.6, 4.2.7, 4.2.11, 4.2.12, 4.2.13,          | Insurance Companies, Consent to Partial Occupancy 0 0 1 11 3 11   |
| 4.3.2, 4.3.6, 4.4.1, 4.4.4, 4.5, 6.3, 7.3.6, 7.3.8, 8.1.3, 8.3.1, 9.2,  | Insurance Companies, Settlement with 11 3 10  |
| 9.4, 9.5.1, 9.8.2, 9.9.1, 10.1.2, 13.5.2, 14.2.2, 14.2.4                | Intent of the Contract Documents 1.2.3, 3.12.4,   |
| cisions to Withhold Certification 9.5, 9.7, 14.1.1.3                    | 4.2.6, 4.2.7, 4.2.12, 4.2.13, 7.4   |
| fective or Nonconforming Work, Acceptance,                              | Interest 13.6   |
| Rejection and Correction of 2.3, 2.4, 3.5.1, 4.2.1,                     | Interpretation 1.2.5, 1.4, 1.5, 4.1.1, 4.3.1, 5.1, 6.1.2, 8.1.4   |
| 4.2.6, 4.3.5, 9.5.2, 9.8.2, 9.9.1, 10.2.5, 12, 13.7.1.3                 | Interpretations, Written  |
| fective Work, Definition of   | Joinder and Consolidation of Claims Required  |
| finitions 1.1, 2.1.1, 3.1, 3.5.1, 3.12.1, 3.12.2, 3.12.3, 4.1.1,        | Judgment on Final Award 4.5.1, 4.5.4.1, 4.5.7   |
| 4.3.1, 5.1, 6.1.2, 7.2.1, 7.3.1, 7.3.6, 8.1, 9.1, 9.8.1                 | Labor and Materials, Equipment 1.1.3, 1.1.6, 3.4, 3.5.1, 3.8.2,   |
| lays and Extensions of Time 4.3.1, 4.3.8.1, 4.3.8.2,                    | 2 12 7 2 12 7 3 12 7 3 12 13 13 13 13 13 13 13 13 13 13 13 13 13  |
| 611 622 721 721 72 6 72 72 72 72 72 72 72 72 72 72 72 72 72             | 3.12.2, 3.12.3, 3.12.7, 3.12.11, 3.13, 3.15.1,  |
| 6.1.1, 6.2.3, 7.2.1, 7.3.1, 7.3.4, 7.3.5, 7.3.8,                        | 4.2.7, 6.2.1, 7.3.6, 9.3.2, 9.3.3, 12.2.4, 14   |
| 7.3.9, 8.1.1, <b>8.3</b> , 10.3.1, 14.1.1.4                             | Labor Disputes  |
| putes   | Laws and Regulations 1.3, 3.6, 3.7, 3.13, 4.1.1, 4.5.5, 4.5.7,  |
| cuments and Samples at the Site   | 9.9.1, 10.2.2, 11.1, 11.3, 13.1, 13.4, 13.5.1, 13.5.2, 13.6   |
| wings Definition of   | Henry State |
| wings, Definition of 1.1.5  | Liens 2.1.2, 4.3.2, 4.3.5.1, 8.2.2, 9.3.3, 9.10.2   |
| awings, Definition of   | Liens   |
| awings, Definition of   | Liens   |
| awings, Definition of   | Liens   |

| Limitations of Liability 2.3, 3.2.1, 3.5.1, 3.7.3, 3.12.8, 3.12.11,  | Owner's Right to Perform Construction and to  |
|--|---|
| 3.17, 3.18, 4.2.6, 4.2.7, 4.2.12, 6.2.2, 9.4.2, 9.6.4, 9.10.4,   | Award Separate Contracts  |
| 10.1.4, 10.2.5, 11.1.2, 11.2.1, 11.3.7, 13.4.2, 13.5.2   | OWNER'S HIGHT to Stop the Work  |
| Limitations of Time, General 2.2.1, 2.2.4, 3.2.1, 3.7.3,   | Owner's Right to Suspend the Work   |
| 3.8.2, 3.10, 3.12.5, 3.15.1, 4.2.1, 4.2.7, 4.2.11, 4.3.2,  | Owner's Right to Terminate the Contract 14.2  Ownership and Use of Architect's Drawings, Specifications |
| 4.3.3, 4.3.4,4.3.6, 4.3.9, 4.5.4.2, 5.2.1, 5.2.3, 6.2.4, 7.3.4, 7.4,   | and Other Documents   |
| 8.2, 9.5, 9.6.2, 9.8, 9.9, 9.10, 11.1.3, 11.3.1, 11.3.2, 11.3.5, 11.3.6, 12.2.1, 12.2.2, 13.5, 13.7  | Parial Occupancy or Use 0.66 o g 11 2 11  |
| Limitations of Time, Specific 2.1.2, 2.2.1, 12.2.2, 15.3, 15.7   | Patching, Cutting and   |
| 3.15.1, 4.2.1, 4.2.11, 4.3, 4.4, 4.5, 5.3, 5.4, 7.3.5, 7.3.9, 8.2,   | Patents, Hovalties and  |
| 9.2, 9.3.1, 9.3.3, 9.4.1, 9.6.1, 9.7, 9.8.2, 9.10.2, 11.1.3, 11.3.6  | Payment, Applications for 4.2.5, 9.2, <b>9.3</b> , 9.4, 9.5.1, 9.8.3, 9.10.1, 9.10.3, 9.10.4, 14.2.4    |
| 11.3.10, 11.3.11, 12.2.2, 12.2.4, 12.2.6, 13.7, 14   | Payment, Centicates for 425 420 022 04 05   |
| Loss of Use Insurance  | 9.6.1. 9.6.6. 9.7.1. 9.8 3 9 10 1 9 10 3 13 7 14 1 1 2 14 2 4   |
| 9.3.1, 9.3.1.2, 9.3.3, 9.4.2, 9.6.5, 9.10.4  | rayment, ranure of  |
| Materials, Hazardous   | 97 9102 14112 14212   |
| Materials, Labor, Equipment and 1.1.3, 1.1.6, 3.4, 3.5.1, 3.8.2.   | Payment, Final  |
| 3.12.2, 3.12.3, 3.12.7, 3.12.11, 3.13, 3.15.1, 4.2.7, 6.2.1,   | 11.1.3, 11.3.5, 12.3.1 Payment Bond, Performance Bond and   |
| 7.3.6, 9.3.2, 9.3.3, 12.2.4, 14  | 9.10.3. 11.3.9. 11.4  |
| Means, Methods, Techniques, Sequences and  | 9.10.3, 11.3.9, <b>11.4</b> Payments, Progress  |
| Procedures of Construction   | 002 0101 12 / 44 - 4  |
| MISCELLANEOUS PROVISIONS 13  | PAYMENTS AND COMPLETION 9, 14.2.3 Payments to Subcontractors  |
| Modifications, Definition of   | 062 063 064 1120 14212  |
| Modifications to the Contract 1.1.1, 1.1.2, 3.7.3, 3.11,   | PCB   |
| 4.1.2, 4.2.1, 5.2.3, 7, 8.3.1, 9.7 Mutual Responsibility 6.2   | Performance Bond and Payment Bond   |
| Nonconforming Work, Acceptance of 12.3   | 0102 1120 117   |
| Nonconforming Work, Rejection and Correction of 2.3.1,   | Permits, Fees and Notices 2.2.3, 3.7, 3.13, 7.3.6.4, 10.2.2 PERSONS AND PROPERTY, PROTECTION OF 10      |
| 4.3.5, 9.5.2, 9.8.2, 12, 13.7.1.3  | Polychlorinated Biphenyl  |
| Notice 2.3, 2.4, 3.2.1, 3.2.2, 3.7.3, 3.7.4, 3.9, 3.12.8,  | Product Data, Definition of   |
| 3.12.9, 3.17, 4.3, 4.4.4, 4.5, 5.2.1, 5.3, 5.4.1.1, 8.2.2, 9.4.1,  | Product Data and Samples, Shop Prawings 2 11 2 40 4 2 7   |
| 9.5.1, 9.6.1, 9.7, 9.10, 10.1.2, 10.2.6, 11.1.3, 11.3, 12.2.2, 12.2.4, 13.3, 13.5.1, 13.5.2, 14  | Progress and Completion 422 424 00  |
| Notice, Written  | Progress Payments 4.2.4, 4,5.4, 6.2 4.3.4, 9.3,   |
| 4.4.4, 4.5, 5.2.1, 5.3, 5.4.1.1, 8.2.2, 9.4.1, 9.5.1, 9.7, 9.10,   | Project, Definition of the  |
| 10.1.2, 10.2.6, 11.1.3, 11.3, 12.2.2, 12.2.4, <b>13.3</b> , 13.5.2, 14   | Project Manual, Delinition of the   |
| Notice of Testing and Inspections 13.5.1, 13.5.2  Notice to Proceed 8.2.2  | Project Manuals   |
| Notice to Proceed  | Project Representatives   |
| Notices, Permits, Fees and 2.2.3, 3.7, 3.13, 7.3.6.4, 10.2.2<br>Observations, Architect's On-Site 4.2.2, 4.2.5,  | Property Insurance  |
| 4.3.6, 9.4.2, 9.5.1, 9.10.1, 13.5  | Regulations and Laws  |
| Observations, Contractor's 1.2.2, 3.2.2  | 4.5.7, 10.2.2, 11.1, 11.3, 13.1, 13.4, 13.5, 13.5, 13.5, 13.6, 14.                                      |
| Occupancy  | Rejection of Work   |
| On-Site Inspections by the Architect 4.2.2, 4.2.9, 4.3.6,  | Releases of Waivers and Liens   |
| 9.4.2, 9.8.2, 9.9.2, 9.10.1<br>On-Site Observations by the Architect 4.2.2, 4.2.5, 4.3.6,  | Representations   |
| 9.4.2, 9.5.1, 9.10.1, 13.5   | Representatives   |
| Orders, Written 2.3, 3.9, 4.3.7, 7, 8.2.2, 11.3.9, 12.1,   | 411 471 4210 511 512 1221   |
| 12.2, 13.5.2, 14.3.1   | nestitution of Claims and Disputes  |
| OWNER  | Responsibility for Those Performing the Work  |
| Owner, Definition of   | 4.2.3, 6.1.3, 6.2, 10<br>Retainage  |
| Owner, Information and Services Required of the 2.1.2,   | Heview of Contract Documents and Field  |
| <b>2.2</b> , 4.3.4, 6, 9, 10.1.4, 11.2, 11.3, 13.5.1, 14.1.1.5, 14.1.3<br>Owner's Authority 3.8.1, 4.1.3, 4.2.9, 5.2.1, 5.2.4, 5.4.1,  | Conditions by Contractor  |
| 7.3.1, 8.2.2, 9.3.1, 9.3.2, 11.4.1, 12.2.4, 13.5.2, 14.2, 14.3.1   | Review of Contractor's Submittals by  |
| Owner's Financial Capability 2.2.1, 14.1.1.5   | Owner and Architect   |
| Owner's Liability Insurance  | 4.2.7, 4.2.9, 5.2.1, 5.2.3, 9.2, 9.8.2<br>Review of Shop Drawings, Product Data                         |
| Owner's Loss of Use Insurance  | and Samples by Contractor   |
| Owner's Relationship with Subcontractors   | Mights and Hemedies   |
| 5.2.1, 5.4.1, 9.6.4<br>Owner's Right to Carry Out the Work 2.4, 12.2.4, 14.2.2.2   | 4.2.6, 4.3.6, 4.5, 5.3, 6.1, 6.3, 7.3.1, 8.3.1, 9.5.1, 9.7, 10.2.5.                                     |
| Owner's Right to Clean Up  | 10 3 17 77 177 47 4 14  |
| 5.55 - 5. | Royalties and Patents 3.17  |
| ALA DOCUMENT ADDS - CONTROL CONTROL  |   |

| Rules and Notices for Arbitration   |              | 4 5 0            |
|---|--------------|------------------|
| Codeby of Demonstrated Process  |              | 4,0,2            |
| Safety of Persons and Property  |              | 10.2             |
| Safety Precautions and Programs   | 423 4        | 27 10.1          |
| Samples, Definition of  |              | 2 17 7           |
| Complex Char Drawless Destroy Date  |              | 5.12.5           |
| Samples, Shop Drawings, Product Data and  | 3.11, 3      | 1.12, 4.2.7      |
| Samples at the Site, Documents and  |              | 3.11             |
| Schedule of Values  | Wallet Break | 00 021           |
| Pahadulas Carataratar   |              | 3.2, 9.3.1       |
| Schedules, Construction   |              | 3.10             |
| Separate Contracts and Contractors 1.1  | .4. 3.1      | 4.2. 4.2.4.      |
| 4.5.5, 6, 11.3.   | 7 12 1       | 2 12 2 5         |
| Chan Desiglar Definition of   | 1, 12.1      | .2, 12.2.)       |
| Shop Drawings, Definition of  |              | 3.12.1           |
| Shop Drawings, Product Data and Samples   | 3.11, 3      | .12. 4.2.7       |
| Site. Use of  | 112 6        | 11621            |
| Site Inspections 1.2.2, 3.3.4, 4.2.2, 4.2.9, 4.3.6, 9.  | 2.10,0       | .1.1, 0.2.1      |
| one mapecdons 1.2.2, 5.5.4, 4.2.2, 4.2.9, 4.5.6, 9.   | 8.2, 9.      | 10.1, 13.5       |
| Site Visits, Architect's 4.2.2, 4.  | 2.5, 4.      | 2.9, 4.3.6.      |
| 9.4.2, 9.5.1, 9.8.2, 9.5  | 07 0         | 101 135          |
| Special Inspections and Testing 4.  | 76 17        | 0.1, 13.5        |
| Special haptenons and resting   | 2.0, 12      | .2.1, 13.5       |
| Specifications, Definition of the   |              | 1.1.6            |
| Specifications, The 1.1.1, 1.1.6, 1.1.7   | 1.2.4        | 13311            |
| Statutes of Limitations 4.5.  | 42 12        | 7.5,5.11         |
| Statutes of Editionations   | 4.2, 12      | .2.6, 13./       |
| Stopping the Work 2.3, 4.3.7, 9.7, 1  | 0.1.2,       | 10.3, 14.1       |
| Stored Materials 6.2.1, 9.3.2, 10.2.1.2   | 1131         | 4 1274           |
| Subcontractor, Definition of  |              | = 1 1            |
| CURCONTRACTORS  |              | 5.1.1            |
| SUBCONTRACTORS  |              | 5                |
| Subcontractors, Work by 1.2   | 4. 3.3.      | 2.3.12.1         |
|   | 137          | F 2 F 4          |
| Subcontractual Dalations F. a. c.   | 4.4.)        | 1 3.3, 3.4       |
| Subcontractual Relations 5.3, 5.4   | 1,9.3.1      | .2, 9.6.2,       |
| Subcontractual Relations  | 14.2.1.      | 2. 14.3.2        |
| Submittals 1.3, 3.2.3, 3.10, 3.11, 3.12, 4.2  | 7 5 2        | 1 5 7 2          |
| 7.3.6, 9.2, 9.3.1, 9.8.2, 9.9.1, 9.10.2, 9.10.3   |              | .1, J.2.J,       |
| 7.3.0, 9.2, 9.3.1, 9.0.2, 9.9.1, 9.10.2, 9.10.3   | , 10.1.      | 2, 11.1.3        |
| Subrogation, Waivers of 6.1.1   | , 11.3.      | 5, <b>11.3.7</b> |
| Substantial Completion 4.2.9, 4.3.5   | 1281         | 1813             |
| 8.2.3, <b>9.8,</b> 9.9.1, 12.2  | 1 17         | 7 7 7 7          |
| Cubetamental Community D. C. 11.  | .1, 12.      | 2.2, 13.7        |
| Substantial Completion, Definition of   |              | 9.8.1            |
| Substitution of Subcontractors  | 5 7          | 2 5 7 4          |
| Substitution of the Architect   |              | 413              |
| Substitutions of Materials  |              | 4.1.5            |
| Substitutions of Materials  |              | 3.5.1            |
| Sub-subcontractor, Definition of  |              | 5.1.2            |
| Subsurface Conditions   |              | 636              |
| Successore and Anglana  | 60% 0000 000 | 4,5,0            |
| Successors and Assigns  |              | . 13.2           |
| Superintendent  | 3.9          | 9, 10.2.6        |
| Supervision and Construction Procedures   | 124          | 3334             |
| 4.2.3, 4.3.4, 6.1.3, 6.2.4, 7.1.3, 7.3.4, 8.2, 8.   | 2 1 1        | 12 14            |
| The Direction of the Control of the | J.1, IC      | 1, 12, 14        |
| Surety 4.4.1, 4.4.4, 5.4.1.2, 9.10.2  | , 9.10.      | 3, 14.2.2        |
| Surety, Consent of 9.9.1  | 9.10         | 9 10 3           |
| Surveys   | 777          | 2 10 2           |
| *************************   | . 2.2.2      | 2, 3.18.3        |
|   |              |                  |

| Suspension by the Owner for Convenience                             | 14.3  |
|---|-------|
| Suspension of the Work 42754214114                                  | 1 1 2 |
| Suspension or Termination of the Contract 4.3.7, 5.4.1.1            | 1.1.  |
| Taxes 3.6, 7.3  | 114   |
| Termination by the Contractor                                       | .0.4  |
| Termination by the Owner for Cause                                  | 4.1   |
| Termination by the Owner for Cause 5.4.1.1, 1                       | 4.2   |
| Termination of the Architect  | .1.3  |
| Termination of the Contractor                                       | .2.2  |
| TERMINATION OR SUSPENSION OF THE CONTRACT                           | 14    |
| Tests and inspections 3.3.3, 4.2.6, 4.2.9, 9.4.2, 12.2 1, 1         | 3.5   |
| TIME  | . 8   |
| Time, Delays and Extensions of 438721                               | PR    |
| Time Limits. Specific 7122213421021121                              | C 1   |
| 4.2.1, 4.2.11, 4.3, 4.4, 4.5, 5.3, 5.4, 7.3.5, 7.3.0, 8.2, 0.2, 0.3 | 3 1   |
| 9.3.3, 9.4.1, 9.6.1, 9.7, 9.8.2, 9.10.2, 11.1.3, 11.3.6, 11.3       | 10    |
| 11.3.11 12.22 12.24 12.26 12.7                                      | 16    |
| Time Limits on Claims   | 15    |
| Title to Work   | 7.7   |
| UNCOVERING AND CORRECTION OF WORK 9.3.2, 9.                         | 2.2   |
| Uncovering of Work  | 12    |
| Heforesgen Conditions   | 2.1-  |
| Unit Prices 4.3.6, 8.3.1, I   | 0.1   |
| Unit Prices   | 3.2   |
| Use of Documents  | 5.3   |
| Use of Site   | 2.1   |
| Values, Schedule of 9.2, 9.   | 3.1   |
| waiver of Claims: Pinal Payment 435 451 011                         | 7 2   |
| Waiver of Claims by the Architect                                   | 1 2   |
| Walver of Claims by the Contractor 0 10 4 11 2 7 12                 | 47    |
| waiver of Claims by the Owner                                       | 3.    |
| 9103 1133 1135 1127 12  | ( )   |
| Walver of Liens 0 10  | 12    |
| Walvers of Subrogation 611 1125 112                                 | 7     |
| Warranty and Warranties 3.5, 4.2                                    | 0     |
| 4.3.5.3, 9.3.3, 9.8.2, 9.9.1, 12.2.2, 13.7.1                        | ٠٧,   |
| Weather Delays  | 2 2   |
| When Arbitration May Be Demanded 4.5                                | 1.2   |
| Work, Definition of   | .4    |
| Written Concept   | .3    |
| Written Consent 1.3.1, 3.12.8, 3.14.2, 4.1.2, 4.3.                  | 4,    |
| 4.5.5, 9.3.2, 9.8.2, 9.9.1, 9.10.2, 9.10.3, 10.1.2, 10.1.           | 3,    |
| 11.3.1, 11.3.1.4, 11.3.11, 13.2, 13.4                               | .2    |
| Written Interpretations   | .7    |
| WITHER NOTICE 23 24 39 312 8 312 8 312 6 43 44                      | A     |
| 4.3. 3.4.1. 3.3. 3.4.1.1. 8.2.2. 9.4.1. 9.5.1. 9.7. 9.10. 10.1      | 2     |
| 10.2.0, 11.1.3, 11.3, 12.2.2, 12.2.4 13.3, 13.5.2 1                 | 4     |
| Written Orders 23 20 42   | 7     |
| 7, 8.2.2, 11.3.9, 12.1, 12.2, 13.5.2, 14.3                          | 1     |
| **************************************                              |       |

# GENERAL CONDITIONS OF THE CONTRACT FOR CONSTRUCTION

#### ARTICLE 1

#### **GENERAL PROVISIONS**

#### 1.1 BASIC DEFINITIONS

#### 1.1.1 THE CONTRACT DOCUMENTS

The Contract Documents consist of the Agreement between Owner and Contractor (hereinafter the Agreement), Conditions of the Contract (General, Supplementary and other Conditions), Drawings, Specifications, addenda issued prior to execution of the Contract, other documents listed in the Agreement and Modifications issued after execution of the Contract. A Modification is (1) a written amendment to the Contract signed by both parties, (2) a Change Order, (3) a Construction Change Directive or (4) a written order for a minor change in the Work issued by the Architect. Unless specifically: enumerated in the Agreement, the Contract Documents do not include other documents such as bidding requirements (advertisement or invitation to bid, instructions to Bidders, sample forms, the Contractor's bid or portions of addenda relating to bidding-requirements).

#### 1.1.2 THE CONTRACT

The Contract Documents form the Contract for Construction. The Contract represents the entire and integrated agreement between the parties hereto and supersedes prior negotiations, representations or agreements, either written or oral. The Contract may be amended or modified only by a Modification. The Contract Documents shall not be construed to create a contractual relationship of any kind (1) between the Architect and Contractor, (2) between the Owner and a Subcontractor or Subsubcontractor or (3) between any persons or entities other than the Owner and Contractor. The Architect shall, however, be entitled to performance and enforcement of obligations under the Contract intended to facilitate performance of the Architect's duties.

#### 1.1.3 THE WORK

The term "Work" means the construction and services required by the Contract Documents, whether completed or partially completed, and includes all other labor, materials, equipment and services provided or to be provided by the Contractor to fulfill the Contractor's obligations. The Work may constitute the whole or a part of the Project.

#### 1.1.4 THE PROJECT

The Project is the total construction of which the Work performed under the Contract Documents may be the whole or a part and which may include construction by the Owner or by separate contractors.

#### 1.1.5 THE DRAWINGS

The Drawings are the graphic and pictorial portions of the Contract Documents, wherever located and whenever issued, showing the design, location and dimensions of the Work, generally including plans, elevations, sections, details, schedules and diagrams.

#### 1.1.6 THE SPECIFICATIONS

The Specifications are that portion of the Contract Documents consisting of the written requirements for materials, equip-

ment, construction systems, standards and workmanship for the Work, and performance of related services.

# 1.1.7 THE PROJECT MANUAL

The Project Manual is the volume usually assembled for the Work which may include the bidding requirements, sample forms, Conditions of the Contract and Specifications.

# 1.2 EXECUTION, CORRELATION AND INTENT

- 1.2.1 The Contract Documents shall be signed by the Owner and Contractor as provided in the Agreement. If either the Owner or Contractor or both do not sign all the Contract Documents, the Architect shall identify such unsigned Documents upon request.
- **1.2.2** Execution of the Contract by the Contractor is a representation that the Contractor has visited the site, become familiar with local conditions under which the Work is to be performed and correlated personal observations with requirements of the Contract Documents.
- 1.2.3 The intent of the Contract Documents is to include all items necessary for the proper execution and completion of the Work by the Contractor. The Contract Documents are complementary, and what is required by one shall be as binding as if required by all; performance by the Contractor shall be required only to the extent consistent with the Contract Documents and reasonably inferable from them as being necessary to produce the intended results.
- **1.2.4** Organization of the Specifications into divisions, sections and articles, and arrangement of Drawings shall not control the Contractor in dividing the Work among Subcontractors or in establishing the extent of Work to be performed by any trade.
- **1.2.5** Unless otherwise stated in the Contract Documents, words which have well-known technical or construction industry meanings are used in the Contract Documents in accordance with such recognized meanings.

#### 1.3 OWNERSHIP AND USE OF ARCHITECT'S DRAWINGS, SPECIFICATIONS AND OTHER DOCUMENTS

1.3.1 The Drawings, Specifications and other documents prepared by the Architect are instruments of the Architect's service through which the Work to be executed by the Contractor is described. The Contractor may retain one contract record set. Neither the Contractor nor any Subcontractor, Subsubcontractor or material or equipment supplier shall own or claim a copyright in the Drawings, Specifications and other documents prepared by the Architect, and unless otherwise indicated the Architect shall be deemed the author of them and will retain all common law, statutory and other reserved rights, in addition to the copyright. All copies of them, except the Contractor's record set, shall be returned or suitably accounted for to the Architect, on request, upon completion of the Work. The Drawings, Specifications and other documents prepared by the Architect, and copies thereof furnished to the Contractor, are for use solely with respect to this Project. They are not to be used by the Contractor or any Subcontractor, Subsubcontractor or material or equipment supplier on other projects or for additions to this Project outside the scope of the

Work without the specific written consent of the Owner and Architect. The Contractor, Subcontractors, Sub-subcontractors and material or equipment suppliers are granted a limited license to use and reproduce applicable portions of the Drawings, Specifications and other documents prepared by the Architect appropriate to and for use in the execution of their Work under the Contract Documents. All copies made under this license shall bear the statutory copyright notice, if any, shown on the Drawings, Specifications and other documents prepared by the Architect. Submittal or distribution to meet official regulatory requirements or for other purposes in connection with this Project is not to be construed as publication in derogation of the Architect's copyright or other reserved rights.

#### 1.4 CAPITALIZATION

1.4.1 Terms capitalized in these General Conditions include those which are (1) specifically defined, (2) the titles of numbered articles and identified references to Paragraphs, Subparagraphs and Clauses in the document or (3) the titles of other documents published by the American Institute of Architects.

#### 1.5 INTERPRETATION

1.5.1 In the interest of brevity the Contract Documents frequently omit modifying words such as "all" and "any" and articles such as "the" and "an," but the fact that a modifier or an article is absent from one statement and appears in another is not intended to affect the interpretation of either statement.

#### **ARTICLE 2**

### **OWNER**

#### 2.1 DEFINITION

- **2.1.1** The Owner is the person or entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. The term "Owner" means the Owner or the Owner's authorized representative.
- **2.1.2** The Owner upon reasonable written request shall furnish to the Contractor in writing information which is necessary and relevant for the Contractor to evaluate, give notice of or enforce mechanic's lien rights. Such information shall include a correct statement of the record legal title to the property on which the Project is located, usually referred to as the site, and the Owner's interest therein at the time of execution of the Agreement and, within five days after any change, information of such change in title, recorded or unrecorded.

# 2.2 INFORMATION AND SERVICES REQUIRED OF THE OWNER

- 2.2.1 The Owner shall, at the request of the Contractor, prior to execution of the Agreement and promptly from time to time thereafter, furnish to the Contractor reasonable evidence that financial arrangements have been made to fulfill the Owner's obligations under the Contract. [Note: Unless such reasonable evidence trere furnished on request prior to the execution of the Agreement, the prospective contractor would not be required to execute the Agreement or to commence the World.]
- 2.2.2 The Owner shall furnish surveys describing physical characteristics, legal limitations and utility locations for the site of the Project, and a legal description of the site.
- 2.2.3 Except for permits and fees which are the responsibility of the Contractor under the Contract Documents, the Owner shall secure and pay for necessary approvals, easements, assess-

ments and charges required for construction, use or occupancy of permanent structures or for permanent changes in existing facilities.

- **2.2.4** Information or services under the Owner's control shall be furnished by the Owner with reasonable promptness to avoid delay in orderly progress of the Work.
- 2.2.5 Unless otherwise provided in the Contract Documents, the Contractor will be furnished, free of charge, such copies of Drawings and Project Manuals as are reasonably necessary for execution of the Work.
- **2.2.6** The foregoing are in addition to other duties and responsibilities of the Owner enumerated herein and especially those in respect to Article 6 (Construction by Owner or by Separate Contractors), Article 9 (Payments and Completion) and Article 11 (Insurance and Bonds).

# 2.3 OWNER'S RIGHT TO STOP THE WORK

**2.3.1** If the Contractor fails to correct Work which is not in accordance with the requirements of the Contract Documents as required by Paragraph 12.2 or persistently fails to carry out Work in accordance with the Contract Documents, the Owner, by written order signed personally or by an agent specifically so empowered by the Owner in writing, may order the Contractor to stop the Work, or any portion thereof, until the cause for such order has been eliminated; however, the right of the Owner to stop the Work shall not give rise to a duty on the part of the Owner to exercise this right for the benefit of the Contractor or any other person or entity, except to the extent required by Subparagraph 6.1.3.

#### 2.4 OWNER'S RIGHT TO CARRY OUT THE WORK

2.4.1 If the Contractor defaults or neglects to carry out the Work in accordance with the Contract Documents and fails within a seven-day period after receipt of written notice from the Owner to commence and continue correction of such default or neglect with diligence and promptness, the Owner may after such seven-day period give the Contractor a second written notice to correct such deficiencies within a second seven-day period. If the Contractor within such second sevenday period after receipt of such second notice fails to commence and continue to correct any deficiencies, the Owner may, without prejudice to other remedies the Owner may have, correct such deficiencies. In such case an appropriate Change Order shall be issued deducting from payments then or thereafter due the Contractor the cost of correcting such deficiencies, including compensation for the Architect's additional services and expenses made necessary by such default, neglect or failure. Such action by the Owner and amounts charged to the Contractor are both subject to prior approval of the Architect. If payments then or thereafter due the Contractor are not sufficient to cover such amounts, the Contractor shall pay the difference to the Owner.

#### ARTICLE 3

# CONTRACTOR

#### 3.1 DEFINITION

**3.1.1** The Contractor is the person or entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. The term "Contractor" means the Contractor or the Contractor's authorized representative.

# 3.2 REVIEW OF CONTRACT DOCUMENTS AND FIELD CONDITIONS BY CONTRACTOR

- 3.2.1 The Contractor shall carefully study and compare the Contract Documents with each other and with information furnished by the Owner pursuant to Subparagraph 2.2.2 and shall at once report to the Architect errors, inconsistencies or omissions discovered. The Contractor shall not be liable to the Owner or Architect for damage resulting from errors, inconsistencies or omissions in the Contract Documents unless the Contractor recognized such error, inconsistency or omission and knowingly failed to report it to the Architect. If the Contractor performs any construction activity knowing it involves a recognized error, inconsistency or omission in the Contract Documents without such notice to the Architect, the Contractor shall assume appropriate responsibility for such performance and shall bear an appropriate amount of the attributable costs for correction.
- **3.2.2** The Contractor shall take field measurements and verify field conditions and shall carefully compare such field measurements and conditions and other information known to the Contractor with the Contract Documents before commencing activities. Errors, inconsistencies or omissions discovered shall be reported to the Architect at once.
- **3.2.3** The Contractor shall perform the Work in accordance with the Contract Documents and submittals approved pursuant to Paragraph 3.12.

#### 3.3 SUPERVISION AND CONSTRUCTION PROCEDURES

- **3.3.1** The Contractor shall supervise and direct the Work, using the Contractor's best skill and attention. The Contractor shall be solely responsible for and have control over construction means, methods, techniques, sequences and procedures and for coordinating all portions of the Work under the Contract, unless Contract Documents give other specific instructions concerning these matters.
- 3.3.2 The Contractor shall be responsible to the Owner for acts and omissions of the Contractor's employees, Subcontractors and their agents and employees, and other persons performing portions of the Work under a contract with the Contractor.
- **3.3.3** The Contractor shall not be relieved of obligations to perform the Work in accordance with the Contract Documents either by activities or duties of the Architect in the Architect's administration of the Contract, or by tests, inspections or approvals required or performed by persons other than the Contractor.
- **3.3.4** The Contractor shall be responsible for inspection of portions of Work already performed under this Contract to determine that such portions are in proper condition to receive subsequent Work.

#### 3.4 LABOR AND MATERIALS

- **3.4.1** Unless otherwise provided in the Contract Documents, the Contractor shall provide and pay for labor, materials, equipment, tools, construction equipment and machinery, water, heat, utilities, transportation, and other facilities and services necessary for proper execution and completion of the Work, whether temporary or permanent and whether or not incorporated or to be incorporated in the Work.
- **3.4.2** The Contractor shall enforce strict discipline and good order among the Contractor's employees and other persons carrying out the Contract. The Contractor shall not permit employment of unfit persons or persons not skilled in tasks assigned to them.

#### 3.5 WARRANTY

**3.5.1** The Contractor warrants to the Owner and Architect that materials and equipment furnished under the Contract will be of good quality and new unless otherwise required or permitted by the Contract Documents, that the Work will be free from defects not inherent in the quality required or permitted, and that the Work will conform with the requirements of the Contract Documents. Work not conforming to these requirements, including substitutions not properly approved and authorized, may be considered defective. The Contractor's warranty excludes remedy for damage or defect caused by abuse, modifications not executed by the Contractor, improper or insufficient maintenance, improper operation, or normal wear and tear under normal usage. If required by the Architect, the Contractor shall furnish satisfactory evidence as to the kind and quality of materials and equipment.

#### 3.6 TAXES

**3.6.1** The Contractor shall pay sales, consumer, use and similar taxes for the Work or portions thereof provided by the Contractor which are legally enacted when bids are received or negotiations concluded, whether or not yet effective or merely scheduled to go into effect.

#### 3.7 PERMITS, FEES AND NOTICES

- 3.7.1 Unless otherwise provided in the Contract Documents, the Contractor shall secure and pay for the building permit and other permits and governmental fees, licenses and inspections necessary for proper execution and completion of the Work which are customarily secured after execution of the Contract and which are legally required when bids are received or negotiations concluded.
- **3.7.2** The Contractor shall comply with and give notices required by laws, ordinances, rules, regulations and lawful orders of public authorities bearing on performance of the Work.
- **3.7.3** It is not the Contractor's responsibility to ascertain that the Contract Documents are in accordance with applicable laws, statutes, ordinances, building codes, and rules and regulations. However, if the Contractor observes that portions of the Contract Documents are at variance therewith, the Contractor shall promptly notify the Architect and Owner in writing, and necessary changes shall be accomplished by appropriate Modification.
- **3.7.4** If the Contractor performs Work knowing it to be contrary to laws, statutes, ordinances, building codes, and rules and regulations without such notice to the Architect and Owner, the Contractor shall assume full responsibility for such Work and shall bear the attributable costs.

#### 3.8 ALLOWANCES

- 3.8.1 The Contractor shall include in the Contract Sum all allowances stated in the Contract Documents. Items covered by allowances shall be supplied for such amounts and by such persons or entities as the Owner may direct, but the Contractor shall not be required to employ persons or entities against which the Contractor makes reasonable objection.
- 3.8.2 Unless otherwise provided in the Contract Documents:
  - .1 materials and equipment under an allowance shall beselected promptly by the Owner to avoid delay in the Work;
  - 2 allowances shall cover the cost to the Contractor of materials and equipment delivered at the site and all required taxes, less applicable trade discounts.

- .3 Contractor's costs for unloading and handling at the site, labor, installation costs, overhead, profit andother expenses contemplated for stated allowance amounts shall be included in the Contract Sum and not in the allowances;
- A whenever costs are more than or less than allowances, the Contract Sum shall be adjusted accordingly by Change Order. The amount of the Change Order chall reflect (1) the difference between actual costs and the allowances under Clause 3.8.2.2 and (2) changes in Contractor's costs under Clause 3.8.2.3.

#### 3.9 SUPERINTENDENT

**3.9.1** The Contractor shall employ a competent superintendent and necessary assistants who shall be in attendance at the Project site during performance of the Work. The superintendent shall represent the Contractor, and communications given to the superintendent shall be as binding as if given to the Contractor. Important communications shall be confirmed in writing. Other communications shall be similarly confirmed on written request in each case.

# 3.10 CONTRACTOR'S CONSTRUCTION SCHEDULES

- **3.10.1** The Contractor, promptly after being awarded the Contract, shall prepare and submit for the Owner's and Architect's information a Contractor's construction schedule for the Work. The schedule shall not exceed time limits current under the Contract Documents, shall be revised at appropriate intervals as required by the conditions of the Work and Project, shall be related to the entire Project to the extent required by the Contract Documents, and shall provide for expeditious and practicable execution of the Work.
- **3.10.2** The Contractor shall prepare and keep current, for the Architect's approval, a schedule of submittals which is coordinated with the Contractor's construction schedule and allows the Architect reasonable time to review submittals.
- **3.10.3** The Contractor shall conform to the most recent schedules.

# 3.11 DOCUMENTS AND SAMPLES AT THE SITE

**3.11.1** The Contractor shall maintain at the site for the Owner one record copy of the Drawings, Specifications, addenda, Change Orders and other Modifications, in good order and marked currently to record changes and selections made during construction, and in addition approved Shop Drawings, Product Data, Samples and similar required submittals. These shall be available to the Architect and shall be delivered to the Architect for submittal to the Owner upon completion of the Work.

# 3.12 SHOP DRAWINGS, PRODUCT DATA AND SAMPLES

- 3.12.1 Shop Drawings are drawings, diagrams, schedules and other data specially prepared for the Work by the Contractor or a Subcontractor, Sub-subcontractor, manufacturer, supplier or distributor to illustrate some portion of the Work.
- **3.12.2** Product Data are illustrations, standard schedules, performance charts, instructions, brochures, diagrams and other information furnished by the Contractor to illustrate materials or equipment for some portion of the Work.
- **3.12.3** Samples are physical examples which illustrate materials, equipment or workmanship and establish standards by which the Work will be judged.
- 3.12.4 Shop Drawings, Product Data, Samples and similar submittals are not Contract Documents. The purpose of their submittal is to demonstrate for those portions of the Work for

- which submittals are required the way the Contractor proposes to conform to the information given and the design concept expressed in the Contract Documents. Review by the Architect is subject to the limitations of Subparagraph 4.2.7.
- **3.12.5** The Contractor shall review, approve and submit to the Architect Shop Drawings, Product Data, Samples and similar submittals required by the Contract Documents with reasonable promptness and in such sequence as to cause no delay in the Work or in the activities of the Owner or of separate contractors. Submittals made by the Contractor which are not required by the Contract Documents may be returned without action.
- **3.12.6** The Contractor shall perform no portion of the Work requiring submittal and review of Shop Drawings, Product Data, Samples or similar submittals until the respective submittal has been approved by the Architect. Such Work shall be in accordance with approved submittals.
- **3.12.7** By approving and submitting Shop Drawings, Product Data, Samples and similar submittals, the Contractor represents that the Contractor has determined and verified materials, field measurements and field construction criteria related thereto, or will do so, and has checked and coordinated the information contained within such submittals with the requirements of the Work and of the Contract Documents.
- **3.12.8** The Contractor shall not be relieved of responsibility for deviations from requirements of the Contract Documents by the Architect's approval of Shop Drawings, Product Data, Samples or similar submittals unless the Contractor has specifically informed the Architect in writing of such deviation at the time of submittal and the Architect has given written approval to the specific deviation. The Contractor shall not be relieved of responsibility for errors or omissions in Shop Drawings, Product Data, Samples or similar submittals by the Architect's approval thereof.
- **3.12.9** The Contractor shall direct specific attention, in writing or on resubmitted Shop Drawings, Product Data, Samples or similar submittals, to revisions other than those requested by the Architect on previous submittals.
- **3.12.10** Informational submittals upon which the Architect is not expected to take responsive action may be so identified in the Contract Documents.
- **3.12.11** When professional certification of performance criteria of materials, systems or equipment is required by the Contract Documents, the Architect shall be entitled to rely upon the accuracy and completeness of such calculations and certifications.

#### 3.13 USE OF SITE

**3.13.1** The Contractor shall confine operations at the site to areas permitted by law, ordinances, permits and the Contract Documents and shall not unreasonably encumber the site with materials or equipment.

#### 3.14 CUTTING AND PATCHING

- **3.14.1** The Contractor shall be responsible for cutting, fitting or patching required to complete the Work or to make its parts fit together properly.
- **3.14.2** The Contractor shall not damage or endanger a portion of the Work or fully or partially completed construction of the Owner or separate contractors by cutting, patching or otherwise altering such construction, or by excavation. The Contractor shall not cut or otherwise alter such construction by the

Owner or a separate contractor except with written consent of the Owner and of such separate contractor; such consent shall not be unreasonably withheld. The Contractor shall not unreasonably withhold from the Owner or a separate contractor the Contractor's consent to cutting or otherwise altering the Work.

#### 3.15 CLEANING UP

- **3.15.1** The Contractor shall keep the premises and surrounding area free from accumulation of waste materials or rubbish caused by operations under the Contract. At completion of the Work the Contractor shall remove from and about the Project waste materials, rubbish, the Contractor's tools, construction equipment, machinery and surplus materials.
- **3.15.2** If the Contractor fails to clean up as provided in the Contract Documents, the Owner may do so and the cost thereof shall be charged to the Contractor.

#### 3.16 ACCESS TO WORK

**3.16.1** The Contractor shall provide the Owner and Architect access to the Work in preparation and progress wherever located.

#### 3.17 ROYALTIES AND PATENTS

**3.17.1** The Contractor shall pay all royalties and license fees. The Contractor shall defend suits or claims for infringement of patent rights and shall hold the Owner and Architect harmless from loss on account thereof, but shall not be responsible for such defense or loss when a particular design, process or product of a particular manufacturer or manufacturers is required by the Contract Documents. However, if the Contractor has reason to believe that the required design, process or product is an infringement of a patent, the Contractor shall be responsible for such loss unless such information is promptly furnished to the Architect.

#### 3.18 INDEMNIFICATION

- 3.18.1 To the fullest extent permitted by law, the Contractor shall indemnify and hold harmless the Owner, Architect, Architect's consultants, and agents and employees of any of them from and against claims, damages, losses and expenses, including but not limited to attorneys' fees, arising out of or resulting from performance of the Work, provided that such claim, damage, loss or expense is attributable to bodily injury, sickness, disease or death, or to injury to or destruction of tangible property (other than the Work itself) including loss of use resulting therefrom, but only to the extent caused in whole or in part by negligent acts or omissions of the Contractor, a Subcontractor, anyone directly or indirectly employed by them or anyone for whose acts they may be liable, regardless of whether or not such claim, damage, loss or expense is caused in part by a party indemnified hereunder. Such obligation shall not be construed to-negate, abridge, or reduce other rights or obligations of indemnity which would otherwise exist as to a party or persondescribed in this Paragraph 3.18.
- **3.18.2** In claims against any person or entity indemnified under this Paragraph 3.18 by an employee of the Contractor, a Subcontractor, anyone directly or indirectly employed by them or anyone for whose acts they may be liable, the indemnification obligation under this Paragraph 3.18 shall not be limited by a limitation on amount or type of damages, compensation or benefits payable by or for the Contractor or a Subcontractor under workers' or workmen's compensation acts, disability benefit acts or other employee benefit acts.
- **3.18.3** The obligations of the Contractor under this Paragraph 3.18 shall not extend to the liability of the Architect, the Archi

tect's consultants, and agents and employees of any of them arising out of (1) the preparation or approval of maps, drawings, opinions, reports, surveys, Change Orders, designs or specifications, or (2) the giving of or the failure to give directions or instructions by the Architect, the Architect's consultants, and agents and employees of any of them provided such giving or failure to give is the primary cause of the injury or damage.

### **ARTICLE 4**

# ADMINISTRATION OF THE CONTRACT

#### 4.1 ARCHITECT

- **4.1.1** The Architect is the person lawfully licensed to practice architecture or an entity lawfully practicing architecture identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. The term "Architect" means the Architect or the Architect's authorized representative.
- **4.1.2** Duties, responsibilities and limitations of authority of the Architect as set forth in the Contract Documents shall not be restricted, modified or extended without written consent of the Owner, Contractor and Architect. Consent shall not be unreasonably withheld.
- 4.1.3 In case of termination of employment of the Architect, the Owner shall appoint an architect against whom the Contractor makes no reasonable objection and whose status under the Contract Documents shall be that of the former architect.
- **4.1.4** Disputes arising under Subparagraphs 4.1.2 and 4.1.3 shall be subject to arbitration.

# 4.2 ARCHITECT'S ADMINISTRATION OF THE CONTRACT

- **4.2.1** The Architect will provide administration of the Contract as described in the Contract Documents, and will be the Owner's representative (1) during construction, (2) until final payment is due and (3) with the Owner's concurrence, from time to time during the correction period described in Paragraph 12.2. The Architect will advise and consult with the Owner. The Architect will have authority to act on behalf of the Owner only to the extent provided in the Contract Documents, unless otherwise modified by written instrument in accordance with other provisions of the Contract.
- **4.2.2** The Architect will visit the site at intervals appropriate to the stage of construction to become generally familiar with the progress and quality of the completed Work and to determine in general if the Work is being performed in a manner indicating that the Work, when completed, will be in accordance with the Contract Documents. However, the Architect will not be required to make exhaustive or continuous on-site inspections to check quality or quantity of the Work. On the basis of onsite observations as an architect, the Architect will keep the Owner informed of progress of the Work, and will endeavor to guard the Owner against defects and deficiencies in the Work.
- **4.2.3** The Architect will not have control over or charge of and will not be responsible for construction means, methods, techniques, sequences or procedures, or for safety precautions and programs in connection with the Work, since these are solely the Contractor's responsibility as provided in Paragraph 3.3. The Architect will not be responsible for the Contractor's failure to carry out the Work in accordance with the Contract Documents. The Architect will not have control over or charge of and will not be responsible for acts or omissions of the Con-

tractor, Subcontractors, or their agents or employees, or of any other persons performing portions of the Work.

- **4.2.4 Communications** Facilitating Contract Administration. Except as otherwise provided in the Contract Documents or when direct communications have been specially authorized, the Owner and Contractor shall endeavor to communicate through the Architect. Communications by and with the Architect's consultants shall be through the Architect. Communications by and with Subcontractors and material suppliers shall be through the Contractor. Communications by and with separate contractors shall be through the Owner.
- **4.2.5** Based on the Architect's observations and evaluations of the Contractor's Applications for Payment, the Architect will review and certify the amounts due the Contractor and will issue Certificates for Payment in such amounts.
- **4.2.6** The Architect will have authority to reject Work which does not conform to the Contract Documents. Whenever the Architect considers it necessary or advisable for implementation of the intent of the Contract Documents, the Architect will have authority to require additional inspection or testing of the Work in accordance with Subparagraphs 13.5.2 and 13.5.3, whether or not such Work is fabricated, installed or completed. However, neither this authority of the Architect nor a decision made in good faith either to exercise or not to exercise such authority shall give rise to a duty or responsibility of the Architect to the Contractor, Subcontractors, material and equipment suppliers, their agents or employees, or other persons performing portions of the Work.
- 4.2.7 The Architect will review and approve or take other appropriate action upon the Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Architect's action will be taken with such reasonable promptness as to cause no delay in the Work or in the activities of the Owner, Contractor or separate contractors, while allowing sufficient time in the Architect's professional judgment to permit adequate review. Review of such submittals is not conducted for the purpose of determining the accuracy and completeness of other details such as dimensions and quantities, or for substantiating instructions for installation or performance of equipment or systems, all of which remain the responsibility of the Contractor as required by the Contract Documents. The Architect's review of the Contractor's submittals shall not relieve the Contractor of the obligations under Paragraphs 3.3, 3.5 and 3.12. The Architect's review shall not constitute approval of safety precautions or, unless otherwise specifically stated by the Architect, of any construction means, methods, techniques, sequences or procedures. The Architect's approval of a specific-item shall not indicate approval of an assembly of which the item is a component-
- **4.2.8** The Architect will prepare Change Orders and Construction Change Directives, and may authorize minor changes in the Work as provided in Paragraph 7.4.
- **4.2.9** The Architect will conduct inspections to determine the date or dates of <del>Substantial</del> Completion and the date of final completion, will receive and forward to the Owner for the Owner's review and records written warranties and related documents required by the Contract and assembled by the Contractor, and will issue a final Certificate for Payment upon compliance with the requirements of the Contract Documents.
- **4.2.10** If the Owner and Architect agree, the Architect will provide one or more project representatives to assist in carrying

- out the Architect's responsibilities at the site. The duties, responsibilities and limitations of authority of such project representatives shall be as set forth in an exhibit to be incorporated in the Contract Documents.
- **4.2.11** The Architect will interpret and decide matters concerning performance under and requirements of the Contract Documents on written request of either the Owner or Contractor. The Architect's response to such requests will be made with reasonable promptness and within any time limits agreed upon. If no agreement is made concerning the time within which interpretations required of the Architect shall be furnished in compliance with this Paragraph 4.2, then delay shall not be recognized on account of failure by the Architect to furnish such interpretations until 15 days after written request is made for them.
- **4.2.12** Interpretations and decisions of the Architect will be consistent with the intent of and reasonably inferable from the Contract Documents and will be in writing or in the form of drawings. When making such interpretations and decisions, the Architect will endeavor to secure faithful performance by both Owner and Contractor, will not show partiality to either and will not be liable for results of interpretations or decisions so rendered in good faith.
- **4.2.13** The Architect's decisions on matters relating to aesthetic effect will be final if consistent with the intent expressed in the Contract Documents.

#### 4.3 CLAIMS AND DISPUTES

- **4.3.1 Definition.** A Claim is a demand or assertion by one of the parties seeking, as a matter of right, adjustment or interpretation of Contract terms, payment of money, extension of time or other relief with respect to the terms of the Contract. The term "Claim" also includes other disputes and matters in question between the Owner and Contractor arising out of or relating to the Contract. Claims must be made by written notice. The responsibility to substantiate Claims shall rest with the party making the Claim.
- 4.3.2 Decision of Architect. Claims, including those alleging an error or omission by the Architect, shall be referred initially to the Architect for action as provided in Paragraph 4.4. A decision by the Architect, as provided in Subparagraph 4.4.4, shall be required as a condition precedent to arbitration or litigation of a Claim between the Contractor and Owner as to all such matters arising prior to the date final payment is due, regardless of (1) whether such matters relate to execution and progress of the Work or (2) the extent to which the Work has been completed. The decision by the Architect in response to a Claim shall not be a condition precedent to arbitration or litigation in the event (1) the position of Architect is vacant, (2) the Architect has not received evidence or has failed to render a decision within agreed time limits, (3) the Architect has failed to take action required under Subparagraph 4.4.4 within 30 days after the Claim is made, (4) 45 days have passed after the Claim has been referred to the Architect or (5) the Claim relates to a mechanic's lien.
- 4.3.3 Time Limits on Claims. Claims by either party must be made within 21 days after occurrence of the event giving rise to such Claim or within 21 days after the claimant first recognizes the condition giving rise to the Claim, whichever is later. Claims must be made by written notice. An additional Claim made after the initial Claim has been implemented by Change Order will not be considered unless submitted in a timely manner.

- **4.3.4 Continuing Contract Performance.** Pending final resolution of a Claim including arbitration, unless otherwise agreed in writing the Contractor shall proceed diligently with performance of the Contract and the Owner shall continue to make payments in accordance with the Contract Documents,
- **4.3.5 Waiver of Claims: Final Payment.** The making of final payment shall constitute a waiver of Claims by the Owner except those arising from:
  - .1 liens, Claims, security interests or encumbrances arising out of the Contract and unsettled;
  - .2 failure of the Work to comply with the requirements of the Contract Documents; or
  - .3 terms of special warranties required by the Contract Documents.
- 4.3.6 Claims for Concealed or Unknown Conditions. If conditions are encountered at the site which are (1) subsurface or otherwise concealed physical conditions which differ materially from those indicated in the Contract Documents or (2) unknown physical conditions of an unusual nature, which differ materially from those ordinarily found to exist and generally recognized as inherent in construction activities of the character provided for in the Contract Documents, then notice by the observing party shall be given to the other party promptly before conditions are disturbed and in no event later than 21 days after first observance of the conditions. The Architect will promptly investigate such conditions and, if they differ materially and cause an increase or decrease in the Contractor's cost of, or time required for, performance of any part of the Work, will recommend an equitable adjustment in the Contract Sum or Contract Time, or both. If the Architect determines that the conditions at the site are not materially different from those indicated in the Contract Documents and that no change in the terms of the Contract is justified, the Architect shall so notify the Owner and Contractor in writing, stating the reasons. Claims by either party in opposition to such determination must be made within 21 days after the Architect has given notice of the decision. If the Owner and Contractor cannot agree on an adjustment in the Contract Sum or Contract Time, the adjustment shall be referred to the Architect for initial determination, subject to further proceedings pursuant to Paragraph
- 4.3.7 Claims for Additional Cost. If the Contractor wishes to make Claim for an increase in the Contract Sum, written notice as provided herein shall be given before proceeding to execute the Work. Prior notice is not required for Claims relating to an emergency endangering life or property arising under Paragraph 10.3. If the Contractor believes additional cost is involved for reasons including but not limited to (1) a written interpretation from the Architect, (2) an order by the Owner to stop the Work where the Contractor was not at fault, (3) a written order for a minor change in the Work issued by the Architect, (4) failure of payment by the Owner, (5) termination of the Contract by the Owner, (6) Owner's suspension or (7) other reasonable grounds, Claim shall be filed in accordance with the procedure established herein.

#### 4.3.8 Claims for Additional Time

- 4.3.8.1 If the Contractor wishes to make Claim for an increase in the Contract Time, written notice as provided herein shall be given. The Contractor's Claim shall include an estimate of cost and of probable effect of delay on progress of the Work. In the case of a continuing delay only one Claim is necessary.
- **4.3.8.2** If adverse weather conditions are the basis for a Claim for additional time, such Claim shall be documented by data

- substantiating that weather conditions were abnormal for the period of time and could not have been reasonably anticipated, and that weather conditions had an adverse effect on the scheduled construction.
- **4.3.9** Injury or Damage to Person or Property. If either party to the Contract suffers injury or damage to person or property because of an act or omission of the other party, of any of the other party's employees or agents, or of others for whose acts such party is legally liable, written notice of such injury or damage, whether or not insured, shall be given to the other party within a reasonable time not exceeding 21 days after first observance. The notice shall provide sufficient detail to enable the other party to investigate the matter. If a Claim for additional cost or time related to this Claim is to be asserted, it shall be filed as provided in Subparagraphs 4.3.7 or 4.3.8.

#### 4.4 RESOLUTION OF CLAIMS AND DISPUTES

- 4.4.1 The Architect will review Claims and take one or more of the following preliminary actions within ten days of receipt of a Claim: (1) request additional supporting data from the claimant, (2) submit a schedule to the parties indicating when the Architect expects to take action, (3) reject the Claim in whole or in part, stating reasons for rejection, (4) recommend approval of the Claim by the other party or (5) suggest a compromise. The Architect may also, but is not obligated to, notify the surety, if any, of the nature and amount of the Claim.
- 4.4.2 If a Claim has been resolved, the Architect will prepare or obtain appropriate documentation.
- **4.4.3** If a Claim has not been resolved, the party making the Claim shall, within ten days after the Architect's preliminary response, take one or more of the following actions: (1) submit additional supporting data requested by the Architect, (2) modify the initial Claim or (3) notify the Architect that the initial Claim stands.
- **4.4.4** If a Claim has not been resolved after consideration of the foregoing and of further evidence presented by the parties or requested by the Architect, the Architect will notify the parties in writing that the Architect's decision will be made within seven days, which decision shall be final and binding on the parties but subject to arbitration. Upon expiration of such time period, the Architect will render to the parties the Architect's written decision relative to the Claim, including any change in the Contract Sum or Contract Time or both. If there is a surety and there appears to be a possibility of a Contractor's default, the Architect may, but is not obligated to, notify the surety and request the surety's assistance in resolving the controversy.

#### 4.5 ARBITRATION

**4.5.1 Controversies and Claims Subject to Arbitration.** Any controversy or Claim arising out of or related to the Contract, or the breach thereof, shall be settled by arbitration in accordance with the Construction Industry Arbitration Rules of the American Arbitration Association, and judgment upon the award rendered by the arbitrator or arbitrators may be entered in any court having jurisdiction thereof, except controversies or Claims relating to aesthetic effect and except those waived as provided for in Subparagraph 4.3.5. Such controversies or Claims upon which the Architect has given notice and rendered a decision as provided in Subparagraph 4.4.4 shall be subject to arbitration upon written demand of either party. Arbitration may be commenced when 45 days have passed after a Claim has been referred to the Architect as provided in Paragraph 4.3 and no decision has been rendered.

- **4.5.2 Flules and Notices for Arbitration.** Claims between the Owner and Contractor not resolved under Paragraph 4.4 shall, if subject to arbitration under Subparagraph 4.5.1, be decided by arbitration in accordance with the Construction Industry Arbitration Rules of the American Arbitration Association currently in effect, unless the parties mutually agree otherwise. Notice of demand for arbitration shall be filed in writing with the other party to the Agreement between the Owner and Contractor and with the American Arbitration Association, and a copy shall be filed with the Architect.
- **4.5.3 Contract Performance During Arbitration.** During arbitration proceedings, the Owner and Contractor shall comply with Subparagraph 4.3.4.
- **4.5.4** When Arbitration May Be Demanded. Demand for arbitration of any Claim may not be made until the earlier of (1) the date on which the Architect has rendered a final written decision on the Claim, (2) the tenth day after the parties have presented evidence to the Architect or have been given reasonable opportunity to do so, if the Architect has not rendered a final written decision by that date, or (3) any of the five events described in Subparagraph 4.3.2.
- **4.5.4.1** When a written decision of the Architect states that (1) the decision is final but subject to arbitration and (2) a demand for arbitration of a Claim covered by such decision must be made within 30 days after the date on which the party making the demand receives the final written decision, then failure to demand arbitration within said 30 days' period shall result in the Architect's decision becoming final and binding upon the Owner and Contractor. If the Architect renders a decision after arbitration proceedings have been initiated, such decision may be entered as evidence, but shall not supersede arbitration proceedings unless the decision is acceptable to all parties concerned.
- **4.5.4.2** A demand for arbitration shall be made within the time limits specified in Subparagraphs 4.5.1 and 4.5.4 and Clause 4.5.4.1 as applicable, and in other cases within a reasonable time after the Claim has arisen, and in no event shall it be made after the date when institution of legal or equitable proceedings based on such Claim would be barred by the applicable statute of limitations as determined pursuant to Paragraph 13.7.
- 4.5.5 Limitation on Consolidation or Joincer. No arbitration arising out of or relating to the Contract Documents shall include, by consolidation or joinder or in any other manner, the Architect, the Architect's employees or consultants, except by written consent containing specific reference to the Agreement and signed by the Architect, Owner, Contractor and any other person or entity sought to be joined. No arbitration shall include, by consolidation or joinder or in any other manner, parties other than the Owner, Contractor, a separate contractor as described in Article 6 and other persons substantially involved in a common question of fact or law whose presence is required if complete relief is to be accorded in arbitration. No person or entity other than the Owner, Contractor or a separate contractor as described in Article 6 shall be included as an original third party or additional third party to an arbitration whose interest or responsibility is insubstantial. Consent to arbitration involving an additional person or entity shall not constitute consent to arbitration of a dispute not described therein or with a person or entity not named or described therein. The foregoing agreement to arbitrate and other agreements to arbitrate with an additional person or entity duly consented to by parties to the Agreement shall be specifically enforceable under applicable law in any court having jurisdiction thereof.

- **4.5.6 Claims and Timely Assertion of Claims.** A party who files a notice of demand for arbitration must assert in the demand all Claims then known to that party on which arbitration is permitted to be demanded. When a party fails to include a Claim through oversight, inadvertence or excusable neglect, or when a Claim has matured or been acquired subsequently, the arbitrator or arbitrators may permit amendment.
- **4.5.7 Judgment on Final Award.** The award rendered by the arbitrator or arbitrators shall be final, and judgment may be entered upon it in accordance with applicable law in any court having jurisdiction thereof.

# **ARTICLE 5**

#### **SUBCONTRACTORS**

#### 5.1 DEFINITIONS

- **5.1.1** A Subcontractor is a person or entity who has a direct contract with the Contractor to perform a portion of the Work at the site. The term "Subcontractor" is referred to throughout the Contract Documents as if singular in number and means a Subcontractor or an authorized representative of the Subcontractor. The term "Subcontractor" does not include a separate contractor or subcontractors of a separate contractor.
- **5.1.2** A Sub-subcontractor is a person or entity who has a direct or indirect contract with a Subcontractor to perform a portion of the Work at the site. The term "Sub-subcontractor" is referred to throughout the Contract Documents as if singular in number and means a Sub-subcontractor or an authorized representative of the Sub-subcontractor.

# 5.2 AWARD OF SUBCONTRACTS AND OTHER CONTRACTS FOR PORTIONS OF THE WORK

- **6.2.1** Unless otherwise stated in the Contract Documents or the bidding requirements, the Contractor, as soon as practicable after award of the Contract, shall furnish in writing to the Owner through the Architect the names of persons or entities (including those who are to furnish materials or equipment fabricated to a special design) proposed for each principal portion of the Work. The Architect will promptly reply to the Contractor in writing stating whether or not the Owner or the Architect, after due investigation, has reasonable objection to any such proposed person or entity. Failure of the Owner or Architect to reply promptly shall constitute notice of no reasonable objection.
- **6.2.2** The Contractor shall not contract with a proposed-person or entity to whom the Owner or Architect has made reasonable and timely objection. The Contractor shall not be required to contract with anyone to whom the Contractor has made reasonable objection.
- 5.2.3 If the Owner or Architect has reasonable objection to a person or entity proposed by the Contractor, the Contractor shall propose another to whom the Owner or Architect has no reasonable objection. The Contract Sum shall be increased or decreased by the difference in cost occasioned by such change and an appropriate Change Order shall be issued. However, no increase in the Contract Sum shall be allowed for such change unless the Contractor has acted promptly and responsively insubmitting names as required.
- 5.2.4 The Contractor shall not change a Subcontractor, person or entity previously selected if the Owner or Architect makes reasonable objection to such change.

#### 5.3 SUBCONTRACTUAL RELATIONS

5.3.1 By appropriate agreement, written where legally required for validity, the Contractor shall require each Subcontractor, tothe extent of the Work to be performed by the Subcontractor, to be bound to the Contractor by terms of the Contract Documents, and to assume toward the Contractor all the obligations and responsibilities which the Contractor, by these Documents, assumes toward the Owner and Architect. Each subcontract agreement shall preserve and protect the rights of the Owner and Architect under the Contract Documents with respect to the Work to be performed by the Subcontractor so that-subcontracting thereof will-not-prejudice such rights, and shall allow to the Subcontractor, unless specifically provided otherwise in the subcontract agreement, the benefit of allrights, remedies and redress against the Contractor that the Contractor, by the Contract Documents, has against the Owner. Where appropriate, the Contractor shall require each Subcontractor to enter into similar agreements with Sub-subcontractors. The Contractor shall make available to each proposed-Subcontractor, prior to the execution of the subcontract agreement, copies of the Contract Documents to which the Subcontractor will be bound, and, upon-written request of the Subcontractor, identify to the Subcontractor terms and conditions of the proposed subcontract agreement which may be at variance with the Contract Documents. Subcontractors shall similarly make copies of applicable portions of such documents available to their respective proposed Sub-subcontractors.

# 5.4 CONTINGENT ASSIGNMENT OF SUBCONTRACTS

- **5.4.1** Each subcontract agreement for a portion of the Work is assigned by the Contractor to the Owner provided that.
  - 1 assignment is effective only after termination of the Contract by the Owner for cause pursuant to Paragraph 14.2 and only for those subcontract agreements which the Owner accepts by notifying the Subcontractor in writing; and
  - 2 assignment is subject to the prior rights of the surery, if any, obligated under bond relating to the Contract.
- 6.4.2 If the Work has been suspended for more than 30 days, the Subcontractor's compensation shall be equitably adjusted

#### **ARTICLE 6**

# CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS

#### 6.1 OWNER'S RIGHT TO PERFORM CONSTRUCTION AND TO AWARD SEPARATE CONTRACTS

- **6.1.1** The Owner reserves the right to perform construction or operations related to the Project with the Owner's own forces, and to award separate contracts in connection with other portions of the Project or other construction or operations on the site under Conditions of the Contract identical or substantially similar to these including those portions related to insurance and waiver of subrogation. If the Contractor claims that delay or additional cost is involved because of such action by the Owner, the Contractor shall make such Claim as provided elsewhere in the Contract Documents.
- **6.1.2** When separate contracts are awarded for different portions of the Project or other construction or operations on the site, the term "Contractor" in the Contract Documents in each case shall mean the Contractor who executes each separate Owner-Contractor Agreement.

- **6.1.3** The Owner shall provide for coordination of the activities of the Owner's own forces and of each separate contractor with the Work of the Contractor, who shall cooperate with them. The Contractor shall participate with other separate contractors and the Owner in reviewing their construction schedules when directed to do so. The Contractor shall make any revisions to the construction schedule and Contract Sum deemed necessary after a joint review and mutual agreement. The construction schedules shall then constitute the schedules to be used by the Contractor, separate contractors and the Owner until subsequently revised.
- **6.1.4** Unless otherwise provided in the Contract Documents, when the Owner performs construction or operations related to the Project with the Owner's own forces, the Owner shall be deemed to be subject to the same obligations and to have the same rights which apply to the Contractor under the Conditions of the Contract, including, without excluding others, those stated in Article 3, this Article 6 and Articles 10, 11 and 12.

#### 6.2 MUTUAL RESPONSIBILITY

- **6.2.1** The Contractor shall afford the Owner and separate contractors reasonable opportunity for introduction and storage of their materials and equipment and performance of their activities and shall connect and coordinate the Contractor's construction and operations with theirs as required by the Contract Documents.
- **6.2.2** If part of the Contractor's Work depends for proper execution or results upon construction or operations by the Owner or a separate contractor, the Contractor shall, prior to proceeding with that portion of the Work, promptly report to the Architect apparent discrepancies or defects in such other construction that would render it unsuitable for such proper execution and results. Failure of the Contractor so to report shall constitute an acknowledgment that the Owner's or separate contractors' completed or partially completed construction is fit and proper to receive the Contractor's Work, except as to defects not then reasonably discoverable.
- **6.2.3** Costs caused by delays or by improperly timed activities or defective construction shall be borne by the party responsible therefor.
- **6.2.4** The Contractor shall promptly remedy damage wrongfully caused by the Contractor to completed or partially completed construction or to property of the Owner or separate contractors as provided in Subparagraph 10.2.5.
- 6.2.5 Claims and other disputes and matters in questionbetween the Contractor and a separate contractor shall be subject to the provisions of Paragraph 4.3 provided the separate contractor has reciprocal obligations.
- **6.2.6** The Owner and each separate contractor shall have the same responsibilities for cutting and patching as are described for the Contractor in Paragraph 3.14.

#### 6.3 OWNER'S RIGHT TO CLEAN UP

**6.3.1** If a dispute arises among the Contractor, separate contractors and the Owner as to the responsibility under their respective contracts for maintaining the premises and surrounding area free from waste materials and rubbish as described in Paragraph 3.15, the Owner may clean up and allocate the cost among those responsible as the Architect determines to be just.

# **ARTICLE 7**

#### CHANGES IN THE WORK

#### 7.1 CHANGES

- **7.1.1** Changes in the Work may be accomplished after execution of the Contract, and without invalidating the Contract, by Change Order, Construction Change Directive or order for a minor change in the Work, subject to the limitations stated in this Article 7 and elsewhere in the Contract Documents.
- **7.1.2** A Change Order shall be based upon agreement among the Owner, Contractor and Architect; a Construction Change Directive requires agreement by the Owner and Architect and may or may not be agreed to by the Contractor; an order for a minor change in the Work may be issued by the Architect alone.
- 7.1.3 Changes in the Work shall be performed under applicable provisions of the Contract Documents, and the Contractor shall proceed promptly, unless otherwise-provided in the Change Order, Construction Change Directive or order for a minor change in the Work.
- 7.1.4 If unit prices are stated in the Contract Documents or subsequently agreed upon, and if quantities originally contemplated are so changed in a proposed Change Order or Construction Change Directive that application of such unit prices to quantities of Work proposed will cause substantial inequity to the Owner or Contractor, the applicable unit prices shall be equitably adjusted.

#### 7.2 CHANGE ORDERS

- 7.2.1 A Change Order is a written instrument prepared by the Architect and signed by the Owner, Contractor and Architect, stating their agreement upon all of the following:
  - -1 a change in the Work;
  - -2- the amount of the adjustment in the Contract Sum, if -any; and-
  - 3 the extent of the adjustment in the Contract Time, if any.
- **7.2.2** Methods used in determining adjustments to the Contract Sum may include those listed in Subparagraph 7.3.3.

#### 7.3 CONSTRUCTION CHANGE DIRECTIVES

- 7.3.1 A Construction Change Directive is a written order prepared by the Architect and signed by the Owner and Architect, directing a change in the Work and stating a proposed basis for adjustment, if any, in the Contract Sum or Contract Time, or both. The Owner may by Construction Change Directive, without invalidating the Contract, order changes in the Work within the general scope of the Contract consisting of additions, deletions or other revisions, the Contract Sum and Contract Time being adjusted accordingly.
- 7.3.2 A Construction Change Directive shall be used in the absence of total agreement on the terms of a Change Orden
- 7.3.3 If the Construction Change Directive provides for an adjustment to the Contract Sum, the adjustment shall be based on one of the following methods:
  - I mutual acceptance of a lump sum-properly itemizedand supported by sufficient substantiating data to permit evaluation;
  - 2-unit prices stated in the Contract Documents or subsequently agreed upon;

- r3 cost to be determined in a manner agreed upon by the parties and a mutually acceptable fixed or percentage fee; or
- 4 as provided in Subparagraph 7.3.6.
- 7.3.4 Upon-receipt of a Construction Change Directive, the Contractor shall promptly proceed with the change in the Work involved and advise the Architect of the Contractor's agreement or disagreement with the method, if any, provided in the Construction Change Directive for determining the proposed adjustment in the Contract Sum or Contract Time.
- 7.3.5 A Construction Change Directive signed by the Contractor indicates the agreement of the Contractor therewith, including adjustment in Contract Sum and Contract Time or the method for determining them. Such agreement shall be effective immediately and shall be recorded as a Change Order.
- 7.3.6 If the Contractor does not respond promptly or disagrees with the method for adjustment in the Contract Sum, the method and the adjustment shall be determined by the Architect on the basis of reasonable expenditures and savings of those performing the Work attributable to the change, including, in case of an increase in the Contract Sum, a reasonable allowance for overhead and profit. In such case, and also under Clause 7.3.3.3, the Contractor shall keep and present, in such form as the Architect may prescribe, an itemized accounting together with appropriate supporting data. Unless otherwise provided in the Contract Documents, costs for the purposes of this Subparagraph 7.3.6 shall be limited to the following.
  - -1 costs of labor, including social security, old age and unemployment insurance, fringe benefits required by agreement or custom, and workers' or workmen's compensation-insurance;
  - 2 costs of materials, supplies and equipment, including cost of transportation, whether incorporated or consumed;
  - 3 rental costs of machinery and equipment, exclusive of hand-tools, whether-rented from the Contractor-orothers;
  - -4 costs of premiums for all bonds and insurance, permit fees, and sales, use or similar taxes related to the Work; and
  - 45 additional costs of supervision and field office personnel directly attributable to the change.
- 7.3.7 Pending final determination of cost to the Owner, amounts not in dispute may be included in Applications for Payment. The amount of credit to be allowed by the Contractor to the Owner for a deletion or change which results in a net decrease in the Contract Sum shall be actual net cost as confirmed by the Architect. When both additions and credits covering related Work or substitutions are involved in a change, the allowance for overhead and profit shall be figured on the basis of net increase, if any, with respect to that change.
- 7.3.8 If the Owner and Contractor do not agree with the adjustment in Contract Time or the method for determining it, the adjustment or the method shall be referred to the Architect for determination.
- 7.3.9 When the Owner and Contractor agree with the determination made by the Architect concerning the adjustments in the Contract Sum and Contract Time, or otherwise reach agreement upon the adjustments, such agreement shall be effective immediately and shall be recorded by preparation and execution of an appropriate Change Order.

#### 7.4 MINOR CHANGES IN THE WORK

**7.4.1** The Architect will have authority to order minor changes in the Work not involving adjustment in the Contract Sum or extension of the Contract Time and not inconsistent with the intent of the Contract Documents. Such changes shall be effected by written order and shall be binding on the Owner and Contractor. The Contractor shall carry out such written orders promptly.

#### **ARTICLE 8**

#### TIME

#### 8.1 DEFINITIONS

- 8.1.1 Unless otherwise provided, Contract Time is the period of time, including authorized adjustments, allotted in the Contract Documents for Substantial Completion of the Work:
- **8.1.2** The date of commencement of the Work is the date established in the Agreement. The date shall not be postponed by the failure to act of the Contractor or of persons or entities for whom the Contractor is responsible.
- **8.1.3** The date of Substantial Completion is the date certified by the Architect in accordance with Paragraph 9.8.
- **8.1.4** The term "day" as used in the Contract Documents shall mean calendar day unless otherwise specifically defined.

#### **B.2** PROGRESS AND COMPLETION

- **8.2.1** Time limits stated in the Contract Documents are of the essence of the Contract. By executing the Agreement the Contractor confirms that the Contract Time is a reasonable period for performing the Work.
- **8.2.2** The Contractor shall not knowingly, except by agreement or instruction of the Owner in writing, prematurely commence operations on the site or elsewhere prior to the effective date of insurance required by Article 11 to be furnished by the Contractor. The date of commencement of the Work shall not be changed by the effective date of such insurance. Unless the date of commencement is established by a notice to proceed given by the Owner, the Contractor shall notify the Owner in writing not less than five days or other agreed period before commencing the Work to permit the timely filing of mortgages, mechanic's liens and other security interests,
- 8.2.3 The Contractor shall proceed expeditiously with adequate forces and shall achieve Substantial Completion within the Contract Time.

#### 8.3 DELAYS AND EXTENSIONS OF TIME

- 8.3.1 If the Contractor is delayed at any time in progress of the Work by an act or neglect of the Owner or Architect, or of an employee of either, or of a separate contractor employed by the Owner, or by changes ordered in the Work, or by labor disputes, fire, unusual delay in deliveries, unavoidable easualties or other causes beyond the Contractor's control, or by delay authorized by the Owner pending arbitration, or by other causes which the Architect determines may justify delay, then the Contract Time shall be extended by Change Order for such reasonable time as the Architect may determine.
- 8.3.2 Claims relating to time shall be made in accordance with applicable provisions of Paragraph 4.3.
- 8.3.3 This Paragraph 8.3 closs not preclude recovery of damages for delay by either party under other provisions of the Contract Documents.

### **ARTICLE 9**

# PAYMENTS AND COMPLETION

#### 9.1 CONTRACT SUM

**9.1.1** The Contract Sum is stated in the Agreement and, including authorized adjustments, is the total amount payable by the Owner to the Contractor for performance of the Work under the Contract Documents.

# 9.2 SCHEDULE OF VALUES

**9.2.1** Before the first Application for Payment, the Contractor, shall submit to the Architect a schedule of values allocated to various portions of the Work, prepared in such form and supported by such data to substantiate its accuracy as the Architect may require. This schedule, unless objected to by the Architect, shall be used as a basis for reviewing the Contractor's Applications for Payment.

#### 9.3 APPLICATIONS FOR PAYMENT

- 9.3.1 At least ten days before the date established for each progress payment, the Contractor shall submit to the Architect an Itemized Application for Payment for operations completed in accordance with the schedule of values. Such application shall be notarized, if required, and supported by such data substantiating the Contractor's right to payment as the Owner or Architect may require, such as copies of requisitions from Subcontractors and material suppliers, and reflecting retainage if provided for elsewhere in the Contract Documents.
- 9.3.1.1 Such applications may include requests for payment onaccount of changes in the Work which have been properly authorized by Construction Change Directives but not yet included in Change Orders.
- **9.3.1.2** Such applications may not include requests for payment of amounts the Contractor does not intend to pay to a Subcontractor or material supplier because of a dispute or other reason.
- 9.3.2 Unless otherwise provided in the Contract Documents, payments shall be made on account of materials and equipment delivered and suitably stored at the site for subsequent incorporation in the Work. If approved in advance by the Owner, payment may similarly be made for materials and equipment suitably stored off the site at a location agreed upon in writing. Payment for materials and equipment stored on or off the site shall be conditioned upon compliance by the Contractor with procedures satisfactory to the Owner to establish the Owner's title to such materials and equipment or otherwise protect the Owner's interest, and shall include applicable insurance, storage and transportation to the site for such materials and equipment stored off the site.
- 9.3.3 The Contractor warrants that title to all Work covered by an Application for Payment will pass to the Owner no later than the time of payment. The Contractor further warrants that upon submittal of an Application for Payment all Work for which Certificates for Payment have been previously issued and payments received from the Owner shall, to the best of the Contractor's knowledge, information and belief, be free and clear of liens, claims, security interests or encumbrances in favor of the Contractor, Subcontractors, material suppliers, or other persons or entities making a claim by reason of having provided labor, materials and equipment relating to the Work.

#### 9.4 CERTIFICATES FOR PAYMENT

9.4.1 The Architect will, within seven-days after receipt of the Contractor's Application for Payment, either issue to the

Owner a Certificate for Payment, with a copy to the Contractor, for such amount as the Architect determines is properly due, or notify the Contractor and Owner in writing of the Architect's reasons for withholding certification in whole or in part as provided in Subparagraph 9.5.1.

9.4.2 The issuance of a Certificate for Payment will constitute a representation by the Architect to the Owner, based on the Architect's observations at the site and the data comprising the Application for Payment, that the Work has progressed to the point indicated and that, to the best of the Architect's knowledge, information and belief, quality of the Work is in accordance with the Contract Documents. The foregoing representations are subject to an evaluation of the Work for conformance with the Contract Documents upon Substantial Completion, to results of subsequent tests and inspections, to minor deviations from the Contract Documents correctable prior to completion and to specific qualifications expressed by the Architect. The issuance of a Certificate for Payment will further constitute a representation that the Contractor is entitled to payment in the amount certified. However, the issuance of a Certificate for Pavment will not be a representation that the Architect has (1) made exhaustive or continuous on-site inspections to check the quality or quantity of the Work, (2) reviewed construction means, methods, techniques, sequences or procedures, (3) reviewed copies of requisitions received from Subcontractors and material suppliers and other data requested by the Owner to substantiate the Contractor's right to payment or (4) made examination to ascertain how or for what purpose the Contractor has used money previously paid on account of the Contract Sum.

#### 9.5 DECISIONS TO WITHHOLD CERTIFICATION

- 9.5.1 The Architect may decide not to certify payment and may withhold a Certificate for Payment in whole or in part, to the extent reasonably necessary to protect the Owner, if in the Architect's opinion the representations to the Owner required by Subparagraph 9.4.2 cannot be made. If the Architect is unable to certify payment in the amount of the Application, the Architect will notify the Contractor and Owner as provided in Subparagraph 9.4.1. If the Contractor and Architect cannot agree on a revised amount, the Architect will promptly issue a Certificate for Payment for the amount for which the Architect is able to make such representations to the Owner. The Architect may also decide not to certify payment or, because of subsequently discovered evidence or subsequent observations. may nullify the whole or a part of a Certificate for Payment previously issued, to such extent as may be necessary in the Architect's opinion to protect the Owner from loss because of:
  - .1 defective Work not remedied;
  - .2 third party claims filed or reasonable evidence indicating probable filing of such claims;
  - .3 failure of the Contractor to make payments properly to Subcontractors or for labor, materials or equipment;
  - .4 reasonable evidence that the Work cannot be completed for the unpaid balance of the Contract Sum;
  - .5 damage to the Owner or another contractor;
  - .6 reasonable evidence that the Work will not be completed within the Contract Time, and that the unpaid balance would not be adequate to cover actual or liquidated damages for the anticipated delay; or
  - .7 persistent failure to carry out the Work in accordance with the Contract Documents.

**9.5.2** When the above reasons for withholding certification are removed, certification will be made for amounts previously withheld.

#### 9.6 PROGRESS PAYMENTS

- 9.6.1 After the Architect has issued a Certificate for Payment, the Owner shall make payment in the manner and within the time provided in the Contract Documents, and shall so notify the Architect.
- **9.6.2** The Contractor shall promptly pay each Subcontractor, upon receipt of payment from the Owner, out of the amount paid to the Contractor on account of such Subcontractor's portion of the Work, the amount to which said Subcontractor is entitled, reflecting percentages actually retained from payments to the Contractor on account of such Subcontractor's portion of the Work. The Contractor shall, by appropriate agreement with each Subcontractor, require each Subcontractor to make payments to Sub-subcontractors in similar manner.
- **9.6.3** The Architect will, on request, furnish to a Subcontractor, if practicable, information regarding percentages of completion or amounts applied for by the Contractor and action taken thereon by the Architect and Owner on account of portions of the Work done by such Subcontractor.
- **9.6.4** Neither the Owner nor Architect shall have an obligation to pay or to see to the payment of money to a Subcontractor except as may otherwise be required by law.
- **9.6.5** Payment to material suppliers shall be treated in a manner similar to that provided in Subparagraphs 9.6.2, 9.6.3 and 9.6.4.
- **9.6.6** A Certificate for Payment, a progress payment, or partial or entire use or occupancy of the Project by the Owner shall not constitute acceptance of Work not in accordance with the Contract Documents.

#### 9.7 FAILURE OF PAYMENT

9.7.1 If the Architect does not issue a Certificate for Payment, through no fault of the Contractor, within seven days after receipt of the Contractor's Application for Payment, or if the Owner does not pay the Contractor within seven days after the date established in the Contract Documents the amount certified by the Architect or awarded by arbitration, then the Contractor may, upon seven additional days' written notice to the Owner and Architect, stop the Work until payment of the amount owing has been received. The Contract Time shall be extended appropriately and the Contract Sum shall be increased by the amount of the Contractor's reasonable costs of shut down, delay and start-up, which shall be accomplished as provided in Article 7.

#### 9.8 SUBSTANTIAL COMPLETION

- 9.8.1 Substantial Completion is the stage in the progress of the Work when the Work or designated portion thereof is sufficiently complete in accordance with the Contract Documents the Owner can occupy or utilize the Work for its intended
- 9.8.2 When the Contractor considers that the Work, or a portion thereof which the Owner agrees to accept separately, is substantially complete, the Contractor shall prepare and submit to the Architect a comprehensive list of items to be completed or corrected. The Contractor shall proceed promptly to complete and correct items on the list. Failure to include an item on such list does not after the responsibility of the Contractor to complete all. Work in accordance with the Contract Documents. Upon receipt of the Contractor's list, the Architect will make an inspection to determine whether the Work or design

nated portion thereof is substantially complete. If the Architect's inspection discloses any item, whether or not included on the Contractor's list, which is not in accordance with the requirements of the Contract Documents, the Contractor shall, before issuance of the Certificate of Substantial Completion, complete or correct such item upon notification by the Architect. The Contractor shall then submit a request for another-inspection-by the Architect-to determine Substantial Completion. When the Work or designated portion thereof is substantially complete, the Architect will prepare a Certificate of Substantial Completion which shall establish the date of Substantial Completion, shall establish responsibilities of the Owner and Contractor for security, maintenance, heat, utilities, clamage to the Work and insurance; and shall fix the time within which the Contractor shall finish all items on the list accompanying the Certificate, Warranties required by the Contract Documents shall commence on the date of Substantial Completion of the Work or designated portion thereof unless otherwise provided in the Gertificate of Substantial Completion. The Certificate of Substantial Completion shall be sub mitted to the Owner and Contractor for their written acceptance of responsibilities assigned to them in such Certificate.

9.8.3 Upon Substantial Completion of the Work or designated portion thereof and upon application by the Contractor and certification by the Architect, the Owner shall make payment, reflecting adjustment in retainage, if any, for such Work or portion thereof as provided in the Contract Documents.

#### 9.9 PARTIAL OCCUPANCY OR USE

9.9.1 The Owner may occupy or use any completed or partially completed portion of the Work at any stage when such portion is designated by separate agreement with the Contractor, provided such occupancy or use is consented to by the insurer as required under Subparagraph 11.3.11 and authorized by public authorities having jurisdiction over the Work. Such partial occupancy or use may commence whether or not the portion is substantially complete, provided the Owner and Contractor have accepted in writing the responsibilities assigned to each of them for payments, retainage if any, security, maintenance, heat, utilities, damage to the Work and insurance, and have agreed in writing concerning the period for correction of the Work and commencement of warranties required by the Contract Documents. When the Contractor considers a portion substantially complete, the Contractor shall prepare and submit a list to the Architect as provided under Subparagraph 9.8.2. Consent of the Contractor to partial occupancy or use shall not be unreasonably withheld. The stage of the progress of the Work shall be determined by written agreement between the Owner and Contractor or, if no agreement is reached, by decision of the Architect-

9.9.2 Immediately prior to such partial occupancy or use, the Owner, Contractor and Architect shall jointly inspect the area to be occupied or portion of the Work to be used in order to determine and record the condition of the Work.

9.9.3 Unless otherwise agreed upon, partial occupancy or use of a portion or portions of the Work shall not constitute acceptance of Work not complying with the requirements of the Contract Documents.

#### 9.10 FINAL COMPLETION AND FINAL PAYMENT

**9.10.1** Upon receipt of written notice that the Work is ready for final inspection and acceptance and upon receipt of a final Application for Payment, the Architect will promptly make

such inspection and, when the Architect finds the Work acceptable under the Contract Documents and the Contract fully performed, the Architect will promptly issue a final Certificate for Payment stating that to the best of the Architect's knowledge, information and belief, and on the basis of the Architect's observations and inspections, the Work has been completed in accordance with terms and conditions of the Contract Documents and that the entire balance found to be due the Contractor and noted in said final Certificate is due and payable. The Architect's final Certificate for Payment will constitute a further representation that conditions listed in Subparagraph 9.10.2 as precedent to the Contractor's being entitled to final payment have been fulfilled.

9.10.2 Neither final payment nor any remaining retained percentage shall become due until the Contractor submits to the Architect (1) an affidavit that payrolls, bills for materials and equipment, and other indebtedness connected with the Work for which the Owner or the Owner's property might be responsible or encumbered (less amounts withheld by Owner) have been paid or otherwise satisfied, (2) a certificate evidencing that insurance required by the Contract Documents to remain in force after final payment is currently in effect and will not be cancelled or allowed to expire until at least 30 days' prior written notice has been given to the Owner, (3) a written statement that the Contractor knows of no substantial reason that the insurance will not be renewable to cover the period required by the Contract Documents, (4) consent of surety, if any, to final payment and (5), if required by the Owner, other data establishing payment or satisfaction of obligations, such as receipts, releases and waivers of liens, claims, security interests or encumbrances arising out of the Contract, to the extent and in such form as may be designated by the Owner. If a Subcontractor refuses to furnish a release or waiver required by the Owner, the Contractor may furnish a bond satisfactory to the Owner to indemnify the Owner against such lien. If such lien remains unsatisfied after payments are made, the Contractor shall refund to the Owner all money that the Owner may be compelled to pay in discharging such lien, including all costs and reasonable attorneys' fees.

9.10.3 If, after Substantial Completion of the Work, final completion thereof is materially delayed through no fault of the Contractor or by issuance of Change Orders affecting final completion, and the Architect so confirms, the Owner shall, upon application by the Contractor and certification by the Architect, and without terminating the Contract, make payment of the balance due for that portion of the Worl: fully completed and accepted. If the remaining balance for Work not fully completed or corrected is less than retainage supulated in the Contract Documents, and if bonds have been furnished, the written consent of surety to-payment of the balance due for that portion of the Work fully completed and accepted shall be submitted by the Contractor to the Architect prior to certification of such payment. Such payment shall be made under terms and conditions governing final-payment, except that it shall-not constitute a waiver of claims. The making of final payment shall constitute a waiver of claims by the Owner as provided in Subparagraph 4.3.5.

**9.10.4** Acceptance of final payment by the Contractor, a Subcontractor or material supplier shall constitute a waiver of claims by that payee except those previously made in writing and identified by that payee as unsettled at the time of final Application for Payment. Such waivers shall be in addition to the waiver described in Subparagraph 4.3.5.

# **ARTICLE 10**

# PROTECTION OF PERSONS AND PROPERTY

#### 10.1 SAFETY PRECAUTIONS AND PROGRAMS

- 10.1.1 The Contractor shall be responsible for initiating, maintaining and supervising all safety precautions and programs in connection with the performance of the Contract.
- 10.1.2 In the event the Contractor encounters on the site material reasonably believed to be asbestos or polychlorinated biphenyl (PCB) which has not been rendered harmless, the Contractor shall immediately stop Work in the area affected and report the condition to the Owner and Architect in writing. The Work in the affected area shall not thereafter be resumed except by written agreement of the Owner and Contractor if in fact the material is asbestos or polychlorinated biphenyl (PCB) and has not been rendered harmless. The Work in the affected area shall be resumed in the absence of asbestos or polychlorinated biphenyl (PCB), or when it has been rendered harmless, by written agreement of the Owner and Contractor, or in accordance with final determination by the Architect on which arbitration has not been demanded, or by arbitration under Article 4.
- **10.1.3** The Contractor shall not be required pursuant to Article 7 to perform without consent any Work relating to asbestos or polychlorinated biphenyl (PCB).
- 10.1.4 To the fullest extent permitted by law, the Owner shall indemnify and hold harmless the Contractor, Architect, Architect's consultants and agents and employees of any of them from and against claims, damages, losses and expenses, including but not limited to attorneys' fees, arising out of or resulting from performance of the Work in the affected area if in fact the material is asbestos or polychlorinated biphenyl (PCB) and has not been rendered harmless, provided that such claim, damage, loss or expense is attributable to bodily injury, sickness, disease or death, or to injury to or destruction of tangible property (other than the Work itself) including loss of use resulting therefrom, but only to the extent caused in whole or in part by negligent acts or omissions of the Owner, anyone directly or indirectly employed by the Owner or anyone for whose acts the Owner may be liable, regardless of whether or not such claim, damage, loss or expense is caused in part by a party indemnified hereunder. Such obligation shall not be construed to negate, abridge, or reduce other rights or obligations of indemnity which would otherwise exist as to a party or person described in this Subparagraph 10.1.4.

# 10.2 SAFETY OF PERSONS AND PROPERTY

- **10.2.1** The Contractor shall take reasonable precautions for safety of, and shall provide reasonable protection to prevent damage, injury or loss to:
  - .1 employees on the Work and other persons who may be affected thereby;
  - .2 the Work and materials and equipment to be incorporated therein, whether in storage on or off the site, under care, custody or control of the Contractor or the Contractor's Subcontractors or Sub-subcontractors; and
  - .3 other property at the site or adjacent thereto, such as trees, shrubs, lawns, walks, pavements, roadways, structures and utilities not designated for removal, relocation or replacement in the course of construction.

- **10.2.2** The Contractor shall give notices and comply with applicable laws, ordinances, rules, regulations and lawful orders of public authorities bearing on safety of persons or property or their protection from damage, injury or loss.
- 10.2.3 The Contractor shall erect and maintain, as required by existing conditions and performance of the Contract, reasonable safeguards for safety and protection, including posting danger signs and other warnings against hazards, promulgating safety regulations and notifying owners and users of adjacent sites and utilities.
- **10.2.4** When use or storage of explosives or other hazardous materials or equipment or unusual methods are necessary for execution of the Work, the Contractor shall exercise utmost care and carry on such activities under supervision of properly qualified personnel.
- 10.2.5 The Contractor shall promptly remedy damage and loss (other than damage or loss insured under property insurance required by the Contract Documents) to property referred to in Clauses 10.2.1.2 and 10.2.1.3 caused in whole or in part by the Contractor, a Subcontractor, a Sub-subcontractor, or anyone directly or indirectly employed by any of them, or by anyone for whose acts they may be liable and for which the Contractor is responsible under Clauses 10.2.1.2 and 10.2.1.3, except damage or loss attributable to acts or omissions of the Owner or Architect or anyone directly or indirectly employed by either of them, or by anyone for whose acts either of them may be liable, and not attributable to the fault or negligence of the Contractor. The foregoing obligations of the Contractor are in addition to the Contractor's obligations under Paragraph 3.18.
- **10.2.6** The Contractor shall designate a responsible member of the Contractor's organization at the site whose duty shall be the prevention of accidents. This person shall be the Contractor's superintendent unless otherwise designated by the Contractor in writing to the Owner and Architect.
- 10.2.7 The Contractor shall not load or permit any part of the construction or site to be loaded so as to endanger its safety.

#### 10.3 EMERGENCIES

**10.3.1** In an emergency affecting safety of persons or property, the Contractor shall act, at the Contractor's discretion, to prevent threatened damage, injury or loss. Additional compensation or extension of time claimed by the Contractor on account of an emergency shall be determined as provided in Paragraph 4.3 and Article 7.

# ARTICLE 11

#### INSURANCE AND BONDS

#### 11.1 CONTRACTOR'S LIABILITY INSURANCE

- 11.1.1 The Contractor shall purchase from and maintain in a company or companies lawfully authorized to do business in the jurisdiction in which the Project is located such insurance as will protect the Contractor from claims set forth below which may arise out of or result from the Contractor's operations under the Contract and for which the Contractor may be legally liable, whether such operations be by the Contractor or by a Subcontractor or by anyone directly or indirectly employed by any of them, or by anyone for whose acts any of them may be liable:
  - .1 claims under workers' or workmen's compensation, disability benefit and other similar employee benefit acts which are applicable to the Work to be performed;

- .2 claims for damages because of bodily injury, occupational sickness or disease, or death of the Contractor's employees;
- .3 claims for damages because of bodily injury, sickness or disease, or death of any person other than the Contractor's employees;
- .4 claims for damages insured by usual personal injury liability coverage which are sustained (1) by a person as a result of an offense directly or indirectly related to employment of such person by the Contractor, or (2) by another person;
- .5 claims for damages, other than to the Work itself, because of injury to or destruction of tangible property, including loss of use resulting therefrom;
- .6 claims for damages because of bodily injury, death of a person or property damage arising out of ownership, maintenance or use of a motor vehicle; and
- .7 claims involving contractual liability insurance applicable to the Contractor's obligations under Paragraph 3.18.
- 11.1.2 The insurance required by Subparagraph 11.1.1 shall be written for not less than limits of liability specified in the Contract Documents or required by law, whichever coverage is greater. Coverages, whether written on an occurrence or claims-made basis, shall be maintained without interruption from date of commencement of the Work until date of final payment and termination of any coverage required to be maintained after final payment.
- 11.1.3 Certificates of Insurance acceptable to the Owner shall be filed with the Owner prior to commencement of the Work. These Certificates and the insurance policies required by this Paragraph 11.1 shall contain a provision that coverages afforded under the policies will not be cancelled or allowed to expire until at least 30 days' prior written notice has been given to the Owner. If any of the foregoing insurance coverages are required to remain in force after final payment and are reasonably available, an additional certificate evidencing continuation for Payment as required by Subparagraph 9.10.2. Information concerning reduction of coverage shall be furnished by the Contractor with reasonable promptness in accordance with the Contractor's information and belief.

#### 11.2 OWNER'S LIABILITY INSURANCE

11.2.1 The Owner shall be responsible for purchasing and maintaining the Owner's usual liability insurance. Optionally, the Owner may purchase and maintain other insurance for self-protection against claims which may arise from operations under the Contract. The Contractor shall not be responsible for purchasing and maintaining this optional Owner's liability insurance unless specifically required by the Contract Documents.

#### 11.3 PROPERTY INSURANCE

11.3.1 Unless otherwise provided, the Owner shall purchase and maintain, in a company or companies lawfully authorized to do business in the jurisdiction in which the Project is located, property insurance in the amount of the initial Contract Sum as well as subsequent modifications thereto for the entire Work at the site on a replacement cost basis without voluntary deductibles. Such property insurance shall be main tained, unless otherwise provided in the Contract Documents or otherwise agreed in writing by all persons and entities who are beneficiaries of such insurance, until final payment has been made as provided in Paragraph 9.10 or until no person or entity

- other than the Owner has an insurable interest in the property required by this Paragraph 11.3 to be covered, whichever is earlier. This insurance shall include interests of the Owner, the Contractor, Subcontractors and Sub-subcontractors in the Work.
- 11.3.1.1 Property insurance shall be on an all-risk policy form and shall insure against the perils of fire and extended coverage and physical loss or damage including, without duplication of coverage, theft, vandalism, malicious mischief, collapse, falsework, temporary buildings and debris removal including demolition occasioned by enforcement of any applicable legal requirements, and shall cover reasonable compensation for Architect's services and expenses required as a result of such insured loss. Coverage for other perils shall not be required unless otherwise provided in the Contract Documents.
- 11.3.1.2 If the Owner does not intend to purchase such property insurance required by the Contract and with all of the coverages in the amount described above, the Owner shall so inform the Contractor in writing prior to commencement of the Work. The Contractor may then effect insurance which will protect the interests of the Contractor, Subcontractors and Subsubcontractors in the Work, and by appropriate Change Order the cost thereof shall be charged to the Owner. If the Contractor is damaged by the failure or neglect of the Owner to purchase or maintain insurance as described above, without so notifying the Contractor, then the Owner shall bear all reasonable costs properly attributable thereto.
- 11.3.1.3 If the property insurance requires minimum deductibles and such deductibles are identified in the Contract Documents, the Contractor shall pay costs not covered because of such deductibles. If the Owner or insurer increases the required minimum deductibles above the amounts so identified or if the Owner elects to purchase this insurance with voluntary deductible amounts, the Owner shall be responsible for payment of the additional costs not covered because of such increased or voluntary deductibles. If deductibles are not identified in the Contract Documents, the Owner shall pay costs not covered because of deductibles.
- 11.3.1.4 Unless otherwise provided in the Contract Documents, this property insurance shall cover portions of the Work stored off the site after written approval of the Owner at the value established in the approval, and also portions of the Work in transit.
- 11.3.2 Boller and Machinery Insurance. The Owner shall purchase and maintain boller and machinery insurance required by the Contract Documents or by law, which shall specifically cover such insured objects during installation and until final acceptance by the Owner; this insurance shall include interests of the Owner, Contractor, Subcontractors and Subsubcontractors in the Work, and the Owner and Contractor shall be named insureds.
- 11.3.3 Loss of Use Insurance. The Owner, at the Owner's option, may purchase and maintain such insurance as will insure the Owner against loss of use of the Owner's property due to fire or other hazards, however caused. The Owner waives all rights of action against the Contractor for loss of use of the Owner's property, including consequential losses due to fire or other hazards however caused.
- 11.3.4 If the Contractor requests in writing that insurance for risks other than those described herein or for other special hazards be included in the property insurance policy, the Owner shall, if possible, include such insurance, and the cost thereof shall be charged to the Contractor by appropriate Change Order.

- 11.3.5 If during the Project construction period the Owner insures properties, real or personal or both, adjoining or adjacent to the site by property insurance under policies separate from those insuring the Project, or if after final payment property insurance is to be provided on the completed Project through a policy or policies other than those insuring the Project during the construction period, the Owner shall waive all rights in accordance with the terms of Subparagraph 11.3.7 for damages caused by fire or other perils covered by this separate property insurance. All separate policies shall provide this waiver of subrogation by endorsement or otherwise.
- 11.3.6 Before an exposure to loss may occur, the Owner shall file with the Contractor a copy of each policy that includes insurance coverages required by this Paragraph 11.3. Each policy shall contain all generally applicable conditions, definitions, exclusions and endorsements related to this Project. Each policy shall contain a provision that the policy will not be cancelled or allowed to expire until at least 30 days' prior written notice has been given to the Contractor.
- 11.3.7 Waivers of Subrogation. The Owner and Contractor waive all rights against (1) each other and any of their subcontractors, sub-subcontractors, agents and employees, each of the other, and (2) the Architect, Architect's consultants, separate contractors described in Article 6, if any, and any of their subcontractors, sub-subcontractors, agents and employees, for damages caused by fire or other perils to the extent covered by property insurance obtained pursuant to this Paragraph 11.3 or other property insurance applicable to the Work, except such rights as they have to proceeds of such insurance held by the Owner as fiduciary. The Owner or Contractor, as appropriate, shall require of the Architect, Architect's consultants, separate contractors described in Article 6, if any, and the subcontractors, sub-subcontractors, agents and employees of any of them, by appropriate agreements, written where legally required for validity, similar waivers each in favor of other parties enumerated herein. The policies shall provide such waivers of subrogation by endorsement or otherwise. A waiver of subrogation shall be effective as to a person or entity even though that person or entity would otherwise have a duty of indemnification, contractual or otherwise, did not pay the insurance premium directly or indirectly, and whether or not the person or entity had an insurable interest in the property damaged.
- 11.3.8 A loss insured under Owner's property insurance shall be adjusted by the Owner as fiduciary and made payable to the Owner as fiduciary for the insureds, as their interests may appear, subject to requirements of any applicable mortgagee clause and of Subparagraph 11.3.10. The Contractor shall pay Subcontractors their just shares of insurance proceeds received by the Contractor, and by appropriate agreements, written where legally required for validity, shall require Subcontractors to make payments to their Sub-subcontractors in similar manner.
- 11.3.9 If required in writing by a party in interest, the Owner as fiduciary shall, upon occurrence of an insured loss, give bond for proper performance of the Owner's duties. The cost of required bonds shall be charged against proceeds received as fiduciary. The Owner shall deposit in a separate account proceeds so received, which the Owner shall distribute in accordance with such agreement as the parties in interest may reach, or in accordance with an arbitration award in which case the procedure shall be as provided in Paragraph 4.5. If after such loss no other special agreement is made, replacement of damaged property shall be covered by appropriate Change Order.

- 11.3.10 The Owner as fiduciary shall have power to adjust and settle a loss with insurers unless one of the parties in interest shall object in writing within five days after occurrence of loss to the Owner's exercise of this power; if such objection be made, arbitrators shall be chosen as provided in Paragraph 4.5. The Owner as fiduciary shall, in that case, make settlement with insurers in accordance with directions of such arbitrators. If distribution of insurance proceeds by arbitration is required, the arbitrators will direct such distribution.
- 11.3.11 Partial occupancy or use in accordance with Paragraph 9.9 shall not commence until the Insurance company or companies providing property insurance have consented to such partial occupancy or use by endorsement or otherwise. The Owner and the Contractor shall take reasonable steps to obtain consent of the insurance company or companies and shall, without mutual written consent, take no action with respect to partial occupancy or use that would cause cancellation, lapse or reduction of insurance.

# 11.4 PERFORMANCE BOND AND PAYMENT BOND

- 11.4.1 The Owner shall have the right to require the Contractor to furnish bonds covering faithful performance of the Contract and payment of obligations arising thereunder as stipulated in bidding requirements or specifically required in the Contract Documents on the date of execution of the Contract.
- 11.4.2 Upon the request of any person or entity appearing to be a potential beneficiary of bonds covering payment of obligations arising under the Contract, the Contractor shall promptly furnish a copy of the bonds or shall permit a copy to be made.

#### **ARTICLE 12**

# **UNCOVERING AND CORRECTION OF WORK**

#### 12.1 UNCOVERING OF WORK

- **12.1.1** If a portion of the Work is covered contrary to the Architect's request or to requirements specifically expressed in the Contract Documents, it must, if required in writing by the Architect, be uncovered for the Architect's observation and be replaced at the Contractor's expense without change in the Contract Time.
- 12.1.2 If a portion of the Work has been covered which the Architect has not specifically requested to observe prior to its being covered, the Architect may request to see such Work and it shall be uncovered by the Contractor. If such Work is in accordance with the Contract Documents, costs of uncovering and replacement shall, by appropriate Change Order, be charged to the Owner. If such Work is not in accordance with the Contract Documents, the Contractor shall pay such costs unless the condition was caused by the Owner or a separate contractor in which event the Owner shall be responsible for payment of such costs.

# 12.2 CORRECTION OF WORK

- 12.2.1 The Contractor shall promptly correct Work rejected by the Architect or failing to conform to the requirements of the Contract Documents, whether observed before or after Substantial Completion and whether or not fabricated, installed or completed. The Contractor shall bear costs of correcting such rejected Work, including additional testing and inspections and compensation for the Architect's services and expenses made necessary thereby.
- **12.2.2.** If, within one year after the date of Substantial Completion of the Work or designated portion thereof, or after the date

for commencement of warranties established under Subparagraph 9.9.1, or by terms of an applicable special warranty required by the Contract Documents, any of the Work is found to be not in accordance with the requirements of the Contract Documents, the Contractor shall correct it promptly after receipt of written notice from the Owner to do so unless the Owner has previously given the Contractor a written acceptance of such condition. This period of one year shall be extended with respect to portions of Work first performed after Substantial Completion by the period of time between Substantial Completion and the actual performance of the Work. This obligation under this Subparagraph 12.2.2 shall survive acceptance of the Work under the Contract and termination of the Contract. The Owner shall give such notice promptly after discovery of the condition.

- **12.2.3** The Contractor shall remove from the site portions of the Work which are not in accordance with the requirements of the Contract Documents and are neither corrected by the Contractor nor accepted by the Owner.
- 12.2.4 If the Contractor fails to correct nonconforming Work within a reasonable time, the Owner may correct it in accordance with Paragraph 2.4. If the Contractor does not proceed with correction of such nonconforming Work within a reasonable time fixed by written notice from the Architect, the Owner may remove it and store the salvable materials or equipment at the Contractor's expense. If the Contractor does not pay costs of such removal and storage within ten days after written notice, the Owner may upon ten additional days' written notice sell such materials and equipment at auction or at private sale and shall account for the proceeds thereof, after deducting costs and damages that should have been borne by the Contractor, including compensation for the Architect's services and expenses made necessary thereby. If such proceeds of sale do not cover costs which the Contractor should have borne, the Contract Sum shall be reduced by the deficiency. If payments then or thereafter due the Contractor are not sufficient to cover such amount, the Contractor shall pay the difference to the Owner
- **12.2.5** The Contractor shall bear the cost of correcting destroyed or damaged construction, whether completed or partially completed, of the Owner or separate contractors caused by the Contractor's correction or removal of Work which is not in accordance with the requirements of the Contract Documents.
- 12.2.6 Nothing contained in this Paragraph 12.2 shall be construed to establish a period of limitation with respect to other obligations which the Contractor might have under the Contract Documents. Establishment of the time period of one year as described in Subparagraph 12.2.2 relates only to the specific obligation of the Contractor to correct the Work, and has no relationship to the time within which the obligation to comply with the Contract Documents may be sought to be enforced, nor to the time within which proceedings may be commenced to establish the Contractor's liability with respect to the Contractor's obligations other than specifically to correct the Work.

#### 12.3 ACCEPTANCE OF NONCONFORMING WORK

**12.3.1** If the Owner prefers to accept Work which is not in accordance with the requirements of the Contract Documents, the Owner may do so instead of requiring its removal and correction, in which case the Contract Sum will be reduced as appropriate and equitable. Such adjustment shall be effected whether or not final payment has been made.

#### **ARTICLE 13**

#### MISCELLANEOUS PROVISIONS

#### 13.1 GOVERNING LAW

**13.1.1** The Contract shall be governed by the law of the place where the Project is located.

#### 13.2 SUCCESSORS AND ASSIGNS

13.2.1 The Owner and Contractor respectively bind themselves, their partners, successors, assigns and legal representatives to the other party hereto and to partners, successors, assigns and legal representatives of such other party in respect to covenants, agreements and obligations contained in the Contract Documents. Neither party to the Contract shall assign the Contract as a whole without written consent of the other. If either party attempts to make such an assignment without such consent, that party shall nevertheless remain legally responsible for all obligations under the Contract.

#### 13.3 WRITTEN NOTICE

**13.3.1** Written notice shall be deemed to have been duly served if delivered in person to the individual or a member of the firm or entity or to an officer of the corporation for which it was intended, or if delivered at or sent by registered or certified mail to the last business address known to the party giving notice.

# 13.4 RIGHTS AND REMEDIES

- **13.4.1** Duties and obligations imposed by the Contract Documents and rights and remedies available thereunder shall be in addition to and not a limitation of duties, obligations, rights and remedies otherwise imposed or available by law.
- 13.4.2 No action or failure to act by the Owner, Architect or Contractor shall constitute a waiver of a right or duty afforded them under the Contract, nor shall such action or failure to act constitute approval of or acquiescence in a breach thereunder, except as may be specifically agreed in writing.

#### 13.5 TESTS AND INSPECTIONS

- 13.5.1 Tests, inspections and approvals of portions of the Work required by the Contract Documents or by laws, ordinances, rules, regulations or orders of public authorities having jurisdiction shall be made at an appropriate time. Unless otherwise provided, the Contractor shall make arrangements for such tests, inspections and approvals with an independent testing laboratory or entity acceptable to the Owner, or with the appropriate public authority, and shall bear all related costs of tests, inspections and approvals. The Contractor shall give the Architect timely notice of when and where tests and inspections are to be made to the Architect may observe such procedures. The Owner shall bear costs of tests, inspections or approvals which do not become requirements until after bids are received or negotiations concluded.
- 13.5.2 If the Architect, Owner or public authorities having jurisdiction determine that portions of the Work require additional testing, inspection or approval not included under Subparagraph 13.5.1, the Architect will, upon written authorization from the Owner, instruct the Contractor to make arrangements for such additional testing, inspection or approval by an entity-acceptable to the Owner, and the Contractor shall give timely notice to the Architect of when and where tests and inspections are to be made so the Architect may observe such procedures.

The Owner shall bear such costs except as provided in Subparagraph 13.5.3.

13.5.3 If such procedures for testing, inspection or approval under Subparagraphs 13.5.1 and 13.5.2 reveal failure of the portions of the Work to comply with requirements established by the Contract Documents, the Contractor shall bear all costs made necessary by such failure including those of repeated procedures and compensation for the Architect's services and expenses.

13.5.4 Required-certificates of testing, inspection or approval shall, unless otherwise required by the Contract Documents, be secured by the Contractor and promptly delivered to the Architect.

**13.5.5** If the Architect is to observe tests, inspections or approvals required by the Contract Documents, the Architect will do so promptly and, where practicable, at the normal place of testing.

**13.5.6** Tests or inspections conducted pursuant to the Contract Documents shall be made promptly to avoid unreasonable delay in the Work.

#### 13.6 INTEREST

**13.6.1** Payments due and unpaid under the Contract Documents shall bear interest from the date payment is due at such rate as the parties may agree upon in writing or, in the absence thereof, at the legal rate prevailing from time to time at the place where the Project is located.

# 13.7 COMMENCEMENT OF STATUTORY LIMITATION PERIOD

# 13.7.1 As between the Owner and Contractor:

- .1 Before Substantial Completion. As to acts or failures to act occurring prior to the relevant date of Substantial Completion, any applicable statute of limitations shall commence to run and any alleged cause of action shall be deemed to have accrued in any and all events not later than such date of Substantial Completion;
- .2 Between Substantial Completion and Final Certificate for Payment. As to acts or failures to act occurring subsequent to the relevant date of Substantial Completion and prior to issuance of the final Certificate for Payment, any applicable statute of limitations shall commence to run and any alleged cause of action shall be deemed to have accrued in any and all events not later than the date of issuance of the final Certificate for Payment; and
- .3 After Final Certificate for Payment. As to acts or failures to act occurring after the relevant date of issuance of the final Certificate for Payment, any applicable statute of limitations shall commence to run and any alleged cause of action shall be deemed to have accrued in any and all events not later than the date of any act or failure to act by the Contractor pursuant to any warranty provided under Paragraph 3.5, the date of any correction of the Work or failure to correct the Work by the Contractor under Paragraph 12.2, or the date of actual commission of any other act or failure to perform any duty or obligation by the Contractor or Owner, whichever occurs last.

#### **ARTICLE 14**

# TERMINATION OR SUSPENSION OF THE CONTRACT

# 14.1 TERMINATION BY THE CONTRACTOR

- **14.1.1** The Contractor may terminate the Contract if the Work is stopped for a period of 30 days through no act or fault of the Contractor or a Subcontractor, Sub-subcontractor or their agents or employees or any other persons performing portions of the Work under contract with the Contractor, for any of the following reasons:
  - .1 issuance of an order of a court or other public authority having jurisdiction;
  - an act of government, such as a declaration of national emergency, making material unavailable;
  - .3 because the Architect has not issued a Certificate for Payment and has not notified the Contractor of the reason for withholding certification as provided in Subparagraph 9.4.1, or because the Owner has not made payment on a Certificate for Payment within the time stated in the Contract Documents;
  - .4 if repeated suspensions, delays or interruptions by the Owner as described in Paragraph 14.3 constitute in the aggregate more than 100 percent of the total number of days scheduled for completion, or 120 days in any 365-day period, whichever is less; or
  - .5 the Owner has failed to furnish to the Contractor promptly, upon the Contractor's request, reasonable evidence as required by Subparagraph 2.2.1.
- **14.1.2** If one of the above reasons exists, the Contractor may, upon seven additional days' written notice to the Owner and Architect, terminate the Contract and recover from the Owner payment for Work executed and for proven loss with respect to materials, equipment, tools, and construction equipment and machinery, including reasonable overhead, profit and damages.
- 14.1.3 If the Work is stopped for a period of 60 days through no act or fault of the Contractor or a Subcontractor or their agents or employees or any other persons performing portions of the Work under contract with the Contractor because the Owner has persistently failed to fulfill the Owner's obligations under the Contract Documents with respect to matters important to the progress of the Work, the Contractor may, upon seven additional days' written notice to the Owner and the Architect, terminate the Contract and recover from the Owner as provided in Subparagraph 14.1.2.

# 14.2 TERMINATION BY THE OWNER FOR CAUSE

- **14.2.1** The Owner may terminate the Contract if the Contractor:
  - .1 persistently or repeatedly refuses or fails to supply enough properly skilled workers or proper materials;
  - .2 fails to make payment to Subcontractors for materials or labor in accordance with the respective agreements between the Contractor and the Subcontractors;
  - .3 persistently disregards laws, ordinances, or rules, regulations or orders of a public authority having jurisdiction; or
  - .4 otherwise is guilty of substantial breach of a provision of the Contract Documents.
- 14.2.2 When any of the above reasons exist, the Owner, upon certification by the Architect that sufficient cause exists to jus-

tify such action, may without prejudice to any other rights or remedies of the Owner and after giving the Contractor and the Contractor's surety, if any, seven days' written notice, terminate employment of the Contractor and may, subject to any prior rights of the surety:

- .1 take possession of the site and of all materials, equipment, tools, and construction equipment and machinery thereon owned by the Contractor;
- .2 accept assignment of subcontracts pursuant to Paragraph 5.4; and
- .3 finish the Work by whatever reasonable method the Owner may deem expedient.
- **14.2.3** When the Owner terminates the Contract for one of the reasons stated in Subparagraph 14.2.1, the Contractor shall not be entitled to receive further payment until the Work is finished.
- **14.2.4** If the unpaid balance of the Contract Sum exceeds costs of finishing the Work, including compensation for the Architect's services and expenses made necessary thereby, such excess shall be paid to the Contractor. If such costs exceed the unpaid balance, the Contractor shall pay the difference to the

Owner. The amount to be paid to the Contractor or Owner, as the case may be, shall be certified by the Architect, upon application, and this obligation for payment shall survive termination of the Contract.

# 14.3 SUSPENSION BY THE OWNER FOR CONVENIENCE

- **14.3.1** The Owner may, without cause, order the Contractor in writing to suspend, delay or interrupt the Work in whole or in part for such period of time as the Owner may determine.
- **14.3.2** An adjustment shall be made for increases in the cost of performance of the Contract, including profit on the increased cost of performance, caused by suspension, delay or interruption. No adjustment shall be made to the extent:
  - .1 that performance is, was or would have been so suspended, delayed or interrupted by another cause for which the Contractor is responsible; or
  - .2 that an equitable adjustment is made or denied under another provision of this Contract.
- 14.3.3 Adjustments made in the cost of performance may have a mutually agreed fixed or percentage fee.

11/01/16

The following supplements modify, change, delete from or add to the Division 1 - General Requirements Section 006001 "General Conditions of the Contract for Construction", AIA Document A201, Fourteenth Edition, 1987. Where any subparagraph or Clause thereof is modified or deleted by these Supplementary Conditions, the unaltered provisions of that Article, Paragraph, Subparagraph or Clause shall remain in effect.

Following Article numbers refer to Articles in the "General Conditions of the Contract for Construction".

# **ARTICLE 1 - GENERAL PROVISIONS**

#### ARTICLE 1.1.1

# Amended as follows:

Change last sentence to read as follows:

"The Contract Documents shall also include, but are not limited to, Notice of Award, the Notice to Contractors, Contract Agreement, Instructions to Bidders, Bid Proposal, All Addenda, Labor and Material and Faithful Performance Bonds and Certificates, General and Supplementary Conditions, all items included in Division 0 -Bidding and Contract Documents" and all items included in Division 1 - "General Requirements". Technical Sections - two through thirty three.

# ARTICLE 1.2.1

#### Change to read:

The Owner has reviewed the Contract Documents issued for bidding for this project. The Contractor by submitting a proposal for this project affirms that he or she has reviewed all the contract documents issued for and during bidding by act of submitting a proposal for this project.

# ARTICLE 1.2.2

#### Add the following to 1.2.2:

No relief, neither extra payment nor time adjustment shall subsequently be made on behalf of the Contractor for any extra expense to which the Contractor may be put due to failure or neglect to make such examination.

# ARTICLE 1.2.6

#### Add the following:

- Larger scale drawings govern smaller scale, and figured dimensions govern scale measurements.
   No measurements shall be scaled. If figures or information on any drawings are insufficient, the Contractor shall secure the Architect's Instructions before proceeding.
- 2. Misplacement, addition, typographical error, or omission of any word, letter, figure or punctuation mark in no way changes the true spirit, intent or meaning of the drawings and specifications.
- 3. Naming any material and/or equipment requires Contractor to furnish and install the named material/equipment, including all incidental and accessory items and/or labor necessary to achieve full and complete functioning of the material and/or equipment according to the best practices of the trade(s) involved, unless specifically noted otherwise.
- 4. In the event of conflicting statements or requirements the Contractor shall notify the Architect prior to bidding and in sufficient time to permit the issuance of written clarification. Failure to notify the Architect prior to bidding shall not relieve the Contractor of the responsibility of so doing before any affected work is performed, at which time he agrees to accept the decision of the Architect and proceed accordingly without additional compensation.
- 5. The Architect's decision shall be binding in the case of determining intents or meanings. Abbreviations and symbols not indicated on legends shall be per Architect's decision regardless of standards of the construction industry.
  - Contractor shall confirm any differences prior to bidding.
- 6. In the event conflicts between Working Drawings and Project Manual, or within either document, the better quality or greater quantity shall prevail. The Contract Documents are complementary, and what is required by one shall be as binding as if required by all. The Contract Documents are intended to encompass all labor and materials, equipment, and transportation necessary for proper execution of the Work. Any item of work mentioned in the Specifications and not shown on the

- drawings, or shown on the drawings and not mentioned in the Specifications, shall be provided by Contractor as if shown in both.
- 7. The general character of the detail work is shown on the Contract Drawings. The Architect may furnish additional details to more fully explain the work, and same shall be considered a part of the Contract. Any work executed before receipt of such details, if not in accordance with same, shall be removed and replaced, or adjusted, as directed, without expense to the Owner. Should any detail, field directive and/or notification through Architect's Activity Report be submitted later than the Contract Drawings be, in the opinion of the Contractor, more elaborate than the Contract Drawings and the Project Manual indicate, written notice thereof shall be given to the Architect within five (5) days of receipt of same. The assertion will then be considered, and, if justified, said detail drawings will be amended or the extra work authorized. Non-receipt of such notice shall relieve the Owner of any additional cost.
- 8. In the event of conflict or inconsistency between the Supplementary Conditions and the General Conditions, the Supplementary Conditions shall control. Where any portion of the Contract Documents are silent and information appears elsewhere in the Contract Documents, such other portions of the Contract Documents shall control.
- 9. Details shown on drawings both typical and specific shall be incorporated in the project at all appropriate locations whether specifically referenced at each location or not. Work not particularly detailed, marked or specified shall be the same at similar parts that are detailed, marked or specified.
- 10. The drawings and Specifications describe the work to be performed by Contractor. Generally, the Specifications describe work which cannot be readily indicated on the drawings and indicate types, qualities, and methods of installation of the various materials and equipment required for the Work. It is not intended to mention every item of work in the Specifications which can be adequately shown on the drawings, or to show on the drawings all items of work described or required by the Specifications even if they could have been shown.
- 11. Locations of electrical conduits, mechanical items, utility lines, landscape irrigation, other lines and plumbing lines are diagramed on the Construction Documents. Plumbing, Mechanical, Landscape Irrigation, Fire Protection and Electrical Plans are diagrammatic, and, therefore, do not necessarily represent the exact as-built installation and or field conditions. Each Contractor shall field verify and coordinate locations prior to proceeding with work to confirm routing and connections. Contractor shall be responsible for the final routing lengths. No extra will be allowed for increases due to routing obstructions, depth deviations, off sets due to above and underground appurtenances, coordination of lines or length of line from POC to POC that occur due to actual field conditions with limits herein. The drawings indicate diagrammatically what is to be done and shall not be a basis for extra cost within limits herein.
  - a. Refer to Architect's Plans and Specifications for construction details which will affect the work and equipment. Examine all trade Plans and Specifications to ensure that this work does not conflict with any trades. It shall remain the Contractor's responsibility to submit Shop Drawings/Coordination Drawings, if he/she has any questions about the final layout. Nothing on these Plans or Project Manual/Specifications shall be construed to permit work not conforming to all applicable codes and regulations.
    - 1) Contractor shall allow for deviance of up to 40 feet for each existing line(s) shown on drawings with reference to line locations, cross overs and rerouting around appurtenances (above and below grade) and related items.
    - No extra cost or credit will be allowed or requested for points of connections 20+/- feet deviating from location(s) of point of connection(s) indicated on drawings.
    - Depths of existing lines shall be field verified with pot holes by contractor prior to excavation for any trenches for utilities. Contractor shall allow for a deviance of 36" for underground lines if a depth is indicated on existing utility lines or utility drawings. If no depth is indicated on plans of utility lines, contractor is responsible for POC regardless of depth of line. Contractor is responsible for making transition to final POC depth of line with required depth of line indicated on drawings.

- 12. The existence and location of any underground utility piping, conduits and or structures shown on the construction documents were obtained by a search of available records. Existing drawings of the site utilities are on file at the office of the Maintenance and Operations. Contractor is to review these drawings prior to starting work. To the best of the architects and consulting engineer's knowledge there are no existing utilities except as shown on these plans. These plans are intended to show locations of electrical lines and landscape irrigation lines that they may be in construction area. The contractor shall take due precautionary measures to protect all utilities and structures shown on these plans and those discovered during construction. The contractor shall be responsible for field verification of the location and depth of existing underground and above ground utilities and shall perform potholing as necessary at all potential and assumed crossings prior to commencing on site construction. The contractor shall be responsible for taking all precautions necessary to protect all existing utilities and structures from damage during and the course of the work, and shall be responsible for repairing or replacing any utilities or structures damage during the course of the work.
  - a. The Contractor is required by Government Code 4216 "California One Call Law" to call dig alert at least two (2) working days before digging. Contractor is required by same code to hand-expose to the point of no conflict 24" on either side of the underground facility so you can determine its exact location before using power equipment.
- 13. Contractor Waives Common Practice: The Contractor shall waive "common practice" and "common usage" as construction criteria wherever the Contract Documents details, plans, specifications, governing codes or ordinances require greater quantity or better quality than common practice or common usage would require.
- 14. Contractor is aware that this Project may be split into several phases, (refer to summary of work herein). If the Project is split into phases, Contractor has made allowances for any delays or damages which may arise from coordination with contractors for other phases. If any delays should arise from a contractor working on a different phase, Contractor's sole remedy for damages, including delay damages, shall be against the contractor who caused such damage and not against Owner. Contractor shall provide access to contractors for other phases as necessary to prevent delays and damages to contractors working on other phases of construction.
- 15. Definitions:
  - a. <u>Equivalent to</u>: Equal or superior in function and quality and approved by the Architect.
  - b. <u>Furnish</u>: Purchase and deliver to job site in new condition as well as "supply and deliver to the project site, ready for unloading, unpacking, assembly, installation, and similar operations."
  - c. <u>Indicated:</u> Refers to graphic representations, notes or schedules on the Drawings, or other Paragraphs or Schedules in Project Manual, and similar requirements in Contract Documents. Where terms such as "shown," "noted," or "scheduled" are used, it is to help locate the reference; no limitation on locations is intended except as specifically noted.
  - d. <u>Install:</u> Receive and store at job site until required; in place secure and connect; furnish required appurtenances. Used to describe operations at the project site, including the actual "unloading, unpacking, assembly, erection, placing, anchoring, applying, working to dimension, finishing, curing, protection, cleaning and similar operations."
  - e. <u>Installer</u>: An entity engaged by CONTRACTOR, either as an employee, subcontractor, or sub-subcontractor for performance of a particular construction activity, including installation, erection, application, and similar required operations. Installers are required to be experienced in the operations they are engaged to perform and licensed as required in the individual specification sections.
  - f. Or Equal: Where named products in specification text are accompanied or are deemed by law to be followed by the term "or equal," or other language of similar effect, CONTRACTOR shall comply with those Contract Document provisions for "substitutions" when obtaining Architect's review and consideration.
  - g. <u>Provide</u>: Includes "provide complete in place," that is, furnish and install.
  - h. <u>Refer:</u> Indicates that the subject is defined or specified in further detail at another location in the Contract Documents or elsewhere as indicated. Except, as otherwise noted, "refer" does not imply that CONTRACTOR must purchase or subcontract the subject work in any special manner.

- i. Section: Refers to a Section of this Project Manual.
- j. <u>Standards</u>: The most recent issue approved and accepted.
- k. As Shown, etc: Where "as shown", "as indicated", "as detailed", or words of similar import are used, reference is made to the Drawings accompanying the Specifications unless otherwise stated. Where "as directed", "as required", "as permitted", "as authorized", "as accepted", "as selected", or words of similar import are used, the direction, requirement, permission, authorization, approval, acceptance or selection by Architect is intended unless otherwise stated.
- I. <u>Plural:</u> Words in singular shall include the plural whenever applicable or the context so indicates.

# Add the following paragraph to Article 1:

# ARTICLE 1.6

## **DSA Requirements:**

- 1. All work shall comply with Title 24, C.C.R., Part 1-6, 9 and 12.
- 2. Title 24, Parts I-5 must be kept on the site during construction.
- 3. Substitutions affecting DSA regulated items, not limited to Structural Safety, Fire and Life Safety and Access Compliance will be considered as changes to approved plans or specifications. They are to be treated as addenda or CCD, and will require DSA's approval prior to fabrication and installation. (Sec. 4-338, Part 1, and IR A-6)
- 4. All addenda must be stamped and signed by A/E on Record and delegated Design Professional when applicable, and approved by DSA. (Sec 4-338 (Sec. 4-338(b), Part 1)
- 5. Construction Change Documents (CCD) and supplementary drawings must be signed by A/E on Record or delegated Design Professional and approved by DSA prior to commencement of work shown. (DSA IR A-6)
- 6. Should any existing conditions such as deterioration or non-confirming construction be discovered which is not covered by the DSA approved documents wherein the finished work will not comply with Title 24, California Code of Regulations, a construction change document or a separate set of plans and specifications, detailing and specifying the required repair work shall be submitted to and approved by DSA before proceeding with the repair work.
- 7. A Project Inspector and testing lab when required must be employed by the Owner and approved by A/E of Record, Structural Engineer when applicable and DSA. (Sec.4-341 (d), Part 1)
- 8. Scope of work: Refer to "Summary of Work" section of the Project Manual is intended to match the scope of work on the cover drawing.
- 1. All work shall comply with Title 24, C.C.R., Part 1-6, 9 and 12.
- 2. Title 24, Parts I-5 must be available during construction.
- 3. Substitutions affecting regulated items, not limited to Structural Safety, Fire and Life Safety and Access Compliance will be considered as changes to approved plans and/or specifications. They are to be treated as addenda or change orders, and will require Architects approval prior to fabrication and installation.
- 4. Construction change directives (preliminary change orders) must be approved by Architect
- 5. Change orders and supplementary drawings must be approved by Architect
- A Project Inspector and testing lab when required must be employed by the Owner and approved by Architect
- 7. Scope of work: Refer to "Summary of Work" section of the Project Manual is intended to match the scope of work on the cover drawing.
- 8. Deferred approvals: "The submittal shall be approved by the Architect. Fabrication and installation of deferred approval item shall not be stated until drawings, specifications and engineering calculations (if needed) for the system have been approved and accepted by the Architect.

# **ARTICLE 2 - OWNER**

# ARTICLE 2.1.2

## Amend as follows:

Change "Mechanic's Lien" to "Stop Notice".

#### ARTICLE 2.2.1

Delete in its entirety.

## ARTICLE 2.2.2

#### Change to read:

The Contractor shall verify surveys describing physical characteristics as required for his work and utility locations for the site of the project. The Owner shall furnish a legal description of the site if requested, in writing, by the Contractor. Available utility drawings the Owner has shall be made available upon written request of Contractor. Contractor shall be responsible for any copies and associated cost of reproduction.

## ARTICLE 2.2.5

#### Change to read:

The Contractor will be furnished, free of charge, (25) copies of Drawings and Project Manuals. Additional copies are available from the Architect at printing cost plus 10% mark-up.

#### ARTICLE 2.3

## Add the following to paragraph 2.3.1:

The waiver by the Owner of any breach of any requirements, condition or provision of this Contract shall not be a waiver of any subsequent or other breach of this Contract nor of any terms, conditions or provisions of this Contract.

#### ARTICLE 2.4.1

## Amended as follows:

Second to last sentence change word "approval" to "review".

## **ARTICLE 2.4.1.1**

# Add the following:

Owner shall have first right of refusal for all items to be demolished and/or removed from project. Owner shall remove said items unless directed otherwise by Construction Documents. Owner shall notify Contractor of items he wishes to salvage within 60 days of contract award.

#### ARTICLE 2.5

# Add the following paragraph to Article 2:

- 2.5.1 The Owner will employ a project inspector to serve as Inspector for the Owner according to the regulations of the Division of the State Architect and subject to the provisions of Section 4-333b, Group 1, Chapter 4, Part 1, Title 24, California Code of Regulations. The Inspector shall be approved by the Division of the State Architect, Architect and Structural Engineer. The term "Inspector" means the project inspector employed by the Owner.
- 2.5.2 Duties, rights and authority of the Inspector will be as set forth in Section 4-342, Group 1, Chapter 4, Part 1, Title 24, California Code of Regulations. The Inspector will be responsible for periodic surveillance of the work in all stages of its progress.
- 2.5.3 The Owner shall have the right to visit the project site at any time. The Owner may bring visitors with them. Contractor shall provide safety devices for Owner, Architect and visitors.

#### ARTICLE 2.6

# Add the following paragraph to Article 2:

# 2.6 Construction Manager.

2.6.1 The Owner, at their option, may employ a Construction Manager and/or Project Manager to be their representative or to assist the Architect. The Construction Manager/Project Manager may, but not limited to, review and negotiate Change Request, order of magnitude cost for Owner-initiated Change Request, evaluate periodic Pay Request, evaluate Contractor's schedule, evaluate Contractor's manpower and other items as directed by Owner, and/or Architect.

# **ARTICLE 3 - CONTRACTOR**

# ARTICLE 3.1

## Add the following to paragraph 3.1:

- 3.1.2 Duties of the Contractor shall include those set forth in Section 4-343, Group 1, Chapter 4, Part 1, Title 24, California Code of Regulations.
  - .1 The Contractor will make and submit Verified Reports in accordance with Sections 4-336, Group 1, Chapter 4, Part 1, Title 24, California Code of Regulations.

#### 3.1.3 Status of Contractor:

- .1 Contractor is and shall at all times be deemed to be an independent contractor and shall be wholly responsible for the manner in which it performs the services required of it by the terms of the Contract Documents.
- .2 Nothing contained in the Contract Documents shall be construed as creating the relationship of employer and employee, or principal and agent, between Owner and Contractor or any of Contractor's agents or employees.
- .3 Contractor exclusively assumes the responsibility for the acts of its employees as they relate to the services to be provided during the course and scope of their employment. Contractor, its agents, and employees shall not be entitled to any rights or privileges of Owner employees and shall not be considered in any manner to be Owner employees.
- .4 Owner shall be permitted to monitor the activities of Contractor to determine compliance with the terms of the Contract Documents.
- .5 Contractors are required by law to be licensed and regulated by the Contractors' State License Board. Any contractor not so licensed is subject to penalties under the law and the Construction Agreement will be considered void pursuant to Business and Professions Code Section 7028.7. Any questions concerning a contractor may be referred to the Registrar, Contractors' State License Board, 3132 Bradshaw Road, Post Office Box 2600, Sacramento, California, 95826.

# 3.1.4 Changes in name or nature of Contractor's legal entity

.1 Before Contractor makes any change in the name or legal nature of the Contractor's entity, Contractor shall first notify Owner in writing and cooperate with Owner in making such changes as Owner may request in the Contract Documents.

# 3.1.5 Debarred Contractor

- .1 Pursuant to Labor Code Sections 1777.1 and 1777.7, a contractor may be prohibited from bidding or performing work as a subcontractor on a public works project. Any contract on a public works project entered into between a contractor and a debarred subcontractor is void as a matter of law. A debarred subcontractor may not receive any public money for performing work as a subcontractor on a public works contract, and any public money that may have been paid to a debarred subcontractor by a contractor on the Project shall be returned to the awarding body. The contractor shall be responsible for the payment of wages to workers of a debarred subcontractor who has been allowed to work on the Project.
- .2 Pursuant to Public Contract Code Section 4701, CONTRACTOR shall request the substitution of any subcontractor who has been debarred by the California Labor Commissioner from working as a subcontractor on public work.
- 3.1.6 Contractor shall be that entity awarded this Construction Agreement by official action of OWNER. Throughout the Contract Documents CONTRACTOR is treated as being of singular number and neuter gender.
- 3.1.7 Contractor selection process and prohibited interest:
  - a. As a means of maintaining the integrity of the formal selection process, contacts with

- individual members of OWNER's Board of Trustees or governing body on behalf of any bidding firm relative to this Project will be considered inappropriate.
- b. No official of OWNER who is authorized in such capacity and on behalf of OWNER to negotiate, make, accept, or approve, or to take part in negotiating, making, accepting or approving, any architectural, engineering, inspection, construction, or material supply contract, or any subcontract in connection with construction of the Project, shall have any direct or indirect financial interest in any part of this Project.
- c. No officer, employee, architect, attorney, engineer. or inspector of or for OWNER who is authorized in such capacity and on behalf of OWNER to exercise any executive, supervisory, or other similar functions in connection with construction of the Project shall have any direct or indirect financial interest in any part of this Project.
- d. CONTRACTOR shall receive no compensation and shall repay OWNER for any compensation received should CONTRACTOR aid, abet, or knowingly participate in any violation of this Article.

# ARTICLE 3.2.1

# Supplement as follows:

3.2.1.1 If Contractor or its subcontractors, material, or equipment suppliers, or any of their officers, agents, and employees performs, permits, or causes the performance of any work to be done under the Contract Documents which it knows, or should have known, to be in error, inconsistent, or ambiguous, or not sufficiently detailed or explained, Contractor shall bear any and all resulting costs, including without limitation the costs of correction without increase or adjustment to the contract price or the time for performance.

# **ARTICLE 3.3**

# Supplement as follows:

- 3.3.5 In accordance with generally accepted construction practices, the Contractor will be solely and completely responsible for conditions of the job site, including safety of all persons and property during performance of the work. This requirement will apply continuously and not be limited to normal working hours.
- 3.3.6 Visits to the site by the Architect, his agents and employees to review the Contractor's performance is not intended to include review of the adequacy of the Contractor's safety measures, in, on or near the construction site.
- 3.3.7 The Contractor shall properly locate all building and other improvements off site and on site, and shall furnish all engineering required for such work, including the locations and identification of bench marks, markers, property lines, lines and grades. He shall check floor elevations and grades against data shown on the Drawings, and shall report any discrepancies to the Architect and Engineer of Record before laying out the work.
  - All work shall be constructed according to the vertical and horizontal measurements of the construction documents, campus standards for on site improvements and standards of local entities for offsite improvements. Adequate stakes shall be set under the direction of a licensed surveyor/engineer for the Contractor. All construction surveying shall be performed in accordance to the surveying procedures of the surveying/engineer of record. If third party subcontractors perform the construction layout and surveying for this project without the coordination of the surveyor/engineer of record, they have assumed the position of the surveyor/engineer of record. The final check of design is to be the construction staking as the design meets the existing control and elevations. Layout surveyor shall provide in his bid time to review existing design and confirm compliance with requirements herein.
  - All existing street cross sections shall be verified for minimum cross fall at time of staking. Any discrepancies to the standard design criteria shall be brought to the architect and surveyor/engineer of record attention prior to proceeding with construction and layout.
  - If construction continues without verification, the Contractor assumes the responsibility of the correctness of the intended design.

- 3.3.8 Where work of one trade joins, is on, adjacent to or covers other work, there shall be no discrepancy when said work is completed. In engaging one kind of work with another, marring or damaging same will not be permitted. Should improper work of any trade be covered by another that results in damage or defects, the whole work affected shall be made good by responsible Contractor without expense to the Owner. Sequencing of work and Contractor's ways and means shall be coordinated and allowances made for coordination between trades as well as ways and means of all trades on project.
- 3.3.9 The Contractor shall consult the other Contractors on the project, if any, and the Architect, regarding the installation of such other Contractor's work and ways and means before starting the various phases of his work, in order to avoid the possibility of the removal of his work to permit others to install their work
- 3.3.10 Assistance required by the Architect in obtaining measurements and/or information on the work shall be furnished fully and efficiently by the Contractor at no cost to the Owner. This includes but not limited to the following:
  - .1 Submissions of details/supplemental information as requested by Architect to clarify field conditions and similar details to that of Construction Documents.
  - .2 Submissions of supplemental data to confirm compliance with construction documents. Architect may request data to confirm ICC/UL/or other approval agencies approval(s) for any assembly.
  - .3 Field dimensions, field topographic grades, and field conditions as requested by architect.
  - .4 Digital photographs, field sketches and or other means requested by architect for use determining contract document compliance, project conformance and responses to field conditions and questions by contractor, owner and architect. Failure to provide this information within two (2) business days shall be grounds for claim for delays by owner to contractor and contractor will be subject to liquidated damages penalty for every calendar day beyond two days indicated herein that information is not provided completely as requested by architect.
  - .5 Information/data required by architect to be used in making aesthetic decisions.
- 3.3.11 Protection of Workers in Trenches: As required by Section 6705 of the California Labor Code, and in addition thereto, whenever work under the Contract involves the excavation of any trench or trenches, or other excavation 5 feet or more in depth, Contractor shall submit for acceptance by the Owner or by a registered civil or structural engineer, employed by the Owner, to whom authority to accept has been delegated, in advance of excavation, a detailed plan showing the design of shoring, bracing, sloping, or other provisions to be made for worker protection from the hazard of caving ground during the excavation, or such trench or trenches. If such plan varies from the shoring system standards established by the Construction Safety Orders of the Division of Industrial Safety, the plan shall be prepared by a registered civil or structural engineer employed by the Contractor, and shall costs therefore shall be included in the price named in the Contract for completion of the Work as set forth in the Contract Documents. Nothing in this Section shall be deemed to allow the use of a shoring, sloping, or other protective system less effective than that required by the Construction Safety Orders. Nothing in this Section shall be construed to impose tort liability on the Owner, Architect and Consulting Engineers, nor any of their officers, agents, representatives, or employees.
  - .1 The Architect/Consulting Engineers shall not be responsible in any way for the contractor's, sub constructor's, Materialmen and lower tier sub contractor's compliance with the "Occupational Health and Safety Regulations" of the US Department of Labor or with State of California Department of Industrial Relations "Construction Safety Orders".

# ARTICLE 3.4

#### Add the following:

3.4.1 No material, supplies, or equipment for work under the Contract Documents shall be purchased subject to any chattel mortgage or under a conditional sale or other agreement by which an interest in all or any part is retained by the seller or supplier. Contractor warrants good title to all material, supplies, and equipment installed or incorporated in the Work, and upon completion of all work agrees to surrender the premises to Owner, together with all improvements and appurtenances

constructed or placed by Contractor, free from any claims, liens, or charges. Contractor further agrees that neither Contractor nor any person, firm, or corporation furnishing any materials or labor for any work covered by the Contract Documents shall have any right to a lien upon the premises or any improvement or appurtenance, except that Contractor may install metering devices or other equipment of utility companies or political subdivisions, title to which is commonly retained by the utility company or political subdivision. In the event of the installation of any metering device or equipment, Contractor shall advise Owner as to its owner. Nothing contained in this article however shall defeat or impair the legal right of persons furnishing material or labor to look to funds due and owing Contractor for payment. This provision shall be inserted in all subcontracts and material contracts and notice of its provisions shall be given to all persons furnishing material for work when no formal contract is entered into for such material.

# ARTICLE 3.5

# Add the following:

- 3.5.2 The term of warranty(s) shall be one year, (or more where noted in project manual), from the date of owner execution of Notice of Completion document. Contractor and Surety expressly agrees to act as co-warrantor of such equipment and materials, and contractor shall supply the Owner with all warranty documents relative to equipment and materials incorporated in the job and warranted by their suppliers or manufacturers.
- 3.5.3 Contractor shall, within five (5) working days of notice of warranty or guarantee item, review warranty or guarantee items, not limited to investigations, diagnosis and review, at the project site with the required related Contractors and suppliers. There shall be no cost to review items to confirm warranty item disposition. Warranty or guarantee item remedy shall be approved by Owner and Architect prior to execution. Correction of items shall be completed within 30 days of determination of items being warranty/guarantee item, unless approved by Owner in writing. Contractor to coordinate work around existing operations and do work during non-occupied times or as directed by Owner. No cost shall be incurred to Owner due to this work without prior written approval by Owner and Architect.
  - a. In the event Contractor or Surety fails to commence and pursue with diligence any replacements or repairs within seven calendar days after being notified in writing, Owner is authorized to proceed to have any defects repaired at the expense of Contractor and Surety, and Contractor and Surety agree to pay the costs and charges immediately on demand.
  - b. If defective work creates a dangerous condition, in the opinion of Owner, or requires immediate correction or attention to prevent further loss to Owner or to prevent interruption or operations of Owner, Owner shall attempt to give the notice required by this Article. If Contractor or Surety cannot be contacted or neither complies with Owner's request for correction within a reasonable time, as determined by Owner, without regard to the provisions of this Article, Owner may proceed to make the correction or provide the attention, and the costs of correction or attention shall be charged against Contractor. Any action by Owner shall not relieve Contractor of the guarantees provided in this Article or elsewhere in the Contract Documents.
  - c. Contractor shall correct any such defective work, as well as any damage to any other part of the Work resulting from such defective work, and provide repair, replacement, or reimbursement, at its sole expense, in a manner approved by Owner and with due diligence and dispatch as required to make the Work ready for use by Owner, ordinary wear and tear, unusual abuse, or neglect excepted. Such corrections shall include but not be limited to any necessary adjustments, modifications, changes of design (unless of Owner's design), removal, repair, replacement, or reinstallation, and shall include all necessary parts, materials, tools, equipment, transportation charges, and labor as may be necessary, and cost of removal. Replacement shall be performed at a time and in such a manner so as to minimize the disruption to Owner's use of the Work.
  - d. Owner may collect its reasonable costs and attorneys' fees in any action to enforce this Article, including additional services required by the Architect.
- 3.5.4 Before and prior to the anniversary of the guarantee-warranty period, there will be a meeting and review of the subject work by the Owner, Architect, and Contractor, or their designated representatives. It shall be the responsibility of the Contractor to arrange for this meeting; the

Contractor's failure to do so will have the effect of extending the guarantee-warranty period until such time as this meeting is held. The guarantee-warranty period will not be complete until such items as noted in the Warranty List generated by this meeting are acted upon to the satisfaction of the Architect.

- 3.5.5 All copies of guarantees and warranties shall be certified by Contractors, Subcontractors and material suppliers.
- 3.5.6 The Contractor, Subcontractor, Materialmen, Supplier, and Manufacturer and/or Fabricator, shall notify the Architect, Owner and Prime Contractor of any requirement(s) for a warranty and/or guarantee to be effective. All contingent conditions of warranty and guarantee shall be met prior to release of retention funds and final acceptance of project as complete. In the event annual and/or follow-up periodic inspections of items under warranty and/or guarantee are required to maintain the warranty and/or guarantee, the provider of said items shall allow for these visits and follow-up reports to Architect, Owner and Prime Contractor in submitting a bid for this project. In the event said required visits are not made by provider of items, the requirements for visit or periodic visit shall be voided.
  - 1. All contingent items on warranty and/or guarantee shall be clearly written as an attachment to submittals and highlighted with yellow marker and yellow tab. Failure to do so voids contingent items on warranty and/or guarantee.
  - 2. All items required by prime Contractor, Owner, Subcontractors and/or Others to be done as a condition of warranty shall be clearly written as an attachment to submittals and highlighted with orange marker and orange tab. Failure to do so voids contingent items on warranty and/or guarantee.
  - 3. The official start date of all warranties and guarantees shall be the date of recording of the Notice of Completion.

## ARTICLE 3.6

# Add the following:

Convict Made Materials: Except as may be provided by law, the Contractor agrees that no materials manufactured or produced in a penal or correctional institution shall be incorporated in the construction under this Contract.

#### ARTICLE 3.8.1

#### Change to read as follows:

The Contractor shall include in the Contract Sum all allowances stated in the Contract Documents. Items covered by allowances shall be supplied for such amounts and by such persons or entities as the Owner may direct. The Owner reserves the right to change the use of allowances at any time during the project. The Owner may select any vendors, contractor, subcontractor or material men for implementation of any allowance items at owner's discretion. Contractor shall make payments from allowance funds as directed by owner. The Architect shall issue a SI, (Supplemental Instruction), for each allowance usage to be used and what specifically it will be used for project or other item as so directed.

# ARTICLE 3.8.2

# Change to read as follows:

Unless otherwise provided in the Contract Documents:

- .1 Materials, labor, equipment, vendors and other contractors under an allowance shall be selected by the Owner. Contractor shall give Owner forty-five (45) calendar day notice prior to when said equipment and material will be required for Work.
- .2 Contractor's costs for unloading and handling at the site including miscellaneous labor and other expenses shall be included in the contract sum and not in the allowances unless approved in advance by Owner and Architect.
- .3 Whenever costs are more than or less than the allowances, the Contract Sum shall be adjusted accordingly by Constructive Cost Directive.
- .4 Contractor shall include in base bid all overhead, profit (all tiers) bonds, insurance and all other non direct cost related items in addition to allowances prescribed herein.

#### ARTICLE 3.10.2

## Add the following:

3.10.2.1 Architect shall be allowed twenty-one (21) calendar days for initial review of submittals and fourteen (14) calendar days each time a submittal is re-submitted.

#### ARTICLE 3.10.4

# Add the following:

- 3.10.4 Contractor shall show anticipated rain, sleet, snow, wind and other weather driven delays on project schedule, as separate line items for each weather delay type, based upon Contractor's proposed ways and means. Line items on Contractor's initial submitted schedule will be used as a basis for additional weather days (time) for aforementioned weather delays only when days indicated on initial schedule for weather delays have been exceeded and weather conditions exceed Contractor's ways and means basis. Conditions shall be based upon weather conditions meeting and exceeding requirements indicated in Article 8.3.2 herein this section.
- 1) Contractor shall indicate on submitted Initial Project Schedule the criteria of each type of weather related delay per 24 hour period that affects the contractor's ways and means. Failure to indicate information on initial project schedule shall mean contractor has reviewed NOAA weather data, highest within last 5 years, for subject project site and based upon that data, weather conditions will have no impact on contractor's ways and means during the construction duration as indicted in herein.
  - a) Contractor shall indicate on Project Schedule the criteria of rain/sleet delay, inches per 24 hour period that affects the contractor's ways and means. Contractor shall use days exceeding contractors ways and means criteria from NOAA data as a basis of days in initial schedule. (Example "Rain delays days 1" per 24 hour period").
  - b) Contractor shall indicate on Project Schedule the criteria of snow delay, inches per 24 hour period that affects the contractor's ways and means. Contractor shall use days exceeding contractors ways and means criteria from NOAA data as a basis of days in initial schedule. (Example "Snow delays days 5" per 24 hour period").
  - c) Contractor shall indicate on Project Schedule the criteria of wind delay, wind speed miles per hour sustained for 2 hours that affects the contractor's ways and means. Contractor shall use days exceeding contractors ways and means criteria from NOAA data as a basis of days in initial schedule. (Example "Wind delays days – 40 MPH for 2 hours per 24 hour period").
  - d) Contractor shall indicate on Project Schedule the criteria of any other anticipated weather delay not indicated herein.
- 2) Contractor shall use weather data as published by NOAA that exceed the actual number of rain/sleet/snow/wind days actually recorded, during the project duration, using highest value within last 5 years at the project site as basis on projected weather related delays that exceed the contractors ways and means criteria indicated on the initial schedule. Using this data along with Contractors Ways and Means criteria will be the basis of consideration for additional weather days beyond those on contractor's initial schedule. This will only be used for additional time when Contractor can show weather delay affected critical path on contractors published initial schedule that exceed weather days indicated on said schedule.
  - a) In the event Contractor uses weather data inconsistent, in initial project schedule or subsequent claims for time extensions, with the aforementioned NOAA data, will be a basis of rejection of claim for additional time due to weather conditions.
  - b) In the event the Contractor fails to list the ways and means criteria on initial project schedule shall be a basis of rejections of claim for additional time due to weather conditions.

# ARTICLE 3.11.1 Add the following:

- .1 Said documents and samples shall be available to the Project Inspector, Architect, Owner and representatives of all agencies having jurisdiction over the work.
- .2 Record Drawings: Contractor shall keep accurate dimensioned locations, on said record copy of

documents, showing actual locations and elevations of all buried, above ground and concealed Work including piping, conduit, valves, stub outs and the like. Elevations shall be referenced to first floor finished elevations as datum. Locating dimensions shall be referenced to permanently fixed accessible and readily identifiable portions of building or site appurtenances. Locations shall be by intersecting coordinate dimensions parallel to and at right angles to building lines. The Contractor must update the Record Drawings as work progresses. At the end of each month, the Architect may review the Record Drawings. If the records are incomplete, or incorrect, an appropriate amount of dollar, equivalent to the cost of uncovering the work to determine the locations of piping and the like, may be deducted from the next progress payment. The deducted sum will be withheld until the Record Drawings are updated and/or corrected to the approval of the Architect and confirmed by Project Inspector.

CONTRACTOR shall keep one copy of all Contract Documents, including addenda, change orders, shop drawings, and other modifications, and Titles 19, 21, and 24 of the California Code of Regulations, on the job at all times. The documents shall be kept in good order and accurately marked to record all changes made during construction. The documents shall be available to the Project Inspector, Owner, Architect and its representatives at all times. CONTRACTOR shall be acquainted with and comply with all statutes and regulations as they relate to this Project. (See particularly the duties of Contractor, 24 California Code of Regulations, Sections 4-343.) CONTRACTOR shall also be acquainted with and comply with all provisions of the California Code of Regulations relating to conditions on this Project, particularly Titles 8 and 17.

# **ARTICLE 3.12.1**

## Amended as follows:

Add the following: "Wherein the Contract Documents reference is made to shop drawings it shall include, as defined below, product data as may be applicable."

## ARTICLE 3.12.4

# Add the following:

3.12.4.1 Contractor shall provide shop drawings of all items that are fabricated and purchased for this project.

# **ARTICLE 3.12.12**

# Add as follows:

Contractor shall submit an itemized list of changes of items different than specified/indicated herein and on construction documents. List shall include items that are different and omitted. In the event items are not included on list, omitted and/or different; Contractor shall be responsible for providing specified item(s). Liabilities subsequent to items omitted/or different shall be responsibility of Contractor and shall be warranted a minimum of five (5) years or greater as prescribed by law. If no list in included with Submittals and Shop Drawings, Architect assumes all items are as specified.

If it is found within five (5) years of Notice of Completion or greater as prescribed by law, that the Contractor has varied from the drawings and/or Specifications in materials, quality, omission, form, or finish, or in the amount or value of the materials and labor used, the Architect shall make a recommendation either: (1) that all such improper work should be removed, remade, and replaced, and all work disturbed by these changes be made good at Contractor's sole expense; or (2) that Owner deduct from any amount due Contractor the sum of money equivalent to the difference in value between the work performed and that called for by the drawings and Specifications if final payment has not been made to contractor. The Architect shall determine such difference in value. At its option, Owner may pursue either recommendation made by the Architect.

# **ARTICLE 3.12.13**

# Add as follows:

Some drawings, shop drawings, product date, samples or other documents may be required of Contractor. If Contractor performs, permits, or causes the performance of any work under the documents provide and/or prepared by or on the behalf of Contractor which document is in error, inconsistent or ambiguous, or not sufficiently detailed or explained, Contractor shall bear any and all resulting costs, including, without limitation, the cost of correction, without increase to or adjustment in the contract price or the time for performance. In no case shall any subcontractor proceed with the work if uncertain without Contractor's written direction

and/or approval.

## **ARTICLE 3.13.1**

# Add as follows:

- 1. Such areas when outside the limits of the contract and project area shall upon completion of the contract be restored to their original condition.
- 2. No pets, not limited to dogs and cats, will be permitted to be on site during construction. Contractor shall post a sign at each project entry indicating requirement (exception shall be service animals).

#### **ARTICLE 3.15.1.1**

# Supplement as follows:

- Each Contractor shall:
  - a. Proceed with due caution to protect the work of others.
  - b. Remove and legally dispose of all excess material, debris, etc., resulting from his work.
  - c. Clean all finger marks, smudges, smears, spatters, drippings, etc., from his work and the work of others, caused by workmen in his employ.
  - d. The subcontractors and all lower tier contractors and vendor, upon completion of their work and prior to leaving the project, shall meet with the Project Inspector, Architects field representative and the General Contractor, to review the work performed and make such corrections as may be found necessary for compliance with contract documents.
  - e. Supplementary cleaning shall be performed as may be required due to work performed pursuant to items as listed on Final Inspection (Punch List).
  - f. Cleaning of various items shall be done in accordance with recommended procedures for respective material, and damage by improper cleaning shall be corrected by the Contractor at his expense. Refer to manufactures written recommendations for cleaning of items manufactured.

## ARTICLE 3.18.

## Replace with the following:

#### 3.18.1

A. CONTRACTOR shall defend, indemnify and hold harmless DISTRICT, Architect, Project Inspector, the State of California and their officers, employees, agents and independent contractors from all liabilities, claims, actions, liens, judgments, demands, damages, losses, costs or expenses of any kind arising from death, personal injury, property damage or other cause based or asserted upon any act, omission, or breach connected with or arising from the progress of Work or performance of service under this Agreement or the Contract Documents. As part of this indemnity, CONTRACTOR shall protect and defend, at its own expense, DISTRICT, Architect, Project Inspector, the State of California and their officers, employees, agents and independent contractors from any legal action including attorneys' fees or other proceeding based upon such act, omission, or breach.

Furthermore, CONTRACTOR agrees to and does hereby defend, indemnify and hold harmless DISTRICT, Architect, Project Inspector, the State of California and their officers, employees, agents and independent contractors from every claim or demand made, and every liability, loss, damage, expense or attorneys' fees of any nature whatsoever, which may be incurred by reason of:

- Liability for (1) death or bodily injury to persons; (2) damage or injury to, loss (including theft), or loss of use of, any property; (3) any failure or alleged failure to comply with any provision of law or the Contract Documents; or (4) any other loss, damage or expense, sustained by any person, firm or corporation or in connection with the Work called for in this Agreement or the Contract Documents, except for liability resulting from the sole or active negligence, or the willful misconduct of the DISTRICT.
- Any bodily injury to or death of persons or damage to property caused by any act, omission or breach of CONTRACTOR or any person, firm or corporation employed

by CONTRACTOR, either directly or by independent contract, including all damages or injury to, loss (including theft), or loss of use of, any property, sustained by any person, firm or corporation, including the DISTRICT, arising out of or in any way connected with Work covered by this Agreement or the Contract Documents, whether said injury or damage occurs either on or off DISTRICT property, but not for any loss, injury, death or damages caused by the sole or active negligence or willful misconduct of the DISTRICT.

- 3. Any dispute between CONTRACTOR and CONTRACTOR's subcontractors/ supplies/sureties, including, but not limited to, any failure or alleged failure of the Contractor (or any person hired or employed directly or indirectly by the Contractor) to pay any Subcontractor or Material man of any tier or any other person employed in connection with the Work and/or filing of any stop notice or mechanic's lien claims.
- 4. Contractor shall require each subcontractor to abide by the terms and conditions indicated herein.

CONTRACTOR, at its own expense, cost, and risk, shall defend any and all claims, actions, suits, or other proceedings that may be brought or instituted against the DISTRICT, its officers, agents, architects or employees, on account of or founded upon any cause, damage, or injury identified herein and shall pay or satisfy any judgment that may be rendered against the DISTRICT, its officers, agents or employees in any action, suit or other proceedings as a result thereof.

- B. In any and all claims against the Owner or the Architect or Architect's Consultants, or any of their employees by any employee of the Contractor, any Subcontractor, anyone directly or indirectly employed by any of them or anyone for whose acts any of them may be liable, the indemnification obligation under this Paragraph shall not be limited in any way by any limitation on the amount or type of damages, compensation or benefits payable by or for the Contractor or by any Subcontractor under workers compensation acts, disability benefit acts or other employee benefit acts.
- C. The obligations of the Contractor under this Paragraph shall not extend to the liability of the Architect, the Architect's consultants or any of their agents or employees arising out of 1) The preparation or approval of maps, drawings, opinions, reports, surveys, change orders, designs or specifications, or, 2) the giving of or the failure to give directions or instructions by the Architect, the Architect's consultants or any of their agents or employees, provided such giving or failure to give is the primary cause of the injury or damage.
- D. See insurance requirements for additional insured scope.
- E. The obligations of this Article expressly include but are not limited to the obligations of indemnification and defense of Owner, the Architect, Project Inspector, and their directors, officials, officers, agents and employees arising in any manner out of any claims against them brought by other contractors, subcontractors, material men, suppliers, or other third parties alleging any of them owe the claimant either time, compensation, or damages due to any act, omission, or occurrence caused or contributed to in any degree by Contractor or any of its subcontractors.
- F. The Contractor shall further indemnify and hold harmless the Owner, Architect and their agents from and against all claims, damages, losses and expenses including, but not limited to, Attorney's fees arising out of the injury to unauthorized persons trespassing upon the premises.
- G. It is the express intent of the parties hereto that the Owner, Architect and Engineers are exculpated from any liability whatsoever occasioned by the Contractor's failure to carry out the Work in accordance with the Contract Documents.

# ARTICLE 3.19

# Add the following paragraph to Article 3:

Right to Audit: The Contractor shall maintain and make available to Owner all books, papers, job cost records, detailed cost estimates, claims, and accounts, including payment, property, payroll, personnel, subcontractors, and financial records related to or which arise out of the Work or under terms of the Contract. The form of record keeping shall be subject to approval by the Owner. These books, papers, records, and accounts shall be made available for examination during normal business hours by Owner and Owner's representative and shall be retained at Contractors principal place of business, in California, for audit during normal business hours at such place of business for four years after recording of the Notice of Completion of the project. Contractor shall provide an office to enable Owner and Owner's representative to conduct such audit during normal business hours.

# **ARTICLE 4 - ADMINISTRATION OF THE CONTRACT**

# ARTICLE 4.2.1

## Add the following:

- a. The Architect shall be OWNER's representative during construction and shall periodically observe the progress and quality of the Work on behalf of OWNER. The Architect shall have the authority to act on behalf of OWNER only to the extent expressly provided in the Contract Documents. The Architect shall have authority to stop work whenever necessary, in the Architect's reasonable opinion, to ensure the proper execution of the Work of the Project.
- b. The Architect shall be, in the first instance, the judge of the performance of the Work. The Architect has the authority under the Contract Documents to enforce CONTRACTOR's faithful performance.
- c. The Architect shall have all authority and responsibility established by law, including Title 24 of the California Code of Regulations. The Architect has the authority to enforce compliance with the Contract Documents and CONTRACTOR shall promptly comply with instructions from the Architect or an authorized representative of the Architect.
- d. On all questions related to quantities, acceptability of material, equipment, or workmanship, execution, progress, or sequence of work, the interpretation of plans, specifications, or drawings, and the acceptable performance of CONTRACTOR, the decision of the Architect shall govern and shall be a condition precedent to any payment, unless otherwise ordered by OWNER. CONTRACTOR shall not impair or delay the progress and completion of the Work by virtue of any question or dispute arising out of or related to the foregoing matters, or the instructions of the Architect relating to them.

## ARTICLE 4.2.7

# Delete the following:

Delete the words "and approve" from the first sentence of subparagraph 4.2.7.

Delete the last sentence from subparagraph 4.2.7 and substitute the following:

The Architect's review of a specific item shall not indicate approval of an assembly of which the item is a component.

## **ARTICLE 4.2.7.1**

## Add the following

The term "approved," where used in conjunction with the Architect's action on the CONTRACTOR'S submittals, applications, and request, is limited to the responsibilities and duties of the Architect stated in General and Supplementary Conditions. Approval shall not release CONTRACTOR from responsibility to fulfill Contract Document requirements, unless otherwise provided in the Contract Documents.

## ARTICLE 4.2.9

Change as follows:

Change "Substantial Completion" to "Completion"

# ARTICLE 4.3; Paragraphs 4.3.3, 4.3.6 and 4.3.9

Change as follows:

Change all citations of 21 days to 10 days.

# ARTICLE 4.3.1

# Add the following:

- 1. Any claim must be served by certified U.S. mail to Owner or bondifed server.
- 2. Any claim/dispute by claimant shall be submitted with references to construction documents that relate to claim/dispute. Provide evidence references that substantiate claim/dispute is valid at time of initial claim submission.
- 3. No claim/dispute shall be considered valid or received unless accompanied with above items.
- 4. Copies of claims shall be delivered to Architect, Consulting Engineers, Owners Legal Representative (Name available on request), related Contractors and Sub Contractors and Project Inspector. This shall be done concurrently with served copy to Owner.

# ARTICLE 4.3.3

# Add the following

- A. If Contractor claims compensation for any damage allegedly sustained by reason of any acts or omissions of Owner or its agents, Contractor shall submit to the Architect a written statement of the damage sustained and potential for damage yet to be sustained. Such written statement shall be submitted, either within ten calendar days after the act or omission by Owner or its agents is discovered or reasonably should have been discovered, or within ten calendar days after first sustaining damage, whichever comes first. On or before the 10th day after the act or omission was discovered, or the first damage was sustained, whichever comes first, Contractor shall file with Owner, Architect and others noted herein, a detailed and itemized statement indicating the factual basis in support of its claim and the amount of damage.
- B. IF CONTRACTOR FAILS TO COMPLY WITH ANY OF THE PROVISIONS OF THIS ARTICLE CONCERNING THE SUBMISSION OF STATEMENTS, ITS CLAIM FOR COMPENSATION SHALL BE FORFEITED AND INVALIDATED AND IT SHALL NOT BE ENTITLED TO CONSIDERATION FOR PAYMENT ON ACCOUNT OF ANY SUCH DAMAGE.
- C. In no event shall Contractor be permitted to reserve rights to make or pursue claims of any kind, whether for compensation in any form, or for time extensions, without the Owner's express written consent. Any attempt to make such reservation or otherwise avoid the effect of this article shall be void and of no force or effect whatsoever.
- D. Any change order executed by Contractor with such reservation or other language of qualified acceptance shall be read and interpreted as though such language did not exist. No action by Owner or Architect is required to invalidate such language, and no oral communication or other act or omission by Owner or anyone acting on Owner's behalf, except Owner's express written consent, shall be construed as acquiescence in or consent to such reservation or other qualified acceptance language.
- E. Contractor shall diligently proceed with performance of the Work, and Owner shall continue to make payment of undisputed amounts, during any time period while claims are pending.

## **ARTICLE 4.3.8.1**

#### Amend to read as follows:

If the Contractor wishes to make claim for an increase in the contract time, written notice as provided herein shall be given. The Contractor's claim shall include effect of delay on progress of the Work. In the case of a continuing delay, only one claim is necessary. Refer herein for requirements for a claim to be valid.

#### ARTICLE 4.3.8.1.1

# Add the following:

No payment or compensation of any kind shall be made to the Contractor for damages because of any hindrance or delay from any cause in the progress of work, whether such hindrance or delay be avoidable or unavoidable. Any finding by any administrative officer and/or judge that a delay was caused either wholly or in part by actions of someone other than the Contractor shall only entitle the Contractor to equivalent extensions of time.

# **ARTICLE 4.3.8.2**

Add the following:

4.3.8.2.1 Refer to requirements herein of initial project schedule submission for criteria of weather conditions (exceeding last 5 years maximum), that were abnormal for period of time that could not have been reasonably anticipated by contractor. Refer to requirements herein of initial project schedule submission of contractor's ways and means criteria for weather conditions for weather delays as well as indicating number of days on initial project schedule submission of anticipated days for rain, snow and wind as well as other weather related project delays.

#### ARTICLE 4.3.8

# Add the following:

## Article 4.3.8.3

The earlier time difference from the date of completion of the project as established by the Contractors Schedule and the later time indicated in the Contract Documents shall be considered float time that can be used by either or both Owner and Contractor. While the Contractor may schedule completion of the Project earlier than the date established by the Contract Documents, no additional compensation shall become due the Contractor for the use of float time between the Contractor's projected early completion date and the completion date established by the Contract Documents. The Owner may use this float time for additional work due to Change Orders, use of Allowance funds and Contractors Change Orders that impact the project schedule.

## ARTICLE 4.3

# Add the following:

#### Article 4.3.10. STOP NOTICES

- The Owner shall withhold funds due to or to become due to Contractor upon receipt of a Stop Notice. The Owner will withhold payment of funds until a Stop Notice Release has been received, except as noted herein. A Stop Notice shall be considered released once claimant has filed a Stop Notice Release with the Owner or based on a determination made through a summary proceedings originated with the Contractor. Contractor may provide a Stop Notice specific bond equal to 125% of claim stated in the Stop Notice. This bond shall be in a form approved by the Owner. Upon approval approving the bond form and receipt of, the Owner will release funds being held by the Stop Notice.
  - a. The Stop Notice and Stop Notice Release must be served by first class mail, registered mail, or certified mail on the contractor and/or its subcontractor, either at his residence or anywhere he maintains an office, or by personal service. It must also be filed with the Owner to such person prescribed in the Instructions to Bidders by any of the same methods aforementioned.
  - b. The Architect shall be copied on all Stop Notices and Stop Notices Releases served to Contractor, Subcontractors and filed with Owner.
  - c. Stop Notices served shall not be considered released until a Stop Notice Release is served to aforementioned parties.
- 2. The Stop Notice must be signed and verified by the claimant or his or her agent, and contain all of the following information:
  - a. The type of labor, services, equipment or materials furnished or agreed to be furnished;
  - b. The name of the person to or for whom the labor, services, equipment or materials were furnished by such claimant;
  - c. The value of the labor, services, equipment or materials already furnished, and the total value of the labor, services, equipment or materials agreed to be furnished;
  - d. The name and address of the claimant;
  - e. A statement of the claimant's demand and the jobsite location.
- 3. The Stop Notice Release must be signed and verified by the claimant or his or her agent, and contain all of the following information:
  - Reference Stop Notice that Stop Notice Release is for specifically. Indicated date of Stop Notice, amount of release and who against. The type of labor, services, equipment or materials furnished or agreed to be furnished by such claimant;

- b. The name of the person to or for whom the labor, services, equipment or materials were furnished:
- c. The value of the labor, services, equipment or materials already furnished, and the total value of the labor, services, equipment or materials agreed to be furnished;
- d. The name and address of the claimant;
- e. A statement of the claimant's demand and the jobsite location.

#### ARTICLES 4.5.1 THROUGH 4.5.7

The 4.5.8 Article shall supersede Articles 4.5.1 through 4.5.7 when a conflict between articles.

## ARTICLE 4.5.8

## Add the following:

This contract is subject to the provisions of Article 1.5 (commencing with Section 20104) of Chapter 1 of Part 3 of Division 2 of the Public Contract Code, and incorporated herein by reference. Those provisions provide a process utilizing informal conferences, non-binding judicially supervised mediation and judicial arbitration to resolve disputes on construction claims of \$375,000 or less. A claim means a separate demand by the Contractor for (a) time extension; (b) payment of money or damages arising from work done by or on behalf of the Contractor pursuant to the Contract and payment of which is not otherwise expressly provided for or the claimant is not otherwise entitled to; or, (c) an amount the payment of which is disputed by the Owner. The claim shall be in writing and include the documents necessary to substantiate the claim. The claim must be submitted prior to the date of final payment within the time limits set forth in Section 20104.2. The Owner may respond to the claim within the time limits set forth in Section 20104.2. If the claim remains in dispute, the parties are obligated to meet and confer for the purposes of settlement of the dispute.

If the dispute is not settled following the meet-and-confer process, the Contractor may file a claim pursuant to provisions of the Government Code. The court shall submit the matter to non-binding mediation. If the matter remains in dispute, the case will be submitted to judicial arbitration as set forth in Section 20104.4.

# ARTICLE 4.5.9

# Add the following:

The Division of the State Architect shall not be subject to arbitration.

# **ARTICLE 5 - SUBCONTRACTORS**

# ARTICLE 5.1

# Add the following:

- 5.1.3 All subcontractors shall be appropriately licensed to perform the work for which employed in conformity with the laws of the State of California.
- 5.1.4 In accordance with California Business and Professions Code Section 7059, if Contractor is designated as a "specialty contractor" (as defined in Public Contract Code Section 7058), all of the work to be performed outside of the Contractor's license specialty, except "incidental" work as that term is used in Section 7059(a), shall be performed by a licensed subcontractor in compliance with the Subletting and Subcontracting air Practices Act, California Public Contract Code Section 4100, et seq.

#### ARTICLE 5.2

#### Amend to read as follows:

Contractor shall comply with the subletting and Subcontracting Fair Practices Act (Government Code S4100 et seq.) including but not limited to submitting with his bid a list of those subcontractors who will perform a portion of the work in an amount in excess of one-half of one percent of the total bid. Said list shall include the name and address of each such subcontractor together with the portion of the work which will be done by that subcontractor.

- a) Listing of Subcontractors shall include subcontractors who exceed the above amounts.
- b) Owner's consent to or approval of any subcontractor shall not in any way relieve Contractor

- of its obligations under the Contract Documents, and no such consent or approval shall be deemed to waive any provision of the Contract Documents.
- c) Failure to list a Subcontractor shall be considered a violation of P.C.C. 4110 and the full 10% penalty will be accessed the prime Contractor due to the using for construction of the Sub or Sub-sub not listed, exceeding amounts above. Per P.C.C. 4110, the prime Contractor shall be entitled to a public hearing and to five (5) days' notice of the time and place thereof.
- d) If Contractor specifies more than one subcontractor for the same portion of work or fails to specify a subcontractor, and such portion of the work exceeds one-half of one percent of the total bid, Contractor agrees that it is fully qualified to perform and shall perform such work itself. The substitution or addition of subcontractors shall be permitted only as authorized by Public Contract Code Sections 4100, et seq.
- e) Substitution of subcontractors.
  - The Contractor, in submitting a request for substitution of subcontractors, acknowledges the following:
  - The Contractor's assumption of that neither the contract time nor the contract price or subject to adjustment solely on account of an approved subcontractor substitution.
  - 2) The Contractor's responsibility for meeting all of its obligations under the Contract Documents is not limited by virtue of the Owner's consent to a subcontractor's substitution.
  - 3) All additional fees or costs associated with re-submittal, review and approval of submittals by the new subcontractor are to be borne by the Contractor.

# ARTICLE 5.3 Amend to read as follows:

- .1 Subcontractors shall perform work under a written agreement with the Contractor which incorporates the Contract Documents. The terms shall obligate the subcontractor to assume towards the Contractor the duties assumed by the Contractor towards the Owner under his contract.
  - a. Contractor agrees to bind each and every subcontractor to the terms of the Contract Documents as far as the terms are applicable to the subcontractor's work. Each subcontract shall contain a reference to Contract Documents, and the terms of the Contract Documents shall be incorporated into and made a part of each subcontract. If Contractor subcontracts any part of its work under the Construction Agreement, Contractor shall be responsible to Owner for any acts and omissions of its subcontractors and of persons either directly or indirectly employed by its subcontractors. Nothing contained in the Contract Documents shall create any contractual relationship between any subcontractor and Owner.
  - b. Contractor shall require subcontractors to include the provisions of this article in their subsubcontracts, if any.
- .2 A copy of each subcontract, if in writing, or if not in writing, then a written statement signed by the Contractor giving the name of the subcontractor and the terms and conditions of such subcontract, shall be filed with OWNER before the subcontractor begins work. Each subcontract will provide for termination in accordance with these General and Supplementary Conditions. Each subcontract shall provide for its annulment by CONTRACTOR at the order of the Architect if in the Architect's opinion the subcontractor fails to comply with the requirements of the Contract Documents insofar as the same may be applicable to this work.
  - a. The Owner shall be afforded the contractual right to review all subcontracts for the project. The Contractor shall provide one (1) copy, at no charge, when requested by the Owner.
- .3 Each subcontract applicable to this Project is hereby assigned to Owner, such assignment to become effective only upon termination of the Construction agreement for cause pursuant to the Contract Documents, and only as to such subcontracts as Owner may, in its sole discretion, select and provide written notice of such assignment, and such assignments are subject to the rights and

- obligations of the surety on any applicable bonds, as detailed in the Contract Documents.
- .4 Each subcontract will provide for termination in accordance with these General Conditions. Each subcontract shall provide for its annulment by Contractor at the order of the Architect if in the Architect's opinion the subcontractor fails to comply with the requirements of the Contract Documents insofar as the same may be applicable to this work.
- .5 CONTRACTOR shall require subcontractors to include the provisions of this article in their subsubcontracts, if any.

## ARTICLE 5.4

Delete in its entirety.

# **ARTICLE 6 - CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS**

# ARTICLE 6.2.5

#### Delete in its entirety and substitute the following:

Should the Contractor or Subcontractors wrongfully cause damage to the work or property of any separate Contractor or Subcontractor, the Contractor shall, upon due notice, promptly attempt to settle with such other Contractor by agreement, or otherwise to resolve the dispute. If such separate Contractor or Subcontractor sues the Owner and/or Architect on account of any damage alleged to have been caused by the Contractor, the Owner shall notify the Contractor, who shall defend such proceedings at the Contractor's expense, and if any judgement or award against the Owner and/or Architect arises therefrom, the Contractor shall pay or satisfy it and shall reimburse the Owner and/or Architect for all attorney's fees and court costs which the Owner and/or Architect has incurred. The Owner and/or Architect may deduct such monies due direct from the Contractor's due monthly payment. The Architect, Architect's staff and consulting engineers shall be reimbursed at current rate sheet rates on file for all time required in the resolution of the above by the Contractor.

# **ARTICLE 7 - CHANGES IN THE WORK**

#### ARTICLE 7.1.3

# Delete subparagraph 7.1.3 and substitute the following:

Changes in the work shall be performed under applicable provisions of the Contract Documents, and the Contractor shall proceed promptly, unless otherwise provided in the Change Order or Construction Change Directive, (CCD).

# ARTICLE 7.2.1

# Delete subparagraph 7.2.1 and substitute the following:

A Change Order is a written instrument prepared by the Architect, signed by the Owner, Contractor and Architect and approved by the Division of the State Architect in accordance with Section 4-338, Group 1, Chapter 4, Part 1, Title 24, California Code of Regulations, stating their agreement upon all of the following:

- .1 A change in the scope of work;
- .2 the amount of the adjustment in the Contract Sum, if any; and
- .3 the extent of the adjustment in the Contract Time, if any.

# ARTICLE 7.2.3

## Add the following:

If additions, deletions or revisions to the work are necessary, and the execution of a written Change Order would delay the project, the Architect may, without invalidating the Contract, issue an Architect's Construction Cost Directive (CCD) directing such work prior to issuance of a Change Order. A preliminary approval from DSA may be obtained, if deemed necessary by the Architect, and a written Change Order will follow. The Contractor shall comply with such instructions without delay.

#### ARTICLE 7.3

Delete this article in its entirety and substitute the following:

- 7.3.1 The Owner, at any time during the progress of the work, shall have the right to order alterations in, additions to, or deviations or omissions from the work contemplated by this Contract, and the same shall in no way make void the Contract. If any such changes involve an increase or decrease in the Contract amount, the Change Order shall state the amount to be added to or deducted from the Contract amount, and shall also state the additional time, if any, needed for the performance of the work and shall comply with 7.2.1.
- 7.3.2 Upon receipt of an Architect's CCD, the Contractor shall promptly proceed with the change in the work involved and advise the Architect of the Contractor's agreement or disagreement with the method, if any, provided in the directive for determining the proposed adjustment in the Contract Sum or Contract Time.
- 7.3.3 An Architect's CCD signed by the Contractor indicates the agreement of the Contractor therewith, including adjustment in Contract Sum and Contract Time or the method for determining them. Such agreement shall be effective immediately and shall be recorded as an item on the next Change Order.
  - 1. The Architect may issue a CCD based upon a proposal from the Contractor including any adjustment for time and cost. The proposal from the Contractor shall indicate the Contractor's agreement with the CCD when they are the same as the Contractor's proposal.
- 7.3.4 If the Contractor does not respond within ten (10) calendar days per Articles 4.3.3 and 4.3.5, or disagrees with the method of adjustment in the Contract Sum and contract time, the method and the adjustment shall be determined by the Architect on the basis of reasonable expenditures and savings of those performing the work attributed to the change, including, in case of an increase in the Contract Sum, a reasonable allowance for overhead and profit. In such case, the Contractor shall keep and present, in such form as the Architect may prescribe, an itemized accounting together with appropriate supporting data.
- 7.3.5 If the Owner and Contractor do not agree with the adjustment in Contract Time or the method for determining it, the adjustment or the method shall be referred to the Architect for determination.
- 7.3.6 The cost or credit to the Owner resulting from a Change in the Work shall be determined in one or more of the following ways:
  - 1. By mutual acceptance of a lump sum amount, properly itemized and substantiated.
  - 2. By unit prices stated in the Contract Documents or subsequently agreed upon.
  - 3. By Alternate bids in Contractor's original proposal authorized prior to expiration date indicated herein.
  - 4. By actual cost and prescribed herein percentage overhead and profit mark up.
  - 5. If none of the preceding methods is agreed upon, then the total cost of such work shall be based on the Contractor's expenditure or savings and a percentage for overhead and profit.
  - 6. Limit labor charges of no greater than the prevailing rate for the classification of labor necessary to complete the change limited to the actual field cost.
  - 7. The labor cost incurred by the Contractor to prepare the estimate of the change, supervision of the performance of the changed work and the like shall be excluded from the labor costs of the change and included in the overhead and profit mark up. (See 01035 for allowed not-to-exceed amounts.)
  - 8. It is expressly understood that the value of such extra work or changes as determined by the any of the methods herein expressly includes any and all of the contractors' costs and expense, both direct and indirect, resulting from delays or additional time required on the project, or resulting from accelerated work to avoid delays to the project

| # | Description  | Extra | Credit |
|---|--|-------|--------|
| а | Materials (attached itemized quantity and unit costs including any sales tax.  |       |        |
| b | Labor (attached itemized hours and rates)  |       |        |
| С | \$ Subtotal  |       |        |
| d | Subcontractor's overhead and profit on the cost of work performed by sub-contractors (one lower tier) shall be a total sum not exceeding five percent (5%) of such work. |       |        |

| е           | \$ Subtotal  | <br> |
|-------------|--|------|
| f<br>g<br>h | Subcontractor's overhead and profit on the cost of work performed by subcontractor shall be a total sum not exceeding ten percent (10%) of the cost of labor, materials, rentals, etc.  \$ Subtotal Contractor's overhead and profit on the cost of work | <br> |
|             | excluding work by Contractor shall be a total sum not exceeding ten percent (10%) of cost of such work.  (Refer to project manual "Allowances" where overhead  |      |
|             | and profit are included in base bid and alternates where occur)  | <br> |
| İ           | Contractor's overhead and profit on the cost of work performed by contractor without subcontractor shall be a total sum not exceeding ten percent (10%) of the cost of labor, materials, rentals, etc  |      |
|             | (Refer to project manual "Allowances" where overhead and profit are included in base bid and alternates where occur)   |      |
| j           | Bond Premium (Submit invoice from Bonding provider) . (Refer to project manual "Allowances" where bond premium are included in base bid and alternates where occur)  |      |
| k           | Insurance (Submit invoice from Insurance provider)   |      |
| 1           | (Refer to project manual "Allowances" where insurance are included in base bid and alternates where occur)  \$ Total   | <br> |
| •           | ψ Fotal  | <br> |
| m           | Number of additional days time extension requested due to this time and material change. (Submit as built critical path schedule validating time extension for review and approval)  |      |
|             | Total Days   |      |
|             | Subcontractor's labor, material, overhead and profit shall<br>be submitted with documentation in original form as<br>submitted to General Contractor.  |      |
|             | (Refer to project manual "Allowances" where overhead and profit are included in base bid and alternates where occur).  |      |

- 7.3.7 Should methods 1, 4 or 5 as stated above be employed, the Contractor shall provide a detailed breakdown including a line-by-line itemization for labor, materials, equipment, taxes, Subcontractor overhead and profit and General Contractor overhead and profit. Subcontractor overhead and profit shall be limited to (refer to 012600 Modification Procedures). General Contractor overhead and profit shall be limited to (refer to 012600 Modification Procedures). This amount shall include supervision, field office expense and home office expenses. Credits, as well as extras, shall be itemized and include overhead costs.
- 7.3.8 All Contractor costs for field supervision, field overhead and home office overhead shall be considered a part of lines "h" and "i" above and shall not in any case exceed percent indicated in 01035. No additional claims beyond the agreed Change Order amount shall be considered valid.

7.3.9 No claims for overhead, burden, or any additional cost shall be considered for time extensions due to weather in any case up to 10% increase in construction allowed days due to weather as allowed herein to be included in contractor proposal.

#### 7.3.10 Allowances:

- 1. The Contractor shall include in his base bid all overhead, profit, supervision, bonds, insurance and all other indirect costs for allowance items. None of these items will be added to lump sum and miscellaneous allowance as it is used by the Owner and directed by the architect. In the event the allowance is required in an Alternate, the contractor shall include in his alternate bid all overhead, profit, supervision, bonds, insurance and all other indirect costs for allowance items specific to that alternate.
- Time and Material basis as changes in the work. For changes that increase the contract price and work is authorized based upon the cost of labor, material, equipment and subcontract prices, plus a percentage for overhead and profit the following requirements shall apply. In the event the costs for changes in the work are not agreed to by the Architect and Contractor the work may be authorized to move forward on a time and material basis. If a Time and Material basis is used and scope of work is being paid thru an allowance overhead and profit are to be included in base bid and alternates where occur.
- 7.3.11 Claims for delays while awaiting instruction or clarification shall not be considered.
- 7.3.12 Costs which shall not be paid in Change Orders under the Contract Documents include but are not limited to interest costs of any type, claim preparation or filing costs, costs in preparing or reviewing proposed change orders or proposals, RFP's, SI's, CCD's etc., lost revenue, lost profit, lost income or earnings, rescheduling costs, costs of idled equipment, lost earnings or interest on unpaid retainage, claims consulting costs, costs of corporate officers or staff visiting the site, fluctuation of foreign currency conversion or exchange rate costs, or loss of other business.
- 7.3.13 If Contractor should claim that any instruction, request, drawing, specification, action, condition, omission, default, or other situation constitutes a change, extra work, or otherwise obligates Owner to pay additional compensation to Contractor or to grant an extension of time, or constitutes a waiver of any provision in the Contract Documents, Contractor shall notify Owner in writing of such claim within five calendar days from the date Contractor has actual or constructive notice of the factual basis supporting the claim. The notice shall state the factual basis for the claim and cite in detail the Contract Documents (including plans and specifications) upon which the claim is based. Contractor's failure to notify Owner and Architect within the five-day period shall be deemed a waiver and relinquishment of such a claim. If the notice is given within the specified time, the procedure for its consideration shall be as stated in these General Conditions. In the event of failure to agree, the matter shall be treated as a claim following the claims procedures in the Contract Documents.
- 7.3.14 Change Orders shall be processed and forwarded to the Contractor promptly for execution. The Contractor shall return the Change Order within ten (10) days fully executed, or shall indicate within seven (7) days any disputes. The Contractor's failure to respond to the Change Order within the prescribed time period shall be deemed a waiver of the Change Order and the Contractor shall relinquish the claim against the Owner for additional monies and/or time.
- 7.3.15 In furtherance of the intent to settle all change orders fully and finally at the issuance date of the change order, the following shall be expressly incorporated in writing and deemed incorporated in all change orders by reference:

The compensation (time and costs) set forth in this change order comprises the total compensation due Contractor for the change(s) defined in the change order, including impact on unchanged work. Acceptance of this change order constitutes a full and complete accord and satisfaction of any and all claims by the Contractor arising out of the or related to the change order, including but not limited to claims for contract balance and retention, time, extended filed or home office, or other overhead, all acceleration, impact, disruption and delay damages, any and all other direct and/or indirect costs, claims by subcontractors and suppliers, and any and all other claims against the owner for time or money, from any source and under any legal theory whatsoever, as to the subject of this change order. No signature under protest or accompanied by reservation or rights or protest language, or any other attempts to avoid such waiver shall be on any force or effect whatsoever. No additions or deletions to this change order shall be allowed, except with written permission of the Owner.

# **ARTICLE 7.4.1.1**

# Add the following:

Supplemental Instructions: The Architect shall issue SI's, (Supplemental Instructions), as the document to provide a written order to the Contractor for items as indicated above. Unless otherwise noted above, this SI does not warrant a cost or time impact to the Contract cost or time. If Contractor does not agree, Contractor has 10 calendar days from date of receipt of SI to file a claim for adjustment in writing to the Architect.

 The Architect shall use SI's for written order for usage of allowance funds for project if any allowances are indicated.

## ARTICLE 7.4.2

## Add the following:

The Contractor shall reimburse Owner for Architect's/Staff plus expenses and consulting engineer(s) time at prevailing office rates for time required to respond to request for information from Contractor of items that are indicated substantially in construction documents, and/or any related construction documents associated with project. Architect shall invoice Owner for costs incurred. Contractor shall reimburse Owner for additional costs incurred by Architect and Consulting Engineers via a negative CCD

- 1. In the event the Architect determines that Contractor's requests for clarification or interpretation are not justified, or do not reflect adequate, competent supervision or knowledge by Contractor, material men, or by the subcontractors, Contractor shall be required to pay the Architect's reasonable and customary fees in processing and responding to such requests.
- 2. Architect shall submit a CCD to Owner and Contractor indicating additional costs incurred by Architect and Consulting Engineers due to actions by the Contractor. Upon Owner approval of CCD, costs will be added to project as CCD to be included in a Change Order.
- Contractor has 10 calendar days from receipt of CCD specific to these items to dispute additional
  costs incurred by Architect and Consulting Engineers. Disputed items will not be considered after 10
  days beyond date of receipt of CCD by Contractor for charges by the Architect and Consulting
  Engineers.

# **ARTICLE 8 - TIME**

# ARTICLE 8.1.1

## Delete paragraph in its entirety and add the following:

The contract time is the period of time, including authorized adjustments, allotted in the Contract Documents from the date of commencement to the Final Completion of the Contract. The Owner shall endeavor to allow the Contractor to commence on site on the date stated in the Notice to Proceed. However, the Owner and/or Architect may delay this start date by up to sixty (60) calendar days at their discretion and the Contractor and his subcontractors must allow in their bid for all financial and scheduling implications of this. This clause shall not affect the Contract "duration". This clause shall not be affected by limit on bid withdrawal dates.

## ARTICLE 8.1.3

# Change the following:

Remove the word "Substantial"

## ARTICLE 8.2.3

# Change to read as follows:

The Contractor shall proceed expeditiously with adequate forces and shall achieve Project Completion within the Contract Time. The Contractor must be use constantly his best endeavors to prevent delays in the progress of the Work and maintain the original days of construction duration unless extended by change order.

# ARTICLE 8.3.1

# Delete in its entirety and substitute the following:

If the Contractor is delayed at any time in progress of the work by an act or neglect of the Owner, Owner's

representative, Project Inspector, Consulting Engineers or Architect, or of an employee of either, or of a separate contractor employed by the Owner, or by changes ordered in the work, or by labor disputes, fire, unusual delay in deliveries, unavoidable casualties or other causes beyond the Contractor's control, or by delay authorized by the Owner pending arbitration, or by other causes which the Architect determines may justify delay, then the scheduled completion date may be extended for a period equal to the length of the delay to the critical path by a No-Cost Change Order.

#### ARTICLE 8.3.2

# Delete in its entirety and substitute the following:

Any claim for extension of time shall be made in writing to the Architect not more than ten (10) calendar days after the commencement of the delay; otherwise, any claim for such an extension of time shall be deemed waived. In the case of a continuing delay, only one claim is necessary, however, that claim must be made within ten (10) calendar days of the commencement of that delay. Claims for delays caused by unforeseeable adverse weather conditions must be filed for each delay. In the event of a continuing continuous weather delay, only one claim is necessary, but must be filed starting with first day of delay and revised to indicate last day of delay no later than (10) calendar days after last day of delay.

Any request for an extension of time must be specific as to the cause of the delay and must specify how many days or portions of days of delay have been caused, or will be caused, by the act, omission or event giving rise to the request for the time extension. If adverse weather conditions are the basis for the Claim for additional time, such Claim shall be documented by data substantiating that weather conditions were abnormal for the period of time and could not have been reasonably anticipated and indicated herein, and that weather conditions had an adverse effect on the initial submitted project schedule of construction beyond the number of days that exceed, during the project duration, the average annual number of rain/sleet /snow/wind days as outlined below.

- 1) The following criteria will be used to evaluate contractors claim for additional time due to weather conditions. Whichever is greater shall be used as a basis of evaluation:
  - Adverse weather conditions shall be considered only as those conditions that exceed the average annual number of rain/snow/wind days, during the project duration, as established by the Annual Local Climatological Summary, and NOAA National Technical Memorandum N R-65 (Revised) as published by the United States Government, National Weather Service, National Climate Center, Ashville, North Carolina.
  - b) Adverse weather conditions shall be considered only as those conditions that exceed the actual number of rain/snow/wind days actually recorded, during the project duration, using highest value within last 5 years at the project site.
- 2) Contractor shall use anticipated rain, sleet, snow, wind and other weather delays indicated in initial project schedule submission prior to submitting a claim for additional time due to weather conditions. Contractor shall document all instances of use of weather delays on project schedule used to date at each subsequent project schedule submission.
- 3) Refer herein for requirements of contractor to provide criteria of other weather delay type not indicated herein based upon Contractor's proposed ways and means in initial project schedule submission.

# ARTICLE 8.3.2

## Add the following:

8.3.2.1 If the Contractor files for an extension of time due to excessive moisture content of the soil, the Claim shall be made within ten (10) calendar days of the discovery. The Owner's Soils Engineer shall determine the level of moisture content allowable for the work to proceed based on the type of work underway and the type of soils. An extension of time may be granted by the Owner only for portions or areas of the work adversely affected by the excessive moisture. The balance of the work shall proceed. The Contractor shall also notify the Owner when conditions appear to be adequately dry so that the moisture content can be tested to determine the length of the time extension.

The standard of review shall be if the soil moisture is different than would be expected at the project site and the general locality of the project based on the last 5 year's most severe weather conditions per NOAA and its impact on soil moisture. If the site and/or general locality, most severe will govern, has soil moisture consistent with the actual site conditions, no time extension due or additional costs due to excessive moisture content will be granted and it will be considered a Contractor's ways and means responsibility and contractor to provide project temporary measures to maintain project schedule not limited to dewatering and other temporary measures.

## ARTICLE 8.3.3

# Delete paragraph in its entirety and substitute the following:

No damages or compensation of any kind shall be paid to a Contractor because of delays in the progress of work, whether such delays be avoidable or unavoidable. The granting of an extension of time because of unavoidable delays shall in no way operate as a waiver on the part of the Owner of the right to collect liquidated damages for other delays or of the right to collect other damages or of any other rights to which the Owner is entitled.

## ARTICLE 8.3.4

# Add the following paragraph:

Delays shall be reviewed against the critical path schedule and shall be considered only if the affected item is within the critical path, at time of delay. Contractors shall make all efforts to identify such items which may cause delay early in the project to avoid such delays. The lack of a critical path schedule, as described elsewhere, shall void claims for delay unless the Architect determines the delay is justified. Critical path delays will only be considered if all related activities, including submittal approvals, are current and in order. In the event of multiple claims for delays only the cumulative impact will be evaluated based upon concurrent activities and only critical path items take precedence over concurrent activities.

## ARTICLE 8.3.5

# Add the following paragraph:

Should the Owner, for any cause, authorize a suspension of work, the time of such suspension will be added to the time allowed for completion. Suspension of work by order of the Owner shall not be deemed a waiver of the claim of the Owner for damages for non-completion of the Work as above required.

# **ARTICLE 9 - PAYMENTS AND COMPLETION**

# **ARTICLE 9.2.1.1**

## Add the following:

The Schedule of Values shall list the following items as a minimum list:

- .1 Overhead and other reoccurring monthly costs related to overhead burden, not limited to the following:
  - a. On-site supervision
  - b. Office supervision (Home office)
  - c. On-site job clerk/support staff
  - d. Office support staff
  - e. Project meetings
  - f. Temporary on-site office
  - g. On-site telephone local, internet and long distance
  - h. On-site utility costs
  - i. Temporary fencing, (Project fencing)
  - j. Temporary toilets, wash areas and related items
- .2 Contractor's profit
- .3 Contractor's direct project costs, not limited to the following:
  - a. Mobilization

- b. Demobilization
- c. Temporary utilities, (direct usage i.e. water, natural gas and etc
- d. Project temporary barricades, specific use
- e. Equipment rental
- f. Contractor-owned equipment
- g. Bonds
- h. Insurance
- .4 Technical section items, Division 2 through Division 33.

## ARTICLE 9.3.1

# Delete in its entirety and substitute the following:

- 9.3.1 At the end of each calendar month, Contractor shall submit to the Architect an itemized Application for Payment for operations completed in accordance with the Schedule of Values. Such application shall be notarized and supported by such data substantiating the Contractor's right to payment as the Owner or Architect may require; such as, copies of requisitions from subcontractors and materials suppliers, and reflecting retainage. The form of application for payment shall be a notarized AIA Document G702, Application and Certificate for Payment, supported by AIA G703, Continuation Sheet.
  - .1 Until final payment, the Owner shall pay 95% of the amount due the Contractor on account of progress payments unless noted otherwise in instructions to bidders and holdouts indicated in the project manual.
  - .2 Payment by Owner of any payments of any payment request is not an indication that the owner has inspected, approved or accepted any part of the work, nor shall payment constitute a waiver in any respect of owner rights.
  - .3 Contractor shall not be entitled to have any payment request processed or any payment for any work performed so long as contractor has failed to comply with any lawful or proper direction the whole or any portion of the work given by the owner or the architect.
  - .4 Contractor shall not be entitled to have any payment for item(s) without architect approved submittals unless agreed in advance by architect in writing.
  - .5 Contractor shall invoice for Overhead, Profit and other reoccurring monthly costs related to overhead burden items indicated herein above. Contractor shall divide the total costs by number of months indicated as time of completion in contract by total cost for a monthly amount to invoice. This amount to be invoiced monthly with each pay application. In the event the time of construction will exceed time of completion the architect reserves the right to reduce the monthly cost by projected time of completion.
    - a. The Contractor may upon prior advanced approval by the architect invoice the Overhead, Profit and other reoccurring monthly costs as a percentage of total project completion if approved by architect prior to start of construction. Under no circumstances may total amount of cumulative costs billed of items exceed Contractors Schedule of Values amount without an approved change order. Once either method is approved no change can be allowed for payment amount determination for duration of project except as noted otherwise herein.
    - b. Profit shall be considered as a fixed item from Contractors Schedule of Values and paid on a monthly basis as indicated above.
  - .6 Contractor may substitute securities, listed in Government Code section 16430, for any amounts withheld by the Owner to insure performance under this Contract pursuant to Public Contract Code Section 22300.
    - a The Contractor shall make the election to utilize the Section 22300 Escrow Procedure on or before 30 days after the Notice to Proceed is given to the Contractor. Request after that date will not be recognized or considered.
    - b The securities or other forms of deposit must be reviewed and approved by the Owner for conformity with the standards prescribed in subdivision (c) of Section

22300 and acceptance by Owner's legal council.

- .7 Contractor shall provide Conditional Lien Releases by Contractor, subcontractors, sub-tier and suppliers for all funds being submitted with each Application for Payment. Prior to next Application for Payment, Contractor shall provide Unconditional Lien Releases by Contractor, subcontractors and suppliers for all funds paid by previous Application for Payment. All contractors and subcontractors must furnish proof that electronic certified payroll records (eCPR) are directly submitted to the Labor Commissioner using the Department of Industrial Relations (DIR) electronic certified payroll reporting system, with each Application for Payment.
  - a. Owner and or Architect have the discretion to require from Contractor any of the following information with the application for payment: (1) certified payroll covering the period of the prior application for payment, (2) receipts or bills of sale for any items
  - In addition, upon submittal of the first payment request, a complete per diem wage rate breakdown for all trades must be submitted in order for the payment request to be processed.
  - c. Contractor not providing Unconditional Lien Release for each subcontractor, sub-tier contractor, supplier and vendor from the previous approved application of payment is grounds for rejection of Application for Payment.
  - d. Conditional and unconditional releases shall be per California Civil Code Sections 8132 and 8134 for progress payment and Sections 8136 and 8138 for final payment.
- .8 Attach the Application for Payment Request-Certification form with the pay request. Form is in Section 006000 "Forms".
- .9 Applications for Payment shall be prepared for only those materials and equipment incorporated into the work in their final locations and shall not include a request for payment for materials and equipment stored on site or off site, unless previously agreed to in Article 9.3.2. below.
- .10 For any payment request the Owner determines not to be a proper payment request suitable for payment, the Owner shall return the request to the Contractor as soon as practicable, but not later than seven (7) days, after the Owner's receipt from Architect. A request returned pursuant to this paragraph shall be accompanied by a document setting forth, in writing, the reasons why the payment request is not proper.
- .11 Contractor shall review Pay Request with Project Inspector (7) days prior to the 25th day of the month. The Owner and Owner's representative shall be invited to review session. Upon mutual agreement, Contractor shall forward Application for Payment Request along with certification and related documents to Architect, completed, see Section 006000 "Forms". Architect shall submit, when approved, application for Payment Request to Owner or reject within seven (7) days of receipt by Architect. Architect shall copy Contractor on Pay Request transmittal to Owner.

# **ARTICLE 9.3.1.1**

Delete this paragraph.

## ARTICLE 9.3.2

# Delete this paragraph and substitute the following:

If approved in advance by the Owner, payment may be made for materials and equipment delivered and suitably stored at the site for subsequent incorporation in the work. If approved in advance by the Owner, payment may similarly be made for materials and equipment suitably stored off the site at a location agreed upon in writing. Payment for materials and equipment stored on or off the site shall be conditioned upon compliance by the Contractor with procedures satisfactory to the Owner to establish the Owner's title to such materials and equipment or otherwise protect the Owner's interest, and shall include applicable insurance, consent of surety, storage and transportation to the site for such materials and equipment stored off site.

- A. The item(s) to be stored off-site must be fabricated products or products specifically for use in the project and cannot be raw materials or other stock that may be used for other purposes without express written approval of the Architect in advance of request.
- B. The Contractor shall furnish the Architect a clear, complete description of the actual fabricated materials for which off-site storage is requested, a bill of sale listing the quantity and the cost of the materials, equipment or other items, F.O.B point of origin and evidence of payment for these materials.
- C. The Project Inspector or other designee shall inspect the materials, equipment or other items at the site, prior to recommending payment for these items. The Contractor shall pay the cost of the Project Inspectors time and any other expenses incurred in performing the inspections. The inspection report must clearly identify what was inspected, where and in what condition it was at the time of inspection. Project Inspector shall photograph all items and confirm each and every item is identified and tagged for use for this project.
- D. The Contractor shall provide evidence that the off-site storage facility is bonded as a warehousing facility.
- E. The Contractor shall obtain and pay for a separate, project specific, broad based comprehensive coverage insurance policy, which provides coverage at replacement value for all loss from any cause, including loss during transport from the off-site storage to the project site as well as all transportation charges.
- F. The Owner shall be listed as "Co-Owner" of the policy. The policy shall not have a deductible since, as the co-owner, Owner must not be liable for payment of any deductible for material for which the Owner will make payment to the Contractor.
- G. The Owner must be provided with the "Original" Policy. Insurance certificates are not acceptable.
- H. Contractor shall obtain consent of surety for approval to store items off-site and provide proof of consent of surety to store items listed off-site.

No application for payment for off-site stored items shall be considered until all of the above has been received, reviewed and approved by the Owner and Architect.

## ARTICLE 9.3.4

# Add the following:

The project may be funded with multiple funding sources. The District may require a supplemental invoice(s) to accompany progress pay requests to document portions of progress work for reimbursement to District from said funding sources. Such supplemental invoices could possibly document that work is encumbered by District but not yet completed. Contractor shall use best endeavors to cooperate with owner's requirements and provide up to five (5) additional copies of complete invoices when so requested.

## ARTICLE 9.4.1

## Amend as follows:

The word "seven" in subparagraph 9.4.1 of the General Conditions is hereby changed to "ten".

# Add the following to paragraph 9.4.1:

If during the course of construction the Architect determines that the application for payment does not reflect the work completed to date, he may re-adjust the billed value to reflect actual construction progress.

A. The Architect may access interim liquidated damages at any time in anticipation of project not being completed on schedule if construction work is behind schedule. Assessment shall not exceed days behind schedule at time of assessment. The amount held for interim liquidated damages will be evaluated periodically and adjusted to actual liquidated damages no later than the end of construction.

- B. If the Contractor can recover construction work back to approved construction schedule based on allowed project duration schedule, Architect has option to release assessment or wait until project completion and release funds if project was completed in contract time with approved adjustments in contract time or maintain assessment retention.
- C. In the event Contractor has no current schedule, Architect shall determine if project is on, ahead, or behind schedule based on allowed project duration herein. Architect may consult with Contractor, Owner and others to make a determination of schedule status in the absence of a current schedule.

## ARTICLE 9.4.2

# Make the following change:

Remove the word "Substantial".

## ARTICLE 9.4.3

# Add the following:

It is recognized that the Certificate of Payment to be issued by the Architect pursuant to paragraph 9.4 of the General Conditions may be based on estimates, and it is agreed that no inaccuracy or error in the estimate will operate to release the Contractor, or any Surety, for liability for damages arising from such work, or stop the Owner from enforcing each and every provision of the Contract, and the Owner shall have the rights, subsequently, to correct any errors made in any estimates. The Contractor shall not be entitled to include any work done or materials furnished for purposes of payment, or be entitled to any payment made on account of any work done or materials furnished so long as any lawful or proper directions for correction in defects of the work, or any portion thereof, given by the Owner or the Architect, shall remain non-complied with.

# ARTICLE 9.5

# Add the following:

- 9.5.3 A. In addition to any amount(s) which Owner may retain under the article entitled herein, Owner may withhold sufficient amount(s) of any payment(s) otherwise due to Contractor, as in its judgment may be necessary to cover the following:
  - 1. Payments which may be past due and payable for claims against Contractor or any Subcontractors at any level for labor or materials furnished in the performance of work under the Contract Documents.
  - Defective work not remedied.
  - 3. Failure of Contractor to make proper payments to its subcontractor(s) or material men for materials and/or labor.
  - 4. Completion of work if there exists a reasonable doubt that the work cannot be completed for the balance then unpaid.
  - 5. Damage to another contractor.
  - 6. All costs and expenses associated with Owner having to acquire alternate educational facilities as temporary measures if Contractor fails to complete the Project within the period of time required by the Contract Documents. The Owner may secure temporary measures, with all costs and expenses associated paid by contractor, if based upon the current construction schedule; completion will be delayed from project duration at fault of Contractor.
  - 7. Project schedule not up-to-date with the current application for payment.
  - 8. Unpaid regular time and overtime charges due consultants, Project Inspectors, the Architect, and Owner or others as a result of extra services that were provided at Contractor's request or as a result of actions of Contractor or those employed by Contractor, including subcontractors, sub-tier contractors, material suppliers, or others will be withheld from current payment requests.
  - 9. Contractor agrees that Owner may withhold 150 percent of the estimated cost of any additional testing or retesting required as a result of the fault or negligence of Contractor, or Subcontractors, sub-tier contractors, vendors, or suppliers, until such time as Owner receives confirmation that payment for such additional testing or

- retesting has been made by Contractor.
- 10. Failure to maintain a current record set of drawings. The record drawings shall be updated to the date current with each application for payment.
- 11. Failure to submit daily reports by Contractor's project superintendent each month or as when requested by Architect and/or Owner.
- 12. Failure to submit items required to accompany payment requests at initial and final completion.
- 13. Failure to submit and keep current any construction schedule required by the Contract Documents.
- 14. Failure to compensate the Architect for substitution review within the required time period.
- 15. Failure to compensate Owner for overtime charges for Owner representatives and employees incurred as a result of services provided during the current payment period.
- 16. Failure to compensate Owner and/or the Architect for the cost of review time to evaluate Contractor's proposed solutions to effect repair of work not in accordance with Contract Documents.
- 17. Failure to submit per diem wage rates for all trades pursuant to appropriate provisions of the General and Supplementary Conditions and Divisions 0 and 1 of Project Manual.
- 18. Penalties for violation of labor laws.
- 19. Cost of site clean-up.
- 20. Required payments to indemnify, hold harmless, or defend Owner.
- 21. Compensation for unpaid extra services for the Architect caused by Contractor.
- 22. Compensation for unpaid extra services for the Testing Agency/Laboratory(s) and Project Inspector(s), including but not limited to reinspection required due to Contractor's failed tests, installation of unapproved or defective materials, or CONTRACTOR's requests for inspection and failure to attend the requested inspection.
- 23. Any liquidated damages, forfeiture of fees, or other damages assessed against Contractor by reason of failure to complete the Project on time.
- 24. All cost related to work in place without submittals approved by the Architect.
- B. If Contractor defaults or neglects to carry out the Work in accordance with the Contract Documents, or fails to perform any provision of the Contract Documents, after 10 days' written notice to Contractor, Owner may make good such deficiencies without prejudice to any other remedy it may have.
  - 1. Owner shall reduce the total contract price by the cost of making good such deficiencies.
  - 2. If Owner deems it expedient to correct work not performed in compliance with the Contract Documents, an equitable deduction from the contract price shall be made.

## ARTICLE 9.6.1

# Delete subparagraph 9.6.1 and substitute the following:

Provided a completed Application for Payment with required attachments is received by the Architect not later than the last day of a month and provided the Architect has issued a Certificate for Payment, the Owner shall make payment in the manner provided in the Contract Documents.

# ARTICLE 9.6.2 Add the following:

## 9.6.2.1 CONTRACTOR shall pay:

- a) All transportation and utility services not later than the 20th day of the calendar month following the month in which the services are rendered;
- b) Ninety five percent, unless noted otherwise in instructions to bidders, of the cost of all materials,

- tools, and other expendable equipment, not later than the 20th day of the calendar month following the month in which the materials, tools, and equipment are delivered to the project site, and the balance of the cost not later than the 30th day following completion of that part of the work in which the materials, tools, and equipment are incorporated or used; and
- c) To each of its subcontractors the respective amounts allowed CONTRACTOR on account of work performed by each subcontractor not later than the seventh day following each payment to CONTRACTOR.

# ARTICLE 9.7.1

## Delete in its entirety and substitute the following:

The Architect will issue a Certificate for Payment after receipt of a proper Application for Payment request from contractor as defined below. Architect, within seven (7) calendar days after receipt of the Contractor's proper Application for Payment will send proper Application for Payment along with certificate for Payment to Owner for approval/payment or send notice to contractor if pay request was not a proper pay request and list reasons Application for Payment was not approved by Architect. Architect shall send/post Contractor's approved Application for Payment with certificate for Payment to Owner via US Mail or equivalent. The Owner will pay the Contractor within 30 calendar days after receipt of Contractor's proper Application for Payment approved by Architect for Payment. The Owner shall send/post payment to Contractor via US Mail or equivalent within time period. The number of days available to Owner to make a timely payment of proper Application for Payment without incurring interest shall be reduced by the number of days by which the Architect exceeds seven (7) calendar days not counting for days until a proper pay request is received by Architect from Contractor.

- 1. The seven (7) calendar days for Architect's review of Application starts after all information is complete and proper as defined herein. In the event information is received in segments, time starts counting after all information is received and no sooner. Definition of "proper" pay request is as follows:
  - a. Dollar values consistent with work completed through date submitted.
  - b. Math correct, including incorrect rounding of retention.
  - c. Approvals and signatures of required parties, including notarized documents.
  - d. Conditional lien releases for current application for payment and unconditional lien releases for previous application for payment included.
  - e. Special hold-out items identified.
  - f. Prevailing wage documentation has been filled electronically with the Labor Commissioner and is current as well as on file at contractor's office and current with each pay request application.
    - 1) Provide proof of submittal of current Prevailing wage documentation on line electronically with the Labor Commissioner when requested.
    - 2) Provide affidavit that records have been filed electronically and are on file and current with each pay request.
  - g. Record drawing documentation approved for period by project inspector or Architect.
  - h. Bonds and insurance in force.
  - i. Reductions for non-conforming work indicated.
  - j. As-built schedule with adjustments showing schedule to complete work and updated completion date.
  - k. 5 copies of each complete pay request application, certifications and attachments.
  - I. Current projected construction schedule indicating as built schedule of work done to date.
  - m. Copies of Bonded Weighmaster Certificates shall be submitted to the Architect along with pay requests for work done. Payment will not be made on non-structural concrete items poured requiring Bonded Weighmaster Certificates without receipt of Certificates of items poured requesting payment.
  - n. Completed Application for Payment Request-Certifications completely filled out by Contractor and approval by Project Inspector. Architect and Owner will review and if approved will approve for payment processing or send back to contractor with reasons not

approved.

# 2. Final Pay Request:

- a. Include all items indicated in #1 above plus the following;
- b. 100% completion of project closeout items.
- c. Notice of completion filed by Owner and 35-calendar day waiting period expired.
- d. Resolution of Stop Notices per Public Contract Code.
- e. Resolution of all claims.
- f. Consent of Surety Company to final payment, AIA Document G707, executed by Surety Company.
- g. 5 copies of each complete pay request application and attachments.
- h. Record schedule.

## ARTICLE 9.8.

Change title to read as follows:

9.8 COMPLETION

#### ARTICLE 9.8.1

Delete in its entirety:

## ARTICLE 9.8.2

Delete in its entirety and substitute the following:

- A. Contractor shall notify the Architect in writing when the project is acceptably completed, all work scope is done and ready for Project Inspector and Contractor to make a list of items to be corrected.
- B. The Project Inspector and Contractor shall make a list of items to be corrected and finish any items discovered incomplete. These items shall be completed and corrected prior to Architect and Consultants preparing a Final Inspection and Punch List Preparation. Contractor shall notify Architect in writing when all items are completed and corrected as well as items completed and corrected from Project Inspector and Contractor list of items previously prepared.
- C. Architect's field representatives (Architectural, Mechanical, Electrical, and other consultants) will make a field survey of the project to confirm that it has reached a state of completion, and items for "B" above have been done, in order to eliminate an unreasonably long Punch List on final inspection.
  - 1. All items in scope of work have been completed and accepted by Contractor and Project Inspector.
  - 2. All building systems are operational, tested and commissioned.
  - 3. Cleaning of building, grounds and related items completed except when approved by Architect in writing in advance of field survey. Contractor's temporary measures may be left in place until final demobilization but a condition on removal for final payment processing.
  - 4. Contractor to provide a written status update of project closeout check list.
- D. If not ready, the Architect will give the Contractor a written brief summary of what must be done for the project to be considered complete, enough for the Punch List preparation.
- E. When, in the opinion of the Architect, the job is ready for the Punch List preparation, a Punch List of items requiring completion and/or correction will be prepared. (This Punch List will be made as specific and complete as possible, and will include the listing of all specified closing items required of the Contractor, such as Record Drawings, maintenance manuals, written guarantees, etc.) Any items observed or noted subsequent to Punch List preparation, shall also be corrected prior to reinspection, unless the Architect determined, in writing, that any such items properly fall into the category of work to be corrected during the warranty/guarantee period.
  - 1. Architect's Punch List preparation will be scheduled and respective contractors, sub-

- contractors and vendors shall be present during the Punch List inspection. Sub-Contractors, contractors and vendors will be required to show operational status of items in project scope of each respective discipline.
- 2. Mechanical, Electrical, and other consultants Punch List preparation will be scheduled and respective sub-contractor, contractors and vendors shall be present during the Punch List inspection. Sub-Contractors, contractors and vendors will be required to show operational status of items in project scope of each respective discipline.
- 3. All Punch List items shall be completed and/or corrected before Contractor calls for Punch List re-inspection.
  - a. The Architect shall be notified in writing by Contractor when all Punch List items are complete and project is ready for punch list back check by Architect and consultants. Architect shall schedule punch list back check with contractor within 10 calendar days from receipt of written notice from Contractor.
  - b. The Architect will allow contractor two, (2), punch list back check visits of all items. Any additional site visits for items not completed and acceptable to Architect after second site visit will charged to the contractor as a negative change order for time spent by Architect to do re-inspections and office related work including reimbursable expenses incurred due to additional site visits.

#### ARTICLE 9.8.3

Delete in its entirety:

## ARTICLE 9.9.1

# Delete in its entirety and add the following:

It is understood and agreed that the Owner shall have the right to occupy the building or use the improvement contemplated by the Contract prior to the completion of the entire work, and that such occupancy or use shall not operate as an acceptance of nor relieve the Contractor from completing any part of the work.

#### ARTICLE 9.9.2

# Delete in its entirety and add the following:

In case that the Contractor has not completed the work, and the contract time including authorized time extensions has expired, the Owner reserves the right to occupy any portion of the Work at any time before completion and while work is in progress. In the event of such occupancy, the Contractor shall provide, without additional cost to the Owner, suitable protection by means of fencing, barriers, posted signs or other temporary measures and methods, as required to prevent persons other than those directly connected with the Work from entering remaining areas where continuing Work is being conducted, vehicles are operating, or materials are stored.

# ARTICLE 9.9.3

# Delete in its entirety and add the following:

Occupancy by the Owner prior to final acceptance shall not be construed by the Contractor as being an acceptance of that part of the Project so occupied, nor shall the Contractor be entitled to, or make demand for, additional compensation or extension of time because of such occupancy.

## ARTICLE 9.9.4

# Add the following:

Occupancy by the Owner prior to final acceptance shall not be deemed to constitute a waiver of existing claims on behalf of the Owner or Contractor against each other.

# ARTICLE 9.9.5

# Add the following:

The metered cost of electricity, water, fuel, etc., for the occupied portions, and the cost of operating the heating and air conditioning systems for the occupied portions will be borne by the Owner except as follows:

- 1. Contractor shall provide water for earthwork activities and related items.
- 2. Contractor shall provide all temporary measures for connecting and disconnection to above items.

The Contractor must save the Owner and his authorized representatives from any liability whatsoever caused by the Contractor, his employees, agents or representatives' refusal or negligence in the payment of all bills, expenses and/or utilities. Should the Contractor fail in this obligation within thirty (30) days of the bill due date, then the Architect may settle all current outstanding balances at his sole discretion direct with the appropriate body, and deduct such monies direct from the Contractor's due monthly payments, retention or other available monies plus charges, including an administration fee by Architect.

## ARTICLE 9.9.6

## Add the following:

Use and occupancy by the Owner prior to final acceptance shall not relieve the Contractor of his responsibility to provide and maintain all insurance and bonds required of the Contractor under the Contract until the final Completion of the Work which shall be defined as thirty five (35) days after filing Notice of Completion, (NOC), contingent upon punch list being 100% complete and all project closeout items 100% complete with no exceptions. In the event the NOC has been filed and there are outstanding punch list items the contractor shall maintain insurance and bonds until such time the punch list is 100% complete, project closeout items 100% complete or 35 days after NOC whichever is greater.

#### ARTICLE 9.10.2.1

## Add the following:

Subject to the provisions of this subparagraph 9.10.2, the final payment will be made thirty-five (35) days after the recording of the Notice of Completion, provided that the Architect shall then have issued the final certificate provided for in subparagraph 9.10.1. The acceptance of said final payment by the Contractor shall constitute a waiver of all claims against the Owner arising under this Contract, and this sentence supersedes subparagraph 9.10.4 of the General Conditions.

- Notice of Completion shall be prepared and approved by Owner prior to recording. The official
  project acceptance date shall be the date of Owner acceptance of the project and authorization to
  filing of the Notice of Completion. Owner shall record Notice of Completion within 10 days of
  acceptance of project as being complete. All warranties start dates will be Date of recordation of
  Notice of Completion.
- 2. The project shall be accepted by the Owner who shall authorize after acceptance of the project the subsequent filing of the Notice of Completion. The final payment shall be made thirty-five (35) days from the date of recordation of the Notice of Completion, provided that: The Contractor shall furnish satisfactory evidence that all claims for labor and materials have been paid and that no claims shall have been presented to the Owner by any person or persons based upon any act or omission of the Contractor, and no Stop Notices or claims shall have been filed against said work or the property whereon it was done.
  - a. The Contractor in direct contract with the owner must record and file with owner an Affidavit of Payment of Debts and Claims and release of Stop Notices prior to request for project acceptance is considered by the Owner. By this document, the contractor hereby certifies that on date of document recordation, he/she has been paid in full less retainage for all materials and equipment furnished, for all labor and services performed, and for all known indebtedness and claims against the undersigned for damages arising in any manner on or against the Project, its land, improvements, and equipment of any kind.
  - All others not limited to sub contractors, lower tier contractors, suppliers, vendors and others providing services, materials, equipment and related items <u>must record their liens and</u> <u>serve owner stop notice within thirty (30) days</u> of the date the Notice of Completion is recorded to place and person indicated in project manual herein. (Civil Code §3116).
  - c. Within 60 days after the date of Notice of Completion of the Project, the retention withheld by the Owner shall be released. In the event of a dispute between the Owner and General Contractor and or sub contractors, the Owner entity may withhold from the final payment an amount not to exceed 150 percent of the disputed amount. Completion is defined as:
    - 1) Owner acceptance of project and filing of Notice of Completion, in the absence of, the following shall govern:

- a) The occupation, beneficial use, and enjoyment of a work of improvement, excluding phased projects completed phases, any operation only for testing, startup, or commissioning, by the public agency, or its agent, accompanied by cessation of labor on the work of improvement.
- b) The acceptance by the public agency, or its agent, of the work of improvements in writing.
- c) After the commencement of a work of improvements, a cessation of labor on the work of improvement for a continuous period of 100 days or more, due to factors beyond the control of the contractor.
- d) After the commencement of a work of improvement, a cessation of labor on the work of improvement for a continuous period of 30 days or more, if the public agency files for record a Notice of Cessation or a Notice of Completion.

9.10.2.1.1 Notice of Completion will not be prepared nor moved forward for Owner approval until all of the following are complete:

- 1. Project Closeout 100% complete.
- 2. Punch list items 100% complete.
- 3. Record documents approved and delivered.
- 4. All warranties and guarantees have been delivered and accepted by Owner and Architect.
- 5. Training complete.
- 6. The contractor(s) final verified report is filed with the Division of State Architect of the Department of General Services.
- 7. OWNER is able to occupy all portions of the project as intended on Construction Documents.
- 8. Contractor in direct contract with the owner upon completion of work shall execute all certifications specified elsewhere.

#### ARTICLE 9.10.3

# Delete in its entirety and substitute the following

If for any justifiable reason, any Contract items cannot be completed prior to the final inspection, they will, with the Owner's concurrence and at the Owner's sole option, be deleted by Change Order to start the Stop Notice filing period as soon as possible. Such Change Order will include the Contractor's agreement to do the work at the prices listed in the Change Order or receipt of Owner's Purchase Order, as soon as the reasons for the delay no longer prevail, and the agreement of the Contractor's surety to be responsible for the work on the terms and condition of the original bond, when such Purchase Order is issued.

1. Scope of work on Purchase Order will be a standalone project meeting the requirements, general conditions and supplementary conditions herein. All technical and completion requirements will be the same as herein.

# ARTICLE 9.11

# Add the following:

- 9.11.1 Should the Contractor fail to complete his/her contract within the time fixed for such completion in the Project, the Contractor shall become liable to the sum per calendar day per phase as indicated, for each day any phase of said Work remains incomplete beyond the time set for completion for agreed and liquidated damages, and not as a penalty, agreed upon by the parties to the contract, it being expressly stipulated that it would be impracticable and extremely difficult to fix the amount of damage. If it appears to the Contractor that he/she will not complete the work provided for in the contract in the time agreed, the Contractor shall make written application to the Owner at least five (5) calendar days prior to the expiration of the time for completion, stating the reasons for the amount of extension which the Contractor believes should be granted. The Owner may then, in its discretion, grant or deny such extension.
- 9.11.2 Any monies due or to become due the Contractor may be retained to cover the said liquidated damages. Should such monies not be sufficient to cover such damages, the Owner shall have the

right to recover the balance from the Contractor and/or his sureties.

# ARTICLE 9.12

Add the following:

# Correction of Work Before Final Payment

- 9.12.1 Contractor shall promptly remove from the premises all work identified by Owner as failing to conform to the Contract Documents, whether incorporated or not. Contractor shall promptly replace and repair its own work to comply with the Contract Documents, without additional expense to OWNER, and shall bear the expense of making good all work of other contractors destroyed or damaged by that removal or replacement, including compensation for the Architect's additional services.
- 9.12.2 If Contractor does not remove work within a reasonable time following written notification, Owner may remove and store the material at Contractor's expense. If Contractor does not pay the expenses of removal within 10 days, Owner may sell the materials at auction or private sale upon 10 days' written notice, and shall account for any net proceeds after deducting all costs and expenses that should have been borne by Contractor.

#### ARTICLE 10 - PROTECTIONS OF PERSONS AND PROPERTY

## ARTICLE 10.1.1

## Add the following at the end of subparagraph:

This requirement will apply continuously and not be limited to normal working hours. Contractor shall verify requirements of working hours allowed by local jurisdictions. Contractor shall make provisions for allowed working hours and securing necessary special permits to work beyond allowed hours allowed by local jurisdictions.

#### **ARTICLE 10.1.1**

# Add the following:

- 10.1.1.1 The duty of the Architect, to conduct periodic surveillance of the Contractor's performance, is not intended to include review of the adequacy of the Contractor's safety measures, in, on, off or near the construction site.
  - a) Comments made in the Architect's field reports with regards to safety items are not intended to relieve the Contractor of his responsibility or intended to set a precedence that the Architect has any responsibility in safety items. The Architect has no training or experience in safety adequacy on construction sites.
  - b) The Contractor shall refer all items to his Safety Officer for disposition that are commented and/or mentioned during any visits by the Architect either by meeting minutes, casual conversations or follow up reports.

# **ARTICLE 10.1.1.2**

## Add the following:

The Contractor shall protect and preserve the Work from all damage or accident, including vandalism providing any temporary roofs, window and door coverings, boxings, or other construction, as required. This shall include any adjoining facility/property of the Owner or others affected by the Work of the Contractor.

## **ARTICLE 10.1.1.3**

#### Add the following:

The Contractor shall assume full responsibility for all glass and plastic glazing installed under this Contract against damage from any source during construction as well as adjacent glass and plastic glazing damaged by contractor, subcontractors and/or lower tier contractors. He shall replace all broken, cracked or scratched glass or plastic without expense to the Owner prior to date of recording of Notice of Completion.

#### ARTICLE 10.1.2

# Add the following:

10.1.2.1 Exception to the above shall be where indicated herein that asbestos, lead, PCB or other hazardous materials to be removed by others. Owner shall remove and dispose of hazardous materials prior to work being done by contractor in affected areas. See Division 2 Demolition Sections for additional requirements.

# ARTICLE 10.1.3

#### Add the following:

10.1.3.1 Exception to the above is if asbestos, lead, PCB or other hazardous material is indicated to be removed by this contractor herein, by submitting a bid contractor is consenting to do work.

# ARTICLE 10.1.4

# Add the following:

10.1.4.1 Exception to the above is if asbestos, lead, PCB or other hazardous material is indicated to be removed by this contractor. Contractor shall remove and dispose of hazardous materials prior to work being done by contractor in affected areas. See Division 2 Demolition Sections for additional requirements.

#### ARTICLE 10.2.1

# Supplement as follows:

- .4 all students, staff, faculty, Owner employees and the general public who are using the site.
- and shall check all projections, off-sets, footings, etc., and properly determine that there are no encroachments of the building or appurtenances on adjoining property or the property of other public entities. Where encroachments occur as a result of the work performed under these contract Documents, the Contractor shall remove such encroachments at his own cost, and at no expense to the Owner; and
- shall give to adjoining landowners all legal notices of excavation that may be required by law or otherwise, and he shall take all steps necessary or required by law to protect the adjoining land and the building and/or other structures situated thereon.
- .7 Shall not dispose of hazardous waste/materials without proper permits-authorizations. Contractor shall provide Owner with copies of permits/approvals of all disposal of hazardous materials.

# ARTICLE 10.2.3

# Add the following:

This project may be constructed at the time facility is in use; therefore, the Contractor shall take specific precautions to prevent injury and access to Staff, faculty and students. Material storage and vehicle access shall be subject to Owner approval. Provide for limited access time for delivery of materials and related items. Confirm and coordinate with Director of Maintenances and Operations and secure all approvals in writing. Refer to Project Summary for more restrictive requirements.

# **ARTICLE 11 - INSURANCE AND BONDS**

#### **ARTICLE 11.1.2**

# Supplement as follows:

1. In the absence of contrary written instructions from the Owner, the Contractor at the Contractor's expense, shall obtain and maintain insurance at all times during the prosecution of the contract, in companies and through agencies approved by the Owner, and within limits not less than those stated in Section 004313 - "Bonds and Certificates Required".

# ARTICLE 11.2

Delete in its entirety.

# ARTICLE 11.3

# Amended as follows:

The Contractor, in lieu of the Owner, shall purchase and maintain insurance, and perform in accordance with the provisions of Articles 11.3 through 11.3.11.

#### **ARTICLE 11.3.1**

#### Shall read as follows:

This insurance does not cover any tools or equipment owned or rented by the Contractor, Subcontractor, Sub-Subcontractor or their employees, the capital value of which is not included in the cost of the work.

#### **ARTICLE 11.3.1.3**

#### Amended as follows:

The Contractor, in lieu of the Owner, shall purchase and maintain insurance, and perform in accordance with the provisions herein.

Omit last sentence "If deductibles are not identified in the contract documents, the owner shall pay costs not covered because of deductibles."

#### **ARTICLE 11.3.2**

# Amended as follows:

Delete in its entirety.

# **ARTICLE 11.3.3**

#### Amended as follows:

Delete in its entirety.

#### **ARTICLE 11.3.6**

#### Amended as follows:

The Contractor will provide a copy of the policies, prior to commencing any work for Owner's records.

# ARTICLE 11.3.9 and 11.3.10

#### Amended as follows:

Delete all reference to Arbitration and Article 4.5.

#### **ARTICLE 11.4.1**

#### Delete in its entirety and substitute the following:

"The Owner will require the Contractor to furnish bonds covering faithful performance of the Contract and payment of obligations arising thereunder as stipulated in bidding requirements or specifically required in the Contract Documents on the date of execution of the Contract". Refer to Section 004313 - "Bonds and Certificates Required" for additional requirements.

# **ARTICLE 12 - UNCOVERING AND CORRECTION OF WORK**

# ARTICLE 12.2

The following material supplements paragraph 12.2 of the General Conditions:

- A. Add to subparagraph 12.2.1 of the General Conditions the following:
  - ". . . such additional services by the Architect, whether required by correction of rejected work, damage to work by actions of contractor(s) and/or to accommodate deviations from the Contract Documents, will be charged for at the prevailing office rate for Architect's/Staff plus expenses, consulting engineer(s) time at prevailing office rates and invoices for associated cost items. Architect shall invoice Owner for costs incurred. Contractor shall reimburse Owner for additional costs incurred by Architect and Consultants via a negative CCD.
  - 1. The Architect shall submit a CCD to Owner and Contractor indicating additional costs

- incurred by Architect and Consulting Engineers due to actions by the Contractor. Upon Owner approval of CCD, costs will be added to project as a negative CCD to be included in a Change Order.
- The Contractor has 10 calendar days from receipt of CCD specific to these items to dispute additional costs incurred by Architect and Consulting Engineers. Disputed items will not be considered after 10 calendar days beyond date of receipt of CCD by Contractor for charges by the Architect and Consulting Engineers.
- B. The warranty and guarantee periods specified in subparagraph 12.2.2 of the General Conditions shall commence on the date recording of the Notice of Completion.
- C. If the Contractor, after notice, fails to proceed promptly to comply with items of the warranty or guarantee as indicated herein, the Owner, may have defects corrected and the Contractor and his Surety shall be liable for all costs incurred.

# **ARTICLE 13 - MISCELLANEOUS PROVISIONS**

ARTICLE 13.1

Add the following subparagraphs to paragraph 13.1:

13.1.2 No discrimination shall be made in the employment of persons upon public works because of race, religious creed, color, national origin, ancestry, physical handicap, medical conditions, marital status, or sex of such person, except as provided in Labor Code Section 1420, and every Contractor violating the provisions of Section 1735 of the Labor Code is subject to all the penalties provided by said section.

#### ARTICLE 13.2.1

Add the following subparagraphs to paragraph 13.2.1:

- 13.2.1.1 Assignment Under Anti-Trust Claims: Pursuant to Public Contract Code Section 7103.5, Contractor and subcontractors shall conform to the following requirements:
  - .1 In entering into a public works contract or a subcontract to supply goods, services, or materials pursuant to a public works contract, contractor or subcontractor offers and agrees to assign to the Owner all rights, title, and interest in all to all causes of action it may have under Section 4 of the Clayton Act (15 U.S.C. Sec. 15) or under the Cartwright Act [Chapter 2 (commencing with Section 16700) of Part 2 of Division 7 of the Business and Professions Code], arising from purchases of goods, services, or materials pursuant to the public works contract or the subcontract. This assignment shall be made and become effective at the time the Owner tenders Final Payment to the Contractor, without further acknowledgment by the parties.

#### **ARTICLE 13.5.1**

Delete the second and third sentences of subparagraph 13.5.1 and substitute the following:

The Owner shall make arrangements for such tests, inspections and approvals with an independent testing laboratory and shall pay the testing laboratory in accordance with Sections 4-333C and 4-335, Group 1, Chapter 4, Part 1, Title 24, California Code of Regulations. The Contractor shall give the Architect and Project Inspector timely notice of when and where tests and inspections are to be made so that the Architect and Project Inspector may observe such procedures.

Delete the last sentence of subparagraph 13.5.1 in its entirety.

#### **ARTICLE 13.5.2**

Delete subparagraph 13.5.2 and substitute the following:

If the Architect, Owner or public authorities having jurisdiction determine that portions of the work require additional testing, inspection or approval not included under subparagraph 13.5.1, the Owner will make arrangements for such additional testing, inspection or approval by the testing laboratory, and the Contractor

shall give timely notice to the Architect and Project Inspector of when and where tests and inspections are to be made so the Architect and Inspector may observe such procedures. The Owner will pay the testing laboratory as provided in subparagraph 13.5.1.

#### **ARTICLE 13.5.3**

#### Delete subparagraph 13.5.3 and substitute the following:

If such procedures for testing, inspection or approval under subparagraphs 13.5.1 and 13.5.2 reveal failure of the portions of the work to comply with requirements established by the Contract Documents, the Owner will deduct compensation for the testing laboratory's and Architect's services and expenses made necessary by such failure from the Contract Sum by Change Order via a negative CCD. The Contractor shall also bear all other related costs made necessary by such failure.

- A. The Contractor shall reimburse Owner for all costs incurred due to failure of testing, inspection or approval by testing lab and related staff, Architect's/Staff plus expenses and consulting engineer(s) time at prevailing office rates for time required to respond to failure of testing, inspection or approval. Contractor shall reimburse Owner for all additional costs incurred via a negative CCD.
- B. The Architect shall submit a CCD to Owner and Contractor indicating additional costs incurred. Upon Owner approval of CCD, costs will be added to project as a negative CCD to be included in a Change Order.
- C. The Contractor has 10 calendar days from receipt of CCD specific to these items to dispute additional costs incurred by Testing laboratory, Architect and Consulting Engineers. Disputed items will not be considered after 10 calendar days beyond date of receipt of CCD by Contractor for charges by the Testing Laboratory, Project Inspector, Architect and Consulting Engineers.
- D. The Contractor shall provide a minimum of five (5) welders where Shop Welding requires welding inspection and/or three (3) welders where Field Welding Inspection requires welding inspection. The purpose is to minimize Inspection cost. If the above requirement is not met, Contractor shall be charged for excess Test/Inspection fees as determined by the Architect as prescribed herein. In the event welding requiring inspection is minimal Contractor shall secure approval in writing from Architect for a lessor number of welders prior to execution of welding.

Delete subparagraph 13.5.4 in its entirety.

#### ARTICLE 13.5.7

#### Add the following:

Whenever the Contractor arranges to work at night or any time when work is conducted other than the normal 40-hr. week, 0700 to 1600 Monday through Friday, or to vary the period during which work is carried on each day, he shall give the Owner a minimum of 48-hr. notice so that Project Inspector and Testing Laboratory may be notified. Such work shall be done without extra compensation to the Contractor. Whenever Contractor arranges to work beyond or before 0700 to 1600 Monday through Friday and test and inspection work is done at a premium time cost, the Contractor shall pay the difference between premium time and non-premium time as a Contractor shall reimburse Owner for all additional costs incurred via a negative CCD.

- A. Architect shall submit a CCD to Owner and Contractor indicating additional costs incurred. Upon Owner approval of CCD, costs will be added to project as a negative CCD to be included in a Change Order.
- B. Contractor has 10 calendar days from receipt of CCD specific to these items to dispute additional costs incurred. Disputed items will not be considered after 10 days beyond date of receipt of CCD by Contractor.

# **ARTICLE 13.5.8**

# Add the following:

Authorized representatives and agents of State and Federal Governments shall be permitted to inspect all

work, materials, payrolls, records and shall be given access to site at all times for such inspection.

#### ARTICLE 13.7.1

Change the following:

Change all "Substantial Completion(s)" references, notes and text to "Notice of Completion"

# **ARTICLE 14 - TERMINATION OR SUSPENSION OF THE CONTRACT**

ARTICLE 14.1

Amend as follows:

- 14.1.1 If the Work is stopped for a period of forty five (45) days under an order of any court or other public authority having jurisdiction, or as a result of an act of government, such as a declaration of a national emergency making materials unavailable, through no act or fault of the Contractor or a Subcontractor or their agents or employees or any other persons performing any of the Work under this contract with a Contractor, then the Contractor may, upon ten (10) additional days' written notice to the Owner and the Architect, terminate the Contract and recover from the Owner payment for all Work executed and for any proven loss sustained upon any materials, equipment, tools and construction equipment and machinery.
- 14.1.2 If the Work is stopped for a period of forty-five (45) days by the Contractor because the Architect has not issued a Certificate for Payment as provided in Paragraph 9.7 or because the Owner has not made payment thereon as provided in Paragraph 9.7, then the Contractor may, upon ten (10) additional days' written notice to the Owner and Architect, terminate the Contract, contingent upon the Owner not paying the Contractor after ten (10) days have elapsed.

# THE FOLLOWING ARTICLES SHALL BE ADDED TO THE GENERAL CONDITIONS:

#### **ARTICLE 15 - SUBSTITUTIONS**

a. Substitutions or Approval of Materials: Whenever any particular brand name or make of material or apparatus is hereinafter called for, every bidder submitting a bid upon this specification and the accompanying drawings, obligates himself to the use of such brands and makes, or of such other brands or makes, as shall have been duly approved by the Architect and the Owner in the manner described herein. Whenever any article of any material is specified by a reference to the name of any manufacturers or dealers or by specific reference to the catalogs of manufacturers or dealers. The intent is to establish a standard of excellence, which the Architect and the Owner have determined upon as requisite and necessary for this project, and subject only therefore, to such modifications as the Architect and Owner may make in accordance with the procedure given in this Article. It is therefore mandatory and binding upon the bidder to abide within the limits of the restrictions imposed. Where the words "or equal", "as selected", "approved", "approved make" or other synonymous terms are used in reference to material, quality, methods or apparatus in lieu of or in addition to other specific reference, it is to be distinctly understood that the approval of any such substitution is vested in the Architect whose decision shall be final and binding upon all concerned. The intent of this specification is NOT TO ELIMINATE PROPERLY QUALIFIED ENTRANTS FROM COMPETITION, but to confine the bidding to the part of the contractors, manufacturers and dealers, to those whose standing and qualifications are such that the Architect and the Owner feel warranted to give them their approval. If however, any bidder desires to have consideration given to INDIVIDUALS, FIRMS, MATERIALS, BRANDS, MAKES, ETC., OTHER THAN SPECIFIED, he may within thirty-five (35) calendar days after the award of the contract, submit or cause to be submitted to the Architect for his approval data substantiating a request for a substitution of "an equal" item. Requests for approval of materials or equipment shall be submitted to the Architect in writing prior to the established deadline except as outlined in sub-paragraph "b" below. Requests received after the established deadline will not be considered. All requests shall clearly define and describe materials or equipment for which approval is requested shall also include the figures received by the Contractor in bid form for the material or materials which are submitted for approval or substitution,

together with the figures in bid form of the specified material or materials for which substitutions are proposed. In case a substitution is offered which is less than the cost of the material or appliance specified, then a credit shall be taken for the difference between the two costs in order that the Owner shall obtain whatever benefits may be derived from the substitution. Requests shall be accompanied by manufacturer's literature, specifications, drawings, cuts, performance data, samples, etc., where same is necessary to completely describe the products. If the Architect does not consider this request for approval of a specified article or material an equal to the specified item, the Contractor shall furnish and install the specified item.

- b. Substitution or Approval of Materials After Expiration of Submittal Period: Since all bids are based upon materials and equipment specified, or substitutions approved in writing prior to bid date, the Architect and the Owner will consider substitutions after bidding where it is necessary to make a substitution because of strikes, lockouts, bankruptcy, discontinuing of a product, etc. Requests for such substitutions of materials after the award of a contract shall be made in writing to the Architect and shall be made within ten (10) days after the date that the Contractor ascertains he cannot obtain the materials or equipment specified. Such requests shall be accompanied by complete description of the materials or apparatus on which a substitution is desired to be made. Substitutions of material or apparatus other than those specified will not be accepted except upon the recommendation of the Architect to the Owner and with written approval of the Owner.
  - 1) In the event the Contractor requests a substitution, no cost increase will be allowed under any circumstances unless recommended by the Architect. A cost decrease will be allowed if item substituted is not equal to specified item as determined by the Architect and recommended in his review of substitution.
  - 2) All substitutions shall be executed by a change order.
- c. If the Substitution is Permitted: The Contractor shall be solely and directly responsible for fitting approved substituted material and equipment into the available space in a manner acceptable to the Architect, and for the proper operation of the substituted equipment with all other equipment with which it may be associated. The Contractor shall bear all costs of meeting the above requirements for presenting a proposed substitution, and if the substitution is accepted, he must bear all cost involved including costs of Architect's and Consulting Engineer's services required in adapting the substituted material or equipment to the installation to the complete satisfaction of the Architect.
- d. Submitting of Bids: It is to be understood that all bids shall be based upon materials and equipment as specified or substitutions approved in writing prior to the bid date.
- e. The Owner reserves the right to sole source items as allowed by Public Contract Code. Refer to Division One Section "Materials and Equipment" for additional requirements and conditions.

# **ARTICLE 16 - TRADE SECTIONS**

- a. For convenience, these specifications are arranged in several trade sections, but such separation shall not be considered as the limits of the work required of any subcontract of trade. The terms and conditions of such limitations are wholly between the Contractor and his subcontractors. It shall be his responsibility to see that all subcontractors and/or material suppliers examine all project manual sections and all contract documents to determine the extent of their work. Neither the Owner nor the Architect will be responsible for Contractor's omission of material and/or labor required by the Contract Documents, Project Manual, Addenda or reasonably inferred therein.
- b. In the event material and/or labor specified under a trade section shall be in contradiction of jurisdiction agreements, it shall be the responsibility of the Contractor, under whose Section the work and/or material has been specified, to inform the General Contractor of such conflict prior to bidding. Future claims for reimbursement due to jurisdiction conflict will not be recognized by either the Owner or the Architect.
- c. Any conflicts which may arise between the trades in regard to the extent of their respective works shall be adjusted by the Contractor.
- d. The requirements of Division 0 Procurement and Contracting Requirements, and Division 1 General Requirements and also including the Instructions to Bidders: General Conditions (AIA), Supplementary Conditions and all documents pertinent thereto, shall apply to each Technical

Section of the Specification as though specifically set forth therein.

#### ARTICLE 17 - CONSTRUCTION TIME AND LIQUIDATED DAMAGES

- a. The Contractor shall construct and execute all the work described in the said Bid Proposal, drawings and Project Manual in accordance with and subject to all of the requirements, covenants, stipulations and restrictions therein contained within the calendar days stipulated in the Bid Proposal.
  - 1) Notice of Completion date will be used for calculating liquidated damages. It will be calculated by the number of days beyond stipulated number of days in project manual verses number of days from first day of construction through date of Notice of Completion.
- b. Should the Contractor fail to complete this Contract and the work provided herein within the time fixed for such completion, the Contractor shall become liable to the Owner for all loss and damage which the latter may suffer on account thereof.
  - The Architect may access interim liquidated damages at any time in anticipation of project not being completed on schedule if construction work is behind schedule. Assessment shall not exceed days behind at time of assessment. The amount held for interim liquidated damages will be evaluated and adjusted to actual liquidated damages no later than the end of construction.
- c. It is HEREBY UNDERSTOOD AND AGREED, that it is and will be difficult and impossible to ascertain and determine the actual damage the Owner will sustain in the event, and if by reason of such delay; and it is therefore agreed that said Contractor will pay the Owner as liquidated damages and not as a forfeiture the sum set forth in the Owner-Contractor Agreement, per day for each and every day of delay beyond the time herein prescribed for finishing the said work, plus such additional costs as may be incurred is not paid, the Contractor agrees that the Owner may deduct the amount thereof from any money due or that may become due the Contractor under this Contract.
- d. This article does not exclude the recovery of damages under other provisions of the Contract Documents.
- e. It is further agreed, when it is obvious by all parties concerned, that the Contractor will not be able to complete the work within the stipulated time limit plus extension, if any, the Owner shall require him to expedite certain buildings, and/or portions of buildings and grounds for occupancy by the end of the stipulated time. This work shall be done and occupancy made without additional compensation to the Contractor or jeopardizing the Owner's right in rejecting or accepting the work at completion of the Contract. Normal wear and tear, due to occupancy, will not be cause for the Contractor to perform additional work over and above the regular Contract requirements, not will the occupancy relieve the Contractor of any responsibility of completing any work in occupied areas.

#### **ARTICLE 18 - INTERRUPTION OF OWNER'S OPERATIONS**

A. The Contractor shall confine his operation and storage of material and equipment within the area of his work and shall cooperate fully to insure the Owner a continued uninterrupted facility. Work requiring interruption of existing services shall be performed at the Owner's convenience on weekends and/or after hours as required. Contractor to provide security as required to protect his work during construction period until project is completed and accepted by Owner/Architect.

#### ARTICLE 19 - USE OF ASBESTOS, PCB OR LEAD MATERIALS/PRODUCTS

A. Contractor shall not use any asbestos, PCB or lead containing products or materials in performing the work under the Contract Documents. Upon completion of the Project, Contractor shall certify in writing to Owner that no asbestos, PCB or lead containing materials or products were used by Contractor or any subcontractor in performing the work required by the Contract Documents.

- B. Should asbestos containing materials be installed by Contractor in violation of this certification, or if removal of asbestos containing materials is otherwise a part of the Project, decontaminations and removals will meet the following criteria:
  - 1. Decontamination and removal of work found to contain asbestos or work installed with asbestos containing equipment shall be done only under the supervision of a qualified consultant, knowledgeable in the field of asbestos abatement and accredited by the Environmental Protection Agency (EPA).
  - 2. Any asbestos removal contractor shall be an EPA accredited contractor qualified in the removal of asbestos and shall be chosen and approved by an owner approved asbestos consultant who shall have sole discretion and final determination in this matter.
  - 3. The asbestos consultant shall be chosen and approved by Owner who shall have sole discretion and final determination in this matter.
  - 4. The work will not be accepted until asbestos contamination is reduced to levels deemed acceptable by the asbestos consultant.
  - Cost of all asbestos removal, including but not limited to the cost of an asbestos removal contractor, the cost of the asbestos consultant, additional Architects costs, analytical and laboratory fees, time delays, and additional costs as may be incurred by Owner shall be borne entirely by Contractor.
  - 6. Interface of work for the Project with work containing asbestos shall be executed by Contractor at Contractor's risk and at Contractor's discretion with full knowledge of the currently accepted standards, hazards, risks, and liabilities associated with asbestos work and asbestos containing products. By execution of the Construction Agreement, Contractor acknowledges the above and agrees to hold harmless Owner, its governing board, or other governing body, employees, agents, and the Architect and assigns for all asbestos liability which may be associated with this work. Contractor further agrees to instruct Contractor's employees with respect to the above standards, hazards, risks, and liabilities. Should lead containing materials be installed by Contractor in violation of this certification, or if removal of lead containing materials is part of the Project, decontaminations and removals will meet the criteria approved by Owner.
  - 7. The cost of all removals or decontaminations resulting from the installation of materials in violation of this certification shall be at the sole expense of Contractor.
- C. Contractor upon completion of work shall execute certifications as follows:
  - 1. Asbestos Certification shall be used for documentation of non-asbestos materials used in project.
  - 2. PCB Certification shall be used for documentation of non-PCB materials used in project.
  - Lead Certification shall be used for documentation of non-lead materials used in project.

#### **ARTICLE 20 - MISCELLANEOUS ITEMS**

- A. The Owner and Architect and their agents and employees will not be responsible for Contractor's means, methods, techniques, sequences of procedures of construction, or the safety precautions and programs incident thereto, and the aforementioned will not be responsible for Contractor's failure to perform the work in accordance with the Contract Documents.
- B. The services to be performed by the Architect under this Contract are intended solely for the benefit of the Owner. Nothing contained herein shall confer any rights upon or create any duties on the part of the Owner/Architect Agreement including, but not limited to any Contractor, subcontractor, supplier, or the agents, officers, employees, insurers, or sureties of any of the aforementioned.

END OF SECTION 006002

04/18/17

# **CONTRACT-AGREEMENT**

|   | CONTINAC   | I-AGINELIVII   | <u>IN_I</u>  |
|---|--|--|--|
| THIS CO   | NTRACT, made and concluded this  | day of   | 2025, A.D., by and   |
| between_  | WEST KERN COMMUNITY COLLEGE  | DISTRICT   | _,hereinafter referred to as the Owner, and  |
|   |  | referr   | red to as the Contractor.  |
|   | pecified and described in certain drawing  |  | a proposal to do certain work for the said ations, and entitled as follows on the Bid  |
|   |  | HEALTH LAB F<br>COLLEGE<br>AFT, CA   | REMODEL  |
| In strict a   | ccordance with drawings and specifications   | prepared ther  | refore by AP Architects.   |
| herewith and has insuch read intent and   | said proposal and specifications and has c<br>nvestigated the character of such work and<br>ding, examination and investigation, the sa  | arefully examing the materials and Contractor a  | carefully read and examined in connection ned the site where said work is to be done, required to be furnished, and by reason of agrees that he thoroughly understands the of said Contract and the requirements,  |
| Owner to  | •  |  | e payments hereinafter to be made by the<br>ling of said Contractor to do said work the  |
| ARTICLE   | <u>.l</u>  |  |  |
| and doing<br>machiner<br>nature of<br>unforesee<br>and befor<br>negligend<br>the mann<br>Engineer | g all the work contemplated and embraced<br>y, implements, apparatus and other means<br>the work to be done under said specific<br>en difficulties or obstructions which may are<br>the acceptance thereof by said Owner, a<br>see or carelessness or discontinuance of the<br>ther and according to the drawings and sp | d in this Contr<br>of construction<br>ations, or froncise or be encount<br>and shall be re<br>work, and for<br>ecifications ar | Il compensation for furnishing all materials act, and for furnishing all necessary tools, n; also all loss or damage arising out of the n the action of the elements, or from any ountered during the progress of said work, sponsible for the consequences of his own r well and faithfully completing the work in ad all requirements of the Architect and/or the whole thereof, the following sum which |
|   |  |  | DOLLARS (\$)   |

# **ARTICLE II - PAYMENTS**

The Owner agrees, in consideration of the performance of this Contract, to pay the Contractor in the following manner:

- (a) Payments will be made only on the certificate of the Architect.
- (b) Monthly payments shall be made to the Contractor in amounts equal to ninety-five percent (95%) of the estimated value of the work done and the materials furnished and incorporated in the work during

the month preceding the date upon which such value is estimated plus ninety-five percent (95%) of the estimated value of all materials which, on the date of estimation of value, are suitably stored on the site for incorporation into the work; provided that no such monthly payment, or payment of any kind, shall theretofore have been made for any such work done or materials furnished and incorporated or materials suitably stored on the site. The aforesaid estimation of value shall be made by the Architect and noted by him upon the certificates furnished by him pursuant to paragraph (2) of this Article II.

- (c) Upon completion of the work, a sum sufficient to increase the total payments to ninety-five percent (95%) of the Contract Price, less an amount determined by Owner to be adequate to complete any unfurnished part of the work by another Contractor should the work not be completed within a reasonable time established by the Owner.
- (d) The project shall be accepted by the Board of Trustees who shall authorize after acceptance of the project the subsequent filing of the Notice of Completion. The final payment shall be made thirty-five (35) days from the date of recordation of the Notice of Completion, provided that: The Contractor shall furnish satisfactory evidence that all claims for labor and materials have been paid and that no claims shall have been presented to the Owner by any person or persons based upon any act or omission of the Contractor, and no Stop Notices or claims shall have been filed against said work or the property whereon it was done.

NOTE:

No certificates given or payments made on account of any Contract shall constitute an acceptance of any equipment, material or work, which may subsequently be found to be defective.

#### ARTICLE III - TIME OF COMPLETION AND LIQUIDATED DAMAGES

The Contractor agrees to commence the work within fifteen (15) calendar days from the date of the Notice to Proceed and prior to receiving Notice to Proceed shall obtain Owner's approval of Insurance and Bonds as required in the supporting documents. Failure to obtain approval of the required documents, within the allotted time, shall not be cause for extension of the time of construction as set forth hereafter.

The Contractor further agrees to construct and execute all of the work described in said drawings, specifications, proposals, addenda and any and all other requirements, covenants, stipulations and restrictions, within 90 calendar days from and after the date of commencement, said date of commencement being agreed upon as the fifteenth (15th) calendar day following the date of Notice to Proceed or first day of construction on Project site, whichever is first. Architect shall issue actual start date to Contractor and Owner.

Both parties agree that the aforementioned stipulated contract period to be a reasonable time scale for completion of the work and Contractor will provide best endeavors to complete the work within the contract period.

If the Contractor shall neglect, fail or refuse to complete the work within the time herein specified, then the contractor does hereby agree, as a part consideration for awarding of this Contract to pay to the Owner the sum of:

Two Hundred Fifty Dollars - No Cents

DOLLARS (\$250.00)

per day plus such additional costs as may be incurred by the Architect because of such delays, not as a penalty but as liquidated damages for such breach of Contract as hereinafter set forth for each and every day that the Contractor shall be in default after the time stipulated in the Contract for completing the work.

The said amount is fixed and agreed upon by and between the Contractor and the Owner because of the impracticability and extreme difficulty of fixing and ascertaining the actual damages the Owner would in such event sustain, and said amount is agreed to be the amount of damages which the Owner would sustain and said amounts shall be retained from time to time by the Owner from the current periodical estimates.

It is further agreed that time is the essence of each and every portion of this Contract and of the specifications

wherein a definite and certain length of time is fixed for the performance of any act whatsoever; and where under the Contract an additional time is allowed for the completion of any work, the new time limit fixed by such extension shall be of the essence of this Contract. Provided, that the Contractor shall not be charged with liquidated damages or any excess cost when the delay in completion of the work is due.

- (a) To any preference, priority or allocation order duly issued by the Government.
- (b) To unforeseeable cause beyond the control and without the fault or negligence of the Contractor, including but not restricted to acts of God, or of the public enemy, acts of the Owner, acts of another Contractor in the performance of the Contract with the Owner, fires, floods, epidemics, quarantine restrictions, strikes, freight embargoes, unusually severe weather (as defined in Construction Documents), and
- (c) To any delays of subcontractors occasioned by any of the causes specified in subsections (a) and (b) of this Article. Provided, further, that the Contractor shall, within seven (7) days from the beginning of such delay, notify the Owner, in writing, of the causes of delay, who shall ascertain the facts and extent of the delay and notify the Contractor within a reasonable time of its decision in the matter.

#### ARTICLE IV - DRAWINGS AND SPECIFICATIONS

This Contract, the drawings and the specifications have been prepared, and are intended to supplement one another. The drawings and specifications shall be deemed by this reference to be incorporated within this Contract, the drawings shall be deemed by this reference to be incorporated within the specifications, and the specifications shall be deemed by this reference to be incorporated with the drawings. In the event a conflict is found to exist between the drawings and specifications, the Architect shall interpret. In the event that the drawings and specifications, or either of them shall be found to conflict with this Contract, then the more restrictive, better quality and/or greater quantity shall govern. Omissions from this Contract of items of provisions present in the specifications or drawings or either of them shall not be deemed a conflict within the meaning of this Article.

#### ARTICLE V - CHANGES, ETC.

Should the Owner, at any time during the progress of the work desire any alterations, or deviations in, or additions to, or omissions from the Contract or the drawings or specifications, said Owner, or representative thereof, shall be at liberty to order them, in writing, and the same shall in no way affect or make void this Contract, but the amount thereof shall be added to, or deducted from, the amount of the Contract Price aforesaid, as the case may be, by a fair and reasonable valuation. This Contract, subject to the provisions of Article II (a) hereof, shall be deemed completed when the work is finished in accordance with the original drawings and specifications, as amended by such changes, whatever may be the nature or extent thereof.

No such changes, whatever may be the nature, or modification shall release or exonerate any surety or sureties upon any guarantee or bond given in connection with this Contract, if required.

# ARTICLE VI - RULE OF PRACTICE

The rule of practice to be observed in this Contract shall be that upon the demand of either the Owner or the Contractor, the character or valuation of any and all changes, omissions, or extra work shall be agreed upon and fixed in writing, signed by the Owner and the Contractor, prior to execution.

#### ARTICLE VII - ACCEPTANCE OF THE WORK

The payment of the progress payments by the Owner shall not be construed as an absolute acceptance of the work done up to the time of such payments, except as to such matters as are open and obvious, but the

entire work, and at the time when it shall be claimed by the Contractor that the Contract and work is completed. Liability under the bonds is to continue for one (1) year from the date of notice of completion and bonds will not be released until such date. In the event any warranties and guarantees exceed the one (1) year, liability under the bonds shall continue until said warranties and guarantees expire.

# ARTICLE VIII - FAILURE TO PROVIDE WORKMEN AND MATERIALS

If the Contractor at any time during the progress of the work should refuse or neglect, without the fault of the Owner, to supply sufficient amount of materials or enough workmen to complete the Contract within the time herein set forth, due allowance being made for the contingencies provided for herein, for a period of more than seven (7) days after having been notified by the Owner in writing to furnish the same, the Owner shall have the power to furnish and provide said materials and/or workmen to finish the said work, and the reasonable expense thereof shall be deducted from the amount of the Contract Price.

# **ARTICLE IX - PENALTIES**

This Contractor shall forfeit, as a penalty to the said Owner, the sum of one hundred dollars (\$100.00) for each laborer, workman, or mechanic employed in the execution of this Contract, or any sub-contractor under him, for each calendar day during which such laborer, workman and/or mechanic is required or permitted to labor in violation of the State of California prevailing wage requirements, and said Owner, when making payments of money due under this Contract, shall withhold and retain there from all sums and amounts which have been forfeited pursuant to the herein said stipulation.

# ARTICLE X - INSURANCE AND BONDS

Insurances and bonds, as set forth in the supporting contract documents, shall be maintained in effect during the period of this Contract.

# **ARTICLE XI - RELATION TO BID PROPOSALS**

Be it further stipulated and agreed that said Owner does promise and agree to employ the said Contractor to provide the materials and do the work according to the terms and conditions herein contained and referred to, for the prices aforesaid, and herein contracts to pay the same at the time, in the manner and upon the conditions set forth above; and the said parties for themselves, their heirs, executors, administrators, successors, and assigns, do hereby agree to the full performance of the covenants herein contained.

It is further agreed by and between the parties hereto that should there be any conflict between the terms of this instrument and the bid proposal of said Contract, then this instrument shall control and nothing herein shall be considered as an acceptance of the said terms of said proposal conflicting herewith.

#### ARTICLE XII - ASBESTOS/PCB-CONTAINING PRODUCTS & LEAD PLUMBING ITEMS

Contractor agrees that lead plumbing domestic water items, asbestos, PCB-containing products or materials will not be used or substituted in performing work under the Agreement.

At the completion of work under the Agreement, Contractor will certify in writing to the Owner that to the best of Contractor's knowledge, no lead plumbing domestic water items, asbestos/PCB-containing products or materials were used or substituted in performing work under the Agreement.

# ARTICLE XIII - COMPLIANCE WITH STORM WATER PREVENTION AND AIR POLLUTION CONTROL RULES

Contractor shall comply with all storm water prevention and air pollution control rules, regulations, ordinances, and statutes which apply to any work performed pursuant to the Contract, including any storm water prevention and pollution control rules, regulations, ordinances, and statutes specified in Section 11017 of the Government Code, as well as local requirements, County, City and local Storm Water Prevention and Air Pollution Control Districts. Contractor shall require all subcontractors to abide by these items.

#### ARTICLE XIV - ANTI-DISCRIMINATION IN EMPLOYMENT

In connection with the performance of work under this Contract, the Contractor agrees (as prescribed in Chapter 6 of Division 3 of Title II of the Government Code of the Sate of California, commencing at Section 12900 and by Labor Code Section 1735) not to discriminate against any employee or applicant for employment because of race, religious creed, color, national origin, ancestry, physical disability, mental disability, medical condition, marital status, gender, or age. The aforesaid provisions shall include, but are not limited, the following: hiring, upgrading, demotion or transfer, recruitment or recruitment advertising, layoff or termination, rates of pay or other forms compensation, and selection for training, including apprenticeship. The Contractor agrees to post hereafter in conspicuous places, available for employees and applicants for employment, notices setting forth the provisions of this anti-discrimination clause. The Contractor further agrees to insert the foregoing provisions in all subcontracts entered into hereunder, except subcontracts for standard commercial supplies of raw materials.

#### ARTICLE XV - CONTRACTOR - EMPLOYEE REQUIREMENTS

By signing this Contract, the Contractor certifies he is aware of the provisions of section 3700 of the Labor Code which require every employer to be insured against liability for workmen's compensation or to undertake self-insurance in accordance with the provisions of that code, and that he will comply with such provisions before commencing the performance of the work of this Contract.

In accordance with the provisions of Section 3700 of the Labor Code, every contractor will be required to secure payment of compensation to his employees.

The Contractor and Subcontractors under him shall comply with the provisions of Division 2, Part 7, Chapter 1, Article 2, Sections 1770-1780 with particular reference to the employment and use of apprentices and other provisions that require him to make travel and subsistence payments to each workman needed to execute the work, as such collective bargaining agreements filed in accordance with the Labor Code, and to pay not less than the minimum per diem wages as determined by the Director of the Department of Industrial Relations, on file in the principal office of the Owner.

Special attention is directed to Sections 1777.5, 1777.6 and 1777.7 of the California Labor Code and Title 8, California Code of Regulations Section 200 et. seq. Each contractor and/or subcontractor must, prior to commencement of the public works contract, contact the Division of Apprenticeship Standards, 455 Golden Gate Avenue, San Francisco, or one of its branch offices regarding apprentices and specifically the required ratio thereunder. Responsibility for compliance with this section lies with the prime Contractor. During the performance of this Contract, the Contractor will not discriminate against any employee or applicant for employment because of race, religion, color, sex or national origin.

#### **ARTICLE XVI - NOTICES**

All notices herein provided to be given or which may be given by either party to the other shall be deemed to have been fully given when made in writing and deposited with the United States Postal Service, Registered or Certified, and postage prepaid and addressed as follows:

To Owner: WEST KERN COMMUNITY COLLEGE DISTRICT

Attention: Leslie Minor, Ph.D.

29 Cougar Court Taft, CA 93268 (661) 763-7700

#### To Contractor:

The address to which the notices shall or may be mailed as aforesaid to either party shall or may be changed by written notice given by such party to the other, as hereinbefore provided, but nothing herein contained shall preclude the giving of any such notice by personal service.

IN WITNESS WHEREOF, the parties to these presents have hereunto set their hands the year and date first herein written.

|                        | WEST KERN COMMUNITY COLLEGE DISTRICT        |  |  |
|------------------------|---|--|--|
| CONTRACTOR             | OWNER                                       |  |  |
| ByAuthorized Signature | By Authorized Signature Leslie Minor, Ph.D. |  |  |
| Date                   | Date  |  |  |

END OF SECTION 006003

#### PART 1 - GENERAL

# 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of Contract, including General and Supplementary Conditions and other Division 0 & 1 Specification Sections, apply to this Section.

# 1.2 SUMMARY

- A. Section includes, but not limited to the following:
  - 1. Project information.
  - 2. Work covered by Contract Documents.
  - 3. Work by Owner.
  - 4. Work under separate contracts.
  - 5. Future work.
  - 6. Purchase contracts.
  - 7. Owner-furnished products.
  - 8. Contractor-furnished, Owner-installed products.
  - 9. Access to site.
  - 10. Coordination with occupants.
  - 11. Work restrictions.
  - 12. Specification and drawing conventions.
  - 13. Miscellaneous provisions.

# B. Related Requirements:

- 1. Division 1 Section 012100 "Allowances": for purchase contracts.
- 2. Division 1 Section 015000 "Temporary Facilities" for limitations and procedures governing temporary use of Owner's facilities.

# 1.3 PROJECT INFORMATION

A. Project Identification:

TC – G-8 ALLIED HEALTH LAB REMODEL Taft College 29 Cougar Court Taft, CA 93268

B. Owner:

West Kern Community College District Justin Madding 29 Cougar Court Taft, CA 93268

C. Architect:

AP Architects
Jose Vargas – Project Manager

SUMMARY OF WORK 011000 - 1

3434 Truxtun Avenue Suite 240, Bakersfield CA 93301

D. Architect's Consultants: The Architect has retained the following design professionals who have prepared designated portions of the Contract Documents:

CHEN Engineers, Inc Richard Chen 729 W. Enterprise Ave. Clovis, CA 93619

#### 1.4 WORK COVERED BY CONTRACT DOCUMENTS

- A. The Work consists of the construction of a remodel of existing modular G8 for nursing Lab use demolition and building improvements are required as part of the work.
- B. Type of Contract:
  - 1. Project will be constructed under a single prime contract.

#### 1.5 WORK SEQUENCE

A. The Work will be conducted in one phase to provide the least possible interference to the activities of the Owner's personnel and to permit an orderly transfer of personnel and equipment to the new facilities. Project completion is scheduled for specific number calendar days, (refer to Bid Proposal). Contractor shall review scope of work, and provide manpower, resources, etc., as required to complete project on or before the date required for project completion. Contractor shall allow in Proposal weekend workers, shifts of workers and additional productivity not limited to workers, materials, temporary facilities and equipment as required to meet project schedule with limited access times as indicated herein.

# 1.6 CONTRACTOR USE OF PREMISES

- A. General: Limit use of the premises to construction activities in areas indicated; allow for Owner occupancy and use by the public.
  - 1. Confine operations to areas within Contract limits indicated. Portions of the site beyond areas in which construction operations are indicated are not to be disturbed.
  - Keep driveways and entrances serving the premises clear and available to the Owner and the Owner's employees at all times. Do not use these areas for parking or storage of materials. Schedule deliveries to minimize space and time requirements for storage of materials and equipment on site.
  - 3. When performing new construction on existing sites, become informed and take into specific account the maturity of the students on the site, and perform work which may interfere with educational facility routine before or after facility hours; enclose the work area with a substantial barricade and arrange work to cause a minimum of inconvenience and danger to students and staff in their regular facility activities.
- B. Use of the Existing Building: Maintain the existing buildings in a weathertight condition throughout the construction period. Repair damage caused by construction operations. Take all precautions necessary to protect the building and its occupants during the construction period.

SUMMARY OF WORK 011000 - 2

# 1.8 OWNER OCCUPANCY

- A. Full Owner Occupancy: The Owner will occupy the site and existing building during the entire construction period. Cooperate with the Owner during construction operations to minimize conflicts and facilitate Owner usage. Perform the Work so as not to interfere with the Owner's operations.
- B. All work shall be complete and approved prior to occupancy not limited to the following:
  - 1. No portion of building may be occupied requiring impaired Required Fire Detection System unless system is installed and approved.
  - 2. All completed work shall be in compliance with CBC 901.5 and CFC 901.5.1 related to acceptance tests.

#### 1.9 WORK RESTRICTIONS:

- A. Work Restrictions, General: Comply with restrictions on construction operations.
  - 1. Comply with limitations on use of public streets and other requirements of authorities having jurisdiction.
  - 2. Smoking is not permitted within the building or within 25 feet (8 m) of entrances, operable windows, or outdoor air intakes.
  - 3. Use of controlled substances on the Project site is not permitted.
- B. On-Site Work Hours: Limit work in the existing building to normal business working hours of 6:00 a.m. to 5:00 p.m., Monday through Friday, except as otherwise indicated.
- C. Construction work that generates noise beyond 90db that will disturb adjacent areas shall be scheduled around class schedule and office hours of occupied rooms within 125 feet of work to be done. This work may have to be done during after hours, evenings and Contractor shall verify class schedules when work will generate noise beyond 90db.
- C. Deliver materials to the building area over the route designated by the facility Maintenance and Operations department. Times of deliveries shall coincide not to be done during 5 minutes before class change time between classes and 5 minutes after if said deliveries path is thru any area students will occupy during class change times. If a delivery is overlapping class change times, cease work, provide temporary barricades and resume 5 minutes after classes resume.

PART 2 - PRODUCTS (Not applicable).

PART 3 - EXECUTION (Not applicable).

END OF SECTION 011000

SUMMARY OF WORK 011000 - 3

#### PART 1 - GENERAL

# 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of Contract, including General and Supplementary Conditions and other Division 0 &1 Specification Sections, apply to this Section.

#### 1.2 SUMMARY

- A. This Section specifies administrative and procedural requirements governing handling and processing allowances.
  - Selected materials, services, equipment, related items and in some cases, their installation and related/non-related work are shown and specified in the Contract Documents by allowances herein. Allowances have been established in lieu of additional requirements and to defer selection of actual materials, miscellaneous additional work scope and equipment to a later date when additional information is available for evaluation.
  - 2. Special allowances have been established for unforeseen conditions, latent conditions and related item to be authorized by the Architect for use.
- B. Types of allowances required include the following:
  - 1. Lump sum allowances in Base Bid.

# C. Related Requirements:

- 1. Division 1 Section "Modification Procedures" for procedures for submitting and handling Change Orders.
- 2. Division 1 Section "Summary of Work" for additional requirements on purchase contracts.

#### 1.3 SUBMITTALS

- A. Submit proposals for purchase of products or systems included in allowances, in the form specified for Change Orders.
- B. Submit invoices or delivery slips to indicate actual quantities of materials delivered to the site for use in fulfillment of each allowance.

# 1.4 ALLOWANCES

- A. Use the allowance only as directed for the Owner's purposes, and only by Supplementary Instructions, which designate amounts to be charged to the allowance.
  - The direct costs for products or equipment ordered by the Owner under the lump sum allowances, including delivery, installation, taxes, and similar costs are part of the allowance. Vendor shall provide insurance as required by the Owner. Contractor shall agree to accept insurance required by Owner for vendor for allowance item. If the contractor requires any special insurance, additional requirements and or bonding of any allowance vendor, contractor shall allow for this cost in his base bid or alternate bid if allowance is tied to an alternate.

ALLOWANCES 012100 - 1

- a. In the event the work under allowance cannot be completed during the duration of the project as prescribed under "Project Summary", contractor may elect to request additional extended overhead. Extended overhead will be determined by actual costs incurred by contractor specific to this project and verified by project schedule.
- 2. Supplementary Instructions (SI) authorizing use of funds from the lump sum allowance will not include the Contractor's related costs and reasonable overhead, supervision, profit margins and other related costs as these costs are already in the contractor's proposal/bid.
- 3. If any individual allowance contains surplus funds or contains deficient funds, Architect may transfer funds between allowances as necessary.
- 4. At Project closeout, credit unused amounts remaining in the allowance to Owner by Negative Change Order amount for unused amounts.
- 5. The contractor shall include in his base bid all overhead, profit, supervision, bonds, insurance and all other indirect costs for allowance items. None of these items will be added to lump sum and miscellaneous allowance as it is used by the Owner and directed by the architect. In the event the allowance is required in an Alternate, the contractor shall include in his alternate bid all overhead, profit, supervision, bonds, insurance and all other indirect costs for allowance items specific to that alternate.

PART 2 - PRODUCTS (Not Applicable)

PART 3 - EXECUTION

# 3.1 INSPECTION

A. Inspect products covered by an allowance promptly upon delivery for damage or defects.

#### 3.2 PREPARATION

- A. Coordinate materials and their installation for each allowance with related materials and installations to ensure that each allowance item is completely integrated and interfaced with related construction activities.
- B. Coordinate scheduling of Owner selected Vendors. Obtain availability schedules from Vendors early in project to coordinate timing of special milestones and products necessary for implementation into overall construction activities.

#### 3.3 SCHEDULE OF BASE BID ALLOWANCES

A. Allowance No. 1: Include a lump sum allowance of \$10,000.00 for use upon the Architect's instructions.

END OF SECTION 012100

ALLOWANCES 012100 - 2

#### PART 1 - GENERAL

# 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of Contract, including General and Supplementary Conditions and other Division 0 & 1 Specification sections, apply to this section.

# 1.2 SUMMARY

A. This section specifies administrative and procedural requirements for handling and processing Contract modifications.

#### 1.3 MINOR CHANGES IN THE WORK and REQUEST FOR INFORMATION (RFI)

- A. Supplemental instructions authorizing minor changes in the Work, not involving an adjustment to the Contract Sum or Contract Time, will be issued by the Architect. The Architect may issue written Supplemental Instructions, (SI), which interpret the Contract Documents or which order minor changes in the work without change in Contract sum or Contract time. The Contractor shall carry out such Supplemental Instructions promptly.
  - Unless otherwise noted, SI, (Supplemental Instructions) does not warrant a cost or time impact to the Contract cost or time. If Contractor does not agree, Contractor has 10 calendar days from date of receipt of SI to file a claim for adjustment in writing to the architect.
  - 2. The Architect shall use SI's for written order for usage of allowance funds for project if any allowances are indicated.
- B. Contractor shall be able to ask valid questions concerning items required to construct project. This shall be done by the following methods in order as follows: (1) Contractor to review plans and determine if information is prescribed therein; (2) Contractor to review question with Project Inspector and determine if information is indicated, intended and/or prescribed in construction documents; (3) Contractor place an informal inquiry with architect and discuss question: and if no answer is determined then (4) Contractor shall prepare a Request for Information (RFI) and deliver to architect for determination of answer and or direction from architect as prescribed herein.
  - Contractor shall submit Request for Information (RFI) on enclosed form at end of this Section.
  - Contractor shall attach to RFI what they consider to be answer to Request for Information.
     Failure to provide this information shall be grounds for Architect to Request for Clarification.
  - 3. An RFI is defined as a request for information for information that cannot be found in the construction documents and related submittals. Items not considered RFI's are as follows:
    - a. A request for a proposed alternative materials, products or colors.
    - b. Substitutions.
    - c. Coordination of Contractor changed/initiated field conditions.

### 1.4 CHANGES IN WORK

A. The Owner, without invalidating the Contract may make changes by altering, adding to, or MODIFICATION PROCEDURES 012600 - 1

deducting from the work, the Contract sum and construction duration being adjusted accordingly. All such work shall be executed under the conditions of the original contract. Unless so authorized, the Contractor shall not deviate from nor alter the work as shown on the drawings or specifications. Additional work may be added to project by using project allowances as prescribed herein.

- 1. In the event additional work is added to the project via allowances the Contractor shall provide an analysis of the schedule impact if any. If additional work is shown to impact the construction schedule the Contractor shall be entitled additional time as agreed to by architect. If additional work is shown by schedule analysis to have no impact, no additional construction duration will added to project.
- 2. Any changes in construction duration shall be documented by a Change Order to Contract.
- B. If Contractor should claim that any instruction, request, drawing, specification, action, condition, omission, default, or other situation constitutes a change, extra work, or otherwise obligates Owner to pay additional compensation to Contractor or to grant an extension of time, or constitutes a waiver of any provision in the Contract Documents, Contractor shall notify Owner in writing of such claim within ten calendar days from the date Contractor has actual or constructive notice of the factual basis supporting the claim. The notice shall state the factual basis for the claim and cite in detail the Contract Documents (including plans and specifications) upon which the claim is based. Contractor's failure to notify Owner and Architect within the ten-day period shall be deemed a waiver and relinquishment of such a claim. If the notice is given within the specified time, the procedure for its consideration shall be as stated in these General Conditions. In the event of failure to agree, the matter shall be treated as a claim following the claims procedures in the Contract Documents.
- C. No change shall be made without such authorization, signed by the Owner, and countersigned by the Architect, or signed by the Architect and stating that the Owner has authorized such changes.
  - 1. Refer to Supplementary Conditions for Construction Change Directive (CCD) procedures.
  - 2. Refer to Supplementary Conditions for Supplemental Instructions (SI) procedures.
- D. Any changes processed by the Contractor or any work performed not in conformance with these plans and specifications which requires extra drawing, specifications, calculations, inspections and any other work by the Architect and/or Engineers shall be paid for by the Contractor. Payment shall be made to the Architect at current hourly rate on file due and payable upon presentation of invoice.

# 1.5 CHANGE ORDER PROCESS-(OWNER AND CONTRACTOR INITIATED PROPOSAL REQUEST, AND ALLOWANCES.)

- A. Owner-Initiated Proposal Requests: Proposed changes in the Work that will require adjustment to the Contract Sum or Contract Time will be issued by the Architect, with a detailed description of the proposed change and supplemental or revised Drawings and Specifications, if necessary.
  - 1. Request for Proposal requests (RFP),issued by the Architect are for information only. Do not consider them an instruction either to stop work in progress, or to execute the proposed change.
  - 2. Unless otherwise indicated in the proposal request, within 20 calendar days of receipt of the proposal request, submit to the Architect for the Owner's review an estimate of cost necessary to execute the proposed change as well as Construction Duration impact.
    - a. Include a list of quantities of products to be purchased and unit costs, along with the total amount of purchases to be made. Where requested, furnish survey data to substantiate quantities.

- Indicate applicable taxes, delivery charges, equipment rental, and amounts of trade discounts.
- c. Include a statement indicating the effect the proposed change in the Work will have on the Contract Duration.
- d. Before Contractor is authorized to proceed with extra work or changes on the basis set forth above, the Owner and the Contractor shall be in complete agreement on what the term "costs" shall include and the amount of overhead and profit the Contractor is to charge.
- e. All unit prices, whether set forth in the Contract or subsequently agreed upon, shall include overhead, profit, supervision, increased premium on all Bonds, increased premium on all insurances and other indirect cost for all tiers of contractors and related material men unless said items are being paid thru an allowance where overhead, profit, supervision, bonds, insurance and related items are included in contractor's base bid.
- f. If there has been no response within 20 calendar days to an Architect's Request for Proposal, the Architect may direct the change to be done Time and Material. Under no circumstance may the contractor increase cost or increase schedule time due to Owner not receiving proposal timely.
- B. Contractor-Initiated Change Order Proposal Requests: When latent or other unforeseen conditions require modifications to the Contract, the Contractor may propose changes by submitting a request for a change to the Architect.
  - 1. Include a statement outlining the reasons for the change and the effect of the change on the Work. Provide a complete description of the proposed change. Indicate the effect of the proposed change on the Contract Sum and Contract Time.
  - 2. Include a list of quantities of products to be purchased and unit costs along with the total amount of purchases to be made. Where requested, furnish survey data to substantiate quantities.
  - 3. Indicate applicable taxes, delivery charges, equipment rental, and amounts of trade discounts.
  - 4. Comply with requirements in Supplementary Conditions Article 15 Substitutions if the proposed change in the Work requires the substitution of one product or system for a product or system specified.
  - 5. Claim submitted to the Architect for extensions of time and extra cost shall be made on forms carrying Contractor's letterhead and shall contain a complete breakdown of all costs and extension of Surety Bonds and Insurance impacts.
  - 6. Before Contractor is authorized to proceed with extra work or changes on the basis set forth above, the Owner and the Contractor shall be in complete agreement on what the term "costs" shall include and the amount of overhead and profit the Contractor is to charge as any Contract Duration impacts.
- C. For changes that increase or decrease the contract price, or being paid by allowance item, the Contractor shall include the following amounts for overhead and profit:
  - 1. Contractor's overhead and profit on the cost of work excluding work by Contractor shall be a total sum not exceeding ten percent (10%) of cost of such work. (See below for allowances)
  - 2. Contractor's overhead and profit on the cost of work performed by contractor without subcontractor shall be a total sum not exceeding ten percent (10%) of the cost of labor, materials, rentals, etc. (See below for allowances)
  - 3. Subcontractor's overhead and profit on the cost of work performed by subcontractor shall be a total sum not exceeding ten percent (10%) of the cost of labor, materials, rentals, etc.
  - 4. Subcontractor's overhead and profit on the cost of work performed by sub-contractors (one lower tier) shall be a total sum not exceeding five percent (5%) of such work.
  - 5. Allowances: The contractor shall include in his base bid all overhead, profit, supervision, bonds, insurance and all other indirect costs for allowance items. None of these items will

be added to lump sum and miscellaneous allowance as it is used by the Owner and directed by the Architect. In the event the allowance is required in an Alternate, the Contractor shall include in his alternate bid all overhead, profit, supervision, bonds, insurance and all other indirect costs for allowance items specific to that alternate.

- D. Time and Material basis as changes in the work. For changes that increase the contract price and work is authorized based upon the cost of labor, material, equipment and subcontract prices, plus a percentage for overhead and profit the following requirements shall apply. In the event the costs for changes in the work are not agreed to by the Architect and Contractor the work may be authorized to move forward on a time and material basis. If a Time and Material basis is used and scope of work is being paid thru an allowance overhead and profit are to be included in base bid and alternates where occur.
  - 1. Daily reports by Contractor, as follows:
    - a. General. At the close of each working day, Contractor shall submit a daily report to the Architect and the Project Inspector on forms approved by Owner, together with applicable delivery tickets listing all labor, materials, and equipment involved for that day, and for other services and expenditures, when authorized, concerning extra work items. An attempt shall be made to reconcile the report daily, and it shall be signed by the Architect and Contractor. In the event of disagreement, pertinent notes shall be entered by each party to explain points which cannot be resolved immediately. Each party shall retain a signed copy of the report. Reports by subcontractors or others shall be submitted through Contractor.
    - b. Labor. The report shall show names of workers, classifications, and hours worked and hourly rate. Project supervision expenses, including for foremen and above, are not allowed. (iii) Materials. The report shall describe and list quantities of materials used and unit cost.
    - c. Equipment. The report shall show the type of equipment, size, identification number, and hours of operation, including loading and transportation, if applicable, and hourly/daily costs.
    - d. Other Services and Expenditures. Other services and expenditures shall be described in such detail as Owner may require.

#### 2. Basis for Establishing Costs

- a Labor. The costs of labor will be the actual cost for wages prevailing locally for each craft classification or type of worker at the time the extra work is done, plus employer payments of payroll taxes and insurance, health and welfare, pension, vacation, apprenticeship funds, and other direct costs resulting from federal, state, or local laws, as well as assessments or benefits required by lawful collective bargaining agreements. The use of labor classifications which would increase the extra work cost will not be permitted unless Contractor establishes the necessity for such additional costs. Labor costs for equipment operators and helpers shall be reported only when such costs are not included in the invoice for equipment rental.
- b. Materials. The cost of materials reported shall be at invoice or lowest current price at which such materials are locally available and delivered to the work site in the quantities involved, plus sales tax, freight, and delivery. Owner reserves the right to approve materials and sources of supply, or to supply materials to Contractor if necessary for the progress of the work. No markup shall be applied to any material provided by the Owner.
- c. Tool and Equipment Rental. No payment will be made for the use of tools which have a replacement value of \$100 or less or where an invoice is not provided. Regardless of ownership, the rates to be used in determining equipment rental costs shall not exceed listed rates prevailing locally at equipment rental sources or distributors at the time the work is performed. The rental rates paid shall include the

cost of fuel, oil, lubrication, supplies, small tools, necessary attachments, repairs and maintenance of any kind, depreciation, storage, insurance, and all incidentals. Necessary loading and transportation costs for equipment used on the extra work shall be included. If equipment is used intermittently, and when not in use could be returned to its rental source at less expense to Owner than holding it at the work site, it shall be returned, unless Contractor elects to keep it at the work site at no expense to Owner. All equipment shall be acceptable to the Architect in good working condition, and suitable for the purpose for which it is to be used. Manufacturer's ratings and manufacturer's approved modifications shall be used to classify equipment and it shall be powered by a unit of at least the minimum rating recommended by the manufacturer.

- d. Other Items. Owner may authorize other items which may be required on the extra work. These items include labor, services, material, and equipment which are different in their nature from those required by the work and which are of a type not ordinarily available from Contractor any of the Subcontractors. Detailed invoices covering all such items shall be submitted with the request for payment. (v) Invoices. Vendors' invoices for material, equipment rental, and other expenditures shall be submitted with the request for payment. If the request for payment is not substantiated by invoices or other documentation, Owner may establish the cost of the item involved at the lowest price which was current at the time of the report.
- Daily worker time sheets shall be approved by the Project Inspector as well as copies of all
  materials invoices delivered to project site for this specific change. Time sheets and
  copies of all material costs shall be provided with pay request for this specific change with
  daily approvals by Project Inspector
- E. The following form shall be used by Contractor as applicable to communicate proposed additions and deductions to the Contract Documents and use of allowances:

| # | Description  | Extra | Credit |
|---|--|-------|--------|
| а | Materials (attached itemized quantity and unit costs including any sales tax.  |       |        |
| b | Labor (attached itemized hours and rates)  | _     |        |
| С | \$ Subtotal  |       |        |
| d | Subcontractor's overhead and profit on the cost of work performed by sub-contractors (one lower tier) shall be a total sum not exceeding five percent (5%) of such work.               |       |        |
| е | \$ Subtotal  |       |        |
| f | Subcontractor's overhead and profit on the cost of work performed by subcontractor shall be a total sum not exceeding ten percent (10%) of the cost of labor, materials, rentals, etc. |       |        |
| g | \$ Subtotal  |       |        |
| h | Contractor's overhead and profit on the cost of work excluding work by Contractor shall be a total sum not exceeding ten percent (10%) of cost of such work.                           |       |        |
|   | (Refer to project manual "Allowances" where overhead and profit are included in base bid and alternates where occur)   |       |        |

|        |          | ·  |               |
|--------|----------|--|---------------|
|        | İ        | Contractor's overhead and profit on the cost of work performed by contractor without subcontractor shall be a total sum not exceeding ten percent (10%) of the cost of labor, materials, rentals, etc.   |               |
|        |          | (Refer to project manual "Allowances" where overhead and profit are included in base bid and alternates where occur)   |               |
|        | j        | Bond Premium (Submit invoice from Bonding provider) .  |               |
|        |          | (Refer to project manual "Allowances" where bond premium are included in base bid and alternates where occur)  |               |
|        | k        | Insurance (Submit invoice from Insurance provider)   |               |
|        |          | (Refer to project manual "Allowances" where bonds and insurance are included in base bid and alternates where occur)   |               |
|        | 1        | \$ Total   |               |
|        | m        | Number of additional days time extension requested due   |               |
|        |          | to this time and material change. (Submit as built critical  |               |
|        |          | path schedule validating time extension for review and approval)   |               |
|        |          | Total Days   |               |
|        |          | Subcontractor's labor, material, overhead and profit shall   |               |
|        |          | be submitted with documentation in original form as submitted to General Contractor.   |               |
|        |          | (Refer to project manual "Allowances" where overhead and profit are included in base bid and alternates where occur)   |               |
| F.     | the me   | pressly understood that the value of such extra work or changes as determine thods herein expressly includes any and all of the contractors' costs and expand indirect, resulting from delays or additional time required on the project coelerated work to avoid delays to the project. | expense, both |
| PART 2 | - PRODUC | CTS (Not Applicable)   |               |
| PART 3 | - EXECUT | TION (Not Applicable)  |               |
| END OF | SECTION  | N 012600   |               |



# CONSTRUCTION REQUEST FOR INFORMATION

| DATE:                                      |  |   |  |  |  | RFI#_   |   |
|--|--|---|--|--|--|---|---|
| ATTN:                                      | Jose Vargas  |   | PROJECT:   | TC – G                                   | -8 ALLIED HEALTI   | H LAB RE  | EMODEL  |
|  | 3434 Truxtun A   | venue Ste 240   | <del>_</del>   | _  |  |   |   |
|  | Bakersfield CA   | 93301   | <del>_</del>   |  |  |   |   |
|  |  |   | PROJECT#:  | 430-00                                   | 83   |   |   |
|  | ,  |   |  |  |  |   |   |
| Subject                                    | :  |   |  |  |  | Section #   |   |
| ,  |  |   |  |  |  | Sheet #:  |   |
| Descrip                                    | otion:   |   |  |  |  |   |   |
| ·  |  |   |  |  |  |   | _   |
|  |  |   |  |  |  |   | _   |
|  |  |   |  |  |  |   |   |
|  |  |   |  |  |  |   |   |
|  |  |   |  |  |  |   |   |
|  |  |   |  |  |  |   |   |
|  |  |   |  |  |  |   |   |
|  |  |   |  |  |  |   |   |
|  |  |   |  |  |  |   |   |
|  |  |   |  |  |  |   |   |
|  |  |   |  |  |  |   |   |
|  |  |   |  |  |  |   |   |
|  | Clarification  | ☐ Unfores   | een Condition  |  | Owner Change   |   | Others  |
|  |  |   |  |  |  |   |   |
|  | esting Sponsor: _  |   |  |  | proved by:   |   |   |
| Totalı                                     | number of pages:_  |   |  | C  | ontractor:   |   |   |
|  | I has been review  | ed with Project   | t Inspector withou   | ıt resoluti                              | ion  |   |   |
| 1 (1                                       | Thas been review   | rea with Frojec   | t mapector withou  | 1030141                                  |  |   |   |
| be recoreques made Archite time at in Cons | ognized on account t for clarification of within 10 calendar act's office of signed current hourly rates struction Documents | t of failure of Arc<br>Request for Info<br>days after occu<br>original claim so<br>s prevailing to re<br>s. | chitect to furnish sommation constitute<br>arrence of the evential constitute date | uch interpes responsint giving received. | respond to Request for<br>retations within that<br>se by Architect. Cla<br>rise to such claim ir<br>Architect shall be r<br>ation that are found t | period. P<br>ims for ac<br>n writing.<br>eimbursed<br>o be subs | artial response or<br>ljustment shall be<br>Date received at<br>by Contractor for<br>tantially answered |
| ARCH                                       | IITECT'S RESI  | PONSE:  |  |  | RFI not valid  |   | RFI valid   |
|  |  |   |  |  | Issued RFI Resp  | onse #:   |   |

#### PART 1 - GENERAL

#### 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and other Division 0 & 1 Specification Sections, apply to this Section.

#### 1.2 SUMMARY

- A. This Section specifies administrative and procedural requirements for project meetings including but not limited to:
  - 1. Pre-Construction Conference.
  - 2. Progress Meetings.
  - Project Coordination Meetings.
- B. Related sections include the following construction schedules are specified in another Division-1 Section.
  - Division 1 Section "Submittals" for construction schedules and related items.

# 1.3 PRE-CONSTRUCTION CONFERENCE

- A. Schedule a pre-construction conference and organizational meeting at the Project site or other convenient location no later than 15 days after Notice to Proceed and prior to commencement of construction activities. Conduct the meeting to review responsibilities and personnel assignments.
- B. Attendees: The Owner, Architect, the Contractor and its superintendent, major subcontractors, manufacturers, suppliers and other concerned parties shall each be represented at the conference by persons familiar with and authorized to conclude matters relating to the Work.
- C. Agenda: Discuss items of significance that could affect progress including such topics as:
  - 1. Tentative construction schedule.
  - 2. Critical Work sequencing.
  - Designation of responsible personnel.
  - 4. Procedures for processing field decisions and Change Orders.
  - 5. Procedures for processing Applications for Payment.
  - 6. Distribution of Contract Documents.
  - 7. Submittal of Shop Drawings, Product Data and Samples.
  - 8. Preparation of record documents.
  - 9. Use of the premises.
  - 10. Office, Work and storage areas.
  - 11. Equipment deliveries and priorities.
  - 12. Safety procedures.
  - 13. First aid.
  - 14. Security.
  - 15. Housekeeping.
  - 16. Working hours.

PROJECT MEETINGS 013100 - 1

17. TRPA compliance.

# 1.4 PROGRESS MEETINGS

- A. Conduct progress meetings at the Project site at periodic scheduled intervals. Coordinate schedules with the Owner and Architect of proposed meeting dates in advance of meetings. Discuss at Pre-construction meeting. Coordinate dates of meetings with preparation of the payment request such Architect and Project Inspector can discuss any issue with Contractor on project site.
- B. Attendees: In addition to representatives of the Owner and Architect, each subcontractor, supplier or other entity concerned with current progress or involved in planning, coordination or performance of future activities shall be represented at these meetings by persons familiar with the Project and authorized to conclude matters relating to progress.
- C. Agenda: Review and correct or approve minutes of the previous progress meeting. Review other items of significance that could affect progress. Include topics for discussion as appropriate to the current status of the Project.
  - Contractor's Construction Schedule: Review progress since the last meeting. Determine
    where each activity is in relation to the Contractor's Construction Schedule, whether on
    time or ahead or behind schedule. Determine how construction behind schedule will be
    expedited; secure commitments from parties involved to do so. Discuss whether schedule
    revisions are required to ensure that current and subsequent activities will be completed
    within the Contract Time.
  - 2. Review the present and future needs of each entity present, including such items as:
    - a. Interface requirements.
    - b. Time.
    - c. Sequences.
    - d. Deliveries.
    - e. Off-site fabrication problems.
    - f. Access.
    - g. Site utilization.
    - h. Temporary facilities and services.
    - i. Hours of Work.
    - j. Hazards and risks.
    - k. Housekeeping.
    - I. Quality and Work standards.
    - m. Safety of Students and Staff.
    - n. Documentation of information for payment requests.
    - o. HCP Impacts on project and work schedules.
- D. Reporting: No later than 3 business days after each progress meeting date, distribute copies of minutes of the meeting to each party present and to other parties who should have been present. Include a brief summary, in narrative form, of progress since the previous meeting and report.
  - 1. Schedule Updating: Revise the construction schedule after each progress meeting where revisions to the schedule have been made or recognized. Issue the revised schedule concurrently with the report of each meeting.
  - 2. Send Architect, Construction Manager, Project Inspector, and Owner digital files of meeting report and revised schedules.
  - 3. Architect shall provide a copy of field report made during visit to Contractor within 10 days of site visit.

PROJECT MEETINGS 013100 - 2

#### 1.5 PROJECT COORDINATION MEETINGS

- A. Conduct coordination meetings at the site before each construction activity that requires coordination with other construction. The various Contractors involved in or affected by the installation, and its coordination or integration with other materials and installations shall attend the meeting. Advise the Architect of scheduled meeting dates.
  - 1. Coordination Meetings Schedule (Minimum):
    - a. Ceiling/Attic Contractors: The following Contractors shall attend and coordinate construction activities and space requirements:
      - 1) Mechanical Systems.
      - 2) Plumbing Systems.
      - 3) Electrical Systems.
      - 4) Ceiling Systems.
      - 5) Structural Systems.
      - 6) Fire Sprinkler Systems.
      - 7) Data Cabling Systems.
      - 8) Other Systems.
    - b. Site/Building Utilities Contractors: The following Contractors shall attend the coordination construction activities and space requirements:
      - 1) All Site/Building Utilities Systems.
      - 2) All Site/Building Electrical Systems.
      - 3) All Site/Building Mechanical Systems.
      - 4) All Site/Building Fire Protection Systems.
      - 5) All Site/Building Data Cabling Systems.
      - 6) Other Site/Building Systems.
      - 7) Landscape/Irrigation Site Systems and Requirements Leaching.
  - 2. Record significant discussions and agreements and disagreements of each meeting, along with the approved coordination drawings. Distribute the record of the meeting to everyone concerned, promptly, including the Owner and Architect.
  - 3. Do not proceed if the coordination meeting cannot be successfully concluded. Initiate whatever actions are necessary to resolve impediments to performance of Work and reconvene the coordination meeting at the earliest feasible date.

PART 2 - PRODUCTS (Not Applicable)

PART 3 - EXECUTION (Not Applicable)

**END OF SECTION 013100** 

PROJECT MEETINGS 013100 - 3

# PART 1 - GENERAL

## 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of Contract, including General and Supplementary Conditions and other Division 0 & 1 Specification Sections, apply to this Section.

#### 1.2 SUMMARY

- A. This Section specifies administrative and procedural requirements for submittals required for performance of the Work, including, but not limited to,;
  - 1. Contractor's construction project schedule.
  - 2. Submittal schedule.
  - 3. List of products/manufacturers.
  - 4. Shop Drawings.
  - 5. Product Data.
  - 6. Samples.
  - 7. Miscellaneous Submittals.
  - 8. Construction Schedule and updates required for each payment request.
  - 9. Coordination Drawings, specified in Division 1 Section "Project Coordination".
  - 10. Test reports.
- B. Administrative Submittals: Refer to other Division-1 Sections and other Contract Documents for requirements for administrative submittals. Such submittals include, but are not limited to:
  - 1. Permits/Public Works.
  - Applications for payment.
  - 3. Performance and payment bonds.
  - 4. Insurance certificates.
  - List of Subcontractors.
  - 6. Schedule of Values.
  - 7. Electronic certified payroll records directly to the Labor Commissioner.
  - 8. Preliminary notices (20-day notice).
  - 9. Procedures for substitutions.
- C. Inspection and test reports are included in Section "Quality Control Services."
- D. Other submittals:
  - 1. See Division 1 Section "Project Schedule and Reports," for additional schedule and report requirements

#### 1.3 SUBMITTAL PROCEDURES

- A. Coordination: Coordinate preparation and processing of submittals with performance of construction activities. Transmit each submittal sufficiently in advance of performance of related construction activities to avoid delay.
  - 1. Coordinate each submittal with fabrication, purchasing, testing, delivery, other submittals and related activities that require sequential activity.
  - 2. Coordinate transmittal of different types of submittals for related elements of the Work so processing will not be delayed by the need to review submittals concurrently for coordination.

- a. The Architect reserves the right to withhold action on a submittal requiring coordination with other submittals until related submittals are received.
- b. No color selections will be made until all color samples for entire project have been received by the Architect and determined to be acceptable.
- 3. Processing: Allow sufficient review time so that installation will not be delayed as a result of the time required to process submittals, including time for re-submittals.
  - a. Allow three weeks for initial review (21 calendar days). Allow additional time if processing must be delayed to permit coordination with subsequent submittals. The Architect will promptly advise the Contractor when a submittal being processed must be delayed for coordination.
  - b. If an intermediate submittal is necessary, process the same as the initial submittal.
  - c. Allow two weeks (14 calendar days) for reprocessing each submittal.
  - d. No extension of Contract Time or claims for additional costs will be approved or authorized because of failure to transmit submittals to the Architect sufficiently in advance of the Work and Critical Path Schedule to permit processing requirements indicated herein. Submittals covering component items and forming an interrelated system of items must be coordinated and submitted concurrently rather than individual submittals for review.
  - e. Time is defined as calendar days, starting from date item is received in architect's office and end with date of transmittal to General Contractor as last day of review.
    - 1) Where Submittal is required to be reviewed by consulting Engineer, time to review will be increased by seven (7) calendar days for review.
    - 2) Where Submittal is required to be reviewed by two parties--i.e., Architect-Engineer, Engineer-Engineer--add fourteen (14) calendar days for review.
    - 3) The Architects' office is closed from December 16 through January 2 of each calendar year. Submittals received less than 21 calendar days before office closing may not be processed before office closed dates noted herein. The balance of the 21 calendar days will resume after January 2 of each closing for submittals and related items. It is the contractor's responsibility to have critical submittals received by the architect 21 calendar days prior to above closing for timely processing.
      - a). Consulting engineering related submittals and re-review items will add time as indicated herein above to the amount of time indicated due to office closing.
  - f. Shop drawings, submittals and related items shall be submitted at a time sufficiently early to allow review by the Architect and the Division of State Architect (DSA) if required, and to accommodate the rate of construction progress required under the Contract Documents. Contractor will be required to pay the Architect's reasonable and customary fees to expedite review of shop drawings which are not submitted in timely fashion.
  - g. Contractor shall have no claim for damages or extension of time due to any delay resulting from contractor having to make required revisions to shop drawings unless the Architect's review of the drawings is delayed beyond the time provided in the contract documents and contractor can establish that the Architect's delay is review actually resulted in a delay in Contractor's construction schedule. Contractor shall provide a record as built schedule as proof of delays within 10 days of the event that gives rise to a delay claim. Contractor shall not be entailed to any claim for damages resulting from DSA review beyond days allowed herein documents after submittal. However, owner may consider an extension of time due to any delay caused by DSA review.
- 4. All submittals shall be cross-referenced to contract documents to expedite checking. Use Project Manual's section designation and Working Drawing's sheet number(s).
- B. Submittal Preparation: Place a permanent label or title block on each submittal for identification.

Indicate the name of the entity that prepared each submittal on the label or title block.

- 1. Provide a space approximately 4" x 4" on the label or beside the title block on Shop Drawings to record the Contractor's review and approval markings and the action taken. Provide on the label or beside the title block on Shop Drawings to record the Architect's review and approval markings and the action taken.
- 2. Include the following information on the label for processing and recording action taken.
  - a. Project name.
  - b. Date.
  - Name and address of Architect.
  - d. Name and address of Contractor.
  - e. Name and address of subcontractor.
  - f. Name and address of supplier.
  - g. Name of manufacturer.
  - h. Number and title of appropriate Specification Section.
  - Drawing number and detail references, as appropriate.
- 3. Submittals shall be stamped and signed by the Prime Contractor to the effect that the contents have been reviewed and approved by him and meet the requirements for this project. Submittals will not be reviewed without this approval by Prime Contractor. Contractor's review and approval of shop drawings shall include the following stamp:
  - a. "Contractor has reviewed and approved not only the field dimensions but the construction criteria and has also made written notation regarding any information in the shop drawings that does not conform to the Contract Documents. This shop drawing has been coordinated with all other shop drawings received to date by Contractor and this duty of coordination has not been delegated to subcontractors, material suppliers, the Architect, or the engineers on this Project.

Signature of Contractor"

- C. Submittal Transmittal: Package each submittal appropriately for transmittal and handling. Transmit each submittal from Contractor to Architect using a transmittal form. Submittals received from sources other than the Contractor will be returned without action.
  - On the transmittal record relevant information and requests for data. On the form, or separate sheet, record deviations from Contract Document requirements, including minor variations and limitations. Include Contractor's certification that information complies with Contract Document requirements.
- D. Failure to provide adequate and correct submittals: Contractor shall make a complete and acceptable submittal to the Architect by the second submission of submittals. Owner shall withhold funds due to Contractor to cover additional costs of the Architect's review beyond the second submission and any other costs incurred by Owner.
  - Architect shall be reimbursed for all time spent in reviewing and processing of re-submittals
    of any submittals after the second submission where items have not been addressed,
    corrected and/or providing a complete submittal as requested by architect in previous
    submissions of a submittal.
- E. Submittal quantities:
  - 1. Provide digital PDF copies of all submittals except as follows:
    - a. Shop Drawings: Provide PDF copies of shop drawings. Architect and Engineer will red line any corrections. Contractor will be sent a PDF file of shop drawing for distribution and printing by Contractor.

#### 1.4 CONTRACTOR'S CONSTRUCTION PROJECT SCHEDULE

- A. Bar-Chart Schedule: Prepare a fully developed, horizontal bar-chart type Contractor's construction project schedule. Identify critical path items throughout project schedule. Submit within 15 days of the award of contract.
  - 1. Provide a separate time bar for each significant construction activity. Provide a continuous vertical line to identify the first working day of each week. Use the same breakdown of units of the Work as indicated in the "Schedule of Values".
  - 2. Within each time bar indicate estimated completion percentage in maximum 10 percent increments. As Work progresses, place a contrasting mark in each bar to indicate Actual Completion.
  - 3. Prepare the schedule on a sheet, or series of sheets of sufficient width to show data for the entire construction period.
  - 4. Secure time commitments for performing critical elements of the Work from parties involved. Coordinate each element on the project schedule with other construction activities; include minor elements involved in the sequence of the Work. Show each activity in proper sequence. Indicate graphically sequences necessary for completion of related portions of the Work.
  - 5. Coordinate the Contractor's construction project schedule with the schedule of values, list of subcontracts, submittal schedule, progress reports, payment requests and other schedules.
  - 6. Indicate completion in advance of the date established for Final Completion. Indicate Final Completion on the schedule to allow time for the Architect's procedures necessary for certification of Final Completion.
- B. Phasing: Provide notations on the schedule to show how the sequence of the Work is affected by requirements for phased completion to permit Work by separate Contractors, partial occupancy by the Owner prior to Final Completion.
- C. Work Stages: Indicate important stages of construction for each major portion of the Work, including testing and installation.
- D Weather Delays: Contractor shall show anticipated rain, sleet, snow, wind and other weather driven delays on initial project schedule, as separate line items for each weather delay type, based upon Contractor's proposed ways and means. Refer to Project Manual herein for additional requirements.
- E. Area Separations: Provide a separate time bar to identify each major construction area for each major portion of the Work. Indicate where each element in an area must be sequenced or integrated with other activities.
- F. Distribution: Following response to the initial submittal, print and distribute copies to the Architect, Owner, subcontractors, and other parties required to comply with scheduled dates. Post copies in the Project meeting room and temporary field office.
- G. Submit with each payment request an updated construction project schedule with the above item plus the following items:
  - 1. As-built project schedule of worked performed to date.
  - 2. Adjusted project schedule showing schedule to complete work.
  - 3. Number of weather delay days and audit of days used for each type of weather delay.
  - 4. Updated Final Completion project schedule dates.

#### 1.5 SUBMITTAL SCHEDULE

A. Prepare a complete schedule of submittals. Submit the schedule within 15 days of Notice to Proceed.

- Coordinate submittal schedule with the list of subcontracts, schedule of values and the list of products as well as the Contractor's construction schedule.
- 2. Prepare the schedule in chronological order. Provide the following information:
  - a. Scheduled date for each first submittal.
  - b. Related Section number.
  - c. Submittal category.
  - d. Name of subcontractor.
  - e. Description of the part of the Work covered.
  - f. Scheduled date for resubmittal.
  - g. Scheduled date the Architect's final release or approval.
- B. Distribution: Following response to initial submittal, print and distribute copies to the Architect, Owner, subcontractors, and other parties required to comply with submittal dates indicated. Post copies in the Project meeting room and field office.
  - 1. When revisions are made, distribute to the same parties and post in the same locations. Delete parties from distribution when they have completed their assigned portion of the Work and are no longer involved in construction activities.
- C. Schedule Updating: Revise the schedule after each meeting or activity, where revisions have been recognized or made. Issue the updated schedule concurrently with report of each meeting.

#### 1.6 LIST OF PRODUCTS/MANUFACTURES

A. Not later than 35 days from the date of the notice to proceed and prior to installation of items, whichever is less, the Contractor shall provide a list showing the name of the manufacturer proposed to be used for each of the products, proposed for installation, and not specified or named in the contract documents including the name of manufacturer of each, for review by the Owner and Architect. The list shall be tabulated by, and be complete for each specification section. Where applicable, subcontractor's names shall be included in such list.

#### 1.7 SHOP DRAWINGS

- A. Submit newly prepared information, drawn to accurate scale. Highlight, encircle, or otherwise indicate deviations from the Contract Documents. Do not reproduce Contract Documents or copy standard information as the basis of Shop Drawings. Standard information prepared without specific reference to the Project is not considered Shop Drawings.
- B. Shop Drawings include fabrication and installation drawings, setting diagrams, schedules, patterns, templates, shop work manufacturing instructions, coordination drawings (for on-site use), design mix information, contractor's engineer calculations, and similar drawings. Include the following information:
  - 1. Dimensions.
  - 2. Identification of products and materials included.
  - 3. Compliance with specified standards.
  - 4. Notation of coordination requirements.
  - 5. Notation of dimensions established by field measurement.
  - 6. Sheet Size: Except for templates, patterns and similar full- size Drawings, submit Shop Drawings on sheets at least 8-1/2" x 11" but no larger than 36" x 48 (See herein for electronic submittals)".
  - 7. Initial Submittal: Submit one black-line print for the Architect's review; the reproducible print will be returned, (see herein for electronic submittals).
    - a. Structural steel, metal deck and miscellaneous steel. Submit black line print and 3 copies for initial submittal.

- 8. Final Submittal: Submit copies per enclosed requirements.
- 9. Do not use Shop Drawings without an appropriate final stamp indicating action taken in connection with construction.
- C. Coordination drawings are a special type of Shop Drawing that show the relationship and integration of different construction elements that require careful coordination during fabrication or installation to fit in the space provided or function as intended.
  - 1. Preparation of coordination Drawings include components previously shown in detail on Shop Drawings or Product Data.
  - 2. Submit coordination Drawings for integration of different construction elements. Show sequences and relationships of separate components to avoid conflicts in use of space.

#### 1.8 PRODUCT DATA

- A. Collect Product Data into a single submittal for each element of construction or system. Product Data includes printed information such as manufacturer's installation instructions, catalog cuts, standard color charts, roughing-in diagrams and templates, standard wiring diagrams and performance curves. Where Product Data must be specially prepared because standard printed data is not suitable for use, submit as "Shop Drawings."
  - 1. Mark each copy to show applicable choices and options. Where printed Product Data includes information on several products, some of which are not required, mark copies to indicate the applicable information. Include the following information:
    - a. Manufacturer's printed recommendations.
    - b. Compliance with recognized trade association standards.
    - c. Compliance with recognized testing agency standards.
    - d. Application of testing agency labels and seals.
    - e. Notation of dimensions verified by field measurement.
    - f. Notation of coordination requirements.
  - Do not submit Product Data until compliance with requirements of the Contract Documents has been confirmed.
  - 3. Preliminary Submittal: Submit a preliminary single-copy of Product Data where selection of options is required.
  - 4. Submittals: Submit copies of each required submittal; submit hardcopies where required for maintenance manuals. The Architect will retain one, and will return the other marked with action taken and corrections or modifications required.
    - a. Unless noncompliance with Contract Document provisions is observed, the submittal may serve as the final submittal.
  - 5. Distribution: Furnish copies of final submittal to installers, subcontractors, suppliers, manufacturers, fabricators, and others required for performance of construction activities. Show distribution on transmittal forms.
    - a. Do not proceed with installation until an applicable copy of Product Data applicable is in the installer's possession.
    - Do not permit use of unmarked copies of Product Data in connection with construction.

#### 1.9 MATERIALS SPECIFIED

- A. The Contract is based on standards of quality established in the Contract Documents.
  - 1. In agreeing to the terms and conditions for the Contract, the Contractor has accepted the responsibility to verify that the specified products will be available and to place orders for

- all required materials in such a timely manner as needed to meet his agreed construction schedule
- Neither the Owner nor the Architect has agreed to the substitution of materials or methods called for in the Contract Documents, except as they may specifically otherwise state in writing and when approved in advance of purchase and installation per requirements herein.
- B. Colors: Provide finish selections indicated in the Project Manual and Plans.
  - 1. Acceptable Manufacturers: The products and manufacturer's specified in the Project Manual and Plans are for purposes of establishing color selection options and quality.
  - 2. Manufacturer's Standard colors and Finishes: Where the Project Manual or Plans specifies a manufacturer's standard color or finish, the Architect makes no guarantee that matching colors or finishes are available as other non-listed manufacturer's "standard colors" from the listing of acceptable manufacturers. The Contractor shall be responsible for providing colors matching those indicated in the Project Manual of listed acceptable manufacturer in the Project Manual at no additional cost.
  - 3. Custom Colors: Where the Finish Schedule Project Manual and or Plan indicates a specific manufacturer's colors, other acceptable manufacturer shall provide matching custom colors where a standard color in not acceptable at no additional cost.

#### 1.10 SAMPLES

- A. Submit full-size, fully fabricated Samples cured and finished as specified and physically identical with the material or product proposed. Samples include partial sections of manufactured or fabricated components, small cuts or containers of materials, color range sets, complete units of repetitively used materials, units of work to be used for independent inspection and testing, and swatches showing color, texture and pattern.
  - 1. Mount, display, or package Samples in the manner specified to facilitate review of qualities indicated. Prepare Samples to match the Architect's Sample. Include the following:
    - a. Generic description of the Sample.
    - b. Sample source.
    - c. Product name or name of manufacturer.
    - d. Compliance with recognized standards.
    - e. Availability and delivery time.
  - 2. Submit Samples for review of kind, color, pattern, and texture, for a final check of these characteristics with other elements, and for a comparison of these characteristics between the final submittal and the actual component as delivered and installed.
    - a. Where variation in color, pattern, texture or other characteristics are inherent in the material or product represented, submit multiple units (not less than 6), that show approximate limits of the variations.
    - b. Refer to other Specification Sections for requirements for Samples that illustrate workmanship, fabrication techniques, details of assembly, connections, operation and similar construction characteristics.
  - 3. Preliminary submittals: Where Samples are for selection of color, pattern, texture or similar characteristics from a range of full selection, submit three (3) full sets of choices for the material or product.
    - a. Color selection shall be done upon complete submittal of materials and/or products needing color selection. It is the responsibility of the general contractor to see to it that required submittals for color selection shall be submitted to the Architect prior to time for implementation to not affect project schedule.
    - b. Provide ninety (90) days time allowance for the Architect to work out preliminary

### TC – G-8 NURSING LAB REMODEL

schemes for College approval from receipt of complete color submittal of all items for this project.

- 4. Submittals: Except for Samples illustrating assembly details, workmanship, fabrication techniques, connections, operation and similar characteristics, submit 4 sets; one will be returned marked with the action taken.
- 5. Maintain sets of Samples, as returned, at the Project site, for quality comparisons throughout the course of construction.
  - a. Unless noncompliance with Contract Document provisions is observed, the submittal may serve as the final submittal.
  - b. Sample sets may be used to obtain final acceptance of the construction associated with each set.
- B. Distribution of Samples: Prepare and distribute additional sets to subcontractors, manufacturers, fabricators, suppliers, installers, and others as required for performance of the Work. Show distribution on transmittal forms.
  - 1. Field Samples specified in individual Sections are special types of Samples. Field Samples are full-size examples erected on site to illustrate finishes, coatings, or finish materials and to establish the standard by which the Work will be judged.
    - a. Comply with submittal requirements to the fullest extent possible. Process transmittal forms to provide a record of activity.

### 1.10 MISCELLANEOUS SUBMITTALS

- A. Miscellaneous submittals are work related submittals (non-administrative) including warranties, maintenance agreements, workmanship bonds, survey data and reports, quality testing and certifying reports, copies of industry standards, record drawings, field measurement data, operating and maintenance materials, overrun stock, and similar work related information and materials not process as shop drawings, product data or samples.
- B. Inspection and Test Reports: Classify and process each as either "shop drawing" or "product data", depending upon whether report is uniquely prepared for project or a standard publication.
- C. Warranties: Refer to "Products" section for specific general requirements on warranties, product/workmanship bonds, and maintenance agreements. In addition to copies desired for Contractor's use, furnish 2 executed copies, plus additional copies where required for maintenance manuals.
- D. Standards: Where submittal of a copy of standards is indicated, submit a single copy for Architect's/Engineer's use. Where workmanship at project site and elsewhere is governed by standard, furnish additional copies to fabricators, installers and others involved in performance of the work.
- E. Closeout Submittals: Refer to "Project Closeout" section for specific general requirements on submittal of closeout information, materials, tools, and similar items.
- F. Schedule of Values: Contractor shall submit Schedule of Values per Instructions to Bidders,. Refer to General Conditions for minimum list of items to be included in Schedule of Values. Changes in Schedule of Values will not be allowed after Notice to Proceed is issued, unless directed by Architect and/or Change Order items which shall carry their own value as a line item on Schedule of Values.
- G. Certified Payroll Records: Refer to "General Conditions" section for specific general and apprentices requirements..

### TC - G-8 NURSING LAB REMODEL

- 1. All contractors and subcontractors must furnish electronic certified payroll records directly to the Labor Commissioner (aka Division of Labor Standards Enforcement).
- 2. General Contractor shall provide certification that all requirements for prevailing wage have been made with each application for payment, not limited to paying prevailing wage, documentation of prevailing wage and furnishing electronically to Labor Commissioner.

### 1.12 ARCHITECT'S ACTION

- A. Except for submittals for record, information or similar purposes, where action and return is required or requested, the Architect will review each submittal, mark to indicate action taken, and return.
  - 1. Compliance with specified characteristics is the Contractor's responsibility.
  - 2. Numerous serious corrections, or incomplete submittals, will necessitate resubmittal, in which case, only one copy will be returned with notations. Contractor shall resubmit required number of sets with corrections made with original mark-up submittal for review.
- B. Conditions of Review: Architect's review is for general conformance with the design concept and contract documents. Review action on a submittal by the Architect does not in any way constitute a change order. Markings or comment shall not be construed as relieving the Contractor from compliance with the project plans and specifications, nor departures therefrom. The Contractor remains responsible for details and accuracy, for conforming an correlating all quantities and dimensions, for selecting fabrication processes for techniques of assembly, and for performing his work in a safe manner.
  - The Contractor is responsible for coordination of his work with and between that of all subcontractors and trades.
  - Absolutely no deviation from the Contract Documents will be permitted without written acknowledgement from Architect of receipt and Review of Written Notification from the Contractor to the Architect accompanying this submittal of all deviations contained in this submittal.
  - 3. The Architect's review is not the final stage of acceptance for any part of the project, nor does it relieve the Contract responsibilities.
  - 4. Contractor shall submit an itemized list of changes of items different than specified/indicated herein and on construction documents. List shall include items that are different and omitted. In the event items are not included on list, omitted from submittal and/or different than specified; Contractor shall be responsible for providing specified item(s). Liabilities subsequent to items omitted/or different shall be the responsibility of Contractor and shall be warranted a minimum of five (5) years or greater as prescribed by law. If no list is included with Shop Drawings, Architect assumes all items are as specified. Items discovered within five (5) years of Notice of Completion shall be corrected and provided by Contractor and Subcontractor at no cost to Owner.
- C. Action: The Architect will identify each submittal with a uniform, self-explanatory action sheet. The sheet will be appropriately marked, as follows, to indicate the action taken:
  - 1. No Exceptions Taken: If this box is marked, the work covered by the submittal may proceed provided it complies with the requirements of the contract documents; acceptance of the work will depend upon that compliance.
  - 2. Make Corrections Noted: If box is marked, the work covered by the submittal may proceed provided it complies with both the Architect's/Engineer's notations or corrections to the submittal and with the requirements of the contract documents; acceptance of the work will depend on that compliance. Submit corrected copy for record if requested by the Architect.
  - 3. Revise and Resubmit: If this box is marked, do not proceed with the work covered by the submittal, including purchasing, fabrication, delivery, or other activity. Revise the submittal in accordance with the Architect's/Engineer's notations and resubmit without delay. Repeat if necessary.
  - 4. Rejected: If this box is marked, do not proceed with the work covered by the submittal, including purchasing, fabrication, delivery, or other activity. Revise the submittal or prepare a

### TC – G-8 NURSING LAB REMODEL

new submittal in accordance with the Architect's/Engineer's notations and resubmit to Architect.

- a. Do not permit submittals marked "Rejected" or, "Revise and Resubmit" to be used at the Project site, or elsewhere where Work is in progress.
- 5. Action Not Required: Where a submittal is primarily for information or record purposes, special processing or other activity, the submittal will be returned, marked "Action Not Required".
- 6. Remarks: The review has occasioned comments that have been attached to the submittal. Process these comments as if they had been written on the submittal itself.
- 7. Requires Intermediate Submittal: Submittal does not meet all requirements for a complete approval. Submittal requires additional information for processing.

# 1.13 SUBSTITUTIONS

# A. Substitution Request:

- Cost to Contractor or Bidder for review of Substitution Request.
  - Each review of a Substitution Request by the Architect will be billed to the submitter (Contractor or Bidder) at the current hourly rate on file at Architect's office, two hour minimum for each review, whether approved or rejected.
  - b. The Contractor will have a thirty-five (35) day period from the contract award date, in which to review the total contract documentation and issue any substitution or clarification requests to the Architect free of any financial charge.
  - c. The Contractor will be expected to identify any omissions, anomalies, divergence or discrepancies in the Contract Documents within this time period and so inform the Architect, in writing.
  - d. Any such omissions, anomalies, divergence or discrepancies not identified to the Architect within this period shall be deemed to be included within the bid sum and not subject to a Change Order by the Architect.
    - 1) The Architect may waive the review cost if, in his sole opinion, the submittal was complete and the time involved in the review was not substantial, and it was in the best interest of the Owner.

### 2. Content of Request:

- a. Complete the attached Substitution Request Form (at the end of this Section), substantiating compliance of proposed substitution with Contract Documents.
- b. For products, attach to the Substitution Request Form:
  - 1) Product identification, including manufacturer's name and address.
  - 2) Manufacturer's literature including product description, performance and complete test data and reference standards.
  - 3) Samples.
- c. For construction methods, attach to the Substitution Request Form:
  - 1) Detailed description of proposed methods.
  - 2) Drawings illustrating methods.
- d. Attach to the Substitution Request Form an itemized comparison of proposed substitution with product or method specified.
- e. Provide long-term serviceability data comparing side by side analysis with specified materials.
- f. Provide manufacturer's experience in years with product with specific product formulation that is to be substituted.

### TC – G-8 NURSING LAB REMODEL

- g. Provide certified warranty issued for this specific project including application-precisely.
- h. Provide system component analysis and statement the product is certified by Contractor to be compatible with all other items of assemblies where product/material/method is specified.
- 3. In making request for substitution, Contractor attests that:
  - a. Contractor has personally investigated proposed product or method, and determined that it is equal or superior in all respects to that specified.
  - b. Contractor will provide the same guarantee or warranty for substitution as for product or method specified.
  - c. Contractor will coordinate installation of accepted substitution into Work, making such changes as may be required for Work to be complete in all respects.
  - d. Contractor waives all claims for additional costs related to substitution which subsequently become apparent.
  - e. Contractor will pay all cost of Consultant to interpret physical properties to compare substitution with specified product, if requested by Architect.
  - f. Colors: The Contractor will match the color and/or finish available for the acceptable manufacturers listed in the Project Manual and/or Plans as a custom color and at no additional cost to the Owner.
- 4. Submit copies of Substitution Request prior to submittals required.

# B. Acceptance of Substitutions

### 1. Procedures:

- The Contract is based on materials, equipment and methods described in the Contract Documents.
- b. Architect will consider proposals submitted in accordance with the Substitution Request.
- c. Substitutions will be considered when submitted within 35 days after date of Contract.
- Architect is solely responsible for judging the acceptance of substitutions.
- e. Substitute materials, equipment or methods shall not be used unless such substitution has been specifically approved for this Work by the Architect and DSA.
- f. Substitutions will not be considered if:
  - 1) They are indicated or implied on product submittals without formal request submitted in accordance with Substitution Request.
  - 2) Acceptance will require substantial revision of Contract Documents.
  - They are submitted more than 35 days after date of Contract, unless the specified or drawing item has been verified to be discontinued or is otherwise unavailable, or the Owner desires a cost savings for the product or system.
- g. Substitutions may be subject to DSA approval if Structural Safety, Fire/Life Safety or Accessibility is impacted.
- 2. Time to review: Architect shall be allowed twenty-eight (28) calendar days to review each substitution submittal. In the event review cannot be completed and more information is requested by the Architect to complete this review, upon receipt of requested information, twenty-eight (28) calendar days will be allowed for additional review after receipt of requested and complete information.

PART 2 - PRODUCTS

# 2.1 SUBSTITUTION REQUEST FORM

# TC - G-8 NURSING LAB REMODEL

See the form attached to the end of this Section.

# PART 3 - EXECUTION

# 3.1 GENERAL

The attached form will be reproduced by the Contractor or any of his Subcontractors for any and all proposed substitutions. No other forms will be accepted.



# **SUBSTITUTION REQUEST**

| DATE         | ::  |                                  | S                                | ubstitution Request #                       | :        |            |           |          |
|--------------|---|----------------------------------|----------------------------------|---|----------|------------|-----------|----------|
| ATTN         | I: Jose Vargas  | PROJECT:                         | TC – G-8 NI                      | JRSING LAB REMOI                            | DEL      |            |           |          |
|              | AP Architects   |                                  |                                  |   |          |            |           |          |
|              | 3434 Truxtun Ave Suite 240  |                                  |                                  |   |          |            |           |          |
|              | Bakersfield, 93301  | PROJECT#:                        | 430-0083                         | NOA DATE                                    | <u> </u> |            |           |          |
|              | hereby submit for your considerat<br>ne Project Manual/ Plans:  | ion the followin                 | ng product/ m                    | anufacturer instead                         | of the   | one(s      | s) spe    | cified   |
| Item         | :   |                                  |                                  | Section                                     | n #:     |            |           |          |
| Man          | ufacturer:  |                                  |                                  | Shee  | t #:     |            |           |          |
| A. R         | eason for not providing specified pro   | oduct:                           |                                  |   |          |            |           |          |
|              |   |                                  |                                  |   |          |            |           |          |
|              |   |                                  |                                  |   |          |            |           |          |
| <u>В</u> . F | Proposed Substitution:  |                                  |                                  |   |          |            |           |          |
|              | •   |                                  |                                  |   |          |            |           |          |
|              | Cost shall be shouldered by the underetailing costs caused by the request ses No Explain:                           |                                  |                                  |   |          |            |           | <u> </u> |
|              |   |                                  |                                  | *NII Not Indicated:                         | Check    | list for A | Architect |          |
| D.           | Provide the following with Substitution   | on Request Pacl                  | kage:                            | *NI –Not Indicated;<br>N/A – Not Applicable | Yes      | o<br>N     | Α×        | Z        |
| 1            | Attached data includes product data approvals and laboratory test data a shall be complete with relevant test(      | dequate for eval                 |                                  |   |          |            |           |          |
| 2            | Attach data includes description of c<br>substitution will have. Include comp<br>project manual which proposed subs | lete information o               | on changes to                    | drawing and/ or                             |          |            |           |          |
| 3            | Does proposed substitution affect di  | ·                                |                                  |   |          |            |           |          |
| 4            | Does proposed substitution affect of  |                                  | s it clear on the                | request form?                               |          |            |           |          |
| 5            | Provide system component compati<br>Does proposed substitution affect lo<br>including where nearest service repr    | cal availability of              |                                  |   |          |            |           |          |
| 6            | Attached cost data with detail break  |                                  |                                  | • •   |          |            |           |          |
| 7            | If substitution is of higher quality, will  | this impact futur                | re replacemen                    | t cost?                                     |          |            |           |          |
| 8            | What is the impact of substitution on   | construction sch                 |                                  |   |          |            |           |          |
| 9            | Provide long term serviceability data   |                                  | hedule?                          |   |          |            |           |          |
| 10           | _   |                                  |                                  | rial.                                       |          |            |           |          |
|              | Provide manufacturing experience product formation substituted.   | compared with                    | specified mate                   |   |          |            |           |          |
| 11           |   | compared with sin years with pro | specified mate<br>oduct with spe | cified material                             |          |            |           |          |

E. The undersigned certifies that the following paragraphs, unless modified by attachments, are correct:

### TC – G-8 NURSING LAB REMODEL

- 1. The undersigned will compensate the Architect, Architect's staff, and consulting engineers at Architect's rate per hour for changes required to the building design, including engineering design, detailing, and construction costs caused by the requested substitution. The Architect is herein defined as any of those firms or individuals listed by reference on the Directory, including all Consultants identified herein.
- 2. Attach all cost data with explanations if different from Specified or Drawing item. Include in that explanation a discussion on quality or proposed substitution and cost differential.
- 3. Attach all cost data with explanations if different from Specified or Drawing item. Include in that explanation a discussion on quality or proposed substitution and cost differential.
  - The undersigned will pay for any subsequent changes in incorporating the proposed substitution that were not
- 4. apparent at the time of approval into the Work, including compensation to the Architect and consulting agent(s) as described in item 2 above.
- 5. The undersigned certifies that the substituted material/product/method is compatible with all the items in the system application's specified use in this project.
- 6. Failure to provide complete substitution package per above requirements are ground for rejection.

The undersigned states that the function, appearance and quality are equivalent or superior to the construction document item. The undersigned agrees to waive all claims for additional costs related to accepted substitution, including cost associated with changes to building design, engineering, or details, which may subsequently become apparent. (Negative response maybe cause for rejection.)

| Submitted by:       |       | Approved by General Contractor:   |   |  |  |
|---------------------|-------|---|---|--|--|
| · -                 |       | Signature   |   |  |  |
| Signature: Address: |       | Signed by: Address:   | Date:   |  |  |
| Phone:              | Fax:  | Phone:  | Fax:  |  |  |
| ARCHITECT'S RESP    | ONSE: |   | bstitution Request #:<br>bmit as construction submittal |  |  |
| Reviewed by:        |       | Accepted as noted – Submit as construction submittal  Rejected – Use specified material |   |  |  |
| Remarks:            |       |   | ate – Use specified material                            |  |  |
|                     |       |   |   |  |  |
|                     |       |   |   |  |  |
|                     |       |   |   |  |  |
|                     |       |   |   |  |  |
|                     |       |   |   |  |  |
|                     |       |   |   |  |  |
|                     |       |   |   |  |  |
|                     |       |   |   |  |  |
|                     |       |   |   |  |  |
|                     |       |   |   |  |  |
|                     |       |   |   |  |  |
|                     |       |   |   |  |  |
|                     |       |   |   |  |  |

END OF SECTION 013300

### PART 1 - GENERAL

# 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and other Division 0 & 1 Specification Sections, apply to this Section.

### 1.2 SUMMARY

- A. This Section specifies requirements for temporary services and facilities, including utilities, construction and support facilities, security and protection.
- B. Temporary construction and support facilities required include but are not limited to:
  - 1. Temporary heat.
  - 2. Waste disposal services.
  - 3. Construction aids and miscellaneous services and facilities.
  - 4. Temporary access to occupied buildings.
  - 5. Temporary roof and envelope of building.
  - Dust control measures.
- C. Security, protection and miscellaneous facilities required include but are not limited to:
  - 1. Temporary fire protection.
  - 2. Barricades, warning signs, lights.
  - 3. Environmental protection.
  - 4. Enclosure fence at site work around each portion of site work.
  - 5. Temporary barriers around rooms with Owner access during construction to keep Staff and students from entering construction area.

### 1.3 QUALITY ASSURANCE

- A. Regulations: Comply with industry standards and applicable laws and regulations if authorities having jurisdiction, including but not limited to:
  - 1. Building Code requirements.
  - 2. Health and safety regulations.
  - 3. Utility company regulations.
  - 4. Police, Fire Department and Rescue Squad rules.
  - 5. Environmental protection regulations.
  - 6. California State Accessibility standards, Title 24.
  - 7. 2022 CBC Chapter 33.
  - 8. 2013, Title 24 CFC Chapter 14.
- B. Standards: Comply with NFPA Code 241, "Building Construction and Demolition Operations", ANSI-A10 Series standards for "Safety Requirements for Construction and Demolition", and NECA Electrical Design Library "Temporary Electrical Facilities."
  - 1. Refer to "Guidelines for Bid Conditions for Temporary Job Utilities and Services", prepared jointly by AGC and ASC, for industry recommendations.

2. Electrical Service: Comply with NEMA, NECA and UL standards and regulations for temporary electric service. Install service in compliance with California Electric Code (CEC).

### 1.4 PROJECT CONDITIONS

- A. Conditions of Use: Keep temporary services and facilities clean and neat in appearance. Operate in a safe and efficient manner. Take necessary fire prevention measures. Do not overload facilities, or permit them to interfere with progress. Do not allow hazardous dangerous or unsanitary conditions, or public nuisances to develop or persist on the site.
- B. Site Access: Shall be coordinated with the Director of Maintenance and Operations. All areas damaged by construction work shall be remedied to his satisfaction. Access by handicapped shall be maintained during construction. Architect shall approve all temporary handicapped access to project area prior to implementing any work.
- C. All on site utilities usage shall be paid by Owner limited to water, natural gas, sewer, storm drain and electricity. Contractor is responsible for temporary connections and disconnection. Return temporary connection to pre-construction conditions.

### PART 2 - PRODUCTS

### 2.1 MATERIALS

A. Open-Mesh Plastic Fencing: Plastic 2-inch fabric fencing, 6 feet high with galvanized steel pipe post, 1-1/2" O.D. for line and corner post.

### 2.2 EQUIPMENT

- A. Electrical Power Cords: Provide grounded extension cords; use "hard-service" cords where exposed to abrasion and traffic. Provide waterproof connectors to connect separate lengths of electric cords, if single lengths will not reach areas where construction activities are in progress.
- B. First Aid Supplies: Comply with governing regulations.
- C. Fire Extinguishers: Provide hand-carried, portable UL-rated, class "A" fire extinguishers for temporary offices and similar spaces. In other locations provide hand-carried, portable, UL-rated, class "ABC" dry chemical extinguishers, or a combination of extinguishers of NFPA recommended classes for the exposures.

### PART 3 - EXECUTION

### 3.1 TEMPORARY CONSTRUCTION AND SUPPORT FACILITIES INSTALLATION

A. Collection and Disposal of Waste: Collect waste from construction areas and elsewhere daily. Comply with requirements of NFPA 241 for removal of combustible waste material and debris. Enforce requirements strictly. Do not hold materials more than 7 days during normal weather or 3 days when the temperature is expected to rise above 80 deg F (27 deg C). Handle hazardous,

dangerous, or unsanitary waste materials separately from other waste by containerizing properly. Dispose of material in a lawful manner.

### 3.2 SECURITY AND PROTECTION FACILITIES INSTALLATION

- A. Temporary Fire Protection: Until fire protection needs are supplied by permanent facilities, install and maintain temporary fire protection facilities of the types needed to protect against reasonably predictable and controllable fire losses. Comply with NFPA 10 "Standard for Portable Fire Extinguishers," and NFPA 241 "Standard for Safeguarding Construction, Alterations and Demolition Operations." Comply with 2013 CFC Chapter 14 during construction.
- B. Barricades, Warning Signs and Lights: Comply with standards and code requirements for erection of structurally adequate barricades. Paint with appropriate colors, graphics and warning signs to inform personnel and the public of the hazard being protected against. The Contractor shall provide, install and maintain for the duration of the work, as required, all lawful or necessary barricades and railings, lights, warning signs and signals, and shall take all other precautions as may be required to safeguard persons, the site and adjoining property, including improvements thereon, against injuries and damages of every nature whatsoever. The Contractor shall not obstruct required exitways of adjacent structures.

### 3.3 OPERATION, TERMINATION AND REMOVAL

- A. Supervision: Enforce strict discipline in use of temporary facilities. Limit availability of temporary facilities to essential and intended uses to minimize waste and abuse.
- B. Termination and Removal: Unless the Architect requests that it be maintained longer, remove each temporary facility when the need has ended, or when replaced by authorized use of a permanent facility, or no later than date of Notice of Completion. Complete or, if necessary, restore permanent construction that may have been delayed because of interference with the temporary facility. Repair damaged Work, clean exposed surfaces and replace construction that cannot be satisfactorily repaired.

END OF SECTION 015050

### PART 1 - GENERAL

# 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of Contract, including General and Supplementary Conditions and other Division 1 Specification Sections, apply to this Section.

### 1.2 SUMMARY

- A. This Section specifies administrative and procedural requirements for project closeout, including but not limited to:
  - 1. Project record document submittal.
  - 2. Operating and maintenance manual submittal.
  - 3. Submittal of warranties.
  - Final cleaning.
  - 5. Compliance forms.
- B. Closeout requirements for specific construction activities are included in the appropriate Sections in Divisions 2 through 33.

### 1.3 PROJECT COMPLETION

- A. Contractor shall notify the Architect in writing when the project is acceptably completed; all work scope is done and ready for Project Inspector and Contractor to make a list of items to be corrected.
- B. The Project Inspector and Contractor shall make a list of items to be corrected and finish any items discovered incomplete. These items shall be completed and corrected prior to Architect and Consultants preparing a Final Inspection and Punch List Preparation. Contractor shall notify Architect in writing when all items are completed and corrected as well as items completed and corrected from Project Inspector and Contractor list of items previously prepared.
- C. Architect's field representatives (Architectural, Mechanical, Electrical, and other consultants) will make a field survey of the project to confirm that it has reached a state of completion, and items for "B" above have been done, in order to eliminate an unreasonably long Punch List on final inspection.
  - 1. All items in scope of work have been completed and accepted by Contractor and Project Inspector.
  - 2. All building systems are operational, tested and commissioned.
  - Cleaning of building, grounds and related items completed except when approved by architect in writing in advance of field survey. Contractor's temporary measures may be left in place until final demobilization but a condition on removal for final payment processing.
  - 4. Contractor to provide a written status update of project closeout check list.
- D. If not ready, the Architect will give the Contractor a written brief summary of what must be done for the project to be considered complete, enough for the Punch List preparation.
- E. When, in the opinion of the Architect, the job is ready for the Punch List preparation, a Punch List

of items requiring completion and/or correction will be prepared. (This Punch List will be made as specific and complete as possible, and will include the listing of all specified closing items required of the Contractor, such as Record Drawings, maintenance manuals, written guarantees, etc.) Any items observed or noted subsequent to Punch List preparation, shall also be corrected prior to re-inspection, unless the Architect determined, in writing, that any such items properly fall into the category of work to be corrected during the warranty/guarantee period.

- Architect's Punch List preparation will be scheduled and respective contractors, subcontractors and vendors shall be present during the Punch List inspection. Sub-Contractors, contractors and vendors will be required to show operational status of items in project scope of each respective discipline.
- 2. Mechanical, Electrical, and other consultants Punch List preparation will be scheduled and respective sub-contractor, contractors and vendors shall be present during the Punch List inspection. Sub-Contractors, contractors and vendors will be required to show operational status of items in project scope of each respective discipline.
- 3. All Punch List items shall be completed and/or corrected before Contractor calls for Punch List re-inspection.
  - a. The Architect shall be notified in writing by Contractor when all Punch List items are complete and project is ready for punch list back check by architect and consultants. Architect shall schedule punch list back check with contractor within 10 calendar days from receipt of written notice from Contractor.
  - b. The Architect will allow contractor two, (2), punch list back check visits of all items. Any additional site visits for items not completed and acceptable to Architect after second site visit will charged to the contractor as a negative change order for time spent by Architect to do re-inspections and office related work including reimbursable expenses incurred due to additional site visits.

# 1.4 DOCUMENT SUBMITTALS

- A. General: Do not use record documents for construction purposes; protect from deterioration and loss in a secure, fire-resistive location; provide access to record documents for the Architect's reference during normal working hours.
- B. Record Drawings: Maintain a clean, undamaged set of blue or black line white-prints of Contract Drawings and Shop Drawings. Mark the set to show the actual installation where the installation varies substantially from the Work as originally shown. Mark whichever drawing is most capable of showing conditions fully and accurately; where Shop Drawings are used, record a cross-reference at the corresponding location on the Contract Drawings. Give particular attention to concealed elements that would be difficult to measure and record at a later date. Transfer all marks to blue line originals provided by Contractor.
- C. Project Closeout Binder(s): Submit requirements per project manual section in a binder(s). See Project Closeout form at the end of this section for additional requirements. Note that this form shall be updated with additional materials and sections once Contractor starts submitting closeout documents to include issued addenda, RFPs, FCDs and other items added in the bid document during construction. All required number of years for warranty shall be provided by the Contractor. In the event this conflicts with manufacturer's warrant, more restrictive will prevail. If manufacturer will not meet requirements, Contractor shall take full responsibility for additional warranty requirement above what manufacturer will provide.
  - 1. Binder(s) shall have cover sheet inserted in front face and side label. Cover sheet shall have the following information:
    - a. Project name
    - b. College name

- c. Date of Notice of Completion
- d. General Contractor's name.
- e. Index of items in binder.

### 1.5 PROJECT CLOSEOUT NOTIFICATION

- A. If requested in writing, the contractor will be given a project closeout check list of items to be completed prior to project being accepted as complete. Contractor shall start project closeout no later than 90% completion of project and be complete prior to project punch list preparation/walk. This is contractor's first notice. Notice of Completion will not be filled until project closeout is complete.
- B. The Contractor will be given a 20 calendar day notification that the project closeout is incomplete and notification that owner will complete project closeout work incomplete and assess contractor additional architectural and engineering fees incurred completing work not complete and per construction documents.
- C. The Contractor will be given a final 10-day notice to complete all project closeout items. The estimated amount of costs will be indicated therein that the owner will be spending for completing the project closeout. Items done by the owner and additional architect's fees will be deducted from funds due contractor. Copy will be sent to surety.
- D. If project closeout is not complete after deadline in Item "C" above, the notice of completion will be filled listing incomplete items. Owner will complete project closeout and deduct cost incurred from funds held. Balance of funds will be distributed per Contract Documents.

PART 2 - PRODUCTS (Not Applicable)

PART 3 - EXECUTION

### 3.1 CLOSEOUT PROCEDURES

- A. Operating and Maintenance Instructions: Arrange for each installer of equipment that requires regular maintenance to meet with the Owner's personnel to provide instruction in proper operation and maintenance. If installers are not experienced in procedures, provide instruction by manufacturer's representatives. Include a detailed review of the following items:
  - 1. Maintenance manuals.
  - 2. Record documents.
  - 3. Warranties and bonds.
  - 4. Maintenance agreements and similar continuing commitments.

# 3.2 FINAL CLEANING

- A. General: General cleaning during construction is required by the General Conditions and included in Section "Temporary Facilities".
- B. Cleaning: Employ experienced workers or professional cleaners for final cleaning. Clean each surface or unit to the condition expected in a normal, commercial building cleaning and maintenance program. Comply with manufacturer's instructions.

- 1. Complete the following cleaning operations before requesting inspection for Certification of Notice of Completion.
  - Clean the site, including landscape development areas, of rubbish, litter and other foreign substances. Sweep paved areas broom clean; remove stains, spills and other foreign deposits.
- C. Removal of Protection: Remove temporary protection and facilities installed for protection of the Work during construction.
- D. Compliance: Comply with regulations of authorities having jurisdiction and safety standards for cleaning. Do not burn waste materials. Do not bury debris or excess materials on the Owner's property. Do not discharge volatile, harmful or dangerous materials into drainage systems. Remove waste materials from the site and dispose of in a lawful manner.

### 3.3 FORMS

- A. In-Service Certification Forms shall be used for all documentation of in-service activities. Copies of forms shall be included in maintenance and operation manuals.
- B. Asbestos Certification shall be used for documentation of non-asbestos materials used in project.
- C. PCB Certification shall be used for documentation of non-PCB materials used in project.
- D. Lead Certification shall be used for documentation of non-lead materials used in project.
- E. Affidavit of Payment of Debts and Claims and Release of Stop Notices shall be notarized and submitted as part of the project closeout requirement.
- F. Extra Materials Receipt shall be used for documentation of extra materials required by construction document; to be submitted to College M&O Director.
- G. Training Attendees Form shall be used for documentation of College staff that participated in the training scheduled and coordinated by the Contractor.

# IN-SERVICE CERTIFICATION

|  | EKIII  | -ICATIO  | <u> </u>                                      | Specificati                                       | ion Section No  |
|--|--|--|---|---|---|
| Pro  | ject:  | G-8 Allied H   | ealth Lab Remodel                             |   | Date:   |
| Owner: West Kern Community College Di Architect: AP Architects  Contractor: Inspector: |  |  | F   | File No:  Project No: 430-0083 cation No:         |   |
| In-S   | Service C  | onducted By:_  |   |   |   |
| Mat  | terials Re   | viewed: (Ched  | ck Applicable Boxes)                          |   |   |
| Atte   | Operation Manuals Special Tendees (F District F Represen | es ance Agreemer n & Maintenace Fools and Parts Please print nar | ☐ Cleaning Identifica  ■ Start-Up □ Control S | rocedures Procedures Ition Systems Sequences Ivin | <ul> <li>☐ Controls Manipulation</li> <li>☐ Emergency Procedures</li> <li>☐ Noise/Vibration</li> <li>Adjustment</li> <li>☐ Effective Energy</li> <li>Utilization</li> </ul> |
| 3.   | District M<br>Represei                                   | faintenance<br>ntatives  | (Plumbing)                                    |   | (Mechanical)  |
| 4.   |  | resent (P.I.<br>Contractor)                                      | (Electrical)                                  |   | (Grounds)   |
| Ме   | eting Date   | e  | Time of Start                                 |   | Time of Completion  |
|  | ·  |  | ed in-service covered a                       | Data  | ne specialty for which it was convened.   |

# ASBESTOS CERTIFICATION

| <u>CERTI</u> | FICATION                                 | Specification Section No   |                              |  |
|--------------|--|----------------------------|------------------------------|--|
| Project:     | G-8 Allied Health Lab Remodel            | Date:                      |                              |  |
| Owner:       | West Kern Community College Dis          | trict                      |                              |  |
| Architect:   | AP Architects                            | File No:                   |                              |  |
| Contractor:  |  | Project No:                | 430-0083                     |  |
| Inspector:   |  | DSA Application No:        |                              |  |
|              |  |                            |                              |  |
|              |  |                            |                              |  |
| To:          |  |                            |                              |  |
| 10.          |  |                            |                              |  |
| From:        |  |                            |                              |  |
| Subject      | Asbestos Containing Building Mat         | erials Letter              |                              |  |
|              | I hereby certify that, to the best of my | / knowledge, the materials | s furnished and/or installed |  |
|              | by(Name of Contractor)                   |                            | or its                       |  |
|              | (Name of Contractor)                     |                            |                              |  |
|              | subcontractor for                        |                            | located at                   |  |
|              | (Name of Project)                        |                            |                              |  |
|              |  | d                          | o not contain Asbestos-      |  |
|              | (Project Address)                        |                            |                              |  |
|              | Containing Building Materials.           |                            |                              |  |
|              |  |                            |                              |  |
| Signed:      |  |                            |                              |  |
| Position:    |  |                            |                              |  |
| Dated:       |  |                            |                              |  |

# PCB CERTIFICATION

Position:

Dated:

| CERII       | FICATION                                 | Specification Section    | n No                         |
|-------------|--|--------------------------|------------------------------|
| Project:    | G-8 Allied Health Lab Remodel            | Date:                    |                              |
| Owner:      | West Kern Community College Dist         | trict                    |                              |
| Architect:  | AP Architects                            | File No:                 |                              |
| Contractor: |  |                          | 430-0083                     |
| Inspector:  |  | DSA Application No:      |                              |
|             |  |                          | _                            |
|             |  |                          |                              |
|             |  |                          |                              |
| То:         |  |                          |                              |
|             |  |                          |                              |
| From:       |  |                          |                              |
| Subject     | PCB Containing Building Materials        | Letter                   |                              |
|             | I hereby certify that, to the best of my | knowledge, the materials | s furnished and/or installed |
|             | by                                       |                          | or its                       |
|             | (Name of Contractor)                     |                          |                              |
|             | subcontractor for                        |                          | located at                   |
|             | (Name of Project)                        |                          |                              |
|             |  | d                        | o not contain PCB-           |
|             | (Project Address)                        |                          |                              |
|             | Containing Building Materials.           |                          |                              |
|             |  |                          |                              |
| Signed:     |  |                          |                              |

# LEAD CERTIFICATION

Position:

Dated:

| CERII       | FICATION                                 | Specification Section No |                              |  |
|-------------|--|--------------------------|------------------------------|--|
| Project:    | G-8 Allied Health Lab Remodel            | Date:                    |                              |  |
| Owner:      | West Kern Community College Dis          | trict                    |                              |  |
| Architect:  | AP Architects                            | File No:                 |                              |  |
| Contractor: |  | Project No:              | 430-0083                     |  |
| Inspector:  |  |                          |                              |  |
|             |  |                          |                              |  |
|             |  |                          |                              |  |
| То:         |  |                          |                              |  |
| From:       |  |                          |                              |  |
| Subject     | Lead Containing Building Materials       | s Letter                 |                              |  |
|             | I hereby certify that, to the best of my | knowledge, the materials | s furnished and/or installed |  |
|             | by                                       |                          | or its                       |  |
|             | (Name of Contractor)                     |                          |                              |  |
|             | subcontractor for                        |                          | located at                   |  |
|             | (Name of Project)                        |                          | located at                   |  |
|             |  | d                        | o not contain PCB-           |  |
|             | (Project Address)                        |                          |                              |  |
|             | Containing Building Materials.           |                          |                              |  |
|             |  |                          |                              |  |
| Signed:     |  |                          |                              |  |

# AFFIDAVIT OF PAYMENT OF DEBTS AND CLAIMS AND RELEASE OF STOP NOTICES

| Project:   | G-8 Allied Health Lab Remodel   | Date:   |
|--|---|---|
| Owner:<br>Architect:<br>Contractor:<br>Inspector:            | West Kern Community College Dis   | File No: Project No: 430-0083 DSA Application No:   |
| retainage for<br>known indeb<br>against the F<br>does hereby | all materials and equipment furnish<br>stedness and claims against the und<br>PROJECT, its land, improvements, an | es that on this date, he/she has been paid in full le<br>ed, for all labor and services performed, and for<br>ersigned for damages arising in any manner on<br>d equipment of any kind. Therefore, the undersign<br>as and/or stop notices against the PROJECT as of<br>O |
|  |   | COMPANY:  |
|  |   | BY:   |
|  |   | TITLE:  |
| STATE OF _   |   |   |
| COUNTY OF  | =   |   |
| Known to m duly sworn, therein expre                         | acknowledged to me that he exectes essed and declared to me that the sta  | scribed to the foregoing instrument, and being futed the same for the purposes and considerat tements contained therein are true.   |
| Sworn and s  | ubscribed to before me this day   | of year of 20   |
|  |   |   |
|  |   | Notary Public in and for said<br>State and County   |
|  |   | My commission expires:  |

# EXTRA MATERIALS RECEIPT

| RECEIPT    |                          |                                    | Specification Section     | n No                    |
|------------|--------------------------|------------------------------------|---------------------------|-------------------------|
| Project:   | G-8                      | Allied Health Lab Remodel          | Date:                     |                         |
| Owner:     | Wes                      | st Kern Community College Distr    | rict                      |                         |
| Architect: |                          | Architects                         | File No:                  |                         |
| Contracto  |                          |                                    | Project No:               |                         |
| Inspector  | :                        |                                    | DSA Application No:       |                         |
|            |                          |                                    |                           |                         |
|            |                          |                                    |                           |                         |
| List of Ma | aterials to              | surrender to M&O office/Directo    | r:                        |                         |
| Item #     | Qty.                     | Description                        |                           |                         |
|            |                          |                                    |                           |                         |
|            |                          |                                    |                           |                         |
|            |                          |                                    |                           |                         |
|            |                          |                                    |                           |                         |
|            |                          |                                    |                           |                         |
|            |                          |                                    |                           |                         |
|            |                          |                                    |                           |                         |
|            |                          |                                    |                           |                         |
|            |                          |                                    |                           |                         |
|            | nat the abo<br>ion docum | ove extra materials are all in goo | od condition and meet the | e required quantity per |
| CONSTRUCT  | ion docum                | ienis.                             |                           |                         |
| Submitted  | d by:                    |                                    |                           |                         |
|            |                          |                                    |                           |                         |
| Signature  | <b>-</b>                 |                                    | Date                      |                         |
| Received   | by:                      |                                    |                           |                         |
|            |                          |                                    |                           |                         |

PROJECT CLOSEOUT 017750 - 10

Signature \_\_\_\_\_ Date \_\_\_\_

# TRAINING ATTENDEES FORM

| Specification Section No. |  |
|---------------------------|--|
|---------------------------|--|

| Project:                                 | G-8 Allied Health Lab Remodel    | Date:                   |           |
|--|----------------------------------|-------------------------|-----------|
| Owner: Architect: Contractor: Inspector: | West Kern Community College Dist | File No:<br>Project No: | 430-0083  |
|  |                                  |                         |           |
| Description o                            | f Training:                      |                         |           |
| List of Attend                           | ees:                             |                         |           |
|  | Name                             | Department              | Signature |
|  |                                  |                         |           |
|  |                                  |                         |           |
|  |                                  |                         |           |
|  |                                  |                         |           |
|  |                                  |                         |           |
| Submitted by                             | :                                |                         |           |
| Signature                                | Contractor                       | Date                    |           |
| Acknowledge                              | ed by:                           |                         |           |
| Signature                                | Project Inspector                | Date                    |           |

END OF SECTION 017750

### PART 1 - GENERAL

### 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of Contract, including General and Supplementary Conditions and other Division 0 & 1 Specification Sections, apply to this Section.

### 1.2 SUMMARY

- A. This Section specifies general administrative and procedural requirements for warranties and bonds required by the Contract Documents, including manufacturer's standard warranties on products and special warranties.
  - 1. Refer to the General Conditions for terms of the Contractor's special warranty of workmanship and materials.
  - 2. General closeout requirements are included in Section "Project Closeout."
  - 3. Specific requirements for warranties for the Work and products and installations that are specified to be warranted, are included in the individual Sections of Divisions 2 through 33.
  - 4. Certifications and other commitments and agreements for continuing services to Owner are specified elsewhere in the Contract Documents.
  - 5. Items that are omitted and/or different than specified/indicated herein and on Construction Documents and items not indicated as a change on Submittals, shall be warrantied as required in Sections 006002 and 013300.
- B. Disclaimers and Limitations: Manufacturer's disclaimers and limitations on product warranties do not relieve the Contractor of the warranty on the Work that incorporates the products, nor does it relieve suppliers, manufacturers, and subcontractors required to countersign special warranties with the Contractor.
- C. Notwithstanding any tests, approvals, certificates, commissioning, inspection or otherwise by the Owner, Architect or any other consultant employed by or on behalf of the employer, the Contractor shall be and remain fully and exclusively responsible and liable for ensuring that his works, and all goods and materials therein are in every respect and detail in accordance with the Contract Documents, and no such tests, approval certificates, commissioning, inspection or otherwise shall in any way diminish or negate the Contractor's responsibility or liability as foresaid.

# 1.3 WARRANTY REQUIREMENTS

- A. Related Damages and Losses: When correcting warranted Work that has failed, remove and replace other Work that has been damaged as a result of such failure or that must be removed and replaced to provide access for correction of warranted Work.
- B. Reinstatement of Warranty: When Work covered by a warranty has failed and been corrected by replacement or rebuilding, reinstate the warranty by written endorsement. The reinstated warranty shall be equal to the original warranty with an equitable adjustment for depreciation.
- C. Replacement Cost: Upon determination that Work covered by a warranty has failed, replace or rebuild the Work to an acceptable condition complying with requirements of Contract Documents. The Contractor is responsible for the cost of replacing or rebuilding defective Work regardless of whether the Owner has benefited from use of the Work through a portion of its anticipated useful service life.
- D. Owner's Recourse: Written warranties made to the Owner are in addition to implied warranties, and shall

not limit the duties, obligations, rights and remedies otherwise available under the law, nor shall warranty periods be interpreted as limitations on time in which the Owner can enforce such other duties, obligations, rights, or remedies.

- 1. Rejection of Warranties: The Owner reserves the right to reject warranties and to limit selections to products with warranties not in conflict with requirements of the Contract Documents.
- E. The Owner reserves the right to refuse to accept Work for the Project where a special warranty, certification, or similar commitment is required on such Work or part of the Work, until evidence is presented that entities required to countersign such commitments are willing to do so.

### 1.4 SUBMITTALS

- A. Submit written warranties to the Architect prior to the date certified for Notice of Completion. If the Architect's Certificate of Notice of Completion designates a commencement date for warranties other than the date of Notice of Completion for the Work, or a designated portion of the Work, submit written warranties upon request of the Architect.
  - 1. When a designated portion of the Work is completed and occupied or used by the Owner, by separate agreement with the Contractor during the construction period, submit properly executed warranties to the Architect within fifteen days of completion of that designated portion of the Work.
- B. When a special warranty is required to be executed by the Contractor, or the Contractor and a subcontractor, supplier or manufacturer, prepare a written document that contains appropriate terms and identification, ready for execution by the required parties. Submit a draft to the Owner through the Architect for approval prior to final execution.
  - 1. Refer to individual Sections of Divisions 2 through 33 for specific content requirements, and particular requirements for submittal of special warranties.
- C. Bind warranties and bonds in heavy-duty, commercial quality, durable 3-ring vinyl covered loose-leaf binders, thickness as necessary to accommodate contents, and sized to receive 8-1/2" by 11" paper.
  - 1. Provide heavy paper dividers with celluloid covered tabs for each separate warranty. Mark the tab to identify the product or installation. Provide a typed description of the product or installation, including the name of the product, and the name, address and telephone number of the installer.
  - 2. Identify each binder on the front and the spine with the typed or printed title "WARRANTIES AND BONDS, the Project title or name, and the name of the Contractor.
  - 3. When operating and maintenance manuals are required for warranted construction, provide additional copies of each required warranty, as necessary, for inclusion in each required manual.

PART 2 - PRODUCTS (Not Applicable).

PART 3 - EXECUTION

# 3.1 FORMS

- A. Project Warranty Form, See attached.
- B. Subcontractor Warranty Form, See attached. END OF SECTION 017800

# PROJECT WARRANTY

| Project: TC – G-8 ALLIED HEALTH   | _AB REMODEL  | Date:  |
|---|--|--|
| Taft College  | District   |  |
| Owner: West Kern Community Colle Architect: AP Architects   | ge District  | File No:   |
| Contractor:   |  | Project No: 430-0083   |
| Inspector:  |  | DSA Áppl No:   |
|   |  |  |
| under the Contract in the required or permitted by the Contract Dopermitted, and that the Work conforms requirements, including substitution not  | (Name of Project) are ocuments, that the Work is free from def with the requirements of the Contract D properly approved and authorized, may   | er that materials and equipment furnished of good quality and new unless otherwise ects not inherent in the quality required or ocuments. Work not conforming to these be considered defective. This warranty down the Contractor impresses as insufficient  |
| maintenance, improper operation, or norm  |  | d by the Contractor, improper or insufficient  |
| applicable special warranty required by the accordance with the requirements of the Contractor expressly agrees to correct it, or his agent to do so unless the Owner have year(s) shall be extended with respetime between Notice of Completion and the special process. | e Contract Documents extending this time e Contract Documents or proves to be described without expense to the Owner, promptly as previously given the Contractor written ect to portions of the Work first performed the actual performance of Work. This oblines   | ignated portion thereof, or by terms of an period, and of the Work is found to be not in defective in materials or workmanship, the fter receipt of written notice from the Owner acceptance of the condition. This period of after Notice of Completion by the period of gation of the Contractor to correct the Work ontract. The Owner shall give such notice |
| the Contractor might have under the Cont<br>time periods required by the Contract Do<br>obligation of the Contractor to correct the   | ract Documents. Establishment of the time ocuments, for correction of the Work as owork, and has no relationship to the time versions and the time within which present the control of the time within which present the control of the time within which present the control of the time within which present the control of the time within which present the control of the | tion with respect to other obligations which e period of year(s), or special extended escribed above relates only to the specific within which the obligation to comply with the occeedings may be commenced to establish cally to correct the Work.   |
|   | or hereby authorizes the Owner to proceed  | within 5 days after being notified in writing to have said defects repaired and made d charges therefore upon demand.  |
|   | d by the Contractor to fulfill the Contractor  | Documents and includes all other labor, 's obligations. The Work may constitute the  |
| Date  |  |  |
| Contractor  |  |  |
| Address   |  |  |
|   |  |  |
| Telephone   |  |  |
| Signature of Contractor   |  |  |
| Title   |  |  |

# SUBCONTRACTOR WARRANTY

| Project: TC – G-8 ALLIED HEA   | LTH LAB REMODEL   | Date:   |  |
|--|---|---|--|
| Taft College<br>Owner: West Kern Community C<br>Architect: AP Architects   | College District  | File No:  |  |
| Contractor: Inspector:   |   | Project No:<br>DSA Appl No:   | 430-0083   |
| Contractor) that materials and in the  |   | rrants to   |  |
| otherwise required or permitted by required or permitted, and that the V these requirements, including substit   | the Contract Documents, that the Vork conforms with the requireme utions not properly approved and fect caused by abuse, modifica   | e Work is free from defects not inher<br>nts of the Contract Documents. Work<br>authorized, may be considered defective<br>tions not executed by the Subcontra  | ent in the quality<br>not conforming to<br>ve. This warranty   |
| special warranty required by the C accordance with the requirements of Subcontractor expressly agrees to c Contractor to do so unless the Owne year (s) shall be extended with respective notice of Completion and the contractor and the contractor to the contractor to do so unless the Owne year (s) shall be extended with respective to the contractor of th | ontract Documents extending thing the Contract Documents or properties it, without expense to the representation has previously given the Contract to portions of the Work first perhecet to performance of Work.  Under the Contract and termination | or designated portion thereof, or by term is time period, and of the Work is for oves to be defective in materials or Owner, promptly after receipt of written overtien acceptance of the condition. In order of the Condition of the Subcontractor to on of the Contract. The Contractor shall | und to be not in workmanship, the n notice from the This period of the period of time correct the Work |
| the Subcontractor might have unde<br>extended time periods required by the<br>specific obligation of the Subcontract<br>comply with the Contract Documer   | r the Contract Documents. Esta<br>ne Contract Documents, for correct<br>tor to correct the work, and has noted to be enforced<br>to may be sought to be enforced.   | eriod of limitation with respect to other ablishment of the time period of yetion of the Work as described above roo relationship to the time within which ed, nor to the time within which procedure the Subcontractor's obligations other the   | ear(s), or special<br>relates only to the<br>n the obligation to<br>ceedings may be                    |
| writing by the Contractor, the Subco   | intractor, hereby authorizes the C  | ons of this warranty within 5 days afte<br>Contractor to proceed to have said def<br>ill honor and pay the costs and charge   | ects repaired and  |
|  | provided by the Subcontractor to  | the Contract Documents and include fulfill the Subcontractor's obligations. e Contract Documents.   |  |
| Date   |   |   |  |
| Contractor   |   |   |  |
| Address  |   |   |  |
| Telephone  |   |   |  |
| Signature of Contractor  |   |   |  |
| Title  |   |   |  |

### PART 1 – GENERAL

### 1.1 RELATED DOCUMENTS

- A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 0 &1 Specification Sections, apply to this Section.
  - Standard details and requirements per jurisdiction requirements, not limited to herein.

### 1.2 SUMMARY

- A. This Section includes, but not limited to, construction waste diversion and related items.
  - Establish a Construction Waste Management (CWM) plan for the diverted materials, or meet local construction and demolition waste management ordinance, whichever is more stringent.
  - 2. Construction Waste Management (CWM) worksheet(s).
  - 3. Construction Waste Management (CWM) acknowledgement.
  - 4. Miscellaneous and related forms as required by local jurisdiction and Owner.
- B. Related Sections: The following Sections contain requirements that relate to this Section:
  - 1. Division 1 Section "Project Closeout" for required closeout documents.
  - 2. Division 2 Section "Building Demolition" for related items.
  - 3. Division 2 Section "Selective Demolition" for related items.

### 1.3 SUBMITTAL

- A. General: Submit for approval prior to initial pay application submission the following in accordance with Conditions of Contract and Division 1 Specification Sections.
  - 1. Construction Waste Management Plan.
  - 2. Construction Waste Management Worksheet(s).
  - 3. Construction Waste Management Acknowledgement.
  - 4. Copies of all applications, permits and related requirements, not limited to, herein and local jurisdiction requirements.
  - 5. Project Closeout Requirements:
    - a. Copies of all completed forms required herein
    - b. Compilation of Construction Waste Management (CWM) worksheet(s).
    - c. Receipts to document diversion.

### 1.4 QUALITY ASSURANCE

- A. Codes and Standards: Comply with provisions of following codes, specifications, and standards, except where more stringent requirements are shown or specified:
  - 1. Title 24, Part 1 CCR 2013, Chapter 5 (California Green Building Standard Code)
- B. Refer to Division 2 Section "Selective Demolition", for additional requirements.
- B. Refer to Division 2 Section "Building Demolition" for additional requirements.

# PART 2 – PRODUCTS (NOT APPLICABLE)

# PART 3 - EXECUTION

### 3.1 CONSTRUCTION WASTE REDUCTION, DISPOSAL AND RECYCLING

- A. Construction waste diversion. Establish a construction waste management plan for the diverted materials, or meet local construction and demolition waste management ordinance, whichever is more stringent.
- B. Construction waste management plan (CWMP). Where a local jurisdiction does not have a construction and demolition waste management ordinance, submit a construction waste management plan for approval by the District that:
  - 1. Identifies the materials to be diverted from disposal by efficient usage, recycling, reuse on the project or salvage for future use or sale.
  - 2. Determine if materials will be sorted on-site or mixed.
  - 3. Identifies diversion facilities where materials collected will be taken.
  - 4. Specifies that the amount of materials diverted shall be calculated by weight or by volume, but not both.
  - 5. See sample forms at end of section.
- C. Documentation. Documentation shall be provided to the enforcing agency which demonstrates compliance with herein. The waste management plan shall be updated as necessary and shall be accessible during construction for examination by the District and Architect.
- D. Construction waste reduction of at least 65 percent. Recycle and/ or salvage for reuse a minimum of 65 percent of the non- hazardous construction and demolition debris, or meet a local construction and demolition waste management ordinance, whichever is more stringent. Calculate the amount of materials diverted by weight or volume, but not both.

# Exceptions:

- 1. Excavated soil and land-clearing debris.
- E. Excavated soil and land clearing debris. 100 percent of trees, stumps, rocks and associated vegetation and soils resulting primarily from land clearing shall be reused or recycled. For a phased project, such material may be stockpiled on site until the storage site is developed, if approved by the District and Architect within 30 days of Notice to Proceed.

# 3.2 DISPOSAL OF WASTE/ DEMOLISHED MATERIALS

- A. Remove from building site debris, rubbish, and other materials resulting from demolition operations. Transport and legally disposed off site per Construction Waste Management Plan.
  - 1. If hazardous materials are encountered during demolition operations, comply with applicable regulations, laws, and ordinances concerning removal, handling, and protection against exposure or environmental pollution.
  - 2. Burning of removed materials is not permitted on project site.
- B. Demolition Permits and Disposal Requirements (Sample forms at end section):
  - 1. The District requires diversion of 65% of discarded materials from landfills, reusing for compliance with AB 939 goals. Contractor shall secure a demolition permit with local jurisdiction as prescribed in Section 002113 "Instructions to Bidders". The following requirements are:

- a. Prepare Construction Waste Management Plan.
- Provide for all Contractor, subcontractor, vendor, workers a Construction Waste Management Acknowledgement. Keep files at project site. Provide copies for Project Closeout.
- c. Complete Construction and Demolition Recycling Program Application as required by local jurisdiction. Verify requirements prior to bidding.
- d. Inform all sub-contractors, vendors and employees about the recycling requirements, as Contractor will be held responsible for any materials they take away from the job site. Document acknowledgement with Construction Waste Management Acknowledgement Report. In order for Contractor to comply with recycling requirements, vendors, employees and subcontractors must take materials to an approved disposal/ recycling site and keep copies of disposal tickets for diversion rate.
  - 1. Track waste by weight or volume, but not both.
- e. Compliance: District will not accept project until it is determined that the Construction and Demolition waste/disposal has fully complied with the diversion requirements and recycling guidelines.
- f. Non-Compliance: If the Contractor fails to adhere to the Construction and Demolition Recycling Plan, the applicant will not receive project acceptance.
- g. Contractor shall complete a Construction Waste Management Worksheet for documenting compliance with the Waste Management Plan after completion of project, as a precedent to project acceptance.

## B. Demolition Permits and Disposal Requirements:

- 1. The local jurisdiction requires diversion of 65% of discarded materials from landfills, reusing for compliance with AB 939 goals. Contractor shall secure a demolition permit as prescribed in Section 002113 "Instructions to Bidders". The following requirements are:
  - a. Complete Construction and Demolition Recycling Program Application.
  - b. The franchise hauler Mid Valley Disposal shall be the exclusive service provider for roll-off service and shall transport the Construction and Demolition material to their transfer station located at 15300W. Jensen Avenue, Kerman, California, for all projects that require a permit to build or deconstruct to spate material and divert a minimum of sixty-five percent (65%) of Construction and Demolition material form projects. Inform all sub-contractors, vendors and employees about the recycling requirements, as Contractor will be held responsible for any materials they take away from the job site. In order for Contractor to comply with recycling requirements, vendors, employees and subcontractors must take materials to Mid Valley Disposal and keep copies of weight tickets for diversion rate.
  - c. Compliance: If the local entity determines that the Construction and Demolition has fully complied with the diversion requirements and recycling guidelines, the local entity will provide final inspection and/or certificate of occupancy.
  - d. Non-Compliance: If the Contractor fails to adhere to the Construction and Demolition Recycling Plan, the applicant will not receive final inspection and/or certificate of occupancy.
  - Contractor shall complete a Waste Reduction and Recycling Report (WRRR) after completion of project, as a precedent to final inspection and/or issuance of any certificate of occupancy is approved by local entity.

END OF SECTION 018000

# CONSTRUCTION WASTE MANAGEMENT (CWM) PLAN Note: This sample form may be used to assist in documenting compliance with the waste managem

|        | Note. This sample form may be used to assist in documenting compilance with the waste management plan.  |
|--------|---|
| Projec | ot Name:  |
| Job #: |   |
| Projec | ot Manager:   |
| vvaste | e Hauling Company:  |
| Conta  | ct Name:  |
|        | bcontractors shall comply with the project's Construction Waste Management Plan.<br>bcontractor foremen shall sign the CWM Plan Acknowledgement Sheet.  |
| payme  | ontractors who fail to comply with the Waste Management Plan will be subject to backcharges or withholding of ent, as deemed appropriate. For instance, Subcontractors who contaminate debris boxes that have been designed single material type will be subject to backcharge or withheld payment, as deemed appropriate.  |
| 1.     | The project's overall rate of waste diversion will be %.  |
| 2.     | This project shall generate the least amount of waste possible by planning and ordering carefully, following all  |
|        | proper storage and handling procedure to reduce broken and damaged material and reusing materials whenever possible. The majority of the waste that is generated on this jobsite will be diverted from the landfill and recycled for other use.   |
| 3.     | Spreadsheet 1, enclosed, identifies the waste materials that will be generated on this project, the diversion strategy for each waste type and the anticipated diversion rate.  |
| 4.     | Waste prevention and recycling activities will be discussed at the beginning of weekly subcontractor meetings.  |
|        | As each new subcontractor comes on-site, the WMP Coordinator will present him/her with a copy of the CWM Plan and provide a tour of the jobsite to identify materials to be salvaged and the procedure for handling jobsite debris. All Subcontractor foremen will acknowledge in writing that they have read and will abide by the CWM Plan. Subcontractor Acknowledgement Sheet enclosed. The CWM Plan will be posted at the jobsite trailer. |
| 5.     | Salvage: Excess materials that cannot be used in the project, nor returned to the vendor, will be offered to site   |
|        | workers, the owner, or donated to charity if feasible.  |
| 6.     | [HAULING COMPANY] will provide a commingled drop box at the jobsite for most of the construction waste. These commingled drop boxes will be taken to [Sorting Facility Name and Location]. The average diversion for commingled waste will be%. As site conditions permit, additional drop boxes will be used for particular phases of construction (e.g. concrete and wood waste) to ensure the highest waste diversion rate possible.         |
| 7.     | In the event that the waste diversion rate achievable via the strategy described in (6) above, is projected to be lower than what is required, then a strategy of source-separated waste diversion and/or waste stream reduction will be implemented. Source separated waste refers to jobsite waste that is not commingled but is instead  |
|        | allocated to a debris box designated for a single material type, such as clean wood or metal.  Notes:   |
|        | a. Waste stream reduction refers to efforts taken by the builder to reduce the amount of waste generated by the project to below four (4) pounds per square foot of building area.  |
|        | b. When using waste stream reduction measures, the gross weight of the product is subtracted from the base  |
|        | weight of four (4) pounds per square foot of building area. This reduction is considered additional diversion and can be used in the waste reduction percentage calculations.   |
| 8.     | [HAULING COMPANY] will track and calculate the quantity (in tons) of all waste leaving the  |
|        | project and calculate the waste diversion rate for the project. [HAULING COMPANY] will provide Project  |
|        | Manager with an updated monthly report on gross weight hauled and the waste diversion rate being achieved on the project. [HAULING COMPANY's] monthly report will track separately the gross weights and diversion rates  |
|        | for commingled debris and for each source —separated waste stream leaving the project. In the event the   |
|        | [HAULING COMPANY] does not service any or all of the debris boxes on the project, the [HAULING COMPANY]   |
|        | will work with the responsible parties to track the material type and weight (in tons) in such debris boxes in order  |
| _      | to determine waste diversion rates for these materials.   |
| 9.     | In the event that the Subcontractors furnish their own debris boxes as part of their scope of work, such  |
|        | Subcontractors shall not be excluded from complying with the CWM Plan and will provide [HAULING COMPANY] weight and waste diversion data for their debris boxes.  |
| 10.    | In the event that site use constraints (such as limited space) restrict the number of debris boxes that can be used   |
|        | for collection of designed waste the project Superintendent will, as deemed appropriate, allocate specific areas  |
|        | onsite where individual material types are to be consolidated. These collection points are not to be contaminated   |
|        | with non-designated waste types.  |

Debris from jobsite office and meeting rooms will be collected by [DISPOSAL SERVICE COMPANY]. [DISPOSAL SERVICE COMPANY] will, at a minimum recycle office paper, plastic, metal and cardboard.

# CONSTRUCTION WASTE MANAGEMENT (CWM) WORKSHEET

Note: This sample form may be used to assist in documenting compliance with the waste management plan.

| Project Name:  |                                |                          |                          |  |  |
|--|--------------------------------|--------------------------|--------------------------|--|--|
| Job Name:  |                                |                          |                          |  |  |
| Project Manager:   |                                |                          |                          |  |  |
| Waste Hauling Company:   |                                |                          |                          |  |  |
| Construction Waste Management (CWM) Plan                                     |                                |                          |                          |  |  |
| DIVERSION METHOD   |                                |                          |                          |  |  |
| WASTE MATERIAL TYPE  | COMMINGLED AND SORTED OFF SITE | SOURCE SEPARATED ON SITE | PROJECTED DIVERSION RATE |  |  |
| Asphalt  |                                |                          |                          |  |  |
| Concrete   |                                |                          |                          |  |  |
| Shotcrete  |                                |                          |                          |  |  |
| Metals   |                                |                          |                          |  |  |
| Wood   |                                |                          |                          |  |  |
| Rigid Insulation   |                                |                          |                          |  |  |
| Fiberglass Insulation  |                                |                          |                          |  |  |
| Acoustic ceiling tile  |                                |                          |                          |  |  |
| Gypsum drywall   |                                |                          |                          |  |  |
| Carpet/ carpet pad   |                                |                          |                          |  |  |
| Plastic pipe   |                                |                          |                          |  |  |
| Plastic buckets  |                                |                          |                          |  |  |
| Plastic  |                                |                          |                          |  |  |
| Hardiplank siding and boards   |                                |                          |                          |  |  |
| Glass  |                                |                          |                          |  |  |
| Cardboard  |                                |                          |                          |  |  |
| Pallets  |                                |                          |                          |  |  |
| Job office trash, paper,<br>glass & plastic bottles, cans,<br>plastic        |                                |                          |                          |  |  |
| Alkaline and rechargeable batteries, toner cartridges and electronic devices |                                |                          |                          |  |  |
| Others:  |                                |                          |                          |  |  |
| Others:  |                                |                          |                          |  |  |
| Others:  |                                |                          |                          |  |  |
| Others:  |                                |                          |                          |  |  |
| Others:  |                                |                          |                          |  |  |

# **CONSTRUCTION WASTE MANAGEMENT (CWM) ACKNOWLEDGE**

Note: This sample form may be used to assist in documenting compliance with the waste management plan. **Project Name:** Job Name: **Project Manager:** Waste Hauling Company: \_\_ **CWM Plan Acknowledgement** The Foreman for each new Subcontractor that comes on site is to receive a copy of the Construction Waste Management Plan and complete this Acknowledgement Form. I have read the Waste Management Plan for the project; I understand the goals of this plan and agree to the follow the procedures described in this plan. SUBCONTRACTOR DATE **FOREMAN NAME SIGNATURE COMPANY NAME** 

### PART 1 - GENERAL

### 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of Contract, including General and Supplementary Conditions and Division 0 & 1 Specification Sections, apply to this Section.

### 1.2 SUMMARY

- A. This Section requires the selective removal and subsequent offsite disposal, unless noted otherwise, but not limited to, the following:
  - 1. Removal of existing items indicated on Drawings and miscellaneous items required to complete project.
- B. Related Sections: The following Sections contain requirements that relate to this Section:
  - 1. Division 1 Section "Construction Waste Reduction, Disposal and Recycling" for waste and disposal requirements.

### 1.3 DEFINITIONS

- A. Remove: Detach items from existing construction and legally dispose of them off-site unless indicated to be removed and salvaged or removed and reinstalled.
- B. Remove and Salvage: Carefully detach from existing construction, in a manner to prevent damage, and deliver to Owner ready for reuse.
- C. Remove and Reinstall: Detach items from existing construction, prepare for reuse, and reinstall where indicated.
- D. Existing to Remain: Existing items of construction that are not to be permanently removed and that are not otherwise indicated to be removed, removed and salvaged, or removed and reinstalled.

### 1.4 MATERIALS OWNERSHIP

- A. Unless otherwise indicated, demolition waste becomes property of Contractor.
- B. Historic items, relics, antiques, and similar objects including, but not limited to, cornerstones and their contents, commemorative plaques and tablets, and other items of interest or value to Owner that may be uncovered during demolition remain the property of Owner.
  - 1. Carefully salvage in a manner to prevent damage and promptly return to Owner.

## 1.5 SUBMITTALS

A. Digital photographs of existing conditions of structure surfaces, equipment, and adjacent improvements that might be misconstrued as damage related to removal operations. Provide copies in submittal; provide on digital storage device or provide download link prior to start of work.

### 1.6 JOB CONDITIONS

- A. Occupancy: Owner will occupy portions of the building immediately adjacent to areas of selective demolition. Conduct selective demolition work in manner that will minimize need for disruption of Owner's normal operations. Provide minimum of 72 hours advance notice to Owner of demolition activities that will affect Owner's normal operations.
  - Refer to Section "Summary of Work" for limitations on noise generations, access and other restrictions
- B. Condition of Facilities: Owner assumes no responsibility for actual condition of items or facilities to be demolished.
  - Conditions existing at time of inspection for bidding purposes will be maintained by Owner insofar as practicable. However, minor variations may occur by Owner's removal and salvage operations prior to start of selective demolition work.
  - Owner has first right of refusal of salvaged items. Contractor shall dispose of all items
     Owner does not want. Owner will walk through project prior to start of demolition and mark
     items for salvage. Contractor to schedule walk within 7 days written advance notification of
     architect, owner, subcontractors and consulting engineers.
- C. Partial Demolition and Removal: Items indicated to be removed but of salvageable value to Contractor may be removed as work progresses. Transport salvaged items from site as they are removed. Document all items on Construction Waste Management Plan and reports.
  - 1. Storage or sale of removed items on site will not be permitted.
- D. Protections: Provide temporary barricades and other forms of protection to protect Owner's personnel and general public from injury due to selective demolition work.
  - 1. Provide protective measures as required to provide free and safe passage of Owner's personnel and general public to occupied portions of building.
  - 2. Erect temporary covered passageways as required by authorities having jurisdiction.
  - 3. Provide interior and exterior shoring, bracing, or support to prevent movement, settlement, or collapse of structure or element to be demolished and adjacent facilities or work to remain
  - 4. Protect from damage existing finish work that is to remain in place and becomes exposed during demolition operations.
  - 5. Protect floors with suitable coverings when necessary.
  - 6. Construct temporary insulated dustproof partitions where required to separate areas where noisy or extensive dirt or dust operations are performed. Equip partitions with dustproof doors and security locks.
  - 7. Provide temporary weather protection during interval between demolition and removal of existing construction on exterior surfaces and installation of new construction to ensure that no water leakage or damage occurs to structure or interior areas of existing building.
  - 8. Close off supply and return air registers in construction area prior to starting any work.

    Maintain fresh air to construction area via fans and other means.
  - 9. Remove protections at completion of work.
  - Equipment, materials and supplies temporarily removed for protection shall be replaced in original locations. Any materials damaged shall be replaced with new materials of life kind and quality.

- 11. Protect wall, trim, floors, equipment, utility lines and materials. When working on finished surfaces limit damage to the smaller area if possible and restore to the pre-construction condition all surfaces which are damaged because of the installation of this work.
- E. Damages: Promptly repair damages caused to adjacent facilities by demolition work.
- F. Traffic: Conduct selective demolition operations and debris removal to ensure minimum interference with roads, streets, walks, and other adjacent occupied or used facilities.
  - 1. Do not close, block, or otherwise obstruct streets, walks, or other occupied or used facilities without written permission from authorities having jurisdiction. Provide alternate routes around closed or obstructed traffic ways if required by governing regulations.
- G. Flame Cutting: Do not use cutting torches for removal until work area is cleared of flammable materials. At concealed spaces, such as interior of ducts and pipe spaces, verify condition of hidden space before starting flame-cutting operations. Maintain portable fire suppression devices during flame-cutting operations.
- H. Utility Services: Maintain existing utilities indicated to remain in service and protect them against damage during demolition operations.
  - 1. Do not interrupt utilities serving occupied or used facilities, except when authorized in writing by authorities having jurisdiction. Provide temporary services during interruptions to existing utilities, as acceptable to governing authorities.
  - 2. Maintain fire protection services during selective demolition operations.
- I. Environmental Controls: Use water sprinkling, temporary enclosures, and other methods to limit dust and dirt migration. Comply with governing regulations pertaining to environmental protection.
  - 1. Do not use water when it may create hazardous or objectionable conditions such as ice, flooding, and pollution.
- J. Fire safety during demolition.
  - Combustible Debris: Combustible debris shall not be accumulated within buildings and site. Combustible debris, rubbish and waste material shall be removed from buildings as often as practical. Combustible debris, waste material and trash shall not be burned on site.
  - 2. Motor Equipment: Internal-combustion-powered construction equipment shall be used in accordance with the following:
    - Equipment shall be located so that exhausts do not discharge against combustible material.
    - b. When possible, exhausts shall be piped to the outside of the building.
    - c. Equipment shall not be refueled while in operation.
    - d. Fuel for equipment shall be stored in an approved area outside of the building and provided for containment for any spills of containers over 5 gallons.
  - 3. Cutting and Welding: Cutting and welding operations shall be in accordance with Chapter 26 California Fire Code.
  - 4. Flame-producing Equipment: The use of torches or flame-producing devices for the sweating of pipe joints shall be in accordance with Chapter 14 and Chapter 26 California Fire Code.
  - 5. Flammable Liquids: The storage, use and handling of flammable liquids shall be in accordance with Chapter 34 California Fire Code. Ventilation shall be provided for

SELECTIVE DEMOLITION 024119 - 3

- operations utilizing the application of materials containing flammable solvents.
- 6. Open-flame Devices: Open-flame devices and other sources of ignition shall not be located in areas where flammable materials are being used.
- 7. Building Access: Access to buildings for the purpose of fire fighting shall be provided per CFC 501.4 and 1412.1 and CBC 3311.4. Construction material shall not block access to buildings, hydrants or fire appliances; maintain areas around device clear at all times.
- 8. General: Demolition of buildings shall be in accordance with Chapter 14 California Fire Code.
- General: Construction of building shall be in accordance with Chapter 14 and Chapter 26

   California Fire Code.

# PART 2 - PRODUCTS (Not Applicable)

### 2.1 PEFORMANCE REQUIREMENTS

- A. Regulatory Requirements: Comply with governing EPA notification regulations before beginning selective demolition. Comply with hauling and disposal regulations of authorities having jurisdiction.
- B. Standards: Comply with ANSI/ASSE A10.6 and NFPA 241.
- C. LEED Requirements for Building Reuse:

## PART 3 - EXECUTION

## 3.1 EXAMINATION

- C. Verify that utilities have been disconnected and capped before starting selective demolition operations.
- D. Review record documents of existing construction provided by Owner. Owner does not guarantee that existing conditions are same as those indicated in record documents.
- E. Survey existing conditions and correlate with requirements indicated to determine extent of selective demolition required.
- F. When unanticipated mechanical, electrical, or structural elements that conflict with intended function or design are encountered, investigate and measure the nature and extent of conflict. Promptly submit a written report to Architect.

### 3.2 PREPARATION

- A. General: Provide interior and exterior shoring, bracing, or support to prevent movement, settlement, or collapse of areas to be demolished and adjacent facilities to remain.
  - 1. Cease operations and notify Owner's Representative immediately if safety of structure appears to be endangered. Take precautions to support structure until determination is made for continuing operations.
  - 2. Cover and protect furniture, equipment, and fixtures from soilage or damage when

- demolition work is performed in areas where such items have not been removed.
- 3. Erect and maintain dust-proof partitions and closures as required to prevent spread of dust or fumes to occupy and non-construction portions of the building.
  - a. Where selective demolition occurs immediately adjacent to occupied portions of the building, construct dust-proof partitions of minimum 4-inch studs, 5/8-inch drywall (joints taped) on occupied side, 1/2-inch plywood on demolition side. Fill partition cavity with 4" sound-deadening insulation.
  - b. Provide weatherproof closures for exterior openings resulting from demolition work.
- 4. Locate, identify, cap off, and disconnect utility services that are not indicated to remain.
  - a. Provide bypass connections as necessary to maintain continuity of service to occupied areas of building. Provide minimum of 72 hours advance notice to Owner if shutdown of service is necessary during changeover. Schedule change over during building/site non-occupied times, and after hours.
- 5. Close off HVAC registers in demolition area to prevent migration of dust and etc. from entering mechanical systems.
- 6. Selective demolition contractor to coordinate with flooring contractor prior to demolition/removal of all items at existing concrete slab as required to receive new floor finishes prior to starting work.

### 3.3 DEMOLITION

- A. General: Perform selective demolition work in a systematic manner. Use such methods as required to complete work indicated on Drawings in accordance with demolition schedule and governing regulations.
  - Demolish concrete and masonry in small sections. Cut concrete and masonry at junctures with construction to remain using power-driven masonry saw or hand tools; do not use power-driven impact tools.
  - Neatly cut openings and holes plumb, square, and true to dimensions required. Use cutting methods least likely to damage construction to remain or adjoining construction. Use hand tools or small power tools designed for sawing or grinding, not hammering and chopping, to minimize disturbance of adjacent surfaces. Temporarily cover openings to remain.
  - 3. Cut or drill from the exposed or finished side into concealed surfaces to avoid marring existing finished surfaces.
  - 4. Remove decayed, vermin-infested, or otherwise dangerous or unsuitable materials and promptly dispose of off-site.
  - 5. Locate demolition equipment throughout structure and promptly remove debris to avoid imposing excessive loads on supporting walls, floors, or framing.
  - 6. Provide services for effective air and water pollution controls as required by local authorities having jurisdiction.
  - 7. Demolish foundation walls to a depth of not less than 12 inches below existing ground surface, unless noted otherwise. Demolish and remove below-grade wood or metal construction. Break up below-grade concrete slabs. (See plans for more restrictive requirements.)
  - 8. For interior slabs on grade, use removal methods that will not crack or structurally disturb adjacent slabs or partitions. Use power saw where possible.
  - 9. Completely fill below-grade areas and voids resulting from demolition work. Provide fill consisting of approved earth, gravel, or sand, free of trash and debris, no stones over 1 inch in diameter, roots, or other organic matter.

SELECTIVE DEMOLITION 024119 - 5

- B. If unanticipated mechanical, electrical, or structural elements that conflict with intended function or design are encountered, investigate and measure both nature and extent of the conflict. Submit report to Owner's Representative in written, accurate detail. Pending receipt of directive from Owner's Representative, rearrange selective demolition schedule as necessary to continue overall job progress without undue delay.
- C. Demolition work that generates noise shall be scheduled around schedule of occupied rooms within 125 feet of work to be done. This work may have to be done during after hours, evenings and weekends, depending on occupancy schedules. Verify with Director of Facilities prior to scheduling any noise-generating demolition. See Section 011000 for additional requirements. Obtain approvals in writing from Director of Facilities for any work during occupied hours within 125 feet of occupied spaces

## 3.4 SALVAGED MATERIALS

- A. Salvaged Items: Where indicated on Drawings as "Salvage Deliver to Owner," carefully remove indicated items, clean, store, and turn over to Owner and obtain receipt.
  - 1. Owner shall remove all items prior to start of demolition of each area of building that they want to salvage. Owner shall have first right of salvage. All items left after Owner has completed salvage shall become property of Contractor.
- B. Where noted herein, Contractor shall remove, package and deliver salvage items to Owner.
  - 1. Clean salvaged items.
  - 2. Pack or crate items after cleaning. Identify contents of containers.
  - 3. Store items in a secure area until delivery to Owner.
  - 4. Transport items to Owner's storage area designated by Owner.
  - 5. Protect items from damage during transport and storage.

### 3.5 DISPOSAL OF WASTE/ DEMOLISHED MATERIALS

- A. Remove from building site debris, rubbish, and other materials resulting from demolition operations. Transport and legally disposed off site per Construction Waste Management Plan.
  - 1. If hazardous materials are encountered during demolition operations, comply with applicable regulations, laws, and ordinances concerning removal, handling, and protection against exposure or environmental pollution.
  - 2. Burning of removed materials is not permitted on project site.

### 3.6 CLEANUP AND REPAIR

- A. General: Upon completion of demolition work, remove tools, equipment, and demolished materials from site. Remove protections and leave interior areas soft broom clean.
  - 1. Repair demolition performed in excess of that required. Return elements of construction and surfaces to remain to condition existing prior to start operations. Repair adjacent construction or surfaces soiled or damaged by selective demolition work.
  - 2. Clean adjacent structures and improvements of dust, dirt, and debris caused by selective demolition operations. Return adjacent areas to condition existing before selective demolition operations began.

END OF SECTION 024119

SELECTIVE DEMOLITION 024119 - 6

## PART 1 - GENERAL

### 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of Contract, including General and Supplementary Conditions and Division 0 & 1 Specification Sections, apply to this Section.

# 1.2 SUMMARY

- A. This Section includes, but is not limited to, the following:
  - 1. Installation of finish hardware.
  - 2. Installation of hollow metal doors / windows / frames.
  - 3. Installation of miscellaneous accessories.
  - 4. Installation of wood paneling and related trim.
  - 5. Installation of flush wood doors/frames.
  - 6. Installation of door and related hardware.
- B. Related Sections: The following Sections contain requirements that relate to this Section:
  - 1. Division 7 Section "Building Insulation" for insulation.
  - 2. Division 7 Section "Joint Sealants" for sealants.
  - 3. Division 8 Section "Door Hardware" for door hardware and related items to be installed.
  - 4. Division 8 Section "Flush Wood Doors," for wood doors to be installed.
  - 5. Division 8 Section "Glass and Glazing" for additional requirements.

## 1.3 SUBMITTALS

- A. General: Submit the following in accordance with Conditions of Contract and Division 1 Specification Sections.
- B. Product data for each type of factory-fabricated product and process specified, including details of construction relative to materials, dimensions of individual components, profiles, textures, and colors.

### 1.4 QUALITY ASSURANCE

A. Installer Qualifications: Arrange for installation of finish carpentry by a firm that can demonstrate successful experience in installing finish carpentry items similar in type and quality to those required for this Project.

# 1.5 DELIVERY, STORAGE, AND HANDLING

- A. Delivery and Storage: Keep materials under cover and dry. Protect against exposure to weather and contact with damp or wet surfaces. Stack lumber as well as plywood and other panels. Provide for air circulation within and around stacks and under temporary coverings including polyethylene and similar materials.
- B. Do not deliver interior finish carpentry until environmental conditions meet requirements specified

for installation areas. If finish carpentry must be stored in other than installation areas, store only in areas where environmental conditions meet requirements specified for installation areas.

### 1.6 PROJECT CONDITIONS

- A. Environmental Conditions: Obtain and comply with finish carpentry manufacturer's and installer's coordinated advice for optimum temperature and humidity conditions for finish carpentry during its storage and installation.
- B. Weather Conditions: Proceed with finish carpentry only when existing and forecasted weather conditions will permit exterior finish carpentry to be installed in compliance with manufacturer's recommendations and when substrate is completely dry.

## PART 2 - PRODUCTS

### 2.1 MISCELLANEOUS MATERIALS

- A. Countersink nails and fill surface where face nailing is unavoidable.
- B. Fasteners for Interior Finish Carpentry: Nails, screws, and other anchoring devices of type, size, material, and finish required for application indicated to provide secure attachment, concealed where possible.
  - 1. Countersink nails, fill surface flush, and sand where face nailing is unavoidable.
- C. Adhesives: Comply with manufacturer's recommendations for adhesives.
- D. Plywood: Plywood where plywood is finish material for room shall be AC grade 5 ply minimum.

### 2.2 FABRICATION

- A. Wood Moisture Content: Comply with requirements of specified inspection agencies and manufacturer's recommendations for moisture content of finish carpentry in relation to relative humidity conditions existing during time of fabrication and in installation areas. Provide finish carpentry with moisture content that is compatible with Project requirements.
- B. Fabricate finish carpentry to dimensions, profiles and details indicated. Ease edges to radius indicated for the following:
  - 1. Lumber less than 1 inch in nominal thickness: 1/16 inch.
  - 2. Lumber 1 inch or more in nominal thickness: 1/8 inch.
- C. Provide steel sleeves (0.065 wall min.) size of bolts, toggles, tackable walls and screws, etc. where load requires sleeve thru gypsum board and other materials. Stainless steel sleeves at wet areas and exterior locations.

### PART 3 - EXECUTION

# 3.1 EXAMINATION

A. Examine substrates for compliance with requirements for installation tolerances and other conditions affecting installation and performance of finish carpentry. Do not proceed with installation until unsatisfactory conditions have been corrected.

## 3.2 PREPARATION

- A. Clean substrates of projections and substances detrimental to application.
- B. Condition finish carpentry to average prevailing humidity conditions in installation areas before installation for a minimum of 24 hours unless longer conditioning recommended by manufacturer.
- C. Backprime lumber for painted finish exposed on the exterior. Comply with requirements for surface preparation and application in Section "Painting."

## 3.3 INSTALLATION, GENERAL

- A. Do not use finish carpentry materials that are unsound, warped, bowed, twisted, improperly treated or finished, not adequately seasoned, or too small to fabricate with proper jointing arrangements.
- B. Install finish carpentry plumb, level, true, and aligned with adjacent materials. Use concealed shims where required for alignment.
  - 1. Scribe and cut finish carpentry to fit adjoining work. Refinish and seal cuts as recommended by manufacturer.
  - 2. Install to tolerance of 1/8 inch in 8 feet for plumb and level. Install adjoining finish carpentry with 1/16 inch maximum offset for flush installation and 1/8 inch maximum offset for reveal installation.
  - 3. Coordinate finish carpentry with materials and systems that may be in or adjacent to standing and running trim and rails. Provide cutouts for mechanical and electrical items that penetrate exposed surfaces of trim and rails.
  - 4. Screws and similar fasteners shall be installed plumb and true. Counter-sunk screws shall seat uniformly in recesses evenly all around.
- C. Finish in accordance with specified requirements.
- D. Refer to Division 6 Section "Interior Architectural Woodwork" for factory finished cabinets and Division 9 Sections for final finishing of finish carpentry.

### 3.4 ADJUSTING

E. Repair damaged or defective finish carpentry where possible to eliminate functional or visual defects. Where not possible to repair, replace finish carpentry. Adjust joinery for uniform appearance.

## 3.5 CLEANING

A. Clean finish carpentry on exposed and semi-exposed surfaces. Touch up factory-applied finishes to restore damaged or soiled areas.

# 3.6 PROTECTION

A. Provide final protection and maintain conditions that ensure finish carpentry is without damage or deterioration at time of Notice of Completion.

END OF SECTION 062000

# SECTION 072100 - BUILDING INSULATION

PART 1 - GENERAL

## 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 0 & 1 Specification Sections, apply to this Section.

### 1.2 SUMMARY

- A. This Section includes the following, but not limited to:
  - Sound attenuation insulation.

## 1.3 QUALITY ASSURANCE

- A. Source Limitations: Obtain each type of building insulation through one source from a single manufacturer.
- B. Provide certification that the insulation material complies with the California Quality Standards of insulating Materials, Title 20, Chapter 4 and Article 3.
- C. Fire-Test-Response Characteristics: Provide insulation and related materials with the fire-test-response characteristics indicated, as determined by testing identical products per test method indicated below by UL or another testing and inspecting agency acceptable to authorities having jurisdiction. Identify materials with appropriate markings of applicable testing and inspecting agency.
  - 1. Surface-Burning Characteristics: ASTM E 84.
  - 2. Fire-Resistance Ratings: ASTM E 119.
  - 3. Combustion Characteristics: ASTM E 136.

### 1.4 DELIVERY, STORAGE, AND HANDLING

- A. Protect plastic insulation as follows:
  - 1. Do not expose to sunlight, except to extent necessary for period of installation and concealment.
  - 2. Protect against ignition at all times. Do not deliver plastic insulating materials to Project site before installation time.
  - 3. Complete installation and concealment of plastic materials as rapidly as possible in each area of construction.

BUILDING INSULATION 072100 - 1

7/7/16

# PART 2 - PRODUCTS

### 2.1 MANUFACTURERS

- A. In other Part 2 articles where titles below introduce lists, the following requirements apply to product selection:
  - 1. Available Products: Subject to compliance with requirements, products that may be incorporated into the Work include, but are not limited to, products specified.

### 2.2 GLASS-FIBER BLANKET INSULATION

- A. Available Manufacturers:
  - 1. CertainTeed Corporation.
  - 2. Guardian Fiberglass, Inc.
  - 3. Johns Manville.
  - 4. Knauf Fiber Glass.
  - 5. Owens Corning.
  - 6. FiberTek

# PART 3 - EXECUTION

### 3.1 EXAMINATION

- A. Examine substrates and conditions, with Installer present, for compliance with requirements of Sections in which substrates and related work are specified and for other conditions affecting performance.
  - 1. Proceed with installation only after unsatisfactory conditions have been corrected.

### 3.2 PREPARATION

A. Clean substrates of substances harmful to insulation or vapor retarders, including removing projections capable of puncturing vapor retarders or of interfering with insulation attachment.

## 3.3 INSTALLATION, GENERAL

- A. Comply with insulation manufacturer's written instructions applicable to products and application indicated.
- B. Install insulation that is undamaged, dry, and unsoiled and that has not been left exposed at any time to ice, rain, and snow.
- C. Extend insulation in thickness indicated to envelop entire area to be insulated. Cut and fit tightly around obstructions and fill voids with insulation. Remove projections that interfere with placement.

BUILDING INSULATION 072100 - 2

- D. Water-Piping Coordination: If water piping is located within insulated exterior walls, coordinate location of piping to ensure that it is placed on warm side of insulation and insulation encapsulates piping.
- E. For preformed insulating units, provide sizes to fit applications indicated and selected from manufacturer's standard thicknesses, widths, and lengths. Apply single layer of insulation units to produce thickness indicated unless multiple layers are otherwise shown or required to make up total thickness.

## 3.4 INSTALLATION OF GENERAL BUILDING INSULATION

- A. Apply insulation units to substrates by method indicated, complying with manufacturer's written instructions. If no specific method is indicated, bond units to substrate with adhesive or use mechanical anchorage to provide permanent placement and support of units.
- B. Set vapor-retarder-faced units with vapor retarder to warm side of construction, unless otherwise indicated.
  - 1. Tape joints and ruptures in vapor retarder, and seal each continuous area of insulation to surrounding construction to ensure airtight installation.
- C. Install insulation in cavities formed by framing members according to the following requirements:
  - 1. Use insulation widths and lengths that fill the cavities formed by framing members. If more than one length is required to fill cavity, provide lengths that will produce a snug fit between ends.
  - 2. Place insulation in cavities formed by framing members to produce a friction fit between edges of insulation and adjoining framing members.
  - 3. For metal-framed wall cavities where cavity heights exceed 96 inches (2438 mm), support unfaced blankets mechanically and support faced blankets by taping stapling flanges to flanges of metal studs.
  - 4. For wood-framed construction, install mineral-fiber blankets according to ASTM C 1320 and as follows:
    - a. With faced blankets having stapling flanges, secure insulation by inset, stapling flanges to sides of framing members.
    - b. With faced blankets having stapling flanges, lap blanket flange over flange of adjacent blanket to maintain continuity of vapor retarder once finish material is installed over it.
- D. Stuff glass-fiber loose-fill insulation into miscellaneous voids and cavity spaces where shown. Compact to approximately 40 percent of normal maximum volume equaling a density of approximately 2.5 lb/cu. ft. (40 kg/cu. m).
- E. Provide faced insulation where exposed in return air plenums. Seal all joints with tape approved by manufacturer for this use.

BUILDING INSULATION 072100 - 3

- 3.5 INSULATION SCHEDULE Refer to plans for additional and more restrictive requirements. in the event of a conflict, the more restrictive requirement will prevail.
  - A. Insulation Type IS1: Glass-fiber sound attenuation blanket insulation, unfaced 4" thick at 4" interior walls where indicated.
  - B. Insulation Type IS2: Glass-fiber sound attenuation blanket insulation, unfaced 6" thick at 6" interior walls where indicated.

## 3.6 INSULATION PROTECTION

A. In the event any insulation becomes damp and/or wet, it shall be removed and replaced.

END OF SECTION 072100

BUILDING INSULATION 072100 - 4

## SECTION 079200 - JOINT SEALANTS

PART 1 - GENERAL

### 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Divisions 0 & 1 Specification Sections, apply to this Section.

### 1.2 SUMMARY

- A. This Section includes, but is not limited to the following sealants for the following applications, including those specified by reference to this Section (refer to Joint Sealer Schedule on construction documents for additional requirements):
  - 1. Interior joints in the following vertical surfaces and horizontal nontraffic surfaces:
    - a. Control and expansion joints on exposed interior surfaces.
    - b. Perimeter joints between interior wall surfaces and frames of interior doors, windows, and elevator entrances.
    - c. Joints between plumbing fixtures and adjoining walls, floors, and counters.
    - d. Other joints as indicated.

### B. Related Sections include the following:

- 1. Division 8 Section "Glazing" for glazing sealants.
- 2. Division 9 Section "Gypsum Board Assemblies" for sealing perimeter joints of gypsum board partitions to reduce sound transmission.
- 3. Division 9 Section "Acoustical Panel Ceilings" for sealing edge moldings at perimeters of acoustical ceilings.
- 4. Division 9 Section "Acoustical Tile Ceiling" for sealing edge moldings at perimeter of acoustical ceilings.

## 1.3 PERFORMANCE REQUIREMENTS

- A. Provide elastomeric joint sealants that establish and maintain watertight and airtight continuous joint seals without staining or deteriorating joint substrates.
- B. Provide joint sealants for interior applications that establish and maintain airtight and water-resistant continuous joint seals without staining or deteriorating joint substrates.

# 1.4 SUBMITTALS

- A. Product Data: For each joint-sealant product indicated.
- B. Samples for Initial Selection: Manufacturer's color charts consisting of strips of cured sealants showing the full range of colors available for each product exposed to view.
- C. Samples for Verification: For each type and color of joint sealant required. Install joint sealants in 1/2-inch- (13-mm-) wide joints formed between two 6-inch- (150-mm-) long strips of material

JOINT SEALANTS 079200 - 1

09/30/22

- matching the appearance of exposed surfaces adjacent to joint sealants.
- D. Product Certificates: Signed by manufacturers of joint sealants certifying that products furnished comply with requirements and are suitable for the use indicated.
- E. Qualification Data: For firms and persons specified in "Quality Assurance" Article to demonstrate their capabilities and experience. Include lists of completed projects with project names and addresses, names and addresses of architects and owners, and other information specified.
- F. Compatibility and Adhesion Test Reports: From sealant manufacturer indicating the following:
  - 1. Materials forming joint substrates and joint-sealant backings have been tested for compatibility and adhesion with joint sealants.
  - 2. Interpretation of test results and written recommendations for primers and substrate preparation needed for adhesion.
- G. Warranties: Special warranties specified in this Section.
- H. Project Closeout Requirements:
  - 1. 5-year installer warranty.
  - 2. 5-year manufacturer's warranty.

# 1.5 QUALITY ASSURANCE

- A. Installer Qualifications: An experienced installer who has specialized in installing joint sealants similar in material, design, and extent to those indicated for this Project and whose work has resulted in joint-sealant installations with a record of successful in-service performance.
- B. Source Limitations: Obtain each type of joint sealant through one source from a single manufacturer.
- C. Preconstruction Field-Adhesion Testing: Before installing elastomeric sealants, field test their adhesion to joint substrates as follows:
  - 1. Locate test joints where indicated or, if not indicated, as directed by Architect.
  - 2. Conduct field tests for each application indicated below:
    - a. Each type of elastomeric sealant and joint substrate indicated.
    - b. Each type of nonelastomeric sealant and joint substrate indicated.
  - 3. Notify Architect seven days in advance of dates and times when test joints will be erected.
  - Arrange for tests to take place with joint sealant manufacturer's technical representative present.
  - 5. Test Method: Test joint sealants by hand-pull method described below:
    - Install joint sealants in 60-inch- (1500-mm-) long joints using same materials and methods for joint preparation and joint-sealant installation required for the completed Work. Allow sealants to cure fully before testing.
    - Make knife cuts from one side of joint to the other, followed by two cuts approximately 2 inches (50 mm) long at sides of joint and meeting cross cut at one end. Place a mark 1 inch (25 mm) from cross-cut end of 2-inch (50-mm) piece.
    - c. Use fingers to grasp 2-inch (50-mm) piece of sealant between cross-cut end and 1-inch (25-mm) mark; pull firmly at a 90-degree angle or more in direction of side cuts

- while holding a ruler along side of sealant. Pull sealant out of joint to the distance recommended by sealant manufacturer for testing adhesive capability, but not less than that equaling specified maximum movement capability in extension; hold this position for 10 seconds.
- d. For joints with dissimilar substrates, check adhesion to each substrate separately. Do this by extending cut along one side, checking adhesion to opposite side, and then repeating this procedure for opposite side.
- Report whether sealant in joint connected to pulled-out portion failed to adhere to joint substrates or tore cohesively. Include data on pull distance used to test each type of product and joint substrate. For sealants that fail adhesively, retest until satisfactory adhesion is obtained.
- 7. Evaluation of Preconstruction Field-Adhesion-Test Results: Sealants not evidencing adhesive failure from testing, in absence of other indications of noncompliance with requirements, will be considered satisfactory. Do not use sealants that fail to adhere to joint substrates during testing.

## 1.6 DELIVERY, STORAGE, AND HANDLING

- A. Deliver materials to Project site in original unopened containers or bundles with labels indicating manufacturer, product name and designation, color, expiration date, pot life, curing time, and mixing instructions for multicomponent materials.
- B. Store and handle materials in compliance with manufacturer's written instructions to prevent their deterioration or damage due to moisture, high or low temperatures, contaminants, or other causes.

# 1.7 PROJECT CONDITIONS

- A. Environmental Limitations: Do not proceed with installation of joint sealants under the following conditions:
  - 1. When ambient and substrate temperature conditions are outside limits permitted by joint sealant manufacturer or are below 40 deg F (4.4 deg C).
  - 2. When joint substrates are wet.
- B. Joint-Width Conditions: Do not proceed with installation of joint sealants where joint widths are less than those allowed by joint sealant manufacturer for applications indicated.
- C. Joint-Substrate Conditions: Do not proceed with installation of joint sealants until contaminants capable of interfering with adhesion are removed from joint substrates.

## 1.8 WARRANTY

- A. General Warranty: Special warranties specified in this Article shall not deprive Owner of other rights Owner may have under other provisions of the Contract Documents and shall be in addition to, and run concurrent with, other warranties made by Contractor under requirements of the Contract Documents.
- B. Special Installer's Warranty: Written warranty, signed by Installer agreeing to repair or replace elastomeric joint sealants that do not comply with performance and other requirements specified in this Section within specified warranty period.

- 1. Warranty Period: Five (5) years from date of Notice of Completion.
- C. Special Manufacturer's Warranty: Written warranty, signed by elastomeric sealant manufacturer agreeing to furnish elastomeric joint sealants to repair or replace those that do not comply with performance and other requirements specified in this Section within specified warranty period.
  - 1. Warranty Period: Five (5) years from date of Notice of Completion.
- D. Special warranties specified in this Article exclude deterioration or failure of elastomeric joint sealants from the following:
  - Movement of the structure resulting in stresses on the sealant exceeding sealant manufacturer's written specifications for sealant elongation and compression caused by structural settlement or errors attributable to design or construction.
  - 2. Disintegration of joint substrates from natural causes exceeding design specifications.
  - 3. Mechanical damage caused by individuals, tools, or other outside agents.
  - Changes in sealant appearance caused by accumulation of dirt or other atmospheric contaminants.

### PART 2 - PRODUCTS

### 2.1 PRODUCTS AND MANUFACTURERS

A. Available Products: Subject to compliance with requirements, products that may be incorporated into the Work include, but are not limited to, the products specified in the Sealant Data Sheets at the end of Part 3.

# 2.2 MATERIALS, GENERAL

- A. Compatibility: Provide joint sealants, backings, and other related materials that are compatible with one another and with joint substrates under conditions of service and application, as demonstrated by sealant manufacturer based on testing and field experience.
- B. Colors of Exposed Joint Sealants: As selected by Architect from manufacturer's full range for this characteristic.

# 2.3 ELASTOMERIC JOINT SEALANTS

- A. Elastomeric Sealant Standard: Comply with ASTM C 920 and other requirements indicated for each liquid-applied chemically curing sealant in the Elastomeric Joint-Sealant Schedule at the end of Part 3, including those referencing ASTM C 920 classifications for type, grade, class, and uses.
- B. Additional Movement Capability: Where additional movement capability is specified in the Elastomeric Joint-Sealant Schedule, provide products with the capability, when tested for adhesion and cohesion under maximum cyclic movement per ASTM C 719, to withstand the specified percentage change in the joint width existing at the time of installation and remain in compliance with other requirements of ASTM C 920 for uses indicated.
- C. Suitability for Contact with Food: Where elastomeric sealants are indicated for joints that will come in repeated contact with food, provide products that comply with 21 CFR 177.2600.

## 2.4 LATEX JOINT SEALANTS (Indicated ES6 on Drawings)

- A. Latex Sealant Standard: Comply with ASTM C 834 for each product of this description.
  - General: Provide manufacturer's standard one-part, non-sag, mildew-resistant, paintable latex sealant of formulation indicated that is recommended for exposed applications on interior and protected exterior locations and that accommodates indicated percentage change in joint width existing at time of installation without failing either adhesively or cohesively (indicated on Drawings as ES6).
  - Acrylic-Emulsion Sealant: Provide product complying with ASTM C 834 that accommodates
    joint movement of not more than 5 percent in both extension and compression for a total of
    10 percent.
  - 3. Available Products" Subject to compliance with requirements, latex joint sealants that may be incorporated in the Work include, but are not limited to, the following:
    - a. Acrylic-Emulsion Sealant:
      - 1) "AC-20," Pecora Corp.
      - 2) "Sonolac," Sonneborn Building Products Div., ChemRex, Inc.
      - 3) "Tremco Acrylic Latex 834," Tremco, Inc.

# 2.5 ACOUSTICAL JOINT SEALANTS (Indicated ES7 on Drawings)

- A. Acoustical Sealant for Exposed and Concealed Joints: For each product of this description indicated in the Acoustical Joint-Sealant Schedule at the end of Part 3, provide manufacturer's standard nonsag, paintable, nonstaining latex sealant complying with ASTM C 834 and the following:
  - 1. Product effectively reduces airborne sound transmission through perimeter joints and openings in building construction as demonstrated by testing representative assemblies according to ASTM E 90.
- B. Acoustical Sealant for Concealed Joints: For each product of this description provide manufacturer's standard, nondrying, nonhardening, nonskinning, nonstaining, gunnable, synthetic-rubber sealant recommended for sealing interior concealed joints to reduce airborne sound transmission.
- C. Available Products: Subject to compliance with requirements, acoustical joint sealants that may be incorporated in the Work include, but are not limited to, the following:
  - 1. Acoustical Sealant:
    - a. "SHEETROCK Acoustical Sealant," United States Gypsum Co.
    - b. "AC-20 FTR Acoustical and Insulation Sealant," Pecora Corp.
  - 2. Acoustical Sealant for Concealed Joints:
    - a. "BA-98," Pecora Corp.
    - b. "Tremco Acoustical Sealant," Tremco, Inc.

# 2.6 PREFORMED JOINT SEALANTS

- A. Preformed Silicone-Sealant System: For each product of this description provide manufacturer's standard system consisting of precured low-modulus silicone extrusion, in sizes to fit joint widths indicated, combined with a neutral-curing silicone sealant for bonding extrusions to substrates.
- B. Preformed Foam Sealants: For each product of this description indicated in the Preformed Joint-Sealant Schedule at the end of Part 3, provide manufacturer's standard preformed, precompressed, impregnated, open-cell foam sealant manufactured from high-density urethane foam impregnated with a nondrying, water-repellent agent; factory produced in precompressed sizes and in roll or stick form to fit joint widths indicated and to develop a watertight and airtight seal when compressed to the degree specified by manufacturer; and complying with the following:
  - 1. Properties: Permanently elastic, mildew resistant, nonmigratory, nonstaining, and compatible with joint substrates and other joint sealants.
  - 2. Impregnating Agent: Manufacturer's standard.
  - 3. Density: Manufacturer's standard.
  - 4. Backing: Pressure-sensitive adhesive, factory applied to one side with protective wrapping.
  - 5. Available Products: Subject to compliance with requirements, preformed foam sealants that may be incorporated in the Work include, but are not limited to, the following:
    - a. "Emseal," Emseal Corp.
    - b. "Emseal Greyflex," Emseal Corp.
    - c. "Wil-Seal 150," Wil-Seal Construction Foams Div., Illbruck.
    - d. "Wil-Seal 250," Wil-Seal Construction Foams Div., Illbruck.

## 2.7 JOINT-SEALANT BACKING

- A. General: Provide sealant backings of material and type that are nonstaining; are compatible with joint substrates, sealants, primers, and other joint fillers; and are approved for applications indicated by sealant manufacturer based on field experience and laboratory testing.
- B. Cylindrical Sealant Backings: ASTM C 1330, of type indicated below and of size and density to control sealant depth and otherwise contribute to producing optimum sealant performance:
  - 1. Type C: Closed-cell material with a surface skin.
  - 2. Type O: Open-cell material.
  - 3. Type B: Bicellular material with a surface skin.
  - Type: Any material indicated above.
- C. Elastomeric Tubing Sealant Backings: Neoprene, butyl, EPDM, or silicone tubing complying with ASTM D 1056, nonabsorbent to water and gas, and capable of remaining resilient at temperatures down to minus 26 deg F (minus 32 deg C). Provide products with low compression set and of size and shape to provide a secondary seal, to control sealant depth, and otherwise contribute to optimum sealant performance.
- D. Bond-Breaker Tape: Polyethylene tape or other plastic tape recommended by sealant manufacturer for preventing sealant from adhering to rigid, inflexible joint-filler materials or joint surfaces at back of joint where such adhesion would result in sealant failure. Provide self-adhesive tape where applicable.

### 2.8 MISCELLANEOUS MATERIALS

A. Primer: Material recommended by joint sealant manufacturer where required for adhesion of sealant to joint substrates indicated, as determined from preconstruction joint-sealant-substrate tests and field tests.

- B. Cleaners for Nonporous Surfaces: Chemical cleaners acceptable to manufacturers of sealants and sealant backing materials, free of oily residues or other substances capable of staining or harming joint substrates and adjacent nonporous surfaces in any way, and formulated to promote optimum adhesion of sealants with joint substrates.
- C. Masking Tape: Nonstaining, nonabsorbent material compatible with joint sealants and surfaces adjacent to joints.

### PART 3 - EXECUTION

### 3.1 EXAMINATION

- A. Examine joints indicated to receive joint sealants, with Installer present, for compliance with requirements for joint configuration, installation tolerances, and other conditions affecting joint-sealant performance.
- B. Proceed with installation only after unsatisfactory conditions have been corrected.

## 3.2 PREPARATION

- A. Surface Cleaning of Joints: Clean out joints immediately before installing joint sealants to comply with joint sealant manufacturer's written instructions and the following requirements:
  - 1. Remove all foreign material from joint substrates that could interfere with adhesion of joint sealant, including dust, paints (except for permanent, protective coatings tested and approved for sealant adhesion and compatibility by sealant manufacturer), old joint sealants, oil, grease, waterproofing, water repellents, water, surface dirt, and frost.
  - 2. Clean porous joint substrate surfaces by brushing, grinding, blast cleaning, mechanical abrading, or a combination of these methods to produce a clean, sound substrate capable of developing optimum bond with joint sealants. Remove loose particles remaining from above cleaning operations by vacuuming or blowing out joints with oil-free compressed air. Porous joint surfaces include the following:
    - a. Concrete.
    - b. Masonry.
    - Unglazed surfaces of ceramic tile.
  - 3. Remove laitance and form-release agents from concrete.
  - 4. Clean nonporous surfaces with chemical cleaners or other means that do not stain, harm substrates, or leave residues capable of interfering with adhesion of joint sealants.
    - a. Metal.
    - b. Glass.
    - c. Glazed surfaces of ceramic tile.
- B. Joint Priming: Prime joint substrates where recommended in writing by joint sealant manufacturer, based on preconstruction joint-sealant-substrate tests or prior experience. Apply primer to comply with joint sealant manufacturer's written instructions. Confine primers to areas of joint-sealant bond; do not allow spillage or migration onto adjoining surfaces.

C. Masking Tape: Use masking tape where required to prevent contact of sealant with adjoining surfaces that otherwise would be permanently stained or damaged by such contact or by cleaning methods required to remove sealant smears. Remove tape immediately after tooling without disturbing joint seal.

### 3.3 INSTALLATION OF JOINT SEALANTS

- A. General: Comply with joint sealant manufacturer's written installation instructions for products and applications indicated, unless more stringent requirements apply.
- B. Sealant Installation Standard: Comply with recommendations of ASTM C 1193 for use of joint sealants as applicable to materials, applications, and conditions indicated.
- C. Acoustical Sealant Application Standard: Comply with recommendations of ASTM C 919 for use of joint sealants in acoustical applications as applicable to materials, applications, and conditions indicated.
- D. Install sealant backings of type indicated to support sealants during application and at position required to produce cross-sectional shapes and depths of installed sealants relative to joint widths that allow optimum sealant movement capability.
  - 1. Do not leave gaps between ends of sealant backings.
  - 2. Do not stretch, twist, puncture, or tear sealant backings.
  - 3. Remove absorbent sealant backings that have become wet before sealant application and replace them with dry materials.
- E. Install bond-breaker tape behind sealants where sealant backings are not used between sealants and back of joints.
- F. Install sealants by proven techniques to comply with the following and at the same time backings are installed:
  - 1. Place sealants so they directly contact and fully wet joint substrates.
  - 2. Completely fill recesses provided for each joint configuration.
  - 3. Produce uniform, cross-sectional shapes and depths relative to joint widths that allow optimum sealant movement capability.
- G. Tooling of Non-sag Sealants: Immediately after sealant application and before skinning or curing begins, tool sealants according to requirements specified below to form smooth, uniform beads of configuration indicated; to eliminate air pockets; and to ensure contact and adhesion of sealant with sides of joint.
  - 1. Remove excess sealants from surfaces adjacent to joint.
  - 2. Use tooling agents that are approved in writing by sealant manufacturer and that do not discolor sealants or adjacent surfaces.
  - 3. Provide concave joint configuration per Figure 5A in ASTM C 1193, unless otherwise indicated.
  - 4. Provide flush joint configuration, per Figure 5B in ASTM C 1193, where indicated.
  - 5. Provide recessed joint configuration, per Figure 5C in ASTM C 1193, of recess depth and at locations indicated.
    - a. Use masking tape to protect adjacent surfaces of recessed tooled joints.
- H. Installation of Preformed Foam Sealants: Install each length of sealant immediately after removing protective wrapping, taking care not to pull or stretch material, to produce seal continuity at ends,

turns, and intersections of joints. For applications at low ambient temperatures where expansion of sealant requires acceleration to produce seal, apply heat to sealant to comply with sealant manufacturer's written instructions.

## 3.4 CLEANING

A. Clean off excess sealants or sealant smears adjacent to joints as the Work progresses by methods and with cleaning materials approved in writing by manufacturers of joint sealants and of products in which joints occur.

# 3.5 PROTECTION

A. Protect joint sealants during and after curing period from contact with contaminating substances and from damage resulting from construction operations or other causes so sealants are without deterioration or damage at time of Project Completion. If, despite such protection, damage or deterioration occurs, cut out and remove damaged or deteriorated joint sealants immediately so installations with repaired areas are indistinguishable from the original work.

## **ELASTOMERIC JOINT SEALANT DATA SHEET**

Elastomeric Joint Sealant Designation: ES #1.

Base Polymer: Polysulfide/polyurthane.

Type: M (multicomponent).

Grade: P (pourable).

Class: 25.

Use[s] Related to Exposure: T (traffic)/40+/-5 shore harnes.

Uses Related to Joint Substrates: M, A, and, as applicable to joint substrates indicated, O.

Use O Joint Substrates: Color anodized aluminum, aluminum coated with a high-performance coating, galvanized steel, brick, granite, marble, ceramic tile, and wood.

Available Products:

1. "Deck-O-Seal #125", D.F.C. Company
Flat to 1/8"/ft. slope
2. "Elasto-Seal 227 - Self-Leveling", Pacific Polymers.

1. "Deck-O-Seal Gun Grade", D.F.C. Co. 1/8"/ft.&greater slope

2. "Elasto-Seal Gun Grade", Pacific Polymers.

3. Sika 2C Non-Sag.

# **ELASTOMERIC JOINT SEALANT DATA SHEET**

Elastomeric Joint Sealant Designation: ES #2.

Base Polymer: Acid-curing silicone.

Type: S (single component).

Grade: NS (nonsag).

Class: 25.

Use[s] Related to Exposure: NT (nontraffic).

Uses Related to Joint Substrates: G, A, and, as applicable to joint substrates indicated, O.

Use O Joint Substrates: Ceramic tile, plastic.

<u>Available Products</u>: 1. "786 Mildew Resistant", Dow Corning.

2. "Sanitary 1700", G.E.

3. "Proglaze White", Tremco.

# **ELASTOMERIC JOINT SEALANT DATA SHEET**

Elastomeric Joint Sealant Designation: ES #3.

Base Polymer: Urethane.

Type: M (multicomponent).

Grade: P (pourable).

Class: 25.

Use[s] Related to Exposure: T (traffic).

Uses Related to Joint Substrates: M, A, and, as applicable to joint substrates indicated, O.

<u>Use O Joint Substrates</u>: Color anodized aluminum, aluminum coated with a high-performance coating, galvanized steel, brick, granite, marble, ceramic tile, and wood.

<u>Available Products</u>: 1. "Chem-Calk 550", Bostik Construction Products Div.

2. "Vulkem 245", Mameco International, Inc.

3. "Urexpan NR-200", Pecora Corp.

4. "THC-900", Tremco Corp.

# **ELASTOMERIC JOINT SEALANT DATA SHEET**

Elastomeric Joint Sealant Designation: ES #4.

Base Polymer: Neutral-curing silicone.

Type: S (single component).

Grade: NS (nonsag).

Class: 25.

Additional Movement Capability: 100 percent movement in extension and 50 percent in compression for a

total of 150 percent movement.

Use[s] Related to Exposure: NT (nontraffic).

Uses Related to Joint Substrates: M, G, A, and, as applicable to joint substrates indicated, O.

<u>Use O Joint Substrates</u>: Coated glass, color anodized aluminum, aluminum coated with a high-performance coating, galvanized steel, brick, granite, marble, ceramic tile, and wood.

Available Products: 1. "790", Dow Corning.

2. "Spectrem 1", Tremco.

# **ELASTOMERIC JOINT SEALANT DATA SHEET**

Elastomeric Joint Sealant Designation: ES #5.

Base Polymer: Neutral-curing silicone.

Type: S (single component).

Grade: NS (nonsag).

<u>Class</u>: 25.

Additional Movement Capability: 50 percent movement in extension and 50 percent in compression for a

total of 100 percent movement.

<u>Use[s] Related to Exposure</u>: NT (nontraffic).

Uses Related to Joint Substrates: M, G, A, and, as applicable to joint substrates indicated, O.

<u>Use O Joint Substrates</u>: Coated glass, color anodized aluminum, aluminum coated with a high-performance coating, galvanized steel, brick, granite, marble, ceramic tile, plastic, and wood.

<u>Available Products</u>: 1. "795", Dow Corning.

2. "895", Pecora Corp.

## **ELASTOMERIC JOINT SEALANT DATA SHEET**

Elastomeric Joint Sealant Designation: ES #6.

Base Polymer: Acrylic emulsion.

Type: S (single component).

Grade: NS (nonsag).

Class: 25.

Additional Movement Capability: 5 percent movement in extension and 5 percent in compression for a

total of 10 percent movement.

<u>Use[s] Related to Exposure</u>: NT (nontraffic).

Uses Related to Joint Substrates: M, G, A, and, as applicable to joint substrates indicated, O.

<u>Use O Joint Substrates</u>: Coated glass, color anodized aluminum, aluminum coated with a high-performance coating, galvanized steel, brick, granite, marble, ceramic tile, plastic, and wood.

Available Products: 1. "Sonolac", Sonneborn Building Products Div., ChemRex, Inc.

2. "AC-20", Pecora Corp.

3. "Tremco Acrylic Latex 834, "Tremco, Inc.

## **ELASTOMERIC JOINT SEALANT DATA SHEET**

Elastomeric Joint Sealant Designation: ES #7.

Base Polymer: Synthetic rubber.

Type: S (single component).

Grade: NS (nonsag).

Class: 25.

Additional Movement Capability: NA.

<u>Use[s] Related to Exposure</u>: NT (nontraffic).

Uses Related to Joint Substrates: M, G, A, and, as applicable to joint substrates indicated, O.

<u>Use O Joint Substrates</u>: Coated glass, color anodized aluminum, aluminum coated with a high-performance coating, galvanized steel, brick, granite, marble, ceramic tile, plastic, and wood.

Available Products:

- 1. Acoustic Sealant:
  - a. "Sheetrock Acoustic Sealant", United States Gypsum Co.
  - b. "AC-20 FTR Acoustic & Insulation Sealant", Pecora Corp.
- 2. Acoustic Sealant for Concealed Joints:
  - a. "BA-98", Pecora Corp.
  - b. "Tremco Acoustic Sealant", Tremco, Inc

# **ELASTOMERIC JOINT SEALANT DATA SHEET**

Elastomeric Joint Sealant Designation: ES #8. Tape Sealant.

Base Polymer: Elastic modified butyl tape.

Type: S (single component).

Grade: NS (nonsag).

Class: 25.

Additional Movement Capability: 100 percent movement in extension and 50 percent in compression for a

total of 150 percent movement.

<u>Use[s] Related to Exposure</u>: NT (nontraffic).

Uses Related to Joint Substrates: M, G, A, and, as applicable to joint substrates indicated, O.

Use O Joint Substrates: Coated glass, color anodized aluminum, aluminum coated with a

high-performance coating, galvanized steel, brick, granite, marble, ceramic tile, and wood.

Available Products: 1. 1-1/2" w x 3/16".

# **ELASTOMERIC JOINT SEALANT DATA SHEET**

Elastomeric Joint Sealant Designation: ES #9.

Base Polymer: Neutral-curing silicone.

Type: S (single component).

Grade: NS (non-sag).

Class: 25.

Additional Movement Capability: 100 percent movement in extension and 50 percent in compression for a

total of 150 percent movement.

Use[s] Related to Exposure: NT (nontraffic).

Uses Related to Joint Substrates: M, G, A, and, as applicable to joint substrates indicated, O.

<u>Use O Joint Substrates</u>: EIFS, coated glass, color anodized aluminum, aluminum coated with a high-performance coating, galvanized steel, brick, granite, marble, ceramic tile, and wood.

Available | Products: 1. Dowsil "790", Dow Corning.

## **ELASTOMERIC JOINT SEALANT DATA SHEET**

Elastomeric Joint Sealant Designation: ES #10.

Base Polymer: Neutral-curing silicone.

Type: S (single component).

Grade: NS (nonsag).

Class: 25.

Additional Movement Capability: ±25 percent movement in extension and in compression for a total of

±25 percent movement.

Use[s] Related to Exposure: NT (nontraffic).

Uses Related to Joint Substrates: M, G, A, and, as applicable to joint substrates indicated, O.

<u>Use O Joint Substrates</u>: Coated glass, color anodized aluminum, aluminum coated with a high-performance coating, galvanized steel, brick, granite, marble, ceramic tile, and wood.

Available Products: 1. "790", Dow Corning.

<u>Additional Notes</u>: Allow for priming the adhesion surface of the copper flashing with Dow Corning 1200 OS or provide field adhesion test results acceptable to the sealant manufacturer.

END OF SECTION 079200

## SECTION 08111 - STANDARD STEEL DOORS AND FRAMES

01/13/14

## PART 1 - GENERAL

## 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of Contract, including General and Supplementary Conditions and Divisions 0 & 1 Specification Sections, apply to this Section.

## 1.2 SUMMARY:

- A. Section includes, but is not limited to the following:
  - 1. Hollow Metal Frames.
  - 2. Hollow Metal Window Frames.
- B. Related Sections:
  - 1. Division 8 "Flush Wood Doors" for wood doors.
  - 2. Division 8 Section "Door Hardware" for door hardware.
  - 3. Division 8 Section "Glass and Glazing" for glazing.
  - 4. Division 9 Section "Painting" for painting.

# 1.3 REFERENCES:

- A. ASTM E152, Fire Tests of Door Assemblies.
- B. ASTM A653, specification for Steel Sheet, Zinc Coated.
- C. ANSI/SDI 100, Recommended Specifications for Standard Steel Doors and Frames.
- D. ANSI/SDI 119, Performance test Procedures for Steel Door Frames and Anchors.
- E. NFPA 80, Standard for Fire Doors and Windows.
- F. ANSI A151.1, Test Procedure and Acceptance Criteria for Physical Endurance, Steel Doors and Frames.
- G. ANSI/SDI A250.6 Recommended Practice for Hardware Reinforcing on Standard Steel Doors and Frames.
- H. ANSI A224.1, Test Procedure and Acceptance Criteria for Prime Painted Steel Surfaces for Steel Doors and Frames.
- I. SDI 107, Hardware on Steel Doors, Reinforcement Application.
- Applicable model building code.
- K. California Title 24.
- L. UL-10C, Fire Tests of Door Assemblies.

M. UL-9, Fire Tests of Window Assemblies.

## 1.4 SUBMITTALS:

- A. General: Submit the following in accordance with Conditions of Contract and Division 1 Specifications.
- B. Shop drawings showing fabrication and installation of standard steel doors and frames. Include details of each frame type, elevations of door design types, conditions at openings, details of construction, location and installation requirements of door and frame hardware and reinforcements, and details of joints and connections. Show anchorage and accessory items.
  - 1. Provide schedule of doors and frames using same reference numbers for details and openings as those on Contract Documents.
  - 2. Indicate coordinate of glazing frames and stops with glass and glazing requirements.
- C. Certification of Compliance: Provide letter of certification that all materials comply with these Specifications.

## 1.5 QUALITY ASSURANCE:

- A. Steel Door and Frame Supplier: direct factory supplier who employs a Certified Door Consultant (CDC) or person with equivalent experience, available at reasonable times during course of Work, for consultation to Owner, Architect and Contractor.
- B. Label Construction: A physical label or approved marking shall be affixed to the fire door or fire door frame at an authorized facility as evidence of compliance with procedures of the labeling agency.

# 1.6 DELIVERY, STORAGE, AND HANDLING:

- A. Delivery: coordinate delivery to the appropriate locations (shop or field) for installation.
- B. Storage of Doors: Doors shall be stored in an upright position under cover. Place the units on at least 4-inch high (101.6 mm) wood sills on floors in a manner that will prevent rust and damage. Do not use non-vented plastic or canvas shelters, which create a humidity chamber and promote rusting. If the corrugated wrapper on the door becomes wet, or moisture appears, remove the wrapper immediately. Provide a 1/4" (6.35 mm) space between the doors to promote air circulation.
- C. Storage of Frames: Frames shall be stored under cover on 4" (101.6 mm) wood sills on floors in a manner that will prevent rust and damage. Do not use non-vented plastic or canvas shelters, which create a humidity chamber and promote rusting. Assembled frames shall be stored in a vertical position, five units maximum in a stack. Provide a 1/4" (6.35 mm) space between frames to promote air circulation.
- D. Inspect delivered items for damage. Minor damage may be repaired provided repaired items are equal to new Work and accepted by the Architect. Provide new items when directed. Comply with VOC regulations when repairing damage.

## 1.7 SEQUENCING AND SCHEDULING

- A. Deliver doors and frames to the jobsite in a timely manner so not to delay progress of other trades.
- B. Issue purchase orders to suppliers so as not to interfere with normal quoted delivery times.

## 1.8 WARRANTY

A. Steel doors and frames supplied with a one (1) year warranty against defects in materials and workmanship.

## 1.9 ENVIRONMENTAL

A. Packaging and Disposal: package in biodegradable packs, paper or cardboard boxes. Dispose of non-biodegradable packs, plastic, styrofoam, polystyrene, and polyurethane to a licensed or authorized collector for proper disposal. Comply with the applicable standards and laws for VOC.

## PART 2 - PRODUCTS

### 2.1 ACCEPTABLE MANUFACTURERS

- A. Subject to compliance with requirements, manufacturers offering doors and frames that may be incorporated in the Work include, but are not limited to, the following:
  - 1. Steelcraft Manufacturing Co., Cincinnati, Ohio
  - 2. Curries Co., Mason City, Iowa
  - 3. Ceco Corp., Oakbrook, Illinois

# 2.2 MATERIALS:

- A. Steel requirements: doors and frames manufactured of commercial quality, stretcher leveled flatness, cold rolled steel per ASTM A366 and A568 general requirements. Galvanized doors and frames to A60 minimum coating weight.
- B. Coating Materials:
  - 1. Primer: Manufacturer's standard rust inhibiting primer to ANSI A250.10.
- C. Core Materials:
  - 1. Doors: non-toxic honeycomb or vertical steel stiffened internal reinforcing manufactured of hot rolled, pickled and oiled steel per ASTM A569.
  - 2. Fire labeled doors with temperature rise rating: Mineral fiber core, temperature rating per code.
- D. Glass Light Frames: flush with door face, fabricated of 18 ga. steel.

## 2.3 FABRICATION

### A. Frames:

### 1. Construction:

- a. 16 gauge cold rolled steel at interior locations; 14 gauge galvanized at exterior locations.
- b. 12 gauge, full width, face and head reinforcement for non-labeled openings over 48" in door width.
- 2. Corner Construction: weld full depth and face, grind smooth and re-prime. Weld includes faces, rabbets, soffit and stops.
- 3. Provide temporary shipping spreaders to help protect frames from damage during transit and handling. Remove spreaders prior to setting frame.

### B. Frame Anchors:

- 1. Attachment to Drywall Construction:
  - a. Manufacturers standard compression type for Drywall type frame.
  - b. Steel or Wood Stud type to accommodate frame jamb depth and face dimension on welded type frame.
- 2. Provide one anchor for every 30 inches of jamb or fraction thereof.
- 3. Floor Anchor: angle clip type:
  - a. 16 Gauge.
  - b. Two fasteners per jamb.
  - a. Weld to bottom of each jamb.
- 4. Head Struts: per details.

## C. Preparation for Hardware:

- 1. Reinforce per SDI 107.
- 2. Lock and Closer reinforcement: box type.
- 3. Door Hinge reinforcement: 7 gauge or equivalent, Manufacturer's standard.
- 4. Punch strike jambs to receive three silencers; double leaf frames to receive manufacturer's standard preparation.
- 5. Hardware locations per "Recommended Locations for Builders' Hardware for Standard Steel Doors and Frames".
- 6. Provide welded in place guards for all hardware cutouts in frame.
- 7. Electrical preps: provide welded-in-place boxes, special designed anchors, raceways and access panels as required.

### PART 3 - EXECUTION

# 3.1 SETTING FRAMES

- A. Set frames in accordance with SDI 105.
- B. Set welded frames in place prior to construction of adjacent partition work. Properly brace

frame until permanent anchors are set.

- C. Install frames plumb and true with only hairline seam at corner joints.
- D. Install fire rated frames in accordance with CBC section 715.4.
- E. Grout all door frames with approved non-shrink grout material, lower 12" each jamb. See plans for details of grouting limits.

### 3.2 DOOR INSTALLATION

## A. Clearances:

- 1. The clearance between the door and frame head and jambs shall be 1/8" (3.2 mm) in the case of both single swing and pairs of doors.
- 2. The clearance between the meeting edges of pairs of doors shall be 1/8" (3.2 mm) to 1/4" (6.3 mm), for fire rated doors 1/8" (3.2 mm)  $\pm 1/16$ " (1.6 mm).
- 3. The clearance at the bottom shall be [3/4" (19.1 mm)] [5/8" (15.8 mm)].
- 4. The clearance between the face of the door and door stop shall be 1/16" (1.6 mm) to 1/8" (3.2 mm).
- 5. All clearances shall be, unless otherwise specified, subject to a tolerance of  $\pm$  1/32" (0.8 mm).

## 3.3 ADJUSTMENT AND CLEANING

- A. Remove dirt and excess sealants, mortar, or glazing compounds from exposed surfaces.
- B. Adjust moving parts for smooth operation. Use shims as required.
- C. Fill dents, holes, etc. with metal filler and sand smooth and flush with adjacent surfaces. Paint to match adjacent surface.

END OF SECTION 081113

# SECTION 081416 - FLUSH WOOD DOORS

04/09/15

### PART 1 - GENERAL

### 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 0 & 1 Specification Sections, apply to this Section.

### 1.2 SUMMARY

- A. This Section includes, but is not limited, the following:
  - 1. Solid-core doors with wood-veneer faces.
  - 2. Factory fitting flush wood doors to frames and factory machining for hardware.
  - 3. Factory finishing transparent with stain.
- B. Related Sections include the following:
  - 1. Division 6 Section "Finish Carpentry" for wood door installation.
  - 2. Division 8 Section "Glass and Glazing" for glass view panels in flush wood door.
  - 3. Division 8 Section "Standard Steel Doors and Frames" for door frames.
  - 4. Division 8 Section "Door Hardware" for door hardware.

# 1.3 SUBMITTALS

- A. Product Data: For each type of door. Include details of core and edge construction, trim for openings, and vision lights.
  - Submit WI Certified Compliance Certificate indicating that doors meet requirements of grades specified.
- B. Shop Drawings: Submit drawings in conformance with the requirements of Architectural Woodwork Standards (AWS). Furnish a Woodwork Institute Certified Compliance Label on the first page of the shop drawing. Indicate location, size, and hand of each door; elevation of each kind of door; construction details not covered in Product Data; location and extent of hardware blocking; and other pertinent data.
  - 1. Indicate dimensions and locations of blocking.
  - 2. Indicate dimensions and locations of mortises and holes for hardware.
  - 3. Indicate dimensions and locations of cutouts.
  - 4. Undercuts
  - 5. Indicate requirements for veneer matching.
  - 6. Indicate doors to be factory finished and finish requirements.
  - 7. Indicate fire-protection ratings for fire-rated doors.
- C. Project Closeout Requirements:
  - 1. Lifetime door warranty.
- D. Samples for Initial Selection: For factory finished doors.
- E. Samples for Verification:

FLUSH WOOD DOORS 081416 - 1

- 1. Factory finished applied to actual door face materials, approximately 8 by 10 inches (200 by 250 mm), for each material and finish. For each wood species and transparent finish, provide set of three samples showing typical range of color and grain to be expected in the finished work.
- 2. Corner sections of doors, approximately 8 by 10 inches (200 by 250mm) with door faces and edges representing actual materials to be used.
  - a. Provide samples for each species of veneer and solid lumber required.
  - b. Provide samples for each color, texture and pattern of plastic laminate required.
  - Finish veneer-faced door samples with same materials proposed for factoryfinished doors.
- 3. Louver blade and frame sections, 6 inches (150 mm) long, for each material and finish specified.
- 4. Frames for light openings, 6 inches (150 mm) long, for each material, type and finish required.
- F. Warranty: Sample of special warranty.

## 1.4 QUALITY ASSURANCE

- A. Source Limitations: Obtain flush wood doors through one source from a single manufacturer.
- B. Monitored Compliance:
  - 1. All millwork and the installation thereof for this project shall be monitored for compliance to the contract documents by a Woodwork Institute Director of Architectural Services.
    - a. Full particulars of the Woodwork Institute Monitored Compliance Program may be found at the Institute web site at <a href="www.woodworkinstitute.com">www.woodworkinstitute.com</a> or by calling the administrative office at (916) 372 9943.
  - 2. Millwork and/or installation found to be non compliant (and not corrected) will be rejected.
  - 3. Issuance of a Monitored Compliance Certificate is a prerequisite of acceptance.
  - 4. All fees charged by the Woodwork Institute for their Monitored Compliance program are the responsibility of the millwork manufacturer and/or installer and shall be included in their Contract proposal.
- C. Quality Standard: Comply with the following standard:
  - 1. WI Quality Standard: WI's "AWS Manual" for grade of door, core, construction, finish, and other requirements.
- D. Manufacturer's Qualification:
  - A qualified manufacturer that is a member in good standing of the Window and Door Manufacturers Association.
- E. Installer Qualification:
  - 1. A qualified installer that is certified for chain of custody by an FSC-accredited certification body when FSC Certified wood is specified.
  - 2. A qualified installer that is a WI member in good standing and qualified user of the AWS method of workmanship.
- F. Product Performance: Provide documents showing compliance to the following WDMA attributes, validating the specified WDMA Performance Duty Level:

FLUSH WOOD DOORS 081416 - 2

- 1. Adhesive Bonding Durability: WDMA TM-6
- 2. Cycle Slam: WDMA TM-7
- 3. Hinge Loading: WDMA TM-8
- 4. Screw Holding: WDMA TM-10
  - a. Door Face
  - b. Vertical Door Edge
  - c. Horizontal Door Edge (applies when hardware is attached)

# 1.5 DELIVERY, STORAGE, AND HANDLING

- A. Deliver materials only when the project is ready for installation and the general contractor has provided a clean storage area.
  - 1. Delivery of architectural millwork shall be made only when the area of operation is enclosed, all plaster and concrete work is dry and the area broom clean.
  - 2. The work area shall be well ventilated and protected from direct sunlight, excessive heat, rain or moisture. Temperature shall be maintained between 60E and 90E F and relative humidity between 45% and 65%. The HVAC system shall be on and functioning.
- B. Protect doors during transit, storage, and handling to prevent damage, soiling, and deterioration. Comply with requirements of referenced standard and manufacturer's written instructions.
  - 1. Individually package doors in cardboard cartons and wrap bundles of doors in plastic sheeting.
  - 2. Comply with WI's Technical Bulletin 420-R for delivery, storage, and handling of doors.
- C. Mark each door with individual opening numbers used on Shop Drawings. Use removable tags or concealed markings.

## 1.6 PROJECT CONDITIONS

A. Environmental Limitations: Do not deliver or install doors until conditions for temperature and relative humidity have been stabilized and will be maintained in storage and installation areas during the remainder of the construction period to comply with requirements of the referenced quality standard for Project's geographical location.

## 1.7 WARRANTY

- A. General Warranty: Door manufacturer's warranty specified in this Article shall not deprive the Owner of other rights the Owner may have under other provisions of the Contract Documents and shall be in addition to, and run concurrent with, other warranties made by the Contractor under requirements of the Contract Documents.
- B. Door Manufacturer's Warranty: Submit written agreement on door manufacturer's standard form, signed by manufacturer, Installer, and Contractor, agreeing to repair or replace defective doors that have warped (bow, cup, or twist) more than 1/4 inch (6.35 mm) in a 42-by-84-inch (1067-by-2134-mm) section or that show telegraphing of core construction in face veneers exceeding 0.01 inch in a 3-inch (0.25 mm in a 75-mm) span, or do not comply with tolerances in referenced quality standard.
  - 1. Warranty shall also include installation and finishing that may be required due to repair or replacement of defective doors.
  - 2. Warranty shall be in effect during the following period of time after the date of Notice of Completion:

FLUSH WOOD DOORS 081416 - 3

a. Solid-Core Interior Doors: Life of installation.

#### PART 2 - PRODUCTS

## 2.1 MANUFACTURERS

- A. Available Manufacturers: Subject to compliance with requirements, manufacturers offering products that may be incorporated into the Work include, but are not limited to, the following:
  - 1. Flush Wood Doors:
    - a. Algoma Hardwoods Inc.
    - b. Eggers Industries; Architectural Door Division.
    - c. Marshfield Door Systems
    - d. Graham Manufacturing Corp.
    - e. V-T Industries Inc.

## 2.2 DOOR CONSTRUCTION, GENERAL

- A. Doors for Transparent Stain Finish: Comply with the following requirements:
  - 1. Grade: Premium, Grade AA face and back.
  - 2. Faces: White birch, plain sliced.
  - 3. Match between Veneer Leaves: Well match.
  - 4. Match within Door Faces: Balance match.
  - 5. Pair and Set Match: Provide for pairs of doors and for doors hung in adjacent sets.
  - 6. Stiles: Same species as face.

# 2.4 SOLID-CORE DOORS

- A. Particleboard Cores: Comply with the following requirements:
  - 1. Particleboard: ANSI A208.1, Grade LD-1.
  - 2. Particleboard: ANSI A208.1, Grade LD-2.
- B. Blocking: Provide wood blocking at particleboard or core doors as follows:
  - 1. 5-inch (125-mm) top-rail blocking.
  - 2. 5-inch (125-mm) bottom-rail blocking.
  - 3. 5-inch (125-mm) midrail blocking, at doors indicated to have exit devices.
  - 4. As necessary to eliminate need for through-bolting hardware.
- C. Interior Veneer-Faced Doors: Comply with the following requirements:
  - 1. Core: Particleboard core, per ANSI 208.1, Grade #1-LD-2-1989.
  - 2. Construction: Five plies with stiles and rails bonded to core, then entire unit abrasive planed before veneering.

## 2.5 FABRICATION

A. Fabricate flush wood doors in sizes indicated for Project site fitting.

FLUSH WOOD DOORS 081416 - 4

- B. Factory fit doors to suit frame-opening sizes indicated, with the following uniform clearances and bevels, unless otherwise indicated:
  - 1. Comply with clearance requirements of referenced quality standard for fitting. Comply with requirements of CBC for fire-rated doors.
- C. Factory machine doors for hardware that is not surface applied. Locate hardware to comply with DHI-WDHS-3. Comply with final hardware schedules, door frame Shop Drawings, DHI A115-W series standards, and hardware templates.
  - 1. Coordinate measurements of hardware mortises in metal frames to verify dimensions and alignment before factory machining.
  - 2. Metal Astragals: Premachine astragals and formed-steel edges for hardware for pairs of fire-rated doors. Prefinished colors as selected. Match Architects sample allow for custom colors.

## 2.6 FACTORY FINISHING

- A. General: Comply with referenced quality standard for factory finishing. Complete fabrication including fitting doors for openings and machining for hardware that is not surface applied, before finishing.
  - 1. Finish faces, all four edges, edges of cutouts and mortises.
- B. Finish doors at factory.
- C. Finish doors at factory that are indicated to receive stain and transparent finish. Field finish doors indicated to receive opaque finish.
- D. Transparent Finish:
  - 1. Grade: Premium Grade.
  - 2. Finish: AWS System #5 catalyzed polyurethane.
  - 3. Staining: Match Architect's sample.
  - 4. Sheen: Semigloss.

## PART 3 - EXECUTION

## 3.1 EXAMINATION

- A. Examine installed door frames before hanging doors.
  - 1. Verify that frames comply with indicated requirements for type, size, location, and swing characteristics and have been installed with plumb jambs and level heads.
  - 2. Reject doors with defects.
- B. Proceed with installation only after unsatisfactory conditions have been corrected.

# 3.2 INSTALLATION

- A. Hardware: For installation, see Division 8 Section "Door Hardware."
  - 1. Hardware shall be installed complete and as required by the manufacturer.

FLUSH WOOD DOORS 081416 - 5

- B. Install all doors in conformance with the Architectural Woodwork Standards, latest edition. Reference AWS Manual Section 9 "Doors".
  - 1. Installation shall conform to the AWS Grade of the items being installed.
  - 2. Doors shall be secured in place, square, plumb and level.
- C. Manufacturer's Written Instructions: Install wood doors to comply with manufacturer's written instructions, referenced quality standard, and as indicated.
  - 1. Install fire-rated doors in corresponding fire-rated frames according to their listing.
- D. Factory-Fitted Doors: Align in frames for uniform clearance at each edge.
- E. Refer to door schedule for requiring factory undercut for return air.
- F. Factory-Finished Doors: Restore finish before installation if fitting or machining is required at Project site.

#### 3.3 ADJUSTING, TOUCH UP AND PROTECTING

- A. Before completion of installation, the installer shall adjust all moving and operating parts to function smoothly and correctly. Do not remove hinge screws after initial insertion. Shims used for alignment purposes must be inserted between hinge and frame. Do not insert shims between hinge and door.
- B. Operation: Rehang or replace doors that do not swing or operate freely.
- C. To prevent stile failure, insure that door closers are properly adjusted and do not limit the door opening swing. Limit door opening swing only with a properly located stop.
- D. Protect doors as recommended by door manufacturer to ensure that wood doors are without damage or deterioration at the time of Notice of Completion.
- E. Finished Doors: Replace doors that are damaged or that do not comply with requirements. Doors may be repaired or refinished if work complies with requirements and shows no evidence of repair or refinishing.
- F. Upon completion of installation, the installer shall clean all installed items of pencil and ink marks, and broom clean the area of operation, depositing debris in containers provided by the Contractor.

END OF SECTION 081416

FLUSH WOOD DOORS 081416 - 6

#### SECTION 087100 - DOOR HARDWARE

#### PART 1 - GENERAL

## 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of contract, including General and Supplementary Conditions and Division 00 & 01 Specification Sections, apply to this Section.

## 1.2 SUMMARY

- A. Section includes, but is not limited to the following:
  - 1. Door Hardware.
- B. Related Sections: The following Sections contain requirements that relate to this Section:
  - 1. Division 06 Section "Interior Architectural Woodwork" for cabinet hardware.
  - 2. Division 06 Section "Finish Carpentry" for installation of hardware.
  - 3. Division 08 Section "Standard Steel Doors and Frames"
  - 4. Division 08 Section "Flush Wood Doors" for factory prefitting and factory premachining of doors for door hardware.
- C. Specific Omissions: Hardware for the following is specified or indicated elsewhere.
  - 1. Windows.
  - 2. Cabinets of all kinds, including open wall shelving and locks.
  - 3. Signs, Except as noted.
  - 4. Installation.
  - 5. Rough hardware.
  - 6. Access doors and panels.

## 1.3 REFERENCES:

- A. Use date of standard in effect as of Bid date.
- B. American National Standards Institute ANSI 156.18 Materials and Finishes.
- C. ADA Americans with Disabilities Act of 2020
- D. BHMA Builders Hardware Manufacturers Association
- E. DHI Door and Hardware Institute
- F. NFPA National Fire Protection Association
  - 1. NFPA 80 Fire Doors and Windows
  - 2. NFPA 101 Life Safety Code
  - NFPA 105 Smoke and Draft Control Door Assemblies
- G. UL Underwriters Laboratories
  - 1. UL 10C Fire Tests of Door Assemblies (Positive Pressure)
  - 2. UL 305 Panic Hardware
- H. WHI Warnock Hersey Incorporated
- I. State of California Building Code
- K. SDI Steel Door Institute
- L. WDI Wood Door Institute
- M. AWI Architectural Woodwork Institute
- N. NAAM National Association of Architectural Metal Manufacturers

## 1.4 SUBMITTALS

- A. General: Submit the following in accordance with Conditions of Contract and Division 01 Specification sections.
- B. Product data including manufacturers' technical product data for each item of door hardware, installation instruction, maintenance of operating parts and finish, and other information necessary to show compliance with requirements.
- C. Organize schedule into "Hardware Sets" with an index of doors and heading, indicating complete designations of every item required for each door or opening. Include the following information:
  - 1. Type, style, function, size, quantity and finish of each hardware item. Use BHMA Finish codes as per ANSI A156.18.
  - 2. Name, part number and manufacturer of each item.
  - 3. Fastenings and other pertinent information.
  - 4. Location of hardware set cross referenced to indications on drawings both on floor plans and in door schedule.
  - 5. Explanation of all abbreviations, symbols, and codes contained in schedule.
  - 6. Mounting locations for hardware.
  - 7. Door and frame sizes and materials and degree of swing.
  - 8. Include a list of all manufacturers used and their nearest representative with address and phone number.
  - 9. Submit manufacturer's technical data and installation instructions for the electronic hardware.
  - 10. Provide data file of final installed hardware sets on compact disk with certification that hardware installed complies with Project requirements.

## 1.5 QUALITY ASSURANCE

#### A. Qualifications:

- 1. Obtain each kind of hardware (latch and lock sets, exit devices, hinges, and closers) from one manufacturer.
- 2. Supplier Qualifications: A recognized architectural door hardware supplier, with warehousing facilities in the Project's vicinity, that has a record of successful inservice performance for supplying door hardware similar in quantity, type, and quality to that indicated for this Project and that employs an experienced hardware consultant who is available to Owner, Architect, and Contractor, at reasonable times during the course of the Work, for consultation.
  - a. Require supplier to meet with Owner to develop and finalize keying requirements and to obtain final instructions in writing.
  - b. Be available for consultation with the Architect at no additional cost to the Owner during progress of construction at job site.
  - c. Be present at substantial completion of construction, and:
    - 1) Inspect installation of all finish hardware items.
    - 2) Make all minor adjustments required.
    - 3) Report to the Architect on completeness of the installation.
- B. Exit Doors: Openable from the inside without the use of a key or any special knowledge or effort.

# 1.6 DELIVERY, STORAGE, AND HANDLING

- A. Acceptance at Site: Individually package each unit of finish hardware complete with proper fastening and appurtenances, clearly marked on the outside to indicate contents and specific locations in the Work.
- B. Deliver packaged hardware items at the times and to the locations (shop or field) for installation, as directed by the Contractor.
- C. Unused Hardware: Return new hardware not installed and existing hardware not indicated to be reinstalled to the Owner.
  - 1. Box and label in original manufacturer's-type boxes. Deliver to Owner to location directed by Architect.
- D. Provide secure lock-up for door hardware delivered to the Project, but not yet installed. Control handling and installation of hardware items that are not immediately replaceable so that completion of the Work will not be delayed by hardware losses both before and after installation.

## 1.7 PROJECT CONDITIONS AND COORDINATION

- A. Coordination: Coordinate hardware with other work. Furnish hardware items of proper design for use on doors and frames of the thickness, profile, swing, security and similar requirements indicated, as necessary for proper installation and function, regardless of omissions or conflicts in the information on the Contract Documents.
  - 1. Embedded and attached items to concrete.
  - 2. Backing in walls for related hardware items.
  - 3. Finish floor materials and floor-mounted hardware, including height differentials.
  - 4. Furnish manufacturers templates to door and frame fabricators.
  - 5. Check Shop Drawings for doors and entrances to confirm that adequate provisions will be made for proper hardware installation.
    - a. Confirm that door manufacturers furnish necessary UBC-7-2 compliant seal system.
- B. Check the Shop Drawings for doors and entrances to confirm that adequate provisions will be made for the proper installation of hardware.
  - 1. Provide written documentation that this requirement has been done prior to installation of hardware.

# 1.8 WARRANTY

- A. Provide guarantee from hardware supplier as follows:
  - 1. Locksets: 3 years.
  - 2. All other Hardware: Two years.
- B. Provide written documentation that the Hardware supplier has made an inspection of the finished installation, including the electrical, electronic and pneumatic hardware, and has verified that the hardware is functioning properly and is installed per manufacturer's written instructions.

#### 1.10 COMMISSIONING:

A. Test door hardware operation with climate control system and stairwell pressurization system both at rest and while in full operation.

#### 1.11 REGULATORY REQUIREMENTS:

- A. Locate latching hardware between 34" to 44" above the finished floor, per California Building Code, Section 1008.1.9.2 and 11B-404.2.7.
  - 1. Locate panic hardware between 36" to 44" above the finished floor.
- B. Handles, pull, latches, locks, other operating devices: readily openable without tight grasping, tight pinching, or twisting of the wrist to operate. California Building Code 1008.1.9.1 and 11B-309.4.
- C. Adjust doors to open with not more than 5.0 lbs pressure to open at exterior doors and 5.0 lbs at interior doors. As allowed per California Building Code, Section 11B-404.2.9, local authority may increase the allowable pressure for fire doors to achieve positive latching, but not to exceed 15 lbs.
- D. Adjust door closer sweep periods so that from an open position of 90 degrees, the door will take at least 5 seconds to move to a point 12 degrees from the latch, measured to the landing side of the door, per California Building Code Section 11B-404.2.8.1.
- E. Smooth surfaces at bottom 10" of push sides of doors, facilitating push-open with wheelchair footrests, per California Building Code Section 11B-404.2.10.
- F. Door opening clear width no less than 32", measured from face of frame stop, or edge of inactive leaf of pair of doors, to door face with door opened to 90 degrees. Hardware projection not a factor in clear width if located above 34" and the hardware projects no more than 4". California Building Code Section 11B-404.2.3, 11B-404.2.4, and 1008.1.1.
- G. Door opening clear height no less than 80" measured from top of sill to bottom of frame header stop. Projections into clear opening height not to exceed 4". California Building Code Section 11B-404.2.3 and 1008.1.1.1.
- H. Thresholds: floor or landing no more than 1/2" below the top of the threshold of the doorway. Change in level between 1/4" and 1/2": beveled to slope no greater than 1:2 (50 percent slope). California Building Code Section 11B-404.2.5.
- I. Floor stops: Do not locate in path of travel. Locate no more than 4" from walls.
- J. Pairs of doors: limit swing of one leaf to 90 degrees to protect persons reading wall-mounted tactile signage.

PART 2 - PRODUCTS

2.1 MANUFACTURERS

- A. Listed Acceptable Substitutes must provide the function and features of the specified product otherwise it will not be considered for substitution or approved.
- B. Items listed with no substitute, manufacturers have been established by the Owner to be the existing standard per public contract code. No substitutions will be considered.

| Item:           | Manufacturer: | Acceptable Substitution: |  |
|-----------------|---------------|--------------------------|--|
| Hinges          | (IVE) Ives    | Bommer                   |  |
| Cylinders       | (SCH)Schlage  | Facility Standard        |  |
| Locks           | (SCH)Schlage  | Facility Standard        |  |
| Silencers       | (IVE) Ives    | Trimco                   |  |
| Kickplates      | (IVE) Ives    | Trimco                   |  |
| Stops & Holders | (IVE) Ives    | Trimco                   |  |
|                 |               |                          |  |

- C. Furnish items of hardware required to complete the work in accordance with these specifications and the manufacturers instructions, including items of hardware inadvertently omitted from this specification. Items omitted to be of equal quality and type as indicated herein.
- D. Where the exact types of hardware specified are not adaptable to the finished shape or size of the members requiring hardware, furnish suitable types having as nearly as practicable the same operation and quality as the type specified, subject to Architect's approval.
- E. Carefully inspect Project for the extent of the finish hardware required to complete the Work. Where there is a conflict between these Specifications and the existing hardware, furnish finish hardware to specification.
  - 1. Provide notification of conflict to Architect thirty (30) days prior to Contractor proceeding with project hardware installation.

## 2.2 HANGING MEANS:

- A. Conventional Hinges: Hinge open widths minimum, but of sufficient throw to permit maximum door swing. Steel or stainless-steel pins and concealed bearings.
  - 1. Three hinges per leaf to 7 foot, 6 inch height. Add one for each additional 30 inches in height or fraction thereof.
  - 2. Extra heavy weight hinges on doors over 3 foot, 5 inches in width.
  - 3. Outswinging exterior doors: non-ferrous with nonremovable (NRP) pins.
  - 4. Non-ferrous material exteriors and at doors subject to corrosive atmospheric conditions.
  - 5. Provide shims and shimming instructions for proper door adjustment.

# 2.3 LOCKSETS, LATCHSETS:

- A. Extra Heavy Duty Cylindrical Locks and Latches: as scheduled.
  - Chassis: cylindrical design, corrosion-resistant plated cold-rolled steel, throughbolted
  - 2. Locking Spindle: stainless steel, interlocking design.
  - 3. Latch Retractors: forged steel. Balance of inner parts: corrosion-resistant plated steel, or stainless steel.
  - 4. Lever Trim: accessible design, independent operation, spring-cage supported, minimum 2" clearance from lever mid-point to door face.
  - 5. Strikes: 16 gage curved steel, bronze or brass with 1" deep box construction, lips of

- sufficient length to clear trim and protect clothing.
- 6. Lock Series and Design: Schlage D series, "Rhodes" design.
- Certifications:
  - a. ANSI A156.2, 1994, Series 4000, Grade 1.
  - b. UL listed for A label and lesser class single doors up to 4ft x 8ft.
- 8. Comply with CBC Section 11B-304.4.

#### 2.4 OTHER HARDWARE

- A. Kick Plates: Four beveled edges, .050 inches minimum thickness, height and width as scheduled. Sheet-metal screws of bronze or stainless steel to match other hardware.
- B. Door stops: Provide stops to protect walls, casework or other hardware.
  - 5. Unless otherwise noted in Hardware Sets, provide wall type with appropriate fasteners. Where wall type cannot be used, provide floor type. If neither can be used, provide overhead type.
- C. Fasteners: Generally, exposed screws to be Phillips or Robertson drive. Pinned TORX drive at high security areas. Flat head sleeve anchors (FHSL) may be slotted drive. Sheet metal and wood screws: full-thread. Sleeve nuts: full length to prevent door compression.
- D. Silencers: Interior hollow metal frames, 3 for single doors, 4 for pairs of doors. Omit where adhesive mounted seal occurs. Leave no unfilled/uncovered prepunched silencer holes.

#### 2.5 FINISH

- A. Generally BHMA 626 Dull Chromium, unless otherwise noted.
- B. Aluminum items: match predominant adjacent material.

## 2.6 KEYING REQUIREMENTS

- A. Initiate and conduct meeting(s) with the Owner and a representative from Ingersoll-Rand Security Technologies to determine the Key System requirements. Keying system shall be approved by Owner's representative in writing. Furnish construction key system in accordance with lock manufacturers standard. Where interchangeable core systems are used, use temporary cores for construction keying. Stamp keys "Do Not Duplicate".
  - 1. Key system: Schlage Primus Classic.
- B. Locks and cylinders: Keyed at the factory of the lock manufacturer where permanent records are maintained. Locks and cylinders shall be of the same manufacturer.
- C. Permanent keys: Deliver only to Owner's representative.
- D. Quantity of Keys:
  - 1. Change Keys: Three per cylinder with a maximum of ten keys per keyed alike group. Furnish balance due as blanks.
  - 2. Master Keys: Six per set.
  - 3. Control Keys: Two.
  - 4. Construction Keys: Twenty.
  - Construction Control Keys: Two.
- E. Keying Schedule: Submit three copies indicating how the Owner's final instructions has been fulfilled.

- F. Key Transcript: Supply to Owner upon completion.
- G. Permanent cylinders and keys shall be delivered directly to the Owner. Upon Project completion, Owner shall install permanent cylinders.
- H. Temporary cylinders shall be furnished and installed as required for security and testing for proper operation of locking hardware.

#### PART 3 - EXECUTION

#### 3.1 HARDWARE LOCATIONS

- A. Locate hardware per SDI-100 and applicable building, fire, life-safety, accessibility and security codes.
  - 1. Where new hardware is to be installed near existing doors/hardware which are scheduled to remain, match locations of the existing hardware.
  - 2. Notify Architect of any code conflicts before ordering material.

## 3.2 INSTALLATION

- A. Mount hardware units at heights indicated in following applicable publications, except as specifically indicated or required to comply with governing regulations and except as otherwise directed by Architect.
  - 1. "Recommended Locations for Builders Hardware for standard Steel Doors and Frames" by Door and Hardware Institute and as indicated herein.
  - 2. NWWDA Industry Standard I.S.1.7, "Hardware Locations for Wood Flush Doors."
- B. Install each hardware item in compliance with the manufacturer's instructions and recommendations. Where cutting and fitting is required to install hardware onto or into surfaces that are later to be painted or finished in another way, coordinate removal, storage, and reinstallation or application of surface protection with finishing work specified in the Division 9 Sections. Do not install surface-mounted items until finishes have been completed on the substrates involved.
- C. Set units level, plumb, and true to line and location. Adjust and reinforce the attachment substrate as necessary for proper installation and operation.
- D. Drill pilot holes for fasteners in wood doors and/or frames. Drill and countersink units that are not factory prepared for anchorage fasteners. Space fasteners and anchors in accordance with industry standards.
- E. Use manufacturers' fasteners furnished with hardware items, or submit Request for substitution with Architect.

#### 3.3 ADJUSTING

- A. Adjust and check each operating item of hardware and each door, to ensure proper operation or function of every component. Replace components which cannot be adjusted, to operate freely and smoothly.
  - 1. Hardware damaged by improper adjustment methods to be repaired or replaced to

# TC – G-8 ALLIED HEALTH LAB REMODEL Owner's satisfaction.

# 3.4 DEMONSTRATION:

A. Demonstrate hardware systems, including adjustment and maintenance procedures.

# 3.5 PROTECTION/CLEANING:

- A. Cover installed hardware, protect from paint, cleaning agents, weathering, carts/barrows, etc. Remove covering materials and clean hardware just prior to substantial completion.
- B. Clean adjacent wall, frame and door surfaces soiled from installation/reinstallation process.

# 3.6 SCHEDULE OF FINISH HARDWARE

A. The Door Schedule on the Drawings indicates which Hardware Set is used with each door.

## HW SET: 01

| 3 | EA | HINGE            | 5BB1HW 5 X 4.5    | 652 | IVE |
|---|----|------------------|-------------------|-----|-----|
| 1 | EA | OFFICE LOCK      | ND91TD RHO        | 626 | SCH |
| 1 | EA | PRIMUS CORE ONLY | 20-740            | 626 | SCH |
| 1 | EΑ | KICK PLATE       | 8400 10" X 2" LDW | 630 | IVE |
| 2 | EA | WALL STOP        | WS401/402CCV      | 626 | IVE |
| 3 | EA | SILENCER         | SR64              | GRY | IVE |

END OF SECTION 087100

10/06/14

## PART 1 - GENERAL

#### 1.1 RELATED DOCUMENTS:

A. Drawings and general provisions of Contract, including General and Supplementary Conditions and Division 0 & 1 Specification sections, apply to work of this section.

## 1.2 SUMMARY:

- A. This Section includes, but not limited to, the following:
  - Extent of glass and glazing work is indicated on drawings. Where glazing is added to existing system, tint of glass shall match existing. Adjacent to be picked from available manufactured tints. Provide safety glazing per CBC Code requirements.
    - a. 6 mm glass one way mirror.
  - 2. Types of work in this section include glass and glazing for:
    - a. Hollow metal window.
- B. Related Sections: The following Sections contain requirements that relate to this Section:
  - 1. Division 8 Section "Standard Steel Doors and Frames" for hollow metal windows.

## 1.3 SYSTEM DESCRIPTION:

- A. Provide glass and glazing that has been produced, fabricated and installed to withstand normal thermal movement, wind loading and impact loading (where applicable), without failure including loss or breakage of glass, failure of sealants or gaskets to remain watertight and airtight, deterioration of glass and glazing materials and other defects in the work.
  - Normal thermal movement is defined as that resulting from an ambient temperature range of 120 deg. F (67 deg. C) and from a consequent temperature range within glass and glass framing members of 180 deg. F (100 deg. C).
  - Deterioration of insulating glass is defined as failure of hermetic seal due to other causes than breakage which results in intrusion of dirt or moisture, internal condensation or fogging, deterioration of protected internal glass coating, if any, resulting from seal failure, and any other visual evidence of seal failure or performance.
  - 3. Deterioration of coated glass is defined as the development of manufacturing defects including peeling, cracking or other indications of deterioration in metallic coating due to normal conditions of use.

#### 1.4 SUBMITTALS:

A. Product Data: Submit manufacturer's technical data for each glazing material and fabricated glass product required, including installation and maintenance instructions.

- B. Samples: Submit, for verification purposes, 12" square samples of each type of glass indicated except for clear single pane units, and 12" long samples of each color required (except black) for each type of sealant or gasket exposed to view. Install sealant or gasket sample between two strips of material representative of adjoining framing system in color.
- Certificate: Submit certificates from respective manufacturers attesting that glass and glazing materials furnished for project comply with requirements.
  - Separate certification will not be required for glazing materials bearing manufacturer's permanent labels designating type and thickness of glass, provided labels represent a quality control program involving a recognized certification agency or independent testing laboratory acceptable to authorities having jurisdiction.
- D. Compatibility and Adhesion Test Report: Submit statement from sealant manufacturer indicating that glass and glazing materials have been tested for compatibility and adhesion with glazing sealants and interpreting test results relative to material performance, including recommendations for primers and substrate preparation needed to obtain adhesion.
- E. Project Closeout Requirements:
  - 1. Insulated glass 5-year warranty.
  - 2. Laminated glass 5-year warranty.

## 1.5 PERFORMANCE REQUIREMENTS

- A. Fire-rated glass ceramic clear and wireless glazing material listed for use in non-impact safety rated location such as transoms and borrowed lites with dire rating requirements ranging from 20 to 90 minutes with required hose stream test.
- B. Passes positive pressure test standards UL10C and UL9.

## 1.6 QUALITY ASSURANCE:

- A. Glazing Standards: Comply with recommendations of Flat Glass Marketing Association (FGMA) "Glazing Manual" and "Sealant Manual" except where more stringent requirements are indicated. Refer to those publications for definitions of glass and glazing terms not otherwise defined in this section or other referenced standards.
- B. Safety Glazing Standard: Where safety glass is indicated by "SG", and/or required by authorities having jurisdiction, provide type of products indicated which comply with CBC 2406 and ANSI Z97.1 and testing requirements of 16 CFR Part 1201 for category II materials.
- C. Fire Resistance Rated Wire Glass: Provide wire glass products that are identical to those tested per ASTM E 163 (UL 9) and are labeled and listed by UL or other testing and inspecting agency acceptable to authorities having jurisdiction.
- D. Insulating Glass Certification Program: Provide insulating glass units permanently marked either on spacers or at least one component pane of units with appropriate certification label of inspecting and testing organization indicated below:
  - 1. Insulating Glass Certification Council (IGCC).

- E. Single Source Responsibility for Glass: To ensure consistent quality of appearance and performance, provide materials produced by a single manufacturer or fabricator for each kind and condition of glass indicated and composed of primary glass obtained from a single source for each type and class required.
- F. Glazing materials and glazing subject to human impact shall comply with CBC Chapter 24.

#### 1.6 DELIVERY, STORAGE, AND HANDLING:

- A. Protect glass and glazing materials during delivery, storage and handling to comply with manufacturer's directions and as required to prevent edge damage to glass, and damage to glass and glazing materials from effects of moisture including condensation, of temperature changes, of direct exposure to sun, and from other causes.
  - Where insulating glass units will be exposed to substantial altitude changes, avoid hermetic seal ruptures by complying with insulating glass fabricator's recommendations for venting and sealing.

#### 1.7 PROJECT CONDITIONS:

A. Environmental Conditions: Do not proceed with glazing when ambient and substrate temperature conditions are outside the limits permitted by glazing material manufacturer or when joint substrates are wet due to rain, frost, condensation or other causes.

## 1.8 WARRANTY:

- A. General: Warranties shall be in addition to, and not a limitation of, other rights the Owner may have under the Contract Documents.
- B. Manufacturer's Special Project Warranty on Insulating Glass: Provide written warranty signed by manufacturer of insulating glass agreeing to furnish f.o.b. point of manufacture, freight allowed project site, within specified warranty period indicated below, replacements for those insulating glass units developing manufacturing defects. Manufacturing defects are defined as failure or hermetic seal of air space (beyond that due to glass breakage) as evidenced by intrusion of dirt or moisture, internal condensation or fogging, deterioration of protected internal glass coatings, if any, and other visual indications of seal failure or performance; provided the manufacturer's instructions for handling, installing, protecting and maintaining units have been complied with during the warranty period.
  - Warranty Period: Manufacturer's standard but not less than 5 years after date of notice of completion. All labor and related work to install replaced glass shall be included in warranty.

#### PART 2 - PRODUCTS

# 2.1 MANUFACTURERS:

A. Available Manufacturers: Subject to compliance with requirements, manufacturers offering

products which may be incorporated in the work include; but are not limited to, the following:

- 1. Clear and Tinted Float Glass:
  - a. PPG Industries, Inc.
- 2. One way mirror:
  - a. Pilkington Building Products "Mirrorpane"

# 2.2 GLASS PRODUCTS, GENERAL:

- A. Primary Glass Standard: Provide primary glass which complies with ASTM C 1036 requirements, including those indicated by reference to type, class, quality, and, if applicable, form, finish, mesh and pattern.
- B. Heat-Treated Glass Standard: (Safety Glazing) Provide heat-treated glass which complies with ASTM C 1048 requirements, including those indicated by reference to kind, condition, type, quality, class, and, if applicable, form, finish, and pattern.
- C. Sizes: Fabricate glass to sizes required for glazing openings indicated, with edge clearances and tolerances complying with recommendations of glass manufacturer. Provide thicknesses indicated or, if not otherwise indicated, as recommended by glass manufacturer for application indicated.
- D. Ceramic-Coated Heat-Treated Spandrel Glass: Condition B (spandrel glass, one surface ceramic coated), Type I (transparent glass, flat), Quality q<sup>3</sup> (glazing select), with ceramic coating applied to second surface and complying with the following requirements:
  - 1. Kind FT (fully tempered).
  - 2. Color: As indicated or, if not otherwise indicated, as selected by Architect from manufacturer's standard colors.
- E. Acid Etched Glass: Primary glass/heat treated where required with frosted appearance.

#### 2.3 PRIMARY GLASS PRODUCTS:

- A. Clear Float Glass: Type I (transparent glass, flat), Class 1 (clear), Quality q3 (glazing select).
  - Low iron glass:
    - a. Thickness 6mm
    - b. Visible light transmittance 91%
    - c. Reflectance 8%
    - d. U-Value 5.8
    - e. Shading Coefficient 1.03
    - f. Basis of Design- The design for each piece of glass is based on the product named. Subject to compliance with requirements, provide either the named product or an approved comparable product by one of the other manufactures specified. Subject to compliance with requirements, manufacturers offering glass that may be incorporated into the Work include, but are not limited to, the following. Model numbers and names for glass manufactured by PPG Starphire below are listed to establish a standard of quality for design, function, materials and workmanship. Other manufacturers may be submitted for evaluation by the Architect. Approval shall be obtained 45 days after Contract date. The Architect shall be the sole judge as to the acceptability of all products submitted for substitution.

- i. Starphire Ultra Clear Glass by PPG
- B. One-Way Mirrors: Transparent mirror formed by chemical vapor deposition on 1/4" grey tinted glass.
  - 1. Visible transmittance: 12%.
  - 2. Visible Transmittance on coated side: 60%.
  - 3. Recommended light ratios: 10:1 Subject side: Observer side.
  - 4. Mirror coating towards subject side.

# 2.4 HEAT-TREATED GLASS PRODUCTS (SAFETY GLAZING):

- A. Manufacturing Process: Manufacture heat-treated glass as follows:
  - 1. By vertical (tong-held) or horizontal (roller hearth) process, at manufacturer's option, except provide horizontal process where indicated as "tong less" or "free of tong marks".
- B. Uncoated Clear Heat-Treated Float Glass: Condition A (uncoated surfaces), Type I (transparent glass, flat), Class 1 (clear), Quality q3 (glazing select), kind as indicated below.
  - 1. Kind FT (fully tempered) where indicated and required by CBC. (SG--Safety Glazing indicated on Construction Documents.)
- C. Uncoated Tinted Heat-Treated Float Glass: Condition A (uncoated surfaces), Type I (transparent glass, flat), Class 2 (tinted heat absorbing and light reducing), Quality q3 (glazing select), with tint color and performance characteristics for 1/4" thick glass matching those indicated for non-heat-treated tinted float glass; kind as indicated below:
  - 1. Kind FT (fully tempered) where indicated and required by CBC.

## 2.5 ELASTOMERIC GLAZING SEALANTS AND PREFORMED GLAZING TAPES:

- A. General: Provide products of type indicated and complying with the following requirements:
  - Compatibility: Select glazing sealants and tapes of proven compatibility with other materials with which they will come into contact, including glass products, seals of insulating glass units, and glazing channel substrates, under conditions of installation and service, as demonstrated by testing and field experience.
  - Suitability: Comply with recommendations of sealant and glass manufacturers for selection of glazing sealants and tapes which have performance characteristics suitable for applications indicated and conditions at time of installation.
  - Elastomeric Sealant Standard: Provide manufacturer's standard chemically curing, elastomeric sealant of base polymer indicated which complies with ASTM C 920 requirements, including those for Type, Grade, Class and Uses.
  - 4. Colors: Provide color of exposed sealants indicated or, if not otherwise indicated, as selected by Architect from manufacturer's standard colors.
- B. One-Part Non-Acid-Curing Silicone Glazing Sealant: Type S; Grade NS, Class 25; Uses NT, G, A, and, as applicable to uses indicated, O; and complying with the following requirements for modulus and additional joint movement capability. (Indicated ES5 on Plans.)
  - 1. Low Modulus: Tensile strength of 45 psi or less at 100 percent elongation when tested per ASTM D 412 after 14 days at 77 deg. F (20 deg. C) and 50 percent relative humidity.

- Additional capability, when tested per ASTM C 719 for adhesion and cohesion under maximum cyclic movement, to withstand the following percentage increase and decrease of joint width, as measured at time of application, and remain in compliance with other requirements of ASTM C 920.
  - a. 50 percent each way for a total 100% movement.
- C. Preformed Butyl-Polyisobutylene Glazing Tape: Provide manufacturer's standard solvent-free butyl-polyisobutylene formulation with a solids content of 100 percent; complying with AAMA A 804.1; in extruded tape form; non-staining and non-migrating in contact with nonporous surfaces; packaged on rolls with a release paper on one side; with or without continuous spacer rod as recommended by manufacturers of tape and glass for application indicated.
- D. Available Products: Subject to compliance with requirements, glazing sealants which may be incorporated in the work include, but are not limited to, the following:
  - 1. One-Part Non-Acid Curing Low-Modulus Silicone Glazing Sealant:
    - a. "Dow Corning 795"; Dow Corning Corp.
    - b. "895"; Pecora Corp.
  - 2. Preformed Butyl-Polyisobutylene Glazing Tape Without Spacer Rod:
    - a. "Chem-Tape 40"; Bostik Construction Products Div.
    - b. "Extru-Seal"; Pecora Corp.
    - c. "PTI 303" Glazing Tape; Protective Treatments, Inc.
    - d. "Tremco 440 Tape"; Tremco Inc.
  - Preformed Butyl-Polyisobutylene Glazing Tape With Spacer Rod:
    - a. "Chem-Tape 60"; Bostik Construction Products Div.
    - b. "Shim-Seal"; Pecora Corp.
    - c. "PTI 303" Shim Tape; Protective Treatments, Inc.
    - d. "Pre-shimmed Tremco 440 Tape"; Tremco Inc.

# 2.6 GLAZING GASKETS:

- A. Lock-Strip Gaskets: Neoprene extrusions of size and shape indicated, fabricated into frames with molded corner units and zipper lock strips, complying with ASTM C 542; black.
- B. Dense Elastomeric Compression Seal Gaskets: Molded or extruded gaskets of material indicated below, complying with ASTM C 864, of profile and hardness required to maintain watertight seal:
  - 1. Neoprene.
  - EPDM.
  - 3. Thermoplastic polyolefin rubber.
  - 4. Any material indicated above.
- C. Available Manufacturers: Subject to compliance with requirements, manufacturers offering products which may be incorporated in the work include, but are not limited to, the following:
  - Manufacturers of Lock-Strip Gaskets:

- a. Cadillac Rubber & Plastics, Inc.
- b. Maloney Precision Products Co.
- c. The Standard Products Co.

#### 2. Manufacturers of Preformed Gaskets:

- a. D. S. Brown Co.
- b. Maloney Precision Products Co.
- c. Tremco.

#### 2.7 MISCELLANEOUS GLAZING MATERIALS:

- D. Compatibility: Provide materials with proven record of compatibility with surfaces contacted in installation.
- E. Cleaners, Primers and Sealers: Type recommended by sealant or gasket manufacturer.
- F. Setting Blocks: Neoprene, EPDM or silicone blocks as required for compatibility with glazing sealants, 80 to 90 Shore A durometer hardness.
- G. Spacers: Neoprene, EPDM or silicone blocks, or continuous extrusions, as required for compatibility with glazing sealant, of size, shape and hardness recommended by glass and sealant manufacturers for application indicated.
- H. Edge Blocks: Neoprene, EPDM or silicone blocks as required for compatibility with glazing sealant, of size and hardness required to limit lateral movement (side-walking) of glass.

#### PART 3 - EXECUTION

# 3.1 EXAMINATION:

A. Require Glazier to inspect work of glass framing erector for compliance with manufacturing and installation tolerances, including those for size, squareness, offsets at corners; for presence and functioning of weep system; for existence of minimum required face or edge clearances; and for effective sealing of joinery. Obtain Glazier's written report listing conditions detrimental to performance of glazing work. Do not allow glazing work to proceed until unsatisfactory conditions have been corrected.

# 3.2 PREPARATION:

A. Clean glazing channels and other framing members to receive glass, immediately before glazing. Remove coatings, which are not firmly bonded to substrates. Remove lacquer from metal surfaces where elastomeric sealants are indicated for use.

# 3.3 GLAZING, GENERAL:

A. Comply with combined printed recommendations of glass manufacturers, of manufacturers of sealants, gaskets and other glazing materials, except where more stringent requirements are indicated, including those of referenced glazing standards.

- B. Glazing channel dimensions as indicated in details are intended to provide for necessary bite on glass, minimum edge and face clearances, and adequate sealant thicknesses, with reasonable tolerances. Adjust as required by job conditions at time of installation.
- C. Protect glass from edge damage during handling and installation; use a rolling block in rotating glass units to prevent damage to glass corners. Do not impact glass with metal framing. Use suction cups to shift glass units within openings; do not raise or drift glass with a pry bar. Rotate glass with flares or bevels along one horizontal edge which would occur in vicinity of setting blocks so that these are located at top of opening. Remove from project and dispose of glass units with edge damage or other imperfections of kind that, when installed, weakens glass and impairs performance and appearance.
- D. Apply primers to joint surfaces where required for adhesion of sealants, as determined by preconstruction sealant-substrate testing.

#### 3.4 GLAZING:

- A. Install setting blocks of proper size in sill rabbet, located one quarter of glass width from each corner, but with edge nearest corner not closer than 6" from corner, unless otherwise required. Set blocks in thin course of sealant, which is acceptable for heel bead use.
- B. Provide spacers inside and out, of correct size and spacing to preserve required face clearances, for glass sizes larger than 50 united inches (length plus height), except where gaskets or glazing tapes with continuous spacer rods are used for glazing. Provide 1/8" minimum bite of spacers on glass and use thickness equal to sealant width, except with sealant tape use thickness slightly less than final compressed thickness of tape.
- C. Provide edge blocking to comply with requirements of referenced glazing standard, except where otherwise required by glass unit manufacturer.
- D. Set units of glass in each series with uniformity of pattern, draw, bow and similar characteristics.
- E. Provide compressible filler rods or equivalent back-up material, as recommended by sealant and glass manufacturers, to prevent sealant from extruding into glass channel weep systems and from adhering to joints back surface as well as to control depth of sealant for optimum performance, unless otherwise indicated.
- F. Force sealants into glazing channels to eliminate voids and to ensure complete "wetting" or bond of sealant to glass and channel surfaces.
- G. Tool exposed surfaces of sealants to provide a substantial "wash" away from glass. Install pressurized tapes and gaskets to protrude slightly out of channel, so as to eliminate dirt and moisture pockets.
- H. Where wedge-shaped gaskets are driven into one side of channel to pressurize sealant or gasket on opposite side, provide adequate anchorage to ensure that gasket will not "walk" out when installation is subjected to movement.
- Miter cut wedge-shaped gaskets at corners and install gaskets in manner recommended by gasket manufacturer to prevent pull away at corners; seal corner joints and butt joints with sealant recommended by gasket manufacturer.

## 3.5 PROTECTION AND CLEANING:

- A. Protect exterior glass from breakage immediately upon installation by use of crossed streamers attached to framing and held away from glass. Do not apply markers to surfaces of glass. Remove nonpermanent labels and clean surfaces.
- B. Protect glass from contact with contaminating substances resulting from construction operations. If, despite such protection, contaminating substances do come into contact with glass, remove immediately by method recommended by glass manufacturer.
- C. Examine glass surfaces adjacent to or below exterior concrete and other masonry surfaces at frequent intervals during construction, but not less often than once a month, for build-up of dirt, scum, alkali deposits or staining. When examination reveals presence of these forms of residue, remove by method recommended by glass manufacturer.
- D. Remove and replace glass which is broken, chipped, cracked, abraded or damaged in other ways during construction period, including natural causes, accidents and vandalism.
- E. Wash glass on both faces not more than 4 days prior to date scheduled for inspections intended to establish date of substantial completion in each area of project. Wash glass by method recommended by glass manufacturer.

END OF SECTION 088000

# SECTION 092900 - GYPSUM BOARD ASSEMBLIES

09/30/22

#### PART 1 - GENERAL

## 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 0 & 1 Specification Sections, apply to this Section.

#### 1.2 SUMMARY

- A. This Section includes, but not limited to, the following:
  - 1. Gypsum board assemblies attached to wood framing and furring members.
  - 2. Patching existing walls match adjacent finish.
  - 3. Interior gypsum board wall panels.
- B. Related Sections include the following:
  - 1. Division 6 Section "Rough Carpentry" for wood framing and furring.
  - 2. Division 7 Section "Building Insulation" for insulation in gypsum board assemblies.
  - 3. Division 7 Section "Joint Sealants" for related sealants.

#### 1.3 DEFINITIONS

A. Gypsum Board Terminology: Refer to ASTM C 11 for definitions of terms for gypsum board assemblies not defined in this Section or in other referenced standards.

## 1.4 SUBMITTALS

- A. Product Data: For each type of product indicated.
- B. Shop Drawings: Show locations, fabrication, and installation of control and expansion joints including plans, elevations, sections, details of components, and attachments to other units of Work.
- C. Samples: For the following products:
  - 1. Trim Accessories: Full-size sample in 12-inch- (300-mm-) long length for each trim accessory indicated.
  - 2. Textured Finishes: For each textured finish indicated and on same backing indicated for Work

## 1.5 QUALITY ASSURANCE

A. Fire-Test-Response Characteristics: For gypsum board assemblies with fire-resistance ratings, provide materials and construction identical to those tested in assembly indicated according to ASTM E 119 by an independent testing and inspecting agency acceptable to authorities having jurisdiction.

- 1. Fire-Resistance-Rated Assemblies: Indicated by design designations from GA-600, "Fire Resistance Design Manual."
- B. Sound Transmission Characteristics: For gypsum board assemblies with STC ratings, provide materials and construction identical to those tested in assembly indicated according to ASTM E 90 and classified according to ASTM E 413 by a qualified independent testing agency.
  - 1. STC-Rated Assemblies: Indicated by design designations from GA-600, "Fire Resistance Design Manual."
- C. Gypsum Board Finish Mockups: Before finishing gypsum board assemblies, install mockups of at least 100 sq. ft. (9 sq. m) in surface area to demonstrate aesthetic effects and qualities of materials and execution.
  - 1. Install mockups for the following applications:
    - a. Surfaces with texture finishes.
    - b. Surfaces indicated to receive nontextured paint finishes.
    - c. Surfaces indicated to receive textured paint finishes.
  - 2. Simulate finished lighting conditions for review of mockups.

# 1.6 DELIVERY, STORAGE, AND HANDLING

- A. Deliver materials in original packages, containers, or bundles bearing brand name and identification of manufacturer or supplier.
- B. Store materials inside under cover and keep them dry and protected against damage from weather, direct sunlight, surface contamination, corrosion, construction traffic, and other causes. Stack gypsum panels flat to prevent sagging.

# 1.7 PROJECT CONDITIONS

A. Environmental Limitations: Comply with ASTM C 840 requirements or gypsum board manufacturer's written recommendations, whichever are more stringent.

## PART 2 - PRODUCTS

## 2.1 MANUFACTURERS

- A. Available Manufacturers: Subject to compliance with requirements, manufacturers offering products that may be incorporated into the Work include, but are not limited to, the following:
  - 1. Gypsum Board and Related Products:
    - a. American Gypsum Co.
    - b. Georgia Pacific Gypsum LLC.
    - c. National Gypsum Company.
    - d. United States Gypsum Co.

# 2.3 INTERIOR GYPSUM WALLBOARD

- A. Panel Size: Provide in maximum lengths and widths available that will minimize joints in each area and correspond with support system indicated.
- B. Glass-Mat Interior Gypsum Board: ASTM C 1658/C 1658M. With fiberglass mat laminated to both sides. Specifically designed for interior use.
  - 1. Products: Subject to compliance with requirements, available products that may be incorporated into the Work include, but not limited t the following:
    - a. Georgia-Pacific Gypsum LLC; DensArmour Plus®
    - b. National Gypsum Company, eXP Extreme
  - 2. Core: 5/8 inch (15.9 mm), Type X and 5/8 inch (15.9 mm), abuse resistant
  - 3. Long Edge: Tapered.
  - 4. Mold Resistance: ASTM D 3273, score of 10.

## 2.4 TRIM ACCESSORIES

- A. Interior Trim: ASTM C 1047.
  - 1. Material: Galvanized or aluminum-coated steel sheet, rolled zinc, plastic, or paper-faced galvanized steel sheet.
  - 2. Shapes:
    - a. Cornerbead: Use at outside corners, unless otherwise indicated.
    - b. Bullnose Bead: Use where indicated.
    - c. LC-Bead (J-Bead): Use at exposed panel edges.
    - d. L-Bead: Use where indicated.
    - e. U-Bead: Use where indicated.
    - f. Expansion (Control) Joint: Use where indicated.
    - g. Curved-Edge Cornerbead: With notched or flexible flanges; use at curved openings.
- B. Aluminum Trim: Extruded accessories of profiles and dimensions indicated.
  - 1. Available Manufacturers: Subject to compliance with requirements, manufacturers offering products that may be incorporated into the Work include, but are not limited to, the following:
    - a. Fry Reglet Corp.
    - b. Gordon, Inc.
    - c. MM Systems Corporation.
    - d. Pittcon Industries.
  - 2. Aluminum: Alloy and temper with not less than the strength and durability properties of ASTM B 221 (ASTM B 221M), alloy 6063-T5.
  - 3. Finish: Class II anodic finishes, clear.

## 2.5 JOINT TREATMENT MATERIALS

A. General: Comply with ASTM C 475.

# B. Joint Tape:

- 1. Interior Gypsum Wallboard: Glassmat or Paper.
- 2. Exterior Gypsum Soffit Board: Glassmat.
- 3. Glass-Mat Gypsum Sheathing Board: 10-by-10 glass mesh.
- 4. Tile Backing Panels: As recommended by panel manufacturer.
- C. Joint Compound for Interior Gypsum Wallboard: For each coat use formulation that is compatible with other compounds applied on previous or for successive coats.
  - 1. Prefilling: At open joints, rounded or beveled panel edges, and damaged surface areas, use setting-type taping compound.
  - 2. Embedding and First Coat: For embedding tape and first coat on joints, fasteners, and trim flanges, use.
    - a. Use setting-type compound for installing paper-faced metal trim accessories.
  - 3. Fill Coat: For second coat, use setting-type, sandable topping compound.
  - 4. Finish Coat: For third coat, use setting-type, sandable topping compound.
  - 5. Skim Coat: For final coat of Level 5 finish, use setting-type, sandable topping compound.
- D. Joint Compound for Exterior Applications:
  - 1. Exterior Gypsum Soffit Board: Use setting-type taping and setting-type, sandable topping compounds.
  - 2. Mat Gypsum Sheathing Board: As recommended by manufacturer.
- 2.6 ACOUSTICAL SEALANT (Indicated as ES7 on drawings)
  - A. Available Products: Subject to compliance with requirements, products that may be incorporated into the Work include, but are not limited to, the following:
    - 1. Acoustical Sealant for Exposed and Concealed Joints:
      - a. Pecora Corp.; AC-20 FTR Acoustical and Insulation Sealant.
      - b. United States Gypsum Co.; SHEETROCK Acoustical Sealant.
    - 2. Acoustical Sealant for Concealed Joints:
      - a. Ohio Sealants, Inc.; Pro-Series SC-170 Rubber Base Sound Sealant.
      - b. Pecora Corp.; BA-98.
      - c. Tremco, Inc.; Tremco Acoustical Sealant.
  - B. Acoustical Sealant for Exposed and Concealed Joints: Nonsag, paintable, nonstaining, latex sealant complying with ASTM C 834 that effectively reduces airborne sound transmission through perimeter joints and openings in building construction as demonstrated by testing representative assemblies according to ASTM E 90.
  - C. Acoustical Sealant for Concealed Joints: Nondrying, nonhardening, nonskinning, nonstaining, gunnable, synthetic-rubber sealant recommended for sealing interior concealed joints to reduce airborne sound transmission.
- 2.7 AUXILIARY MATERIALS

- A. General: Provide auxiliary materials that comply with referenced installation standards and manufacturer's written recommendations.
- B. Laminating Adhesive: Adhesive or joint compound recommended for directly adhering gypsum panels to continuous substrate.
- C. Fastening Adhesive:
  - 1. Wood: ASTM C 557.
  - 2. Steel: Adhesive recommended for attaching panels to steel framing.
- D. Steel Drill Screws: ASTM C 1002, unless otherwise indicated.
  - 1. Use screws complying with ASTM C 954 for fastening panels to steel members from 0.033 to 0.112 inch (0.84 to 2.84 mm) thick.
- E. Sound Attenuation Blankets: ASTM C 665, Type I (blankets without membrane facing) produced by combining thermosetting resins with mineral fibers manufactured from glass, slag wool, or rock wool. See below for materials at plenum ceiling areas.
  - 1. Fire-Resistance-Rated Assemblies: Comply with mineral-fiber requirements of assembly.
  - 2. Plenum areas use faced insulation mineral fiber ASTM C665 Type III Class A Category.
- F. Thermal Insulation: As specified in Division 7 Section "Building Insulation."

#### 2.8 TEXTURE FINISHES

- A. Available Products: Subject to compliance with requirements, products that may be incorporated into the Work include, but are not limited to, the following:
  - 1. Texture Finish:
    - a. Goldbond Building Products, "Perfect Spray Em."
    - b. United States Gypsum Co.; SHEETROCK Wall and Ceiling Spray Texture.
    - c. Georgia Pacific, "GyProc Wall and Ceiling Texture."
  - 2. Primer:
    - a. Hamilton Materials, "Prep Coat Plus."
    - b. PPG SpeedHide® Max-Build High Build Drywall Surfacer 6-1.
    - c. Sherwin Williams Builders Solution A63W100 Latex Surfacer.

#### PART 3 - EXECUTION

# 3.1 EXAMINATION

A. Examine areas and substrates, with Installer present, and including welded hollow-metal frames, cast-in anchors, and structural framing, for compliance with requirements and other conditions affecting performance. Proceed with installation only after unsatisfactory conditions have been corrected.

# 3.2 APPLYING AND FINISHING PANELS, GENERAL

- A. Gypsum Board Application and Finishing Standards: ASTM C 840 and GA-216.
- B. Install sound attenuation blankets before installing gypsum panels, unless blankets are readily installed after panels have been installed on one side.
- C. Install gypsum panels with face side out. Butt panels together for a light contact at edges and ends with not more than 1/16 inch (1.5 mm) of open space between panels. Do not force into place.
- D. Locate edge and end joints over supports, except in ceiling applications where intermediate supports or gypsum board back-blocking is provided behind end joints. Do not place tapered edges against cut edges or ends. Stagger vertical joints on opposite sides of partitions. Do not make joints other than control joints at corners of framed openings.
- E. Attach gypsum panels to framing provided at openings and cutouts.
- F. Cover both faces of stud partition framing with gypsum panels in concealed spaces (above ceilings, etc.), except in chases braced internally.
  - 1. Unless concealed application is indicated or required for sound, fire, air, or smoke ratings, coverage may be accomplished with scraps of not less than 8 sq. ft. (0.7 sq. m) in area.
  - 2. Fit gypsum panels around ducts, pipes, and conduits.
  - 3. Where partitions intersect open concrete coffers, concrete joists, and other structural members projecting below underside of floor/roof slabs and decks, cut gypsum panels to fit profile formed by coffers, joists, and other structural members; allow 1/4- to 3/8-inch-(6.4- to 9.5-mm-) wide joints to install sealant.
- G. Space fasteners in gypsum panels according to referenced gypsum board application and finishing standard and manufacturer's written recommendations.
  - 1. 12" OC maximum, see drawings for more restrictive locations.
- H. Clean all stud cavities of debris, excess materials and dust prior to closing wall in.

#### 3.3 PANEL APPLICATION METHODS

- A. Single-Layer Application:
  - 1. On partitions/walls, apply gypsum panels horizontally (perpendicular to framing), unless otherwise indicated or required by fire-resistance-rated assembly, and minimize end joints.
    - Stagger abutting end joints not less than one framing member in alternate courses of board.
- B. Single-Layer Fastening Methods: Apply gypsum panels to supports with steel drill screws.
  - 1. #6 x 1" minimum.
- C. Laminating to Substrate: Where gypsum panels are indicated as directly adhered to a substrate (other than studs, joists, furring members, or base layer of gypsum board), comply with gypsum

board manufacturer's written recommendations and temporarily brace or fasten gypsum panels until fastening adhesive has set.

#### 3.4 INSTALLING TRIM ACCESSORIES

- A. General: For trim with back flanges intended for fasteners, attach to framing with same fasteners used for panels. Otherwise, attach trim according to manufacturer's written instructions.
- B. Control Joints: Install control joints at locations indicated on Drawings

#### 3.5 FINISHING GYPSUM BOARD ASSEMBLIES

- A. General: Treat gypsum board joints, interior angles, edge trim, control joints, penetrations, fastener heads, surface defects, and elsewhere as required to prepare gypsum board surfaces for decoration. Promptly remove residual joint compound from adjacent surfaces.
- B. Prefill open joints, rounded or beveled edges, and damaged surface areas.
- C. Apply joint tape over gypsum board joints, except those with trim having flanges not intended for tape.
- D. Gypsum Board Finish Levels: Finish panels to levels indicated below, according to ASTM C 840, for locations indicated:
  - 1. Level 1: Embed tape at joints in ceiling plenum areas, concealed areas, and where indicated, unless a higher level of finish is required for fire-resistance-rated assemblies and sound-rated assemblies.
  - 2. Level 2: Embed tape and apply separate first coat of joint compound to tape, fasteners, and trim flanges where [panels are substrate for acoustical tile].
  - 3. Level 3: Embed tape and apply separate first and fill coats of joint compound to tape, fasteners, and trim flanges where indicated.
  - 4. Level 4: Embed tape and apply separate first, fill, and finish coats of joint compound to tape, fasteners, and trim flanges at panel surfaces that will be exposed to view, unless otherwise indicated.
  - 5. Level 5: Embed tape and apply separate first, fill, and finish coats of joint compound to tape, fasteners, and trim flanges, and apply skim coat of joint compound over entire surface where indicated.
- E. Glass-Mat Gypsum Sheathing Board: Finish according to manufacturer's written instructions for use as exposed soffit board.
- F. Glass-Mat, Water-Resistant Backing Panels: Finish according to manufacturer's written instructions.

#### 3.6 APPLYING TEXTURE FINISHES

A. Surface Preparation and Primer: Prepare and apply primer to gypsum panels and other surfaces receiving texture finishes prior to applying texture finish. Apply primer to surfaces that are clean, dry, and smooth. The primer shall be applied to a dry film thickness of 1.7 to 1.8 mils.

- B. Texture Finish Application: Mix and apply finish using powered spray equipment, to produce a uniform texture matching approved mockup and free of starved spots or other evidence of thin application or of application patterns.
  - Texture: Medium stipple at walls/ceilings painted eggshell.
- C. Prevent texture finishes from coming into contact with surfaces not indicated to receive texture finish by covering them with masking agents, polyethylene film, or other means. If, despite these precautions, texture finishes contact these surfaces, immediately remove droppings and overspray to prevent damage according to texture finish manufacturer's written recommendations. Light stipple at restrooms and selected rooms painted semi-gloss.

## 3.7 FIELD QUALITY CONTROL.

- A. Above-Ceiling Observation: The Architect will conduct an above-ceiling observation before installing gypsum board ceilings and report deficiencies in the Work observed. Do not proceed with installation of gypsum board to ceiling support framing until deficiencies have been corrected.
  - 1. Notify the Architect seven days in advance of date and time when Project, or part of Project, will be ready for above-ceiling observation.

END OF SECTION 092900

# SECTION 095113 - ACOUSTICAL PANEL CEILINGS

02/22/16

## PART 1 - GENERAL

#### 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Divisions 0 & 1 Specification Sections, apply to this Section.

## 1.2 SUMMARY

- A. This Section includes, but is not limited to, ceilings consisting of acoustical panels and exposed suspension systems.
  - 1. Lay-in acoustical panels with suspension systems.

#### 1.3 SUBMITTALS

- A. Product Data: For each type of product specified.
- B. Detail major elements, components and systems of Acoustical Panel Ceilings and materials in relationship with other systems, installations and building components. Show space requirements for installation and access. Show where sequence and coordination of installations are important to the efficient flow of the Work. Include the following:
- C. Qualification Data: For firms and persons specified in "Quality Assurance" Article to demonstrate their capabilities and experience. Include lists of completed projects with project names and addresses, names and addresses of architects and owners, and other information specified.
- D. Product Test Reports: Indicate compliance of acoustical panel ceilings and components with requirements based on comprehensive testing of current products.
- E. Project Closeout Requirement:
  - Extra materials.

# 1.4 QUALITY ASSURANCE

- A. Installer Qualifications: Engage an experienced installer who has completed acoustical panel ceilings similar in material, design, and extent to that indicated for this Project and with a record of successful in-service performance.
- B. Source Limitations for Ceiling Units: Obtain each acoustical ceiling panel from one source with resources to provide products of consistent quality in appearance and physical properties without delaying the Work.
- C. Source Limitations for Suspension System: Obtain each suspension system from one source with resources to provide products of consistent quality in appearance and physical properties without delaying the Work.
- D. Fire-Test-Response Characteristics: Provide acoustical panel ceilings that comply with the following requirements:

- 1. Fire-response tests were performed by UL, ITS/Warnock Hersey, or another independent testing and inspecting agency that is acceptable to authorities having jurisdiction and that performs testing and follow-up services.
- 2. Surface-burning characteristics of acoustical panels comply with ASTM E 1264 for Class Imaterials as determined by testing identical products per ASTM E 84.
- Fire-resistance-rated assemblies, which are indicated by design designations from UL's "Fire
  Resistance Directory," from ITS/Warnock Hersey's "Directory of Listed Products," or from
  the listings of another testing and inspecting agency, are identical in materials and
  construction to those tested per ASTM E 119.
- 4. Products are identified with appropriate markings of applicable testing and inspecting agency.
- E. Testing of embeds and wires will be field tested by Owner's laboratory to values accepted by DSA and quantity as required to substantiate conformance with specifications.

## 1.5 DELIVERY, STORAGE, AND HANDLING

- A. Deliver acoustical panels and suspension system components to Project site in original, unopened packages and store them in a fully enclosed space where they will be protected against damage from moisture, direct sunlight, surface contamination, and other causes.
- B. Before installing acoustical panels, permit them to reach room temperature and a stabilized moisture content.
- C. Handle acoustical panels carefully to avoid chipping edges or damaging units in any way.

# 1.6 PROJECT CONDITIONS

- A. Environmental Limitations: Do not install acoustical panel ceilings until spaces are enclosed and weatherproof, wet-work in spaces is complete and dry, work above ceilings is complete, and ambient temperature and humidity conditions are maintained at the levels indicated for Project when occupied for its intended use.
  - Do not install suspension system until vinyl tack surface has been installed.

#### 1.7 COORDINATION

A. Coordinate layout and installation of acoustical panels and suspension system with other construction that penetrates ceilings or is supported by them, including light fixtures, HVAC equipment, fire-suppression system, and partition assemblies.

# 1.8 EXTRA MATERIALS

- A. Furnish extra materials described below that match products installed, are packaged with protective covering for storage, and are identified with labels describing contents.
  - 1. Acoustical Ceiling Units: Full-size units equal to 2.0 percent of amount installed, or minimum of three (3) unopened cases of each type and color used.

#### PART 2 - PRODUCTS

#### 2.1 MANUFACTURERS

A. Available Products: Subject to compliance with requirements, products that may be incorporated into the Work include, but are not limited to, those indicated for each designation in the Acoustical Panel Ceiling Data Sheets at the end of Part 3.

# 2.2 ACOUSTICAL PANELS, GENERAL

- A. Acoustical Panel Standard: Provide manufacturer's standard panels of configuration indicated that comply with ASTM E 1264 classifications as designated by types, patterns, acoustical ratings, and light reflectances, unless otherwise indicated.
- B. Acoustical Panel Colors and Patterns: Match appearance characteristics indicated for each product type.
  - Where appearance characteristics of acoustical panels are indicated by referencing ASTM E 1264 pattern designations and not manufacturers' proprietary product designations, provide products selected by Architect from each manufacturer's full range of products that comply with requirements indicated for type, pattern, color, light reflectance, acoustical performance, edge detail, and size.
- C. Panel Characteristics: Comply with requirements indicated in the Acoustical Panel Data Sheets at the end of Part 3, including those referencing ASTM E 1264 classifications.

#### PART 3 - EXECUTION

## 3.1 EXAMINATION

- A. Examine substrates and structural framing to which acoustical panel ceilings attach or abut, with Installer present, for compliance with requirements specified in this and other Sections that affect ceiling installation and anchorage, and other conditions affecting performance of acoustical panel ceilings.
  - 1. Proceed with installation only after unsatisfactory conditions have been corrected.

## 3.2 PREPARATION

- Coordination: Furnish layouts for cast-in-place anchors, clips, and other ceiling anchors whose installation is specified in other Sections.
- B. Measure each ceiling area and establish layout of acoustical panels to balance border widths at opposite edges of each ceiling. Avoid using less-than-half-width panels at borders, and comply with layout shown on reflected ceiling plans.

# 3.3 INSTALLATION

- A. General: Install acoustical panel ceilings to comply with manufacturer's written instructions, CISCA's "Ceiling Systems Handbook", ASTM E580-08, CBC Section 1615A.1.16 and Interpretation of Regulation IR25-2.10.
  - Standard for Ceiling Suspension System Installations: Comply with ASTM C 636.
    - a. CBC Sections 1632A, Table 16 A-0 and IR-25-5.
    - b. DSA IR 25-5.
- B. Install acoustical panels with undamaged edges and fitted accurately into suspension system runners and edge moldings. Scribe and cut panels at borders and penetrations to provide a neat, precise fit.
  - 1. Arrange directionally patterned acoustical panels as follows:
    - As indicated on reflected ceiling plans.
  - 2. For square-edged panels, install panels with edges fully hidden from view by flanges of suspension system runners and moldings.
  - 3. For reveal-edged panels on suspension system runners, install panels with bottom of reveal in firm contact with top surface of runner flanges.
  - 4. For reveal-edged panels on suspension system members with box-shaped flanges, install panels with reveal surfaces in firm contact with suspension system surfaces and panel faces flush with bottom face of runners.
  - 5. Install hold-down clips in areas indicated, in areas required by authorities having jurisdiction, and for fire-resistance ratings; space as recommended by panel manufacturer's written instructions, unless otherwise indicated or required.
  - 6. Protect lighting fixtures and air ducts to comply with requirements indicated for fire-resistance-rated assembly.

#### 3.4 CLEANING

- A. Clean exposed surfaces of acoustical panel ceilings, including trim, edge moldings, and suspension system members. Comply with manufacturer's written instructions for cleaning and touchup of minor finish damage. Remove and replace ceiling components that cannot be successfully cleaned and repaired to permanently eliminate evidence of damage.
- 3.5 ACOUSTICAL PANEL CEILING PANEL (Data Sheets) See Data Sheets attached herein.

## **ACOUSTICAL PANEL CEILING DATA SHEET**

Water-Felted, Mineral-Base Acoustical Panels for Acoustical Panel Ceiling "Type B":

Where this designation is indicated, provide acoustical panels complying with the following:

Available Manufacturers: Subject to compliance with requirements, manufacturers offering products that may be incorporated in the Work include, but are not limited to, the following:

Cortega 747 Armstrong World Industries, Inc.

Classification: Panels fitting ASTM E 1264 for Type III, mineral base with painted finish; Form 2, water felted.

Pattern: Panels fitting ASTM E 1264 pattern designation (CD) as specified by product designation.

Color: White, unless noted otherwise on Drawings.

Light Reflectance Coefficient: Not less than LR 0.83.

Noise Reduction Coefficient: NRC 0.50.

Ceiling Attenuation Class: Not less than CAC 35.

Edge Detail: Square Lay-in

Thickness: 15/16"

Size: 24" x 48"

Flamespread: Class I (25 or less).

Smoke Density: 50 or under.

END OF SECTION 095113

# SECTION 096513 - RESILIENT WALL BASE AND ACCESSORIES

05/24/16

## PART 1 - GENERAL

#### 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Divisions 0 & 1 Specification Sections, apply to this Section.

## 1.2 SUMMARY

- A. This Section includes, but is not limited to, the following:
  - 1. Resilient wall base.
- B. Related Sections include the following:
  - 1. Division 9 Section "Resilient Tile Flooring."

## 1.3 SUBMITTALS

- A. Product Data: For each type of product specified.
- B. Samples for Initial Selection: Manufacturer's standard sample sets consisting of sections of units showing the full range of colors and patterns available for each type of product indicated.
- C. Samples for Verification: In manufacturer's standard sizes, but not less than 12 inches (300 mm) long, of each product color and pattern specified.
- D. Product Certificates: Signed by manufacturers of resilient wall base and accessories certifying that each product furnished complies with requirements.
- E. Project Closeout Requirement:
  - 1. Cleaning and maintenance data.

## 1.4 QUALITY ASSURANCE

A. Source Limitations: Obtain each type and color of product specified from one source with resources to provide products of consistent quality in appearance and physical properties without delaying the Work.

## 1.5 DELIVERY, STORAGE, AND HANDLING

- A. Deliver products to Project site in manufacturer's original, unopened cartons and containers, each bearing names of product and manufacturer, Project identification, and shipping and handling instructions.
- B. Store products in dry spaces protected from the weather, with ambient temperatures maintained

- between 50 and 90 deg F (10 and 32 deg C).
- C. Move products into spaces where they will be installed at least 48 hours before installation, unless longer conditioning period is recommended in writing by manufacturer.

#### 1.6 PROJECT CONDITIONS

- A. Maintain a temperature of not less than 70 deg F (21 deg C) or more than 95 deg F (35 deg C) in spaces to receive resilient products for at least 48 hours before installation, during installation, and for at least 48 hours after installation, unless manufacturer's written recommendations specify longer time periods. After postinstallation period, maintain a temperature of not less than 55 deg F (13 deg C) or more than 95 deg F (35 deg C).
- B. Do not install products until they are at the same temperature as the space where they are to be installed.
- C. For resilient products installed on traffic surfaces, close spaces to traffic during installation and for time period after installation recommended in writing by manufacturer.
- D. Coordinate resilient product installation with other construction to minimize possibility of damage and soiling during remainder of construction period. Install resilient products after other finishing operations, including painting, have been completed.

## 1.7 EXTRA MATERIALS

- A. Furnish extra materials described below that match products installed, are packaged with protective covering for storage, and are identified with labels describing contents.
  - Furnish not less than 10 linear feet (3 linear m) for each 500 linear feet (150 linear m) or fraction thereof, of each different type, color, pattern, and size of resilient wall base product installed.
  - Deliver extra materials to Owner.
  - 3. Furnish one (1) stair tread for each 25 treads, or fraction thereof.

## PART 2 - PRODUCTS

#### 2.1 MANUFACTURERS

A. Available Products: Subject to compliance with requirements, products that may be incorporated into the Work include, but are not limited to, those indicated in the Rubber Wall Base Product Data Sheets at the end of this Section.

#### 2.2 RESILIENT WALL BASE

- A. Rubber Wall Base: Products complying with ASTM F-1861, Type TS or TP and with requirements specified in the Rubber Accessory Product Data Sheets.
  - 1. Minimum critical radiant flux and specific optical density smoke rating for CBC 804.4.1.

## 2.3 RESILIENT ACCESSORIES

- A. Rubber Accessories: Products complying with requirements specified in the Rubber Wall Base Product Data Sheets and Accessory Schedule.
  - 1. Minimum critical radiant flux and specific optical density smoke rating for CBC 804.4.1.

#### 2.4 INSTALLATION ACCESSORIES

- A. Trowelable Leveling and Patching Compounds: Latex-modified, portland-cement-based formulation provided or approved by resilient product manufacturer for applications indicated.
- B. Stair-Tread-Nose Filler: Two-part epoxy compound recommended by resilient tread manufacturer to fill nosing substrates that do not conform to tread contours.
- C. Adhesives: Water-resistant type recommended by manufacturer to suit resilient products and substrate conditions indicated.

#### PART 3 - EXECUTION

## 3.1 EXAMINATION

A. Examine substrates, areas, and conditions where installation of resilient products will occur, with Installer present, for compliance with manufacturer's requirements, including those for maximum moisture content. Verify that substrates and conditions are satisfactory for resilient product installation and comply with requirements specified. Do not proceed with installation until unsatisfactory conditions have been corrected.

## 3.2 PREPARATION

- A. General: Comply with manufacturer's written installation instructions for preparing substrates indicated to receive resilient products.
- B. Use trowelable leveling and patching compounds, according to manufacturer's written instructions, to fill cracks, holes, and depressions in substrates.
- C. Use stair-tread-nose filler, according to resilient tread manufacturer's written instructions, to fill nosing substrates that do not conform to tread contours.
- D. Remove coatings, including curing compounds, and other substances that are incompatible with adhesives and that contain soap, wax, oil, or silicone, using mechanical methods recommended by manufacturer. Do not use solvents.
- E. Broom and vacuum clean substrates to be covered immediately before installing resilient products. After cleaning, examine substrates for moisture, alkaline salts, carbonation, or dust. Do not proceed with installation until unsatisfactory conditions have been corrected.

## 3.3 INSTALLATION

- A. General: Install resilient products according to manufacturer's written installation instructions.
- B. Apply resilient wall base to walls, columns, pilasters, casework and cabinets in toe spaces, and other permanent fixtures in rooms and areas where base is required.
  - Install wall base in full lengths as long as practicable without gaps at seams and with tops of adjacent pieces aligned. Pre-mold corners to be installed at inside/outside corners. Use of multiple small pieces will not be accepted.
  - 2. Tightly adhere wall base to substrate throughout length of each piece, with base in continuous contact with horizontal and vertical substrates.
  - 3. Do not stretch base during installation.
  - 4. On masonry surfaces or other similar irregular substrates, fill voids along top edge of resilient wall base with manufacturer's recommended adhesive filler material.
- C. Place resilient products so they are butted to adjacent materials and bond to substrates with adhesive. Install reducer strips at edges of flooring that would otherwise be exposed. Form resilient accessories to curves as indicated on plans.
- D. Apply resilient products to stairs as indicated and according to manufacturer's written installation instructions.

#### 3.4 CLEANING AND PROTECTING

- A. Perform the following operations immediately after installing resilient products:
  - 1. Remove adhesive and other surface blemishes using cleaner recommended by resilient product manufacturers.
  - Sweep or vacuum horizontal surfaces thoroughly.
  - 3. Do not wash resilient products until after time period recommended by resilient product manufacturer.
  - 4. Damp-mop or sponge resilient products to remove marks and soil.
- B. Protect resilient products against mars, marks, indentations, and other damage from construction operations and placement of equipment and fixtures during the remainder of construction period. Use protection methods indicated or recommended in writing by resilient product manufacturer.
  - 1. Apply protective floor polish to vinyl resilient products installed on floors and stairs that are free from soil, visible adhesive, and surface blemishes, if recommended by manufacturer.
    - a. Coordinate selection of floor polish with Owner's maintenance service.
  - 2. Cover resilient products installed on floors and stairs with undyed, untreated building paper until inspection for Notice of Completion.
- C. Clean resilient products not more than 4 days before dates scheduled for inspections intended to establish date of Notice of Completion in each area of Project. Clean products according to manufacturer's written recommendations.
  - Before cleaning, strip protective floor polish that was applied to vinyl products on floors and stairs after completing installation only if required to restore polish finish and if recommended by resilient product manufacturer.
  - After cleaning, reapply polish on vinyl products on floors and stairs to restore protective floor finish according to resilient product manufacturer's written recommendations. Coordinate with Owner's maintenance program.

# RUBBER WALL BASE PRODUCT DATA SHEET

Rubber Wall Base Designation: RWB

Style: Cove with top-set toe.

Minimum Nominal Thickness: 1/8 inch.

Height: 4 inches.

Lengths: Cut lengths 4 feet long.

Exterior Corners: Manufactured Pre-molded Corners.

Interior Corners: Manufactured Pre-molded or formed on job.

<u>Color and Pattern</u>: As selected by Architect from manufacturer's full range of colors and patterns produced for rubber wall base complying with requirements indicated.

Available Products: Burke

Armstrong Roppe Johnsonite

# RUBBER ACCESSORY PRODUCT DATA SHEET

Rubber Accessory Designation: RA

<u>Product Description</u>: See Reducer Strip/Floor Accessory Schedule.

Profile and Dimensions: As indicated.

<u>Color</u>: As selected by Architect from manufacturer's full range of colors produced for rubber accessories complying with requirements indicated.

Available Products: Mercer, Musson, Schluter, - see Schedule on Drawings.

END OF SECTION 096513

#### PART 1 - GENERAL

## 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Divisions 0 & 1 Specification Sections, apply to this Section.

#### 1.2 SUMMARY

- A. This Section includes, but is not limited to, the following:
  - 1. Vinyl composition floor tile.
- B. Related Sections include the following:
  - Division 9 Section "Resilient Wall Base and Accessories" for resilient wall base, reducer strips, and other accessories installed with resilient floor tiles.

#### 1.3 SUBMITTALS

- A. Product Data: For each type of product specified.
- B. Samples for Initial Selection: Manufacturer's color charts consisting of units or sections of units showing the full range of colors and patterns available for each type of product indicated.
- C. Samples for Verification: Full-size tiles of each different color and pattern of resilient floor tile specified, showing the full range of variations expected in these characteristics.
- D. Product Certificates: Signed by manufacturers of resilient products certifying that each product furnished complies with requirements.
- E. Maintenance Data: For resilient floor tile to include in the maintenance manuals specified in Division 1.
- F. Project Closeout Requirement:
  - 1. Extra material.
  - 2. Cleaning and maintenance data.

### 1.4 QUALITY ASSURANCE

- A. Installer Qualifications: Engage an experienced installer to perform work of this Section who has specialized in installing resilient products similar to those required for this Project and with a record of successful in-service performance.
- B. Source Limitations: Obtain each type, color, and pattern of product specified from one source with resources to provide products of consistent quality in appearance and physical properties without delaying the Work.
- C. Fire-Test-Response Characteristics: Provide products with the following fire-test-response characteristics as determined by testing identical products per test method indicated below by a

testing and inspecting agency acceptable to authorities having jurisdiction.

- 1. Critical Radiant Flux: 0.45 W/sq. cm or greater when tested per ASTM E 648.
- 2. Smoke Density: Maximum specific optical density of 450 or less when tested per ASTM E 662.

## 1.5 DELIVERY, STORAGE, AND HANDLING

- A. Deliver products to Project site in manufacturer's original, unopened cartons and containers, each bearing names of product and manufacturer, Project identification, and shipping and handling instructions.
- B. Store products in dry spaces protected from the weather, with ambient temperatures maintained between 50 and 90 deg F (10 and 32 deg C).
- C. Store tiles on flat surfaces.
- D. Move products into spaces where they will be installed at least 48 hours before installation, unless longer conditioning period is recommended in writing by manufacturer.

#### 1.6 PROJECT CONDITIONS

- A. Maintain a temperature of not less than 70 deg F (21 deg C) or more than 95 deg F (35 deg C) in spaces to receive products for at least 48 hours before installation, during installation, and for at least 48 hours after installation, unless manufacturer's written recommendations specify longer time periods. After postinstallation period, maintain a temperature of not less than 55 deg F (13 deg C) or more than 95 deg F (35 deg C).
- B. Do not install products until they are at the same temperature as the space where they are to be installed.
- C. Close spaces to traffic during flooring installation and for time period after installation recommended in writing by manufacturer.
- D. Install tiles and accessories after other finishing operations, including painting, have been completed.
- E. Do not install flooring over concrete slabs until slabs have cured and are sufficiently dry to bond with adhesive, as determined by flooring manufacturer's recommended bond and moisture test.

## 1.7 EXTRA MATERIALS

- A. Furnish extra materials described below that match products installed, are packaged with protective covering for storage, and are identified with labels describing contents.
  - 1. Furnish not less than one box for each 50 boxes or fraction thereof, of each type, color, pattern, class, wearing surface, and size of resilient tile flooring installed.
  - 2. Deliver extra materials to Owner.

#### PART 2 - PRODUCTS

#### 2.1 MANUFACTURERS

A. Available Products: Subject to compliance with requirements, products that may be incorporated into the Work include, but are not limited to, those indicated in the Resilient Tile Flooring Data Sheet at the end of this Section.

#### 2.2 RESILIENT TILE

- A. Rubber Floor Tile: Products complying with ASTM F 1344 and with requirements specified in the Resilient Tile Flooring Schedule.
- B. Vinyl Composition Floor Tile: Products complying with ASTM F 1066 and with requirements specified in the Resilient Tile Flooring Schedule.
  - Coefficient of friction of .6 per ASTM 2047.

## 2.3 INSTALLATION ACCESSORIES

- Trowelable Leveling and Patching Compounds: Latex-modified, portland-cement-based formulation provided or approved by flooring manufacturer for applications indicated.
- B. Adhesives: Water-resistant type recommended by manufacturer to suit resilient products and substrate conditions indicated.

## PART 3 - EXECUTION

## 3.1 EXAMINATION

- A. Examine substrates, areas, and conditions where installation of resilient products will occur, with Installer present, for compliance with manufacturer's requirements. Verify that substrates and conditions are satisfactory for resilient product installation and comply with requirements specified.
- B. For wood subfloors, verify the following:
  - 1. Underlayment over subfloor complies with requirements specified in Division 6 Section "Rough Carpentry."
  - 2. Underlayment surface is free of irregularities and substances that may interfere with adhesive bond, show through surface, or stain flooring.
- C. Do not proceed with installation until unsatisfactory conditions have been corrected.

## 3.2 PREPARATION

- A. General: Comply with resilient product manufacturer's written installation instructions for preparing substrates indicated to receive resilient products.
- B. Use trowelable leveling and patching compounds, according to manufacturer's written instructions, to fill cracks, holes, and depressions in substrates.
- C. Remove coatings, including curing compounds, and other substances that are incompatible with

flooring adhesives and that contain soap, wax, oil, or silicone, using mechanical methods recommended by manufacturer. Do not use solvents.

D. Broom and vacuum clean substrates to be covered immediately before product installation. After cleaning, examine substrates for moisture, alkaline salts, carbonation, or dust. Do not proceed with installation until unsatisfactory conditions have been corrected.

## 3.3 TILE INSTALLATION

- A. General: Comply with tile manufacturer's written installation instructions.
- B. Lay out tiles from center marks established with principal walls, discounting minor offsets, so tiles at opposite edges of room are of equal width. Adjust as necessary to avoid using cut widths that equal less than one-half of a tile at perimeter.
  - 1. Lay tiles square with room axis, unless otherwise indicated.
- C. Match tiles for color and pattern by selecting tiles from cartons in the same sequence as manufactured and packaged, if so numbered. Cut tiles neatly around all fixtures. Discard broken, cracked, chipped, or deformed tiles.
  - 1. Lay tiles, match checkerboard pattern.
- D. Scribe, cut, and fit tiles to butt neatly and tightly to vertical surfaces and permanent fixtures, including built-in furniture, cabinets, pipes, outlets, edgings, door frames, thresholds, and nosings.
- E. Extend tiles into toe spaces, door reveals, closets, and similar openings.
- F. Maintain reference markers, holes, and openings that are in place or marked for future cutting by repeating on finish flooring as marked on subfloor. Use chalk or other nonpermanent, nonstaining marking device.
- G. Install tiles on covers for data, telephone and electrical ducts, and similar items in finished floor areas. Maintain overall continuity of color and pattern with pieces of flooring installed on covers. Tightly adhere edges to perimeter of floor around covers and to covers.
- H. Adhere tiles to flooring substrates using a full spread of adhesive applied to substrate to comply with tile manufacturer's written instructions, including those for trowel notching, adhesive mixing, and adhesive open and working times.
  - 1. Provide completed installation without open cracks, voids, raising and puckering at joints, telegraphing of adhesive spreader marks, and other surface imperfections.
- I. Hand roll tiles according to tile manufacturer's written instructions.

#### 3.4 CLEANING AND PROTECTING

- A. Perform the following operations immediately after installing resilient products:
  - 1. Remove adhesive and other surface blemishes using cleaner recommended by resilient product manufacturers.
  - 2. Sweep or vacuum floor thoroughly.
  - 3. Do not wash floor until after time period recommended by flooring manufacturer.
  - 4. Damp-mop floor to remove marks and soil.

- B. Protect flooring against mars, marks, indentations, and other damage from construction operations and placement of equipment and fixtures during the remainder of construction period. Use protection methods indicated or recommended in writing by flooring manufacturer.
  - 1. Apply protective floor polish to floor surfaces that are free from soil, visible adhesive, and surface blemishes, if recommended in writing by manufacturer.
    - Coordinate selection of floor polish with Owner's maintenance service.
  - 2. Cover products installed on floor surfaces with undyed, untreated building paper until inspection for Notice of Completion.
  - 3. Do not move heavy and sharp objects directly over floor surfaces. Place plywood or hardboard panels over flooring and under objects while they are being moved. Slide or roll objects over panels without moving panels.
- C. Clean floor surfaces not more than 4 days before dates scheduled for inspections intended to establish date of Notice of Completion in each area of Project. Clean products according to manufacturer's written recommendations.
  - 1. Before cleaning, strip protective floor polish that was applied after completing installation only if required to restore polish finish and if recommended by flooring manufacturer.
  - After cleaning, reapply polish to floor surfaces to restore protective floor finish according to flooring manufacturer's written recommendations. Coordinate with Owner's maintenance program.

# VINYL COMPOSITION FLOOR TILE PRODUCT DATA SHEET

Vinyl Composition Floor Tile Designation: VCT-1

Class: Class 2 through pattern tile, Type IV, Comp. I.

Wearing Surface: Smooth with a minimum coefficient of friction of at least 0.6 per ASTM D-2047.

Gage: 1/8 inch.

Size: 12 X 12.

<u>Color and Pattern</u>: As selected by Architect from manufacturer's full range of colors and patterns produced for tile of class, wearing surface, thickness, size, and pattern specified. Allow for four accent colors per room -60%/20% 20% - pattern as directed by the Architect.

Available Products: Armstrong - Imperial Texture - Standard Excelon.

END OF SECTION 096519

## PART 1 - GENERAL

### 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 0 & 1 Specification Sections, apply to this Section.

#### 1.2 SUMMARY

- A. This Section includes, but is not limited to, surface preparation and field painting of the following:
  - 1. Exposed interior items and surfaces.
- B. Paint exposed surfaces, except where the paint schedules indicate that a surface or material is not to be painted or is to remain natural. If the paint schedules do not specifically mention an item or a surface, paint the item or surface the same as similar adjacent materials or surfaces whether or not schedules indicate colors. If the schedules do not indicate color or finish, the Architect will select from standard colors and finishes available.
- C. Do not paint prefinished items, concealed surfaces, finished metal surfaces, operating parts, and labels.
  - 1. Prefinished items include the following factory-finished components:
    - Acoustical wall panels.
    - b. Finished mechanical and electrical equipment.
    - c. Light fixtures.
    - d. Acoustical ceiling panels.
    - e. Wall covering/VTS board.
    - f. Wood doors prefinished from door supplier
    - g. Casework prefinished by cabinet supplier.
  - 2. Concealed surfaces include walls or ceilings in the following generally inaccessible spaces:
    - a. Foundation spaces.
    - b. Furred areas.
    - c. Ceiling plenums.
    - d. Pipe spaces.
    - e. Duct shafts.
  - 3. Finished metal surfaces include the following:
    - a. Anodized aluminum.
    - b. Stainless steel.
    - c. Chromium plate.
    - d. Copper.
    - e. Bronze and brass.
  - 4. Operating parts include moving parts of operating equipment and the following:
    - a. Valve and damper operators.

- b. Linkages.
- c. Sensing devices.
- Motor and fan shafts.
- 5. Labels: Do not paint over Underwriters Laboratories (UL), Factory Mutual (FM), or other code-required labels or equipment name, identification, performance rating, or nomenclature plates.

## 1.3 DEFINITIONS

- A. General: Standard coating terms defined in ASTM D 16 apply to this Section.
  - 1. Flat refers to a lusterless or matte finish with a gloss range below 15 when measured at an 85-degree meter.
  - 2. Eggshell refers to low-sheen finish with a gloss range between 5 and 20 when measured at a 60-degree meter.
  - 3. Satin refers to low-sheen finish with a gloss range between 15 and 35 when measured at a 60-degree meter.
  - 4. Semigloss refers to medium-sheen finish with a gloss range between 30 and 65 when measured at a 60-degree meter.
  - 5. Full gloss refers to high-sheen finish with a gloss range more than 65 when measured at a 60-degree meter.

#### 1.4 SUBMITTALS

- A. Product Data: For each paint system specified. Include block fillers and primers.
  - 1. Material List: Provide an inclusive list of required coating materials. Indicate each material and cross-reference specific coating, finish system, and application. Identify each material by manufacturer's catalog number and general classification.
  - 2. Certification by the manufacturer that products supplied comply with federal, state and local regulations controlling use of volatile organic compounds (VOCs).
- B. Samples for Initial Selection: Manufacturer's color charts showing the full range of colors available for each type of finish-coat material indicated.
- C. Samples for Verification: Of each color and material to be applied, with texture to simulate actual conditions, on representative Samples of the actual substrate.
- D. Project Closeout Requirement:
  - 1. Extra materials.

## 1.5 QUALITY ASSURANCE

A. Source Limitations: Obtain block fillers, primers, and undercoat materials for each coating system from the same manufacturer as the finish coats.

## 1.6 DELIVERY, STORAGE, AND HANDLING

A. Deliver materials to the Project Site in manufacturer's original, unopened packages and containers

bearing manufacturer's name and label, and the following information:

- Product name or title of material.
- 2. Product description (generic classification or binder type).
- 3. Manufacturer's stock number and date of manufacture.
- 4. Contents by volume, for pigment and vehicle constituents.
- 5. Thinning instructions.
- 6. Application instructions.
- 7. Color name and number.
- 8. VOC content.
- B. Store materials not in use in tightly covered containers in a well-ventilated area at a minimum ambient temperature of 45 deg F (7 deg C). Maintain containers used in storage in a clean condition, free of foreign materials and residue.
  - 1. Protect from freezing. Keep storage area neat and orderly. Remove oily rags and waste daily. Take necessary measures to ensure that workers and work areas are protected from fire and health hazards resulting from handling, mixing, and application.

#### 1.7 PROJECT CONDITIONS

- A. Apply water-based paints only when the temperature of surfaces to be painted and surrounding air temperatures are between 50 and 90 deg F (10 and 32 deg C).
- B. Apply solvent-thinned paints only when the temperature of surfaces to be painted and surrounding air temperatures are between 45 and 95 deg F (7.2 and 35 deg C).
- C. Do not apply paint in snow, rain, fog, or mist; or when the relative humidity exceeds 85 percent; or at temperatures less than 5 deg F (3 deg C) above the dew point; or to damp or wet surfaces.
  - Painting may continue during inclement weather if surfaces and areas to be painted are enclosed and heated within temperature limits specified by manufacturer during application and drying periods.

### 1.8 EXTRA MATERIALS

- A. Furnish extra paint materials from the same production run as the materials applied in the quantities described below. Package paint materials in unopened, factory-sealed containers for storage and identify with labels describing contents. Deliver extra materials to the Owner.
  - 1. Quantity: Furnish the Owner with extra paint materials in the quantities indicated below:
    - a. Wall Paint: Two (2) gallons unopened of each color applied.
    - b. Door/Trim: Two (2) gallons unopened of each color applied.
    - c. Stain: Two (2) gallons unopened of each color applied.

## PART 2 - PRODUCTS

### 2.1 MANUFACTURERS

A. Available Products: Where Specifications name a product, or refer to a product indicated on Drawings, and include a list of manufacturer(s), provide the specified or indicated product or a

comparable product.

- B. Manufacturers Names: The following manufacturers are referred to in the paint schedules by use of shortened versions of their names, which are shown in parentheses:
  - 1. Sherwin-Williams Co. (S-W).
  - 2. Trinity Coating Co.
  - 3. Gemini Coating Co.
  - 4. Evonik

## 2.2 PAINT MATERIALS, GENERAL

- A. Material Compatibility: Provide block fillers, primers, undercoats, and finish-coat materials that are compatible with one another and the substrates indicated under conditions of service and application, as demonstrated by manufacturer based on testing and field experience.
- B. Material Quality: Provide manufacturer's best-quality paint material of the various coating types specified. Paint-material containers not displaying manufacturer's product identification will not be acceptable.
  - 1. Proprietary Names: Use of manufacturer's proprietary product names to designate colors or materials is not intended to imply that products named are required to be used to the exclusion of equivalent products of other manufacturers. Furnish manufacturer's material data and certificates of performance for proposed substitutions.
- C. Colors: Provide color selections made by the Architect.
  - 1. Allow for colors as follows:
    - a. Each room shall have one (1) field color and one (1) accent wall color. (This does not include doors, frames and miscellaneous items.
    - b. Rooms over 250 square feet, allow for one (1) field color and two (2) accent wall colors
    - a. Refer to plans for more restrictive requirement where more colors per room are required.
    - b. Rooms over 500 square feet, allow for one (1) field color and three (3) accent wall colors.

### PART 3 - EXECUTION

#### 3.1 EXAMINATION

- A. Examine substrates, areas, and conditions, with the Applicator present, under which painting will be performed for compliance with paint application requirements.
  - 1. Do not begin to apply paint until unsatisfactory conditions have been corrected and surfaces receiving paint are thoroughly dry.
  - 2. Start of painting will be construed as the Applicator's acceptance of surfaces and conditions within a particular area.
- B. Coordination of Work: Review other Sections in which primers are provided to ensure compatibility of the total system for various substrates. On request, furnish information on characteristics of finish materials to ensure use of compatible primers.

1. Notify the Architect about anticipated problems using the materials specified over substrates primed by others.

## 3.2 PREPARATION

- A. General: Remove hardware and hardware accessories, plates, machined surfaces, lighting fixtures, and similar items already installed that are not to be painted. If removal is impractical or impossible because of the size or weight of the item, provide surface-applied protection before surface preparation and painting.
  - 1. After completing painting operations in each space or area, reinstall items removed using workers skilled in the trades involved.
- B. Cleaning: Before applying paint or other surface treatments, clean the substrates of substances that could impair the bond of the various coatings. Remove oil and grease before cleaning.
  - 1. Schedule cleaning and painting so dust and other contaminants from the cleaning process will not fall on wet, newly painted surfaces.
- C. Surface Preparation: Clean and prepare surfaces to be painted according to manufacturer's written instructions for each particular substrate condition and as specified.
  - 1. Provide barrier coats over incompatible primers or remove and reprime.
  - Cementitious Materials: Prepare concrete, concrete masonry block, cement plaster, and mineral-fiber-reinforced cement panel surfaces to be painted. Remove efflorescence, chalk, dust, dirt, grease, oils, and release agents. Roughen as required to remove glaze. If hardeners or sealers have been used to improve curing, use mechanical methods of surface preparation.
    - a. Use abrasive blast-cleaning methods if recommended by paint manufacturer.
    - b. Determine alkalinity and moisture content of surfaces by performing appropriate tests. If surfaces are sufficiently alkaline to cause the finish paint to blister and burn, correct this condition before application. Do not paint surfaces where moisture content exceeds that permitted in manufacturer's written instructions.
    - c. Clean concrete floors to be painted with a 5 percent solution of muriatic acid or other etching cleaner. Flush the floor with clean water to remove acid, neutralize with ammonia, rinse, allow to dry, and vacuum before painting.
  - 3. Wood: Clean surfaces of dirt, oil, and other foreign substances with scrapers, mineral spirits, and sandpaper, as required. Sand surfaces exposed to view smooth and dust off.
    - a. Scrape and clean small, dry, seasoned knots, and apply a thin coat of white shellac or other recommended knot sealer before applying primer. After priming, fill holes and imperfections in finish surfaces with putty or plastic wood filler. Sand smooth when dried.
    - b. Prime, stain, or seal wood to be painted immediately on delivery. Prime edges, ends, faces, undersides, and backsides of wood, including cabinets, counters, cases, and paneling.
    - c. When transparent finish is required, backprime with spar varnish.
    - Backprime paneling on interior partitions where masonry, plaster, or other wet wall construction occurs on backside.
    - e. Seal tops, bottoms, and cutouts of unprimed wood doors with a heavy coat of varnish or sealer immediately on delivery.

- 4. Ferrous Metals: Clean ungalvanized ferrous-metal surfaces that have not been shop coated; remove oil, grease, dirt, loose mill scale, and other foreign substances. Use solvent or mechanical cleaning methods that comply with the Steel Structures Painting Council's (SSPC) recommendations.
  - a. Treat bare and sandblasted or pickled clean metal with a metal treatment wash coat before priming.
  - b. Touch up bare areas and shop-applied prime coats that have been damaged. Wirebrush, clean with solvents recommended by paint manufacturer, and touch up with the same primer as the shop coat.
  - c. Apply stripe coat (second coat of primer) on edges, bolts, angles, and welds.
- 5. Galvanized Surfaces: Clean galvanized surfaces with nonpetroleum-based solvents so surface is free of oil and surface contaminants. Remove pretreatment from galvanized sheet metal fabricated from coil stock by mechanical methods.
- D. Materials Preparation: Mix and prepare paint materials according to manufacturer's written instructions.
  - 1. Maintain containers used in mixing and applying paint in a clean condition, free of foreign materials and residue.
  - 2. Stir material before application to produce a mixture of uniform density. Stir as required during application. Do not stir surface film into material. If necessary, remove surface film and strain material before using.
  - 3. Use only thinners approved by paint manufacturer and only within recommended limits.
- E. Tinting: Tint each undercoat a lighter shade to simplify identification of each coat when multiple coats of the same material are applied. Tint undercoats to match the color of the finish coat, but provide sufficient differences in shade of undercoats to distinguish each separate coat.

## 3.3 APPLICATION

- A. General: Apply paint according to manufacturer's written instructions. Use applicators and techniques best suited for substrate and type of material being applied.
  - 1. Paint colors, surface treatments, and finishes are indicated in the schedules.
  - 2. Do not paint over dirt, rust, scale, grease, moisture, scuffed surfaces, or conditions detrimental to formation of a durable paint film.
  - 3. Provide finish coats that are compatible with primers used.
  - 4. The term "exposed surfaces" includes areas visible when permanent or built-in fixtures, convector covers, covers for finned-tube radiation, grilles, and similar components are in place. Extend coatings in these areas, as required, to maintain the system integrity and provide desired protection.
  - 5. Paint surfaces behind movable equipment and furniture the same as similar exposed surfaces. Before the final installation of equipment, paint surfaces behind permanently fixed equipment or furniture with prime coat only.
  - 6. Paint interior surfaces of ducts with a flat, nonspecular black paint where visible through registers or grilles.
  - 7. Paint back sides of access panels and removable or hinged covers to match exposed surfaces
  - 8. Finish exterior doors on tops, bottoms, and side edges the same as exterior faces.
  - 9. Finish interior of wall and base cabinets and similar field-finished casework to match existing.
  - 10. Sand lightly between each succeeding enamel or varnish coat.

- B. Scheduling Painting: Apply first coat to surfaces that have been cleaned, pretreated, or otherwise prepared for painting as soon as practicable after preparation and before subsequent surface deterioration.
  - 1. The number of coats and the film thickness required are the same regardless of application method. Do not apply succeeding coats until the previous coat has cured as recommended by the manufacturer. If sanding is required to produce a smooth, even surface according to manufacturer's written instructions, sand between applications.
  - 2. Omit primer on metal surfaces that have been shop primed and touchup painted.
  - 3. If undercoats, stains, or other conditions show through final coat of paint, apply additional coats until paint film is of uniform finish, color, and appearance. Give special attention to ensure edges, corners, crevices, welds, and exposed fasteners receive a dry film thickness equivalent to that of flat surfaces.
  - 4. Allow sufficient time between successive coats to permit proper drying. Do not recoat surfaces until paint has dried to where it feels firm, does not deform or feel sticky under moderate thumb pressure, and where application of another coat of paint does not cause the undercoat to lift or lose adhesion.
- C. Application Procedures: Apply paints and coatings by brush, roller, spray, or other applicators according to manufacturer's written instructions.
  - 1. Brushes: Use brushes best suited for the type of material applied. Use brush of appropriate size for the surface or item being painted.
  - 2. Rollers: Use rollers of carpet, velvet back, or high-pile sheep's wool as recommended by the manufacturer for the material and texture required, except on smooth metal surfaces.
  - 3. Spray Equipment: Use airless spray equipment with orifice size as recommended by the manufacturer for the material and texture required. Use spray equipment on smooth metal surfaces.
- D. Minimum Coating Thickness: Apply paint materials no thinner than manufacturer's recommended spreading rate. Provide the total dry film thickness of the entire system as recommended by the manufacturer.
- E. Mechanical and Electrical Work: Painting of mechanical and electrical work is limited to items exposed in equipment rooms and in occupied spaces.
- F. Mechanical items to be painted include, but are not limited to, the following:
  - 1. Piping, pipe hangers, and supports exposed and not galvanized.
  - 2. Heat exchangers.
  - 3. Tanks
  - 4. Ductwork exposed except do not paint in mechanical rooms.
  - Access doors.
  - 6. Grilles to match ceiling color--spray application, unless ceiling is white.
  - 7. Motors and mechanical equipment.
  - 8. Accessory items.
- G. Electrical items to be painted include items may be prefinished, if so, match Architect's color selected regardless of factory options, but are not limited to, the following:
  - 1. Exposed conduit and fittings that are not galvanized or plastic.
  - 2. Switchgear spray application interior/ exterior
  - 3. Panel boards spray application interior/exterior
  - 4. Transformers exterior
- H. Block Fillers: Apply block fillers to concrete masonry block at a rate to ensure complete coverage

with pores filled.

- I. Prime Coats: Before applying finish coats, apply a prime coat of material, as recommended by the manufacturer, to material that is required to be painted or finished and that has not been prime coated by others. Recoat primed and sealed surfaces where evidence of suction spots or unsealed areas in first coat appears, to ensure a finish coat with no burn through or other defects due to insufficient sealing.
- J. Pigmented (Opaque) Finishes: Completely cover surfaces as necessary to provide a smooth, opaque surface of uniform finish, color, appearance, and coverage. Cloudiness, spotting, holidays, laps, brush marks, runs, sags, ropiness, or other surface imperfections will not be acceptable.
- K. Transparent (Clear) Finishes: Use multiple coats to produce a glass-smooth surface film of even luster. Provide a finish free of laps, runs, cloudiness, color irregularity, brush marks, orange peel, nail holes, or other surface imperfections. Provide stain coat on all transparent finishes.
  - 1. Provide satin finish for final coats, unless noted otherwise.
- L. Stipple Enamel Finish: Roll and redistribute paint to an even and fine texture. Leave no evidence of rolling, such as laps, irregularity in texture, skid marks, or other surface imperfections.
- M. Completed Work: Match approved samples for color, texture, and coverage. Remove, refinish, or repaint work not complying with requirements.

## 3.4 CLEANING

- A. Cleanup: At the end of each workday, remove empty cans, rags, rubbish, and other discarded paint materials from the site.
  - 1. After completing painting, clean glass and paint-spattered surfaces. Remove spattered paint by washing and scraping. Be careful not to scratch or damage adjacent finished surfaces.

## 3.5 PROTECTION

- A. Protect work of other trades, whether being painted or not, against damage by painting. Correct damage by cleaning, repairing or replacing, and repainting, as approved by Architect.
- B. Provide "Wet Paint" signs to protect newly painted finishes. Remove temporary protective wrappings provided by others to protect their work after completing painting operations.
  - 1. At completion of construction activities of other trades, touch up and restore damaged or defaced painted surfaces. Comply with procedures specified in PDCA P1.

## 3.6 SHEEN SCHEDULE - OPAQUE FINISH

ITEM SHEEN

Interior:

Walls, Type A finish Ceiling, Type A finish Walls, Type B finish Ceiling, Type B finish

Eggshell Eggshell Semi-gloss Semi-gloss

HM Doors and HM Frames Semi-gloss

# 3.7 SHEEN SCHEDULE - TRANSPARENT FINISH/WITH STAIN

ITEM SHEEN

Prefinished

#### 3.8 INTERIOR PAINT SCHEDULE

Wood Doors

- A. Gypsum Board: Provide the following finish systems over interior gypsum board surfaces:
  - 1. Flat Acrylic Finish: 2 finish coats over a primer.
    - Primer: Latex-based, interior primer applied at spreading rate recommended by the manufacturer to achieve a total dry film thickness of not less than 1.2 mils (0.031 mm).
      - 1) PrepRite 200 Primer B28W2600.
    - b. First and Second Coats: Flat, acrylic-latex-based, interior paint applied at spreading rate recommended by the manufacturer to achieve a total dry film thickness of not less than 2.5 mils (0.064 mm).
      - 1) S-W: Pro Mar 200 Flat B30W251.
  - 2. Eggshell, Acrylic-Enamel Finish: 2 finish coats over a primer.
    - Primer: Latex-based, interior primer applied at spreading rate recommended by the manufacturer to achieve a total dry film thickness of not less than 1.2 mils (0.031 mm).
      - 1) S-W: Promar 200 Primer B28W2600.
    - b. First and Second Coats: Eggshell, acrylic-latex, interior enamel applied at spreading rate recommended by the manufacturer to achieve a total dry film thickness of not less than 2.8 mils (0.071 mm).
      - 1) S-W: Pro Mar 200 HP Eggshell.
  - 3. Semi-gloss, Acrylic-Enamel Finish: 2 finish coats over a primer.
    - a. Primer: Latex-based, interior primer applied at spreading rate recommended by the manufacturer to achieve a total dry film thickness of not less than 1.2 mils (0.031 mm).
      - 1) S-W: Promar 200 Primer B28W2600.
    - b. First and Second Coats: Semi-gloss, acrylic-latex, interior enamel applied at spreading rate recommended by the manufacturer to achieve a total dry film thickness of not less than 2.8 mils (0.071 mm).
      - 1) S-W: Pro Mar 200 HP Semi-Gloss.

END OF SECTION 099000